This electronic thesis or dissertation has been downloaded from the King's Research Portal at https://kclpure.kcl.ac.uk/portal/



Changing Patterns of Incidence, Service Engagement and Outcomes Among Black African and Black Caribbean Patients With Psychotic Illness

Oduola, Sheri

Awarding institution: King's College London

The copyright of this thesis rests with the author and no quotation from it or information derived from it may be published without proper acknowledgement.

END USER LICENCE AGREEMENT



Unless another licence is stated on the immediately following page this work is licensed

under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International

licence. https://creativecommons.org/licenses/by-nc-nd/4.0/

You are free to copy, distribute and transmit the work

Under the following conditions:

- Attribution: You must attribute the work in the manner specified by the author (but not in any way that suggests that they endorse you or your use of the work).
- Non Commercial: You may not use this work for commercial purposes.
- No Derivative Works You may not alter, transform, or build upon this work.

Any of these conditions can be waived if you receive permission from the author. Your fair dealings and other rights are in no way affected by the above.

Take down policy

If you believe that this document breaches copyright please contact <u>librarypure@kcl.ac.uk</u> providing details, and we will remove access to the work immediately and investigate your claim.

Changing Patterns of Incidence, Service Engagement and Outcomes Among Black African and Black Caribbean Patients With Psychotic Illness

Sherifat Abimbola Oduola

Thesis submitted for the degree of

Doctor of Philosophy in Psychiatric Epidemiology

Institute of Psychiatry, Psychology and Neuroscience,

King's College London,

University of London

June 2017

Abstract

Strong and consistent evidence of higher incidence of psychosis, crisis pathways to care and poor outcomes following first episode of psychosis (FEP) in ethnic minority populations have been reported particularly among Black African and Caribbean populations in the UK. Further, there is ongoing debate about the role of early intervention services which have shown promising effects on reducing treatment delays and a range of outcomes following FEP. This study aimed to estimate ethnic differences in the incidence of psychosis, pathways to care and two-year service use outcomes in three ethnic groups (namely, White British, Black African and Black Caribbean). An investigation of pathways to care and clinical outcomes comparing early intervention service (EIS) and non-EIS users was also conducted.

An administrative incidence study design with a follow up cohort study of outcomes of psychoses was employed to investigate the aims of this study. First episode psychosis cases were identified and their data were drawn from the South London and Maudsley NHS Foundation (SLaM) Biomedical Research Centre case register. Pathways to care, sociodemographic and clinical information were collected using the Personal and Psychiatric History Schedule (PPHS, WHO, 1996) the Medical Research Council Socio-demographic schedule (Mallett et al., 1997) and the Life Chart schedule (WHO, 1992) respectively for the purpose of case note data extraction. Data on 558 FEP patients were analysed using standardised incidence rate (SIR) / Poisson regression, logistic regression and negative binomial regression.

Higher incidence rates were found among Black African (adj. IRR =3.59; 95% CI 2.8 – 4.55) and Black Caribbean (adj. IRR = 2.81; 95% CI 2.15 – 3.68) ethnic groups compared with the White British group.

At first contact for psychosis, Black African (adj. OR = 3.23; 95% CI 1.57–6.63) patients were more likely to be compulsorily admitted, but there were no differences between Black Caribbean (adj. OR = 1.78; 95% CI 0.75–4.24) and White British patients. Comparison of pathways to care between the AESOP study and this study data showed that there were no ethnic differences in GP or criminal justice agency referral in this study compared with the AESOP study. However, there was evidence that Black African (adj. OR = 7.34; 95% CI1.15–

46.74) and Black Caribbean (adj. OR = 48.89; 95% CI3.49–684.71) patients were more likely to be referred by the accident and emergency department.

At two-year follow-up Black African patients experienced worse service use outcomes, as well as higher rates of compulsory admissions (adj. IRR =3.01; 95% Cl1.33–6.80) and hospital admissions (adj. IRR = 1.98; 95% Cl1.12–3.48). Ethnic differences were not evident between Black Caribbean and White British patients on compulsory admission or hospital admission outcomes at follow-up. Further, compared with patients who accessed early intervention service, patients in the non-EIS group were more likely to be hospitalised (adj. IRR = 3.30; 95% Cl 1.98–5.49), and compulsorily admitted (adj. IRR = 2.22; 95% Cl1.18–4.18).

This study provides a novel insight into the associations between ethnicity and incidence of psychosis, pathways to care and outcomes of two-year follow-up service use, and hints at a possibility that the service engagement disparities between Black Caribbean and White British patients may be diminishing.

Abbreviations

AESOP Aetiology and Ethnicity in Schizophrenia and Other Psychoses study

- ARMS At Risk Mental State
- BME Black and Minority Ethnic groups
- BRC Biomedical Research Centre of the SLaM NHS Trust
- BRCID Unique identifying number for each record of the Case Register
- CI Confidence Interval
- CRIS Case Register Interactive Search system of the Biomedical Research Centre
- EI Early intervention
- EIS Early intervention Service
- ELFEP East London First Episode Psychosis study
- FEP First episode psychosis
- ICD-10 WHO International Classification of Diseases (10th/11th edition)
- IOPPN Institute of Psychiatry, Psychology & Neuroscience
- IQR Inter-quartile range
- IR Incidence rate
- IRR Incidence rate ratio
- MHA Mental Health Act
- MRC Medical Research Council
- N Number / sample size
- ONS UK Office of National Statistics
- OPCRIT Operational Criteria Checklist for Affective and Psychotic Disorders
- OR Odds Ratio
- RCT Randomised Control Trial
- RR Relative Risk
- SLaM South London and Maudsley NHS Trust

Table of Content

A	Abstract2					
Abbreviations						
Li	List of Tables					
Li	st of Fig	gures		14		
A	cknowle	edger	nents	15		
1	Cha	pter 1	L: General Introduction	16		
	1.1	Sync	opsis	16		
	1.1.	Ethn	icity	17		
	1.2.	Psyc	hosis	20		
	1.2.	1.	Diagnosis and Classification	20		
	1.3.	Obje	ectives, Aims and Hypotheses	23		
	1.4.	Stru	cture of Thesis	25		
	1.5.	Cano	didate's Distinct and Original Contributions	27		
2	Cha	pter 2	2: Literature Review – Incidence of First Episode of Psychosis and Ethnicity	28		
	2.1	Sync	opsis	28		
	2.2	Exar	nining the Role of Ethnicity and Social Risk Factors in the Aetiology of Psychosis	29		
	2.2.	1	Psychosis, ethnicity and migration	30		
	2.2.2	2	Ethnic density, neighbourhood and psychosis	46		
	2.2.	3	Social disadvantage and psychosis	53		
	2.3	Synt	hesis of current evidence	61		
	2.3.	1	Definition of ethnic groups	61		
	2.3.	2	Study design and sampling	62		
	2.3.	3	Sources used to collect data	63		
	2.4	Sum	mary	64		
 Chapter 3: Literature Review – Early Intervention for Psychosis, Pathways to Care ar 65 		3: Literature Review – Early Intervention for Psychosis, Pathways to Care and Ethnic	ity			
	3.1	Sync	opsis	65		
	3.2	Exar	nining the Role of Early Intervention Services in the Pathways into Care	66		
	3.2.2	1	Duration of untreated psychosis, mode of onset and ethnicity – the UK evidence	67		
	3.2.2 evid	2 lence	Duration of untreated psychosis, mode of onset and ethnicity – international 74			
3.3 Source of referral and mode of contact with mental health services			80			

3.4	4	Patl	hways to Care – Compulsory Admission and Ethnicity	84
3.	5	Methodological consideration and gaps in current evidence		
3.	6	Sum	nmary	99
4	Cha	pter	4: Literature Review – Ethnicity, Course and Service Use Outcomes Following First	
Episo	ode l	Psych	nosis	101
4.:	1	Syn	opsis	101
4.: cli	2 nica	Earl I oute	y intervention for psychosis – the promise of improved pathways to care and bette comes	er 102
4.3	3	Exa	mining course and outcome of psychosis and ethnicity	103
4.4	4	Lim	itation of current research	106
4.	5	Sum	nmary	106
5	Cha	pter	5: At risk population and the mental health provider	108
5.3	1	Bac	kground	108
5.2	2	Con	texts	108
5.3	3	The	Denominator: population of Lambeth and Southwark – Socioeconomic context	109
	5.3.	1	London Borough of Lambeth	110
	5.3.	2	The London Borough of Southwark	111
5.4	4	The	Mental Health Provider	112
5.	5	Cas	e registers: History	113
6	Cha	pter	6: Study Methodology	115
6.:	1	Syn	opsis	115
6.2	2	Aim	is and justification of study design	116
6.3	3	Just	ification of administrative incidence with a follow up cohort study design	118
6.4	4	CRIS	S: a case register and research platform	120
6.5	5	Met	thodology of the CRIS-FEP study	123
	6.5.	1	Background and study design	123
	6.5.	2	Screening and case ascertainment of psychosis cases	124
	6.5.	3	Reliability	126
	6.5.	4	Main data collection measures	127
	6.5.	5	Ethical approval	131
6.	6	Met	thodology of the follow-up study	131
6.	7	Stat	tistical analysis	133
7	7 Chapter 7: Results of Identifying First Episode Psychosis, Incidence of Psychosis and Ethnicity 137			:y
7.	1	Syn	opsis	137

7.2	Bac	kground: the AESOP findings	137
7.3	Res	ults	138
7.3	.1	Screening results	138
7.3.	.2	Demographic characteristics of total sample at baseline	140
7.3.	.3	Comparison between sociodemographic variables and ethnic groups	141
7.4	Esti	mating incidence rate of first episode psychosis	145
7.4	.1	Unadjusted and adjusted incidence rate ratios for psychosis by ethnic groups	148
7.4.	.2	Unadjusted and adjusted age-specific incidence rate ratios by ethnic groups	150
7.5	Sum	imary	153
8 Cha Ethnicity	apter 3 /	8: Results of Pathways to Care, Compulsory Admission at First Episode Psychosis	s and 155
8.1	Syne	opsis	158
8.2	Patł	nway to care variables at first contact for psychosis (baseline)	159
8.2.	.1	Comparison between ethnicity and pathways to care variables	161
8.3	Asso	ociations between sociodemographic, pathways to care variables and compulso	ry
admis	sion .		166
8.3	.1	Unadjusted odds ratios for hospital admission at first contact	166
8.3.	.2	Unadjusted odds ratios for compulsory admission at first contact	168
8.3.	.3	Unadjusted odds ratios for 'GP' referral at first contact	172
8.3.	.4	Unadjusted odds ratios for A&E referral at first contact	175
8.3.	.5	Unadjusted odds ratios for Police referral at first contact	179
8.3	.6	Unadjusted odds ratios for 'Other' referral at first contact	181
8.4 at firs	Adjı t cont	usted Odds of admission (hospital and compulsory) and source of referral by eth act	nicity 185
8.4.	.1	Adjusted odds ratios for hospital admission by ethnicity at first contact	186
8.4.	.2	Adjusted odds ratios for compulsory admission by ethnicity at first contact	188
8.4. con	.3 Itact	Adjusted odds ratios for compulsory admission by ethnicity and gender at first 188	
8.4.	.4	Adjusted odds ratios for 'Other' referral by ethnicity at first contact	193
8.4	.5	Adjusted odds ratios for GP referral by ethnicity at first contact	195
8.4	.6	Adjusted odds ratios for A&E referral by ethnicity at first contact	195
8.4	.7	Adjusted odds ratios for Police/CJA referral by ethnicity at first contact	195
8.5	Sum	ımary	199

9 Chapter 9: Results of the Impact of Early Intervention Services on Pathways to Care for Minority Ethnic Groups during First Episode Psychosis: comparing data from the AESOP and present (CRIS-						
FEP) studies						
9.1	Synopsis					
9.2	Analysis					
9.3	Results					
9.3.	1 Sample characteristics					
9.3. GP r	2 Unadjusted associations between sociodemographic/pathways to care variables and referral by use of EIS					
9.3. acci	3 Unadjusted associations between sociodemographic/pathways to care variables and dent and emergency referral by use of EIS					
9.3. CJA	4 Unadjusted associations between sociodemographic/pathways to care variables and referral by use of EIS					
9.3. com	5 Unadjusted associations between sociodemographic/pathways to care variables and apulsory admission by use of EIS					
9.4	Multivariable logistic regression models224					
9.4.	Associations between ethnicity and GP referral by use of EIS					
9.4.	Associations between ethnicity and A&E referral by use of EIS					
9.4.	Associations between ethnicity and criminal justice agency referral by use of EIS 229					
9.4.	Associations between ethnicity and compulsory admissions by use of EIS					
9.5	Summary					
10 C up 2	hapter 10: Results of Associations between Ethnicity and Service Use Outcomes at Follow- 33					
10.1	Synopsis					
10.2	Sample characteristics at follow-up236					
10.3	Comparison between ethnic groups and clinical characteristics at follow-up					
10.4	Rate of hospital admission per year by ethnic group during the follow-up period244					
10.5	Rate of compulsory admission per year by ethnic group during the follow-up period246					
10.6	Associations between ethnicity and number of hospital admissions during the follow-up 248					
10.7 follow	Associations between ethnicity and the number of compulsory admissions during the -up period252					
10.8	Associations between length of hospital stay and ethnicity during the follow-up period.255					
10.9	Summary					
11 C during th	hapter 11: Results: Ethnicity and Service Use Outcomes by Early Intervention Service Status ne Follow-up Period					

	11.1	Syno	opsis	259
	11.2	Com	nparisons between EIS use, sociodemographic and follow-up clinical characteristics.	260
	11.2	2.1	EIS eligible versus Non-eligible comparisons	260
	11.2	2.2	EIS versus non-EIS comparisons	260
	11.3	Asso	ociations between EIS status and number of hospital admissions during follow-up	264
	11.4	Asso	ociations between number of compulsory admissions and EIS use during follow-up	267
	11.5	Asso	ociations between length of hospital stay and EIS use during follow-up	269
	11.6 status	Asso 271	ociations between ethnicity and service use outcomes by early intervention service	
	11.7 service	Asso e stat	ociations between ethnicity and number of hospital admissions by early intervention	า 271
	11.8 interve	Asso entio	ociations between ethnicity and number of compulsory admissions by early n service status	273
	11.9 status	Asso 275	ociations between ethnicity and length of hospital stay by early intervention service	
	11.10	Sum	nmary	277
1	2 Cl	hapte	er 12: General Discussion and Methodological Consideration	278
	12.1	Syno	opsis	279
	12.2	Sum	nmary of findings	279
	12.3	Met	hodological Strengths and Limitations	281
	12.3	8.1	Strengths	281
	12.3	8.2	Limitations	283
	12.4	Met	hodological considerations	285
	12.4	1.1	The role of bias	285
	12.4	1.2	Selection bias	286
	12.4	1.3	Information bias (observer)	286
	12.4	1.4	Chance	287
	12.4	1.5	Confounding	288
	12.5	Expl	laining the findings and relationship with previous studies	288
	12.5	5.1	Interpreting the Incidence findings and relationship to previous studies	289
	12.5 relat	5.2 tions	Interpreting the Pathways to care characteristics at first contact for psychosis and hip to previous studies	294
	12.5 stud	5.3 lies	Interpreting the Service Use Outcomes at follow-up and relationship to previous 300	
	12.6	Imp	lications for research, policy and clinical agenda	302

12.7	Future research	308
12.8	Conclusions	310
REFEREN	NCES	313
APPEI	NDIX 1: FEP Screening Search Commands	327
APPEI	NDIX 2: Psychosis Screening Schedule	329
APPEI	NDIX 3: Data Collection Booklet	331
APPEI	NDIX 4: Follow-up Data Extraction	340

List of Tables

Table 1.1: Race, culture and ethnicity (from Fernando, 2002)	17
Table 1.2: ICD-10 classifications of psychotic disorders	22
Table 2.1: Studies of incidence of psychosis, ethnicity and migration.	34
Table 2.2: Studies of psychosis and ethnic density	48
Table 2.3: Studies of psychosis and social disadvantage and ethnicity	55
Table 3.1: UK studies of duration of untreated psychosis and ethnicity	70
Table 3.2: International studies of duration of untreated psychosis and ethnicity	75
Table 3.3: Studies of psychosis compulsory admission and ethnicity	88
Table 5.1: Ethnic breakdown of at risk population 2011 census	109
Table 6.1: Study time points	116
Table 6.2: Inter-rater reliability for rating scales	127
Table 7.1: Overall sociodemographic variables	140
Table 7.2: Comparison between sociodemographic variables and ethnic groups	143
Table 7.3: Summary of overall study population and population at risk	145
Table 7.4: 4 Crude and age-standardised incidence per 100,000 person-years with 95% CI for	[.] first
episode psychosis by demographic factors	147
Table 7.5: Incidence rate ratios (IRR) with 95% CI in ethnic minority groups for first episode p	osychosis
	149
Table 7.6: Incidence rate ratios (IRR) with 95% CI in ethnic minority groups for first episode p	sychosis
by gender	149
Table 7.7: Adjusted age-specific incidence rate ratios (IRR) with 95% CI and p-value in ethnic	minority
groups for first episode all psychosis	151
Table 7.8: Comparison of findings from AESOP and current study by ethnicity	152
Table 8.1: Basic characteristics of pathways to care variables	160
Table 8.2: Comparison between clinical, pathways to care variables and ethnic groups	163
Table 8.3: Unadjusted odds ratios of hospital admission by sociodemographic and pathways	to care
variables at first contact	166
Table 8.4: Unadjusted odds ratios of compulsory admission by socio-demographic and pathy	vays to
care variables at first contact	169
Table 8.5: Unadjusted odds ratios of GP referral by socio-demographic and pathways to care	!
variables at first contact	172
Table 8.6: Unadjusted odds ratios of A&E referral by socio-demographic and pathways to car	re
variables at first contact	176
Table 8.7: Unadjusted odds ratios of police/CJA referral by socio-demographic and pathways	to care
variables at first contact	179
Table 8.8: Unadjusted odds ratios of 'Other' source of referral by socio-demographic and part	thways
to care variables at first contact	
Table 8.9: Adjusted odds ratios of associations between hospital admission and ethnicity at f	irst
contact (n=367)	187
Table 8.10: Adjusted odds ratios of associations between compulsory admission and ethnicit	y for
men at first contact (n=191)	189

Table 8.11: Adjusted odds ratios of associations between compulsory admission and ethnicity for women at first contact (n=176) 189
Table 8.12: Adjusted odds ratios of associations between ethnicity and compulsory admission at first contact
Table 8.13: Adjusted odds ratios of associations between ethnicity and 'Other' referral at first contact. (n=338)
Table 8.14: Adjusted odds ratios of associations between ethnicity and GP referral at first contact 196
Table 8.15: Adjusted odds ratios of associations between ethnicity and A&E referral at first contact
Table 8.16: Adjusted odds ratios of associations between police/CJA referral and ethnicity at first contact 198
Table 9.1: Demographic and pathways to care characteristics of AESOP and CRIS – FEP samples 205
Table 9.2: Unadjusted odds ratios of associations between sociodemographic/pathways to care
Table 0.2: Unadjusted adds ratios of associations between sociadomographic (nathways to save
variables and A&E referral by use of EIS
Table 9.4: Unadjusted odds ratios of associations between sociodemographic/pathways to care
variables and CJA referral by use of EIS
Table 9.5: Unadjusted odds ratios of associations between sociodemographic/pathways to care
variables and compulsory admission by use of EIS221
Table 9.6: Adjusted odds ratios of associations between ethnicity and GP referral by use of EIS 226
Table 9.7: Adjusted odds ratios of associations between ethnicity and A&E referral by use of EIS 228
Table 9.8: Adjusted odds ratios of associations between ethnicity and CJA referral by use of EIS 230
Table 9.9: Adjusted odds ratios of associations between ethnicity and compulsory admission by use
of EIS
Table 10.1: Service Use and Clinical characteristics during the follow up period 236
Table 10.2: Comparisons between clinical and service use characteristics at follow up and ethnic groups 241
Table 10.3: Rate of hospital admission per year with 95% confidence intervals by ethnic group during
the follow up period
Table 10.4: Rate of compulsory admission per year with 95% confidence intervals by ethnic group
during the follow up period
Table 10.5: Adjusted rate ratios of associations between ethnicity and number of hospital admissions
during the follow up period (n=367)250
Table 10.6: Adjusted rate ratios of associations between ethnicity and number of compulsory
admissions during the follow up period (n=367)
Table 10.7: Adjusted rate ratios of associations between ethnicity and length of hospital stay during
the follow up period (n=367)
Table 11.1: Comparisons between baseline socio-demographic, pathways to care and follow up
clinical characteristics by EIS status
Table 11.2: Unadjusted and adjusted rate ratios for number of hospital admission and EIS use
(n=366)
Table 11.3: Unadjusted and adjusted rate ratios for number of compulsory admission and EIS use(n=366)

Table 11.4: Unadjusted and adjusted rate ratios of length of stay during hospital admission and EIS
use (n=366)
Table 11.5: Associations between ethnicity and number of hospital admissions by use of El services
Table 11.6: Associations between ethnicity and number of compulsory admissions by use of El
services
Table 11.7: Associations between ethnicity and length of hospital stay by use of EI services276

List of Figures

Figure 3.1: FEP pathways to care	81
Figure 5.1: The study catchment areas (Source: Google Wikipedia, Se16boy)	109
Figure 6.1: CRIS front end	121
Figure 6.2: SQLCRIS front end	122
Figure 6.3: Structure of data collection	123
Figure 7.1: Screening flowchart	139
Figure 7.2: Standardised incidence rates by age and gender	148
Figure 10.1: Length of hospital stay by ethnicity	239
Figure 10.2: Histogram of number of hospital admissions	248
Figure 10.3: Histogram of number of compulsory admissions	252
Figure 10.4: Histogram of length of hospital stay	255

Acknowledgements

Many people have contributed to the work presented in this thesis.

Firstly, I am grateful to the patients whose anonymised clinical data were central to this study.

I am hugely grateful to my supervisors, Professor Craig Morgan, Professor Tom Craig, Dr Jay Das-Munshi and Professor Dinesh Bhugra, who provided me with their unfailing support and academic assistance over the last six years.

I am indebted and grateful to Professor Tom Craig, who has been my mentor, for providing me with invaluable support and advice since starting my career in academic research and for supervising this study. To him, I owe any success in my academic career.

Particular thanks go to Professor Craig Morgan, whose earlier work inspired the work presented in this thesis and for his support and his contagious passion for the subject.

My employers, the NIHR South London & Maudsley Biomedical Research Centre for allowing me time to carry out this work.

My colleagues Dr Francois Bourque, Dr Charlotte Gayer-Anderson and Dr Jason Tsang for their huge contributions to the screening of the case register to identify first episode psychosis patients.

My thanks to colleagues at the Biomedical Research Centre Nucleus, particularly Mr Matthew Broadbent, Mr Hitesh Shetty, Dr Alex Tulloch and Professor Rob Stewart for their generosity in providing time and advice in setting up this study and its data extraction.

Dr Peter Schofield, who was generous to provide time and gave invaluable statistical advice. Thanks also to my colleague Dr Julie Williams who happily read and offered suggestions on initial drafts.

My friends: Mar Rus-Calafell, Sarah Ojugo, Surajudeen Olowolayemo, Rosie Padley, Iris Seechurn, Adeyinka Omomo and Rana Rasheed for offering listening ears and great distractions.

I am indebted to my family: my children Rilewa, Kofo, Lanre for their patience, being my strength and incredibly supportive throughout this journey. My brothers Yinka, Niyi, Dipo and my grandmother, Sara, for their love, belief in me and unwavering support. My uncle, Ayoade, my aunt, Mojisola, and cousins Oladayo, Ayodeji and Etimbuk for their infinite support and encouragements. And finally my mum Kemi, who has been my rock; I could not have succeeded without her.

This thesis is dedicated to my children and the loving memory of my dad.

1 Chapter 1: General Introduction

1.1 Synopsis

The aetiological role of social factors in the onset of psychosis has been extensively studied, with particular attention paid to the role of ethnicity. Several studies have consistently reported a higher incidence of psychosis among black and ethnic minority populations in the Western nations (Harrison et al., 1997, Fearon et al., 2006a, Veling et al., 2006).

As the proportion of people living outside their country of birth is increasing globally, recognising the health needs of minority ethnic groups in the host society is paramount. For instance, compared to the 2001 census, data from the 2011 census suggest that 13% of the population in England and Wales are people born outside the UK (ONS, 2011b). In addition, with the passage of time, some ethnic minority groups and their succeeding generations are more integrated into the UK society than others. With the continuous social changes, it is unclear to what extent ethnicity still contributes to the onset and course of psychotic disorder.

Furthermore, there have been major mental healthcare provision changes in the UK. In the last two decades, early intervention services (EIS) were introduced specifically to detect symptoms, reduce the duration of untreated psychosis and improve access to care and treatment (Craig et al., 2004a, Birchwood, 1995, Kuipers et al., 2004, McGorry et al., 2009). Further, psychiatric hospital beds have substantially reduced, community-based crisis services; home treatment and crisis resolution teams were introduced and there is greater emphasis on inter-professional working culture (Johnson et al., 2001, Priebe et al., 2008, Peck and Norman, 1999). These changes are likely to have an impact on mental health of the overall population and in particular ethnic minority populations.

The overall purpose of this thesis is to describe the incidence rates of psychosis, pathways to care at first contact and service use outcomes at two-year follow-up by ethnicity with a focus on three ethnic groups (namely, Black African, Black Caribbean and White British) in two well-defined catchment areas of south London.

Before outlining the structure of this thesis, it is important to consider the exposure and outcome (ethnicity and psychosis) variables that are under investigation in this study. These are briefly discussed below.

1.1. Ethnicity

The terms ethnicity, race and culture are often used interchangeably and so it is important to disentangle them. Fernando (2002) provides a summary of the interplay and connections between the three in Table 1.1 below.

Table 1.1: Race, culture and ethnicity (from Fernando, 2002)

	Characterised by	Determined by	Perceived as
Race	Physical Appearance	Genetic Ancestry	Permanent
			(genetic / biological)
Culture	Behaviour	Upbringing	Changeable
	Attitudes	Choice	(assimilation /
			acculturation)
Ethnicity	Sense of Belonging	Social Pressures	Partially Changeable
	Group Identity	Psychological Need	

The distinctions of race and ethnicity are socially constructed, as the question of who belongs to which groups and the status attached to different groups are matters of social convention (Blakemore and Boneham, 1994). Race may be considered as externally motivated, stemming more from the need of human beings to categorise each other than the need to form inclusive social groups (Karlsen et al., 2006). It becomes important to maintain group boundaries and group identity around the perceived similarities of members of the group. And so, more emphasis is placed on the process of stereotype by others which inherently contains a judgement of value (Karlsen et al., 2006). Physical appearance, such as skin colour, hair colour etc. become important in categorising racial groups particularly in the Western states. These racial judgements are in turn shaped by folk-myth and beliefs about the intrinsic features of 'Black' and 'White' (Blakemore and Boneham, 1994). These characterisations originated during the voyage of Europeans to West Africa in the 1500s, and the striking characteristic they discovered was the African peoples' skin colour (Jordan, 1974). However, the 'Black' and 'White' characterisations are emotionally loaded in the West, particularly in the English language, not only because of the contrast that one denotes the polar opposite of the other, but because the two have been argued to carry a connotation of hierarchy (Jordan, 1974, Mason, 1995). Therefore racism is a form of biological reductionism, which stems from an observation of physical features and preconceived stereotypes about a racial group (Blakemore and Boneham, 1994). The use of racial categories has largely reduced in scientific research and been replaced by the use of ethnicity, a more acceptable term. However, researchers often categorise people into 'ethnic' categories in such a way that it is difficult to discern them from racial categories, e.g. the crude dichotomy between Black and White is used in some research studies' ethnic categorisations. People from diverse cultural and ethnic backgrounds are consequently often grouped together on the basis primarily of skin colour, with the underlying assumption that this marks an important and meaningful distinction that might explain an outcome of interest, for example, rates of psychosis, compulsory admission and hospitalisation.

Ethnicity is not an easily definable concept, because it is subjective and includes a range of factors in its definition. These include physical appearance, geography, language and cultural systems (Fernando, 2002). According to (Karlsen et al., 2006), an ethnic group implies a categorisation which may be defined by personal choice, establishment of common identity as well as certain shared economic, social or cultural characteristics at a given time. Ethnic identity may be a resource; it is also another way for 'outsiders' to make

judgements about a group (Blakemore and Boneham, 1994). These may be positive stereotypes (for example, the outgoing, extended Black family) or terms of abuse. So to put it simply, ethnicity can be described as a fluid concept encompassing an individual's sense of belonging to a particular group on the basis of geographical origin, skin colour, physical appearance, language, religion and cultural practice (Sewell, 2009, Fernando, 2002). Ethnicity is flexible and shifting on different levels according to situation and context. This is particularly important as far as minority status is concerned because opinions vary as to whether ethnic loyalties fade over time as migrants adjust to their adopted country, or whether they are retained (Blakemore and Boneham, 1994).

Culture on the other hand provides a set of socially shared guidelines or rules that shape and constrain beliefs, attitudes and behaviour. Like ethnicity, culture is not static or homogenous. As Blakemore and Boneman (1994) note, 'culture itself is an umbrella term referring broadly to a way of life: distinctive social institution (marriage, religion), social norms, manners, attitudes and ways of thinking' (p.6). Beliefs, attitudes and so on inevitably undergo modifications as individuals are exposed to and experience other cultural systems and ideas. Although cultural heritage may form a significant component of ethnic identity (as above), it does not define it and those who perceive themselves as belonging to an ethnic group may well differ markedly in terms of the cultural orientation that informs their beliefs and actions (Fernando, 2002). This highlights caution against confusing culture and ethnicity.

For the purpose of this thesis, capitalising on the key component of ethnicity, i.e. a sense of belonging, suggests that ethnicity can only be accurately ascertained by self-ascription. While this is true, it is nonetheless recognised that cultures vary considerably across the Caribbean Islands as well as in sub-Saharan Africa. Therefore, sense of identity, sense of belonging and self-ascription themselves will be influenced by the cultural context of the individual. In clinical practice, information on ethnicity is routinely collected and recorded in clinical records. However, it is acknowledged that the information may not be available or is incorrectly recorded. Therefore it is important to ensure that ethnicity is assigned consistently across the sample in this study. To do this, in addition to the routinely recorded ethnic groups, myself and colleagues manually checked and assigned ethnicity for individual patients included in the study. Ethnicity was ascribed independently by researchers using all

available information from the clinical records including country of birth, nationality, language spoken at home, parents' country of birth in conjunction with country of birth, geographical region (e.g. Saharan and sub-Saharan Africa), racial groups and religious groups (ONS, 2003). For the three ethnic groups under investigation in this study, 'Black African' includes people born in sub-Saharan Africa and people whose parents and grandparents were born in Africa. Similarly, 'Black Caribbean' includes people born or whose parents and grandparents were born in the Caribbean. Individuals of mixed parentage (e.g. Caribbean and other parentage or African and other parentage) were not included in the Black African and Black Caribbean groups. Instead, they were included in the 'Mixed' ethnic group. For the White British group, this includes people born in rest of Europe or elsewhere to White parents were excluded from the White British group.

1.2. Psychosis

Psychosis is one of the most challenging psychiatric conditions to define due to the subjective nature of its clinical presentation which is often self-reported and as yet there is no objective test or validated biological marker for the condition (Jones, 2011). Recent studies show that men are 1.5 times more likely to develop psychosis than women, although later onset is reported in women with better course and outcome (WHO, 2005, Fearon et al., 2006a, Kirkbride et al., 2012a). It is a costly condition both to sufferers, their family and the economy (van Os and Kapur, 2009). Incidence rates ranging between 0.7 to 1.4 per 10,000 per year have been in reported in studies where schizophrenia is narrowly defined (Jablensky et al., 1992b) and a median of 1.5 per 10,000 per year in a recent meta-analysis (McGrath et al., 2004).

1.2.1. Diagnosis and Classification

Psychotic symptoms include delusions (firmly-held, false beliefs about external reality despite alternative evidence), hallucinations (false perception in any modality in the absence of eternal stimuli) and thought disorder (disturbance in conscious thoughts, shown through verbal processes). The International Classification of Disease (ICD-10) (World Health Organization, 1993) provides groups of diagnoses that come under the auspices of psychosis as shown in Table 1.2 below. These diagnoses are used in clinical practice and research and

they are at times referred to in two broad terms: schizophrenia spectrum disorders (F20-29) or affective disorders (F30-33).

For the purpose of this thesis, a broad and pragmatic definition of psychosis was employed, using the Psychosis Screening Schedule to determine whether a person screens positive. Individuals screened positive if they experienced at least one of the following:

- Hallucinations or pseudo-hallucinations in any modality
- Delusions
- Marked thought and speech disorder (e.g. incoherence, irrelevance, thought blocking, neologisms, incomprehensibility of speech) other than simple retardation or acceleration
- Marked psychomotor disorder (e.g. negativism, mutism or stupor, catatonic excitement, constrained attitudes or unnatural posture maintained for long periods) other than simple retardation or acceleration
- Emergence of marked exacerbation of bizarre and grossly inappropriate behaviour (e.g. talking or giggling to self, acts incomprehensible to others, loss of social constraints etc.)

Or two of the following:

- Marked reduction or loss of interests, initiative and drive, leading to a serious deterioration of the performance of usual activities and tasks
- Emergence or marked exacerbation of social withdrawal (active avoidance of communication with others)
- Severe excitement, purposeless destructiveness or aggression
- Episodic or persistent states of overwhelming fear or severe anxiety
- Gross and persistent self-neglect

Table 1.2: ICD-10 classifications of psychotic disorders

ICD-	Disorder	ICD-	Disorder
Code		Code	
F20- F29	delusional disorders	F30- F39	Niood (affective) disorders
F20	Schizophrenia	F30	Manic Episode
F20.0	Paranoid Schizophrenia	F30.2	Mania with psychotic symptoms
F20.1	Hebephrenic Schizophrenia	F31	Bipolar affective disorder
F20.2	Catatonic Schizophrenia	F31.2	Bipolar affective disorder, current
F20.3	Undifferentiated Schizophrenia		episode manic with psychotic
F20.4	Post-schizophrenic depression		symptoms
F20.5	Residual Schizophrenia	F31.5	Bipolar affective disorder, current
F20.6	Simple Schizophrenia		episode severe depression with
F20.8	Other Schizophrenia		psychotic symptoms
F20.9	Schizophrenia, unspecified		
F21	Schizotypal Disorder	F32	Depressive episode
F22	Persistent Delusional Disorders	F32.3	Severe depressive episode with
F23	Acute and Transient Psychotic		psychotic symptoms
	Disorders	F33	Recurrent depressive disorder
F24	Induced Delusional Disorder	F33.3	Recurrent depressive disorder,
F25	Schizoaffective Disorders		current episode severe with
F25.0	Schizoaffective Disorder, manic		psychotic symptoms
	type		
F25.1	Schizoaffective Disorder,		
	depressive type		
F25.2	Schizoaffective Disorder, mixed		
F25.8	type		
F25.9	Other Schizoaffective Disorders		
F28	Schizoaffective Disorder,		
F29	unspecified Other Non-organic		
	Psychotic Disorders		

1.3. Objectives, Aims and Hypotheses

As mentioned above, the overall objective of this thesis is to describe (at baseline) the incidence rates of psychosis, pathways to care at first contact and (at follow-up) the service use outcomes by ethnic group at two years following first episode of psychosis.

At baseline, cross-sectional data on sociodemography, clinical presentation, compulsory detention and pathways to care at first contact were collected. In addition, a comparison of pathways to care is made between data from this study and the Aetiology and Ethnicity in Schizophrenia and Other Psychoses (ASEOP) study. Two questions will be addressed at this stage: (a) is the discrepancy in incidence rates between ethnic groups any larger (or smaller) now than 15 years ago? and (b) are there any differences in the duration of untreated psychosis (DUP) and pathways to care for FEP cases compared to 15 years ago and is this associated with the introduction of EIS?

The aims of the first phase (baseline) are:

- To estimate the incidence of psychosis by ethnic groups during a two-year period May 2010 – April 2012 in a well-defined catchment area in south London.
- 2. To compare pathways to care, hospital admission (compulsory vs. noncompulsory) and source of referral (GP, A&E and others) at first contact for psychosis by ethnic groups.
- 3. To compare these pathways to care (compulsory vs. non-compulsory) and source of referral (GP, A&E and others) with those reported in the earlier AESOP sample and between those using early intervention service or not in the current study sample.

In relation to these aims, the following hypotheses will be tested:

Compared with those of White British ethnicity:

Incidence

1.1 The incidence of all psychoses will be higher in those of Black Caribbean and Black African ethnicity.

1.2 The magnitude of relative risk will be smaller for the minority ethnic groups (Black African and Black Caribbean) compared with those reported in previous studies (e.g. AESOP study, Fearon et al. 2006).

Pathways to care at first contact for psychosis

- 2.1 Higher rates of hospital admissions will be associated with Black African and Black Caribbean ethnicity.
- 2.2 Increased risk of compulsory admissions will be associated with Black African and Black Caribbean ethnicity.
- 2.3 Higher rates of police involvement will be associated with Black African and Black Caribbean ethnicity.
- 2.4 Lower levels of GP referral will be associated with Black African and Black Caribbean ethnicity.
- 2.5 Higher rates of accident and emergency referral will be associated with Black African and Black Caribbean ethnicity.
- 2.6 Compared with 15 years ago (i.e. AESOP vs. CRIS FEP samples), among patients aged 18-35 years old, ethnic differences in GP referral and crisis source of referral (criminal justice agency, accident and emergency) and compulsory admission at first contact for psychosis will be smaller for Black African and Black Caribbean patients.

At follow up, data relating to service use (i.e. number of hospitalisations, number of compulsory admissions and total number of days spent in hospital) were collected longitudinally during the two-year follow-up period. Two questions were the addressed: (a) are there differences by ethnic group in the rate of admission, compulsory admission and length of time patients spend in hospital during the follow up period? and (b) are there differences in rates of hospital admission, length of hospital stay and compulsory detention by early intervention service status during the follow up period?

The follow-up phase aims are:

- 1. To estimate rates and rate ratios of hospital admission and compulsory detention during the two-year follow-up by ethnic group.
- 2. To estimate differences in length of hospital stay by ethnic group during the twoyear follow-up.
- To compare service use (hospital admission, compulsory admission) between those using early intervention services and standard community care during the two-year follow-up.
- To determine whether there are ethnic differences in service use outcomes during the two-year follow-up in relation to early intervention service use status.

Course and outcome

The following hypotheses will be tested:

- 3.1 Over the two-year follow-up period, worse service use outcomes (characterised by increased rates of hospital admissions, compulsory admission and longer duration of hospital stay) will be observed among those of Black Caribbean and Black African ethnicity compared with White British patients.
- 3.2 Over the two-year follow-up period, worse service use outcomes will be associated with non-EIS use compared with EIS use and these will vary by ethnic groups.

1.4. Structure of Thesis

This thesis is structured as follow:

Chapter 2 presents a review of literature and methodological consideration of research studies that have examined the relationships between ethnicity and the incidence of psychosis.

Chapter 3 presents an overview and critique of research studies that investigated associations between ethnicity and pathways to care and source of referral. It also considers the role of early intervention services in the pathways to care and help-seeking during first episode psychosis.

Chapter 4 reviews and reports on studies that have investigated the associations between ethnicity and course and outcome of psychosis.

Chapter 5 gives a contextual overview of the population at risk within which this thesis study was conducted. It provides an in-depth detail of the case register from which psychosis cases were identified. It also discusses the sociodemographic and socioeconomic characteristics of the population at risk.

Chapter 6 presents the general methodology of the study from which the data for this thesis were drawn (Section 6.5). Section 6.5.3 provides an in-depth detail of the screening process for identifying first episode psychosis cases. Section 6.5.5 gives details of the measures used to assess sociodemographic characteristics, pathways to care and service use outcomes.

Chapter 7 describes the results of the screening procedure, sociodemographic characteristics and findings from the analysis of incidence of psychosis by ethnic groups (Hypotheses 1.1 and 1.2).

Chapter 8 presents the findings of the associations between ethnicity and various sources of referral, compulsory and hospital admissions during first contact for psychosis (Hypotheses 2.1 to 2.5).

Chapter 9 presents the results of the early intervention-stratified associations between ethnicity and pathways to care comparing data from the Aetiology and Ethnicity in Schizophrenia and Other Psychoses (ASEOP) study and data from this thesis (Hypothesis 2.6).

Chapter 10 presents findings of the relationships between ethnicity and service use outcomes at the two-year follow-up (Hypothesis 3.1).

Chapter 11 explores associations between early intervention service and service use outcomes. It also describes early intervention services-specific ethnic variations in service use outcomes (Hypothesis 3.2).

Chapter 12 presents a summary of findings presented in the preceding chapters, alongside a discussion of methodological strengths and limitations of the study. It then offers suggestions for direction of future research and the implications of the findings.

1.5. Candidate's Distinct and Original Contributions

This thesis was formed as part of a larger on-going study, the Clinical Records Interactive Search-First Episode Psychosis (CRIS-FEP) led by Professor Craig Morgan. Within this study, I was involved from the outset in all aspects of screening clinical records of over nine thousand patients, identifying first episode cases and collecting data. I led on the follow-up phase of this thesis, whereby key variables of interest were defined and extracted from CRIS. In preparing data for analysis in this thesis, I was responsible for checking data integrity and completeness on key variables such as ethnicity, sociodemographic characteristics, pathways to care, DUP and service use. I contributed extensively to data entry and data cleaning. The aims and hypotheses of this thesis, which I believe are novel, were developed by me and I carried out all the analyses presented in this thesis.

2 Chapter 2: Literature Review – Incidence of First Episode of Psychosis and Ethnicity

2.1 Synopsis

Higher incidence rates of psychosis are consistently reported among black and ethnic minority groups particularly in Western countries (Bhugra et al., 1997, Fearon et al., 2006a, Chorlton et al., 2012, Kirkbride et al., 2012b, Fung et al., 2009). Several possible explanations have been put forward to understand the excess of psychosis in these groups, including biological, environmental and social factors.

Much attention has been given to biological risk factors and bio-markers in understanding the aetiology of psychosis, even more so because of the emergence of sophisticated technologies for studying the brain and genes (Morgan et al., 2008b). So far, biological risk factors alone have not been sufficient to explain the aetiology of psychosis; even attempts made at explaining psychosis through hereditary from the twins studies fail to fully account for the aetiology of the illness (Cardno and Gottesman, 2001). Similarly, other biological factors such as obstetric complication have been found to be associated with increased risk of schizophrenia in offspring (Clarke et al., 2006) and the prevalence of such complications tends to be higher in migrant groups (Sharpley et al., 2001). However, this evidence is not consistent as other studies have found no association of higher obstetric complication among migrant groups (Hutchinson et al., 1997, Morgan et al., 2010). Another biological argument for increased rates of psychosis in the migrant groups (particularly black migrants) is that of vitamin D deficiency, since moving to colder climates may reduce exposure to sunlight (McGrath, 1999). Morgan et al. in a review contextualised this notion and argued that the potential racial discrimination that individuals from Black migrant and minority groups experience, who are most visible in predominately white societies, may be equally plausible as low vitamin D (Morgan et al., 2010).

The role of social factors and experiences has been a controversial debate in understanding the aetiology of psychosis. However, social changes whether at societal, familial or individual level may be at play in providing some promising indication in understanding the aetiology of psychosis. In the past two decades many efforts have been spent examining the role of social factors in the onset of psychosis with evidence suggesting that there are strong

links between the environment and social factors and psychosis (van Os et al., 2010). For example, an excess of psychosis is widely reported among migrant populations in Western countries such as the UK, USA, Australia, Europe and also among migrants' children born in the host country (Selten et al., 2001, Veen et al., 2002, Cantor-Graae and Selten, 2005, Fearon et al., 2006a, McKenzie et al., 2008). This suggests there is something about living as a member of a migrant or minority ethnic group in a host context that increases the risk of developing psychosis.

The demographic profile of the UK population has undergone a noticeable change with the proportion of people born outside the UK living in England and Wales increasing and rising unemployment, but with the number of people without educational qualifications falling by around 6% (ONS, 2011b). These factors may impact on the mental health of the overall population and in particular migrant populations.

This chapter provides a detailed review of research from social perspectives, addressing specifically the question, is the discrepancy in incidence rates between ethnic groups any larger (or smaller) now than they were 15 years ago?

2.2 Examining the Role of Ethnicity and Social Risk Factors in the Aetiology of Psychosis.

In this chapter, I will review the literature concerning associations between minority ethnicity (namely Black African and Black Caribbean) and social factors through the prism of migration (section 2.2.1), ethnic density (section 2.2.2) and social disadvantage (section 2.2.3) as predictors of psychotic disorder. To clarify, this is not a systematic review, since a number of such reviews and meta-analyses have been carried out recently (McGrath et al., 2004, Kirkbride et al., 2012a). Instead, a careful search was undertaken to scope the main literature on the topic. Using bibliographic databases (including Embase, Psycholnfo, Pubmed and Web of Science) and search terms including 'ethnicity', 'psychosis', 'African', 'Caribbean', 'migration', 'social' and 'density', I have predominately considered studies conducted in the last decade, as studies in this time period employed more robust

methodology than earlier studies. For example, many recent studies defined ethnicity according to standardised categories of self-ascription (e.g. ONS census categories) and many have moved away from crude comparison of 'Black and White' ethnicities. I have also only included studies that have employed population-based epidemiological methods including those that identified first episode psychosis patients in both inpatient and community settings in well-defined geographical areas. Forty-seven studies were included and their key findings are outlined in Tables 2.1, 2.2, and 2.3 below. I have acknowledged some older studies that measure the rates of psychosis among minority ethnic groups compared to the native population because these studies provide underpinning evidence and theories upon which the review in this chapter is based. Limitations of current evidence are presented in section 2.3 and finally a summary of the review is outlined in section 2.4.

2.2.1 Psychosis, ethnicity and migration

Migration is defined as the process of going from one country, region or place of residence to settle in another and there are three key stages to the process: pre-migration, migration and post-migration (Bhugra et al., 2011b). Migration has been recognised as an important factor in understanding the aetiology of psychosis for many decades. Evidence of its effects began to be observed in the 1930s when Odegaard examined the rates of hospital admissions among Norwegian migrants to Minnesota and found double the rate among the migrants compared to the indigenous Minnesotans (Odegaard, 1932). Since then, endeavours have continued in understanding the role of migration as a predictor of psychosis. Over the years, a number of hypotheses have been put forward as possible explanations for excess of psychosis in migrant groups. For example, Cochrane and Bal (Cochrane and Bal, 1987) argued for four key theories which research studies have attempted to test. They are reviewed below.

1. Countries of origin have higher rates of schizophrenia. A few older studies have compared rates of psychosis between Western and developing countries, but evidence shows no increase in rates in developing countries (Hickling and Rodgers-Johnson, 1995, Bhugra et al., 1996, Jablensky et al., 1992b, Hanoeman et al., 2002, Bhugra et al., 2000, Tafari et al., 1991). Yet the rate of psychosis among migrant groups to Western countries,

especially Black African and Black Caribbean (first and second generation), is higher (Fearon et al., 2006a, Kirkbride et al., 2012a). This may suggest that it is factors in the host counties that most explain the observed discrepancies.

2. Schizophrenia (or related precursors) predisposes individuals to migration; that is to say, individuals with schizophrenia or at risk of schizophrenia are more likely to migrate than people who do not. This is synonymous with the hypothesis of selection and drift, whereby people with schizophrenia either drift into or are selected into urban areas or lower social class groups because of disability or discrimination (March et al., 2008). Selten and colleagues (2002) tested this hypothesis by envisaging that the entire population of Surinam had migrated to the Netherlands. They calculated denominator population for the Surinamese by combining the Surinamese-born population in the Netherlands and that of Surinamese 1980 census figures and then compared incidence rates of schizophrenia to the native Dutch population. The result revealed a markedly increased relative risk of psychosis, adj. RR of 1.46 (95% Cl 1.35 – 1.57) among the Surinamese (Selten et al., 2002). However, Lundberg and colleagues (2007) in a Ugandan study sought to examine the impact of psychotic-like experiences on the likelihood of migration. They compared prevalence of psychosis between those planning to emigrate and those with no intention to emigrate and found no difference between the two groups (Lundberg et al., 2007).

3. Migration produces stress. Cochrane and Bal argued that difficulties and traumas surrounding migration and the social disadvantage of being a migrant (e.g. experiencing discrimination) increases the risk of psychosis (Cochrane and Bal, 1987). Odegaard (1932) demonstrated that psychosis is at its peak during the post-migration phase. Indeed, post-migratory experiences such as securing housing, building social relationships/networks and understanding the way of life in the host country, which are often negative (Johns et al., 2004), may also increase stress and the risk of developing psychosis. Historically, migrants tend to settle in city areas of host countries and much effort has been spent in understanding the relationship between inner-city living (urbanicity) and psychosis. It is well documented that social factors such as stressful life events, social isolation, higher levels of crime, higher unemployment rates and overcrowding are more commonplace in inner cities and these disadvantages are often experienced by migrants (Boydell et al., 2012, Morgan et al., 2009, Reininghaus et al., 2008).

4. Misdiagnosis of schizophrenia (psychosis). Arguments have been put forward that migrant patients have psychological conditions which have been misdiagnosed as psychosis, and that the diagnosis of schizophrenia has been used to impose social control on minority groups. Hickling and colleagues' (1999) study compared diagnoses using the Present State Examination (PSE), a research diagnostic interview, made independently by a British psychiatrist and a Jamaican psychiatrist and showed a 55% agreement between the British psychiatrist and the Jamaican psychiatrist in the proportion of black patients diagnosed with schizophrenia by the British psychiatrist or the Jamaican psychiatrist. However, there was disagreement in the diagnosis of affective psychoses, where the British psychiatrist were more likely to diagnosed patients with affective psychoses as having schizophrenia. However, the overwhelming evidence in the last few decades shows stubbornly higher rates of psychosis among Black African and Black Caribbean patients in the UK, Europe and the USA (Fearon et al., 2006a, Cantor-Graae and Selten, 2005, March et al., 2008, Veling, 2013). Many of these studies have employed standardised scales for assessing symptoms, and even blinded to ethnicity they found elevated rates of psychosis in the minority groups. Therefore, misdiagnosis is unlikely to play a major role, given numerous well designed studies consistently show a higher rate of psychosis in black Caribbean and black African minorities, and migrants; but other social and environmental factors may be more relevant.

Considering research in the past ten years, a number of studies have reported various ways of assessing the discrepancy of incidence rates of psychosis between ethnic groups as shown in Table 2.1. A diverse range of incidence rates per 100,000 person-years were reported with incidence rates (IR) ranging between 16.4 (Tarricone et al., 2012) and 50.0 (Cheng et al., 2011a). Only two studies reported IR by ethnic group. Fearon et al. (2006) reported IRs of 20.2 for White British, 140.8 for Black Caribbean and 80.6 for Black African (Fearon et al., 2006a), while Cheng and colleagues (2011) reported IRs of 47.0 for White British and 94.2 for Black groups (Cheng et al., 2011a, Fearon et al., 2006a). Meanwhile, all studies reporting incidence rate ratios (IRR) among migrant groups compared to the native populations found higher rates in the migrant groups ranging between 1.2 (Veling et al., 2006) (Cheng et al., 2011a, Coid et al., 2008b) and 18.2 (Lloyd et al., 2005). Considering the variation of effect by ethnic groups, IRR were reported as follows: for Black Caribbean, IRRs ranged from 2.1 (Cheng et al., 2011a) to 18.2 (Lloyd et al., 2005); Black African IRRs from 2.6

(Kirkbride et al., 2008) to 11.9 (Lloyd et al., 2005); Other White from 1.2 (Tortelli et al., 2014) to 2.3 (Lloyd et al., 2005). Similarly, higher relative risks were reported in migrant groups particularly among Moroccan, RR= 2.3 (Veling et al., 2006) to 7.8 (Zandi et al., 2010), and Surinamese, RR= 1.46 (Selten et al., 2002) to 1.5 (Zandi et al., 2010).

In addition, the fact that excess rates are not confined to first generation migrants as new evidence, suggests that both first and second generation migrants are at risk. Cantor-Graae and Selten (2005) in a systematic review concluded that personal or family migration is a risk factor of psychosis among both first and second generation migrants. The relative risk of psychosis is highest 4.8 (95% CI 3.7–6.2) in migrants of Black ethnicity compared with other migrant groups (Cantor-Graae and Selten, 2005). Bourque and colleagues (2011) in a metaanalysis, also reported IRR of 5.4 (95% CI 3.2-8.8) among second generation migrants categorised as 'Black'. Two other studies reviewed in Table 2.1 below have also reported incidence rate ratios by generational status, for example Coid and colleagues (2008) reported IRRs of 3.2 and 3.7 among first and second generation Black Africans respectively. Meanwhile among the Black Caribbean group, IRR of 3.2 and 1.6 were reported respectively. This suggests that rates of psychosis are elevated in both generations but appears to be diminishing in the among second generation Black Caribbean group. Further, Veling and colleagues reported IRRs from 2.3 to 5.8 among first and second generation Moroccans and 4.0 to 2.9 among first and second generation Surinamese(Veling et al., 2006). Similarly, rates appear to be reducing among the second generation Surinamese in the Netherlands.

Other demographic factors have also been considered in estimating incidence of psychosis; one study looked at age of migration and reported that young migrants (aged 20 and below) are particularly at risk (Veling et al., 2011). Gender has also been reported as a contributory factor with Asian women reported to be more at risk compared with White British women, while first generation Black African men were reported to be at higher risk compared with White British men (Coid et al., 2008b). Furthermore, variation in magnitude of risk by diagnosis were reported, for schizophrenia IRR ranged from 1.5 (Smith et al., 2006) to 9.1 (Fearon et al., 2006a); affective psychosis IRR from 1.7 (Fearon et al., 2006a) to 18.2 (Lloyd et al., 2005); depressive psychosis IRR from 3.1 to 5.6 (Fearon et al., 2006a).

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
Selten et al. (2002) (the Netherlands)	10 yr retrospective Hospital admissions	697 Surinamese 10,726 Dutch	The Bureau of Statistics Netherlands and The Bureau of Statistics Paramaribo Country of birth	ICD-8 and ICD-9 Clinical diagnosis only	Schizophrenia	Reference group: Native Dutch RR = 1.46(95% CI 1.35–1.57) Surinamese
Lloyd et al. (2005) (UK)	Incidence and Case-control (AESOP)	75 FEP	MRC Socio- demographic schedule (Mallett 1997). 2001 UK Census ONS classification Self-ascribed Other informants Researcher – ascribed (via case notes)	ICD-10 SCAN SCAN - IGC	Mania - with or without psychotic symptom Bipolar affective disorder	Reference group: White IRR = 12.3 (95% CI 8.3–17.6) all BME IRR = 18.2 (95% CI 10.8–28.8) African-Caribbean IRR = 11.9 (95% CI 5.9–21.3) Black African IRR = 12.7 (95% CI 4.6–27.8) mixed ethnicity IRR = 2.3 (95% CI 1.6–3.2) White group
Fearon et al. (2006) (UK)	Incidence and Case-control	568 FEP patients	MRC Socio- demographic	ICD-10 SCAN	All psychoses:	Adj. IR = 20.2 (95% 17.8 – 22.7) White British

 Table 2.1: Studies of incidence of psychosis, ethnicity and migration.

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
	(AESOP)		schedule (Mallett 1997) 2001 UK Census ONS classification Self-ascribed Other informants Researcher – ascribed (via case notes)	SCAN - IGC		Adj. IR = 140.8 (95% 114.4 – 167.2) Black Caribbean Adj. IR = 80.6 (95% 60.0 – 101.2) Black African Adj. IR = 31.6 (95% 16.7 – 46.5) Asian Adj. IR = 55.0 (95% 30.9 – 79.1) Other Adj. IR = 45.9 (95% 26.4 – 65.5) Mixed Adj. IR = 33.1 (95% 22.0 – 44.2) Other White Adjusted IRR for gender and age Reference group: White British Schizophrenia: IRR = 9.1 (95% CI 6.6 – 12.6) African Caribbean IRR = 5.8 (95% CI 3.9 – 8.4) Black African IRR = 1.4 (95% CI 0.7 – 3.1) Asian IRR = 2.6 (95% CI 1.9 – 6.5) Other IRR = 2.5 (95% CI 1.6 – 3.9) White
Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
----------------------	--------------	--------	--	-------------------------------------	------------	---
						Manic psychosis: IRR = 8.0 (95% Cl 4.3 - 14.8) African Caribbean IRR = 6.2 (95% Cl 3.1 - 12.1) Black African IRR = 2.7 (95% Cl 0.9 - 7.6) Asian IRR = 3.0 (95% Cl 0.9 - 10.0) Other IRR = 6.2 (95% Cl 2.6 - 15.0) Mixed IRR = 1.7 (95% Cl 0.6 - 4.3) White Other Depressive psychosis: IRR = 3.1 (95% Cl 0.6 - 4.3) White Other Depressive psychosis: IRR = 3.1 (95% Cl 1.5 - 6.1) African Caribbean IRR = 2.1 (95% Cl 0.9 - 5.0) Black African All Psychoses IRR = 6.7 (95% Cl 5.4 - 8.3) African-Caribbean
						IRR = 4.1 (95% Cl 3.2 – 5.3) Black

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
						African IRR = 1.5 (95% CI 0.9 – 2.4)Asian IRR = 2.6 (95% CI 1.7 – 3.9) Other IRR = 2.7 (95% CI 1.8 – 4.2) Mixed IRR = 1.6 (95% CI 1.1 – 2.2) White Other
						IRR = 3.0 (95% CI 1.3 - 7.1) Asian IRR = 5.6 (95% CI 2.5 - 12.4) Other IRR = 4.0 (95% CI 1.6 - 10.2) Mixed IRR = 1.3 (95% CI 0.5 - 3.2) White Other
						Other psychosis: IRR = 5.5 (95% Cl 3.8 - 8.0) African Caribbean IRR = 2.7 (95% Cl 1.6 - 4.5) Black African IRR = 0.6 (95% Cl 0.2 - 1.9) Asian IRR = 0.3 (95% Cl 0.1 - 2.2) Other IRR = 1.3 (95% Cl 0.5 - 3.5) Mixed IRR = 0.8 (95% Cl 0.4 - 1.7) White Other

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
Kirkbride et al. (2006) (UK)	Incidence and Case-control (AESOP)	568 FEP patients	MRC Socio- demographic schedule (Mallett 1997). 2001 UK Census ONS classification Self-ascribed Other informants Researcher – ascribed (via case notes)	ICD-10 SCAN SCAN- IGC	All psychoses Schizophrenia: Substance- induced psychoses: Affective psychosis: Other psychosis:	IRR = 2.3 (95% CI 1.7 – 3.1) Schizophrenia IRR =3.6 (95% CI 3.0 – 4.2) All psychoses among BME No ethnic difference in affective psychoses Variation by study centre IRR = 49.4 (95% CI 43.6 – 55.3) Southeast London IRR = 23.9 (95% CI 20.6 – 27.2) Nottingham IRR = 20.4 (95% CI 15.1 – 25.7) Bristol
Veling et al. (2006) (The Netherlands)	Cohort incidence study	197 FEP patients	The Municipality of The Hague classification Country of birth Country of parents' birth	CASH IRAOS DSM-IV	Schizophrenia	First generation Adj. IRR = 2.3 (95% Cl 1.7 – 3.0) Moroccans Adj. IRR = 4.0 (95% Cl 2.5 – 6.3) Surinamese Adj. IRR = 1.4 (95% Cl 0.7 – 2.6) Turks Adj. IRR = 1.2 (95% Cl 0.5 – 2.5)

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
						Immigrants from other non- Western countries
						Second generation Adj. IRR = 5.8 (95% CI 2.9 – 11.4) Moroccans Adj. IRR = 2.9 (95% CI 1.6 – 5.0) Surinamese Adj. IRR = 2.3 (95% CI 1.0 – 5.4) Turks
						Adj. IRR = 3.5 (95% Cl 1.8 – 6.8) Immigrants from other non- Western countries
Veling et al. (2007) (the Netherlands)	Cross-sectional survey	361 first contact patients	The Municipality of the Hague classification Country of birth Country of parents' birth	CASH DSM-IV IRAOS	Depressive disorder Bipolar disorder Brief psychotic disorder	Positive symptoms Moroccan vs. native Dutch Adj. OR = 3.44 (95% Cl 1.49 – 7.93) Persecutory delusion Adj. OR = 3.14 (95% Cl 1.43 – 6.87) Bizarre behaviour Adj. OR = 3.08 (95% Cl 1.36 – 7.00) Visual hallucinations Adj. OR = 0.29 (95% Cl 0.08 – 1.06) Religious delusions

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
						Negative symptoms Moroccan vs. native Dutch Adj. OR = 2.69 (95% CI 1.23 – 5.90) Anhedonia Adj. OR = 2.61 (95% CI 1.12 – 6.09) poverty of speech Adj. OR = 2.31 (95% CI 1.04 – 5.12) self-care
Coid et al. (2008) (UK)	Population- based incidence survey (ELFEP)	484 FEP	2001 UK ONS census	SCAN - IGC DSM- IV	Non-affective psychoses	Incidence rates Adj. IR = 20.9 (95% CI 16.2 – 25.6) White British Adj. IR = 42.4 (95% CI 28.6 – 56.2) White Other Adj. IR = 90.8 (95% CI 67.8 – 113.8) Black Caribbean Adj. IR = 73.6 (95% CI 54.4 – 92.7) Black African Adj. IR = 37.7 (95% CI 28.1 – 47.2) Asian Adj. IR = 25.1 (95% CI 13.7 – 36.4) Other Incidence rate ratios IRR = 4.2 (95% CI 3.0 – 5.8) Black Caribbean

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
					Affective psychoses	IRR = 3.4 (95% CI 2.4 – 4.7) Black African IRR = 1.6 (95% CI 1.1 – 2.4) 2^{nd} generation Black Caribbean IRR = 3.2 (95% CI 2.3 – 4.6) 1^{st} generation Black African IRR = 3.7 (95% CI 2.2 – 6.4) 2^{nd} generation Black African IRR = 2.3 (95% CI 1.0 – 5.3) Asian women IRR = 3.1 (95% CI 2.0 – 4.7) 1^{st} generation Black African men
Kirkbride et al (2008) (UK)	Population- based incidence survey (ELFEP)	484 FEP	2001 UK ONS census Self-ascribed Other informants Researcher – ascribed (via case notes)	SCAN - IGC DSM-IV	Schizophrenia Affective psychoses	IRR = 3.1 (95% CI 2.1 – 4.5) Black Caribbean IRR = 2.6 (95% CI 1.8 – 3.8) Black African IRR = 3.1 (95% CI 1.2 – 8.1) Pakistani IRR = 2.3 (95% CI 1.1 – 4.7) Bangladeshi (women). IRR = 7.7 (95% CI 3.2 – 18.8) Mixed White and Black

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
						Caribbean IRR = 2.1 (95% CI 1.2 – 3.8) White Other
Kirkbride et al. (2009) (UK)	Incidence study over 3 time periods	347 FEP	UK Census ONS- 1981, 1991, 2001 classification Self-ascribed Other informants Researcher – ascribed (via case notes)	PSE PPHS SCAN SANS ICD-9 ICD-10	Schizophrenia Affective psychosis Substance – induced psychoses Affective psychosis	IR of all psychosis per 100,000 IR = 23.4 (95% CI 18.6 – 28.2) 1978-80 IR = 26.0 (95% CI 21.2 – 30.8) 1992-94 IR = 27.1 (95% CI 22.3 – 31.9) 1997-99 Linear change in incidence over time by diagnosis IRR = 1.15 (95% CI 1.05 – 1.25) substance induced psychosis. IRR = 0.98 (95% CI 0.96 – 1.00) schizophrenia IRR = 1.00 (95% CI 0.98 – 1.03) affective psychoses
Cheng et al. (2011) (UK)	Longitudinal cohort study CAMEO study	285 FEP	UK ONS Census 2001 and 2002 estimates Self-ascribed	ICD-10 PANSS	All psychotic disorders	IR = 50.0/100,000 overall IR = 47.0 (95% CI 41.0 – 53.9) White British IR = 49.4 (95% CI 32.8 – 71.4) non- British white IR = 94.2 (95% CI 47.0 – 168.5) Black

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
						IR = 42.5 (95% CI 28.0 – 61.9) Other Adj. IRR = 2.1 (95% CI 1.1 – 3.8) Black Adj. IRR = 1.1 (95% CI 0.8 – 1.7) White non-British Adj. IRR = 0.9 (95% CI 0.6 – 1.4) Mixed and other
Veling et al. (2011) (the Netherlands)	7-yr incidence study	173 immigrant 119 second- generation citizens 226 Dutch citizens	The Municipality of the Hague Native First generation Country of birth Country of parents' birth	CASH DSM-IV IRAOS	All psychotic disorders	IRR among non-Western Immigrants by age of migration IRR = 3.0 (95% Cl 2.1 – 4.2) Age 0- 4 IRR =2.3 (95% Cl 1.6 – 3.3) Age 5- 9 IRR =1.5 (95% Cl 1.0 – 2.3) Age 10-14 IRR = 1.85 (95% Cl 1.35 – 2.53) Age 15-19 IRR = 1.40 (95% Cl 0.96 – 2.03) Age 20-24 IRR = 1.14 (95% Cl 0.71 – 1.83) Age 20-29 IRR = 1.82 (95% Cl 1.43 – 2.32) Second generation citizens

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
Tarricone et al. (2012) (Italy)	Prospective cohort study – 8-yr	163 FEP	Municipality Registry Migrant vs. Native Country of birth	ICD-10 ICG-SCAN	Affective psychosis Non-affective psychosis Substance- induced psychosis Schizophrenia	Lower IR of FEP compared to other study centres IR = 16.4/100,000 (95% CI 13.9 – 18.9) IR higher in men, migrants and young people IRR = 2.53 (95% CI 2.19 – 2.89) all psychosis – migrants IRR = 3.38 (95% CI 2.98 – 3.79) non-affective psychosis – migrants IRR = 4.04 (95% CI 3.55 – 4.53) SCZ – migrants
Tortelli et al.	Hospital cross-	258 patients	2008 French census	ICD-10	Non-affective	
(2014) (France)	sectional study	– hospital admission	categories (natives vs. migrants) Country of birth	clinical diagnosis	psychosis Affective psychosis	Adj. IR = 28.0 (95% CI 12.5 – 62.5) Overall Adj. IR = 17.5 (95% CI 6.5 – 48.0) Natives Adj. IR = 22.0 (95% CI 9.0 – 54.0) Europe Adj. IR = 125.0 (95% CI 9.0 – 54.0) Europe Adj. IR = 125.0 (95% CI 84.5 – 184.5) Sub-Saharan Africa Adj. IR = 25.0 (95% CI 11.0 – 59.0) North Africa IRR = 2.9 (95% CI 0.9 – 9.8)

Authors (Country)	Study Design	Sample	Measure of population denominator / Ascertainment of ethnicity	Scale for measuring psychosis	Outcome(s)	Main findings
						Migrants IRR = 7.1 (95% CI 2.3 – 21.8) sub- Saharan African IRR = 1.2 (95% CI 0.35 – 5.1) European IRR = 1.4 (95% CI 0.4 – 5.6) North African

IRR, incidence rate ratio. IR, incidence rate. RR, relative risks. OR, odds ratio. MRC Socio-demographic schedule (Mallett, 1997). DUP, Duration of untreated psychosis. ICC, interclass correlation coefficient. ONS, Office of National Statistics. BME, Black and minority ethnic. HSE, health survey England. CASH, comprehensive assessment of symptoms and history. IROAS, instrument for the retrospective assessment of the onset of schizophrenia. SCAN, schedule of clinical assessment in neuropsychiatry. SCAN - IGC, schedule of clinical assessment in neuropsychiatry - item group checklist SCAN. CIDI, composite international diagnostic interview. PDI, Peters et al. 1999 delusions inventory. ICD, international classification of diseases. SANS, schedule for the assessment of negative symptoms. PPHS, personal and psychiatric history schedule. PSE, present state examination. CASH-CS, comprehensive assessment of symptoms and history – cultural sensitive.

2.2.2 Ethnic density, neighbourhood and psychosis

A few years after Odegaard's findings, Faris and Dunham (1939) explored the relationship between the spatial distribution of psychosis and social organisation. They examined ethnic density (i.e. proportion and concentration of a given ethnic group in a defined area) and social class and argued that social isolation and language barriers among different communities could explain higher rates of psychosis (Faris, 1939). Their Chicago study revealed that the least socially organised inner city area had higher rates of schizophrenia. Faris and Dunham's work inspired interest in the investigation of ethnicity, ethnic density, social class and mental illness – psychosis in particular. As such, much attention has been given to exploring these areas in recent years. Table 2.2 shows studies carried out in the last ten years investigating ethnic density effects on rates of psychosis.

From the table, Kirkbride and colleagues (2007) reported a variance in the incidence of schizophrenia (23% (95% CI 9.9 – 42.2)) and non-affective psychoses (15% (95% CI 2.9 – 55.0)) at both individual and area levels. They argued that the variance could be attributed to neighbourhood-level risk factors and found that social capital (measured by voter turnout) was associated with the raised incidence (Kirkbride et al., 2007). Similar trends were observed in the Netherlands, where Veling et al. (2008) compared incidence rates for native Dutch and immigrants; they reported an IRR of 2.36 (95% CI 1.89 – 2.95) in low-ethnic density neighbourhoods whereas this was not so in high-ethnic density areas (IRR 1.25 (95% CI 0.66 – 2.37)) (Veling, Susser et al. 2008). Schofield et al. (2011) explored the ethnic density effect in explaining raised incidence among minority ethnic populations in a UK inner city area. They found that in the neighbourhoods where Black people comprised more than 25% of the population, there was no statistically significant ethnic difference, but differences were found in areas where Black people were less represented; the OR was 2.88 (95% CI 1.89 – 4.39) (Schofield et al., 2011). This is consistent with more recent findings, for example Das-Munshi (2012) and colleagues reported an overall OR of 1.07 (95% CI 1.01 -1.14) for all minority ethnic groups in areas of lower own-group density and that residents were more likely to experience greater social adversity, which is consistent with reports of psychotic experiences. They argued that ethnic density acts as a buffer against the development of psychotic experiences (Das-Munshi et al., 2012). In a narrative review of ethnic density and impact on mental disorders, Shaw et al. (2012) reported a consistent

association between ethnic density and psychosis and even stronger associations between minority status, area-level material deprivation and low socioeconomic status. They concluded that there might be potential psychosocial benefits drawn from living among people of the same ethnic group (Shaw et al., 2012). Termorshuizen et al. (2014) echoed this phenomena when they found a decreasing rate ratio of non-affective psychotic disorder with increasing minority ethnic density (from RR = 2.36 to 1.24) in non-Western immigrants and (from RR = 1.63 to 1.01) in Western immigrants compared with native Dutch patients (Termorshuizen et al., 2014).

Several studies of ethnic density have used individual-level variables such as ethnicity, age and gender. However, more recently, Kirkbride and colleagues (2014) investigated whether incidence of psychosis varied at neighbourhood level, looking at absolute neighbourhood level factors (deprivation, social fragmentation, and social cohesion) and relative factors (inequality, ethnic density and ethnic separation). Using a Bayesian hierarchical spatial model, they reported associations with level of income inequality (RR = 1.25; 95% CI 1.04– 1.49), absolute deprivation (RR = 1.28; 95% CI 1.08 – 1.51) and population density (RR= 1.18; 95% CI 1.00 – 1.41). They also found that the neighbourhood ethnic composition effects were associated with the incidence of non-affective psychosis for people of Black Caribbean and Black African ethnicities (Kirkbride et al., 2014). Table 2.2: Studies of psychosis and ethnic density.

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring	Type of Psychosis/Diagnosis	Main findings
			-	psychosis		
Kirkbride et al.	Incidence and	218 FEP	MRC Socio-	ICD-10	Non-affective	23% of variance in
(2007) (UK)	Case-control		demographic	SCAN	psychoses	incidence of schizophrenia
	(AESOP)		schedule (Mallett	SCAN - IGC	Schizophrenia	attributed to
			1997)		Other non-affective	neighbourhood level risk
			2001 LIK Consus ONS		psychoses	Adi ICC = 22 5% (95% CI
			classification			9.9 – 42.2) SCZ
						Adj. ICC = 15.3% (95% CI
			Self-ascribed			2.9 – 55.0) Non-affective
			Other informants			psychoses
			Researcher – ascribed			
			(via case notes)			
Veling et al.	Prospective	226 Dutch	The Municipality of	CASH	Schizophrenia	Adj. IRR = 2.36 (95% Cl
(2008) (the	incidence	Natives	the Hague	DSM-IV	Major depressive	1.89 – 2.95) Low-ethnic
Netherlands)	study	240	Native	IRAOS	with psychosis	density neighbourhoods
		immigrants	First generation		Bipolar disorder	Adj. IRR = $1.25 (95\% \text{ Cl})$
		FEP	Country of hirth		Delusional disorder	0.66 – 2.37) High-ethnic
			Country of paronts'			density areas
			birth			
Schofield et al.	Clinical record	60971 records	2001 UK Census ONS	First diagnosis	Non-organic	OR = 2.88 (95% Cl 1.89 –
(2011) (UK)	survey	of Black and	classification	of psychotic	psychosis	4.39) Black patients in less
		White	LSOA	illness		represented areas
		patients		GP area codes		OR = 5.24 (95% Cl 1.95 –
			Self-ascribed			14.07) Black patients in

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis/Diagnosis	Main findings
						lowest density quintiles
Das-Munshi et al. (2012) (UK)	Cross- sectional survey Ethnic density	351 people	Self-ascribed Generational status determined by country of birth MSOA	PSQ	Psychotic symptoms: Hallucinations, Delusions, Hypomania	For 10% point reduction in own-group density OR = 1.07 (95% Cl 1.01 – 1.14) combined minority ethnic group Adj. OR = 0.91(95% Cl0.72 – 1.14) White British Adj. OR = 5.44 (95% Cl 0.77 – 38.3) Irish Adj. OR = 1.05 (95% Cl 0.68 – 1.61) Black Caribbean Adj. OR = 1.26 (95% Cl 1.00 – 1.60) Bangladeshi Adj. OR = 1.38 (95% Cl 1.02 – 1.86) Indian Adj. OR = 1.17 (95% Cl 0.95 – 1.45) Pakistani Association of own group density and social disadvantage Adj. OR = 2.26 (95% Cl 1.62 – 3.14) interpersonal racism

Authors	Study Design	Sample	Ascertainment of	Scale for	Type of	Main findings
(Country)			ethnicity	measuring	Psychosis/Diagnosis	
				psychosis		
						Adj. OR = 1.46 (95% Cl
						1.06 – 2.00) Work-related
						discrimination
						Adi. ICC = 0.06 (95% CI
						0.01 – 0.20) variation in
						area-level reporting
						psychotic experiences
						ICC = 0.03 (95% CI 0.00 –
						0.30) individual-level
						factors accounting for
						variation (adjusted for
						discrimination, social
						support)
Bhavsar et al.	Historical	405 First	IMD	OPCRIT	Schizophrenia	IRR = 1.03 (95% CI 1.004 –
(2014) (UK)	case register	episode				1.04) Electoral ward
	cohort study	schizophrenia				deprivation
	(LEO Case					IRR = 1.04 (95% Cl 1.02 –
	register)					1.06) Super output
						deprivation
Kirkbride et al.	Population-	427 FEP	2001 UK Census ONS	DSM-IV	Non-affective	RR = 1.25 (95% Cl 1.04 –
(2014) (UK)	based		classification	SCAN	psychosis	1.49) multiple deprivation
	incidence				Affective psychosis	RR = 1.28 (95% CI 1.08 –
	survey		Self-ascribed			1.15) income inequality
	ELFEP		Other informants			deprivation
			Researcher – ascribed			RR = 1.18 (95% CI 1.00 –

Authors	Study Design	Sample	Ascertainment of	Scale for	Type of	Main findings
(Country)			ethnicity	measuring	Psychosis/Diagnosis	
			(via case notes)			1.41) Population density
						Association with own group density RR = 1.54 (95% CI 1.12 – 2.03) lower density of own group RR = 0.70 (95% CI 0.48 – 0.99) higher density of own group
Termorshuizen et al. (2014) (the Netherlands)	Retrospective Case register cohort	2,064 patients	Population register of Statistics Netherlands (Central Bureau voor de Statistics) Native. Migrant	DSM-IV	Non-affective psychosis	High vs. low density Adj. RR = 2.36 (95% Cl 2.02 – 2.76) < 21.8% ethnic density Adj. RR = 1.71 (95% Cl 1.14 – 2.07) 21.8% to 43.3% ethnic density Adj. RR = 1.24 (95% Cl 0.96 -1.61) > 43.3% ethnic density RR decreased from 2.36 to 1.24 – non-Western immigrants RR from 1.63 to 1.01 – Western immigrants

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis/Diagnosis	Main findings

IRR, incidence rate ratio. RR, relative risks. OR, odds ratio. MRC Socio-demographic schedule (Mallett, 1997). ICC, interclass correlation coefficient. IMD, index of multiple deprivation (2004). MSOA, middle super output area (ONS 2001). LSOA, lower super output area. OPCRIT, operational criteria checklist for psychotic and affective illness (McGuffin, et al. 1991). DSM-IV, diagnostic and statistical manual of mental disorders (American Psychiatric Association and Association, 1980). GP, general practitioner.

2.2.3 Social disadvantage and psychosis

Evidence of an association between social disadvantages and increased risk of psychosis is well documented (Morgan et al., 2009, Stilo et al., 2013, Cooper et al., 2008). As such, the relationships between current or long-term social disadvantage indicators (such as unemployment, living alone or childhood adversity) and psychosis have been consistently reported over the last decade. Table 2.3 shows recent work on this topic. Different dimensions of adversity have been reported, but only a few studies looked at these specifically in relation to ethnicity. For instance, the odds of unemployment among psychosis cases compared to controls ranged between OR 3.64 (Morgan et al., 2008a) to 20.92 (Mallett et al., 2002). This was highest among Black Caribbean patients, whose OR was 20.92, compared to White British patients, whose OR was 3.13 to 7.50 (Mallett et al., 2002, Morgan et al., 2008a). Furthermore, Boydell and colleagues (2012), in an incidence study of first episode psychosis, calculated the standardised incidence rates (SIR) among unemployed patients and found a higher incidence of psychosis among the unemployed Black Caribbean (SIR = 12.05) compared to White patients (SIR = 11.75) but slightly lower among Black African patients (SIR = 6.78) (Boydell et al., 2012). Ramsay and colleagues (2012) also reported a higher proportion of unemployment among FEP patients, at 65.0% (Ramsay et al., 2012). In addition, indicators such as living alone have been reported to be higher among psychosis cases compared to controls; for example, in the last decade the reported odds ratios ranged from 0.59 (Mallett et al., 2002) to 4.55 (Stilo et al., 2013).

As can be seen in Table 2.3, nine of the sixteen studies included used cross-sectional designs which mean that direction of causality may not be established. For example, it is possible that individuals were unemployed before onset of psychosis or vice versa.

Perceived discrimination (Cooper et al., 2008) and strong identification with an ethnic minority group (Reininghaus et al., 2010a) have also been reported to be associated with increased risk of psychosis in Black and minority ethnic (BME) groups. For example, Reininghaus and colleagues found that the higher the level of ethnic identification, the higher the odds of psychosis (Reininghaus et al., 2010a). Meanwhile, Morgan and Fearon (2007) argued that psychosis is an outcome of a series of interactions (biological and social) over the life course. Negative social experiences among the migrant groups such as those

noted above may be major factors in understanding the excess of psychosis among African Caribbean patients (Morgan and Fearon, 2007). This argument was crystallised when the same authors investigated the prevalence and social correlates of psychotic-like experiences in a general population sample of Black and White British participants; they reported a 19% prevalence of unusual (psychosis-like) experiences in the sample and these were more likely to be reported among the Black participants, with an OR of 2.08 among the Black Caribbean participants and an OR of 4.59 among the Black Africans (Morgan et al., 2009).

More recently, Morgan et al. (2014) sought to investigate the associations between two putative risk factors, i.e. childhood and adult disadvantage, to further our understanding of the aetiology of psychosis. The findings were consistent with previous studies as the authors found that although complex, the combination of both predictors have a potential to set some individuals on a predominantly socio-developmental path to psychosis (Morgan et al., 2014). The contributory mediating factors they found included parental separation, education, adult disadvantage, self-esteem and parental death (Morgan et al., 2014).

Furthermore, there is growing evidence that childhood adversity is a putative risk factor for psychosis (see Table 2.3). Matheson and colleagues (2013) in a recent meta-analysis reported that increased rates of childhood adversity were more likely to be reported among patients with schizophrenia compared to controls (OR = 3.60; 95% Cl 2.08 – 6.23) (Matheson et al., 2013). More specifically, Gayer-Anderson (2014) found increased risk of psychosis associated with the following experiences of abuse: severe household discord (adj. OR = 5.06;; 95% Cl 2.11 – 12.09), psychological abuse (adj. OR = 6.27;; 95% Cl 2.48 – 15.84) and sexual abuse (adj. OR = 2.62; 95% Cl 1.06 - 6.48) (Gayer-Anderson, 2013). Similarly, Fisher et al. (2010) reported that patients with psychosis were three times more likely to report physical abuse compared to controls. Meanwhile, studies reporting odds ratios (OR) of parental separation (both or one parent) among cases and controls have found higher odds in cases, particularly in relation to ethnicity. Parental separation among Black African and Caribbean patients with psychosis has demonstrated effect sizes ranging from 2.16 (Morgan et al., 2014) to 5.0 (Mallett et al., 2002). For parental death among African and Caribbean patients the odds ratios range from 1.62 to 4.71 (Morgan et al., 2007).

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis	Main findings
Mallett et al. (2002) (UK)	Case-control study	100 cases 100 controls	UK 2001 ONS Census classifications Self-ascribed	PSE CATEGO	Schizophrenia	Unemployment OR = 7.50 (95% CI 1.69 – 33.47) White British OR = 20.92 (95% CI 2.04 – 214.29) African- Caribbean OR = 6.39 (95% CI 0.54 – 76. 14) Asian OR = 5.5 (95% CI 2.59 – 11.68) cases vs. controls Living alone OR = 0.59 (95% CI 0.12 – 2.84) White British OR = 2.16 (95% CI 0.13 – 15.58) African- Caribbean OR = 3.35 (95% CI 0.13 – 86.72) Asian OR = 1.75 (95% CI 0.86 – 3.56) cases vs. controls Separation from both parents OR = 5.00 (95% CI 1.09 – 22.82) African-Caribbean OR = 0.33 (95% CI 0.35 – 3.20)Asian
Fossion et al. (2004) (the	Cross-sectional survey	341 Moroccan patients	Not recorded		Schizophrenia	Belgian v. Moroccan OR = 1.45 (95% Cl 1.16 – 1.81) Lives with parent

 Table 2.3: Studies of psychosis and social disadvantage and ethnicity.

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring	Type of Psychosis	Main findings
Netherlands)	Hospital admission data	341 Belgian patients				OR = 0.77(95% CI 0.62 – 0.96) Registered with psychiatrist OR = 0.72 (95% CI 0.52 – 0.98) Salaried employment OR = 0.70 (95% CI 0.52 – 0.94) Alcohol related problems
Morgan et al. (2007) (UK)	Case-control survey	390 FEP cases 391 controls	MRC Socio- demographic schedule (Mallett 1997) UK 2001 ONS classification Self-ascribed Other informants Researcher – ascribed (via case notes)	ICD-10 SCAN SCAN-IGC	Affective psychosis Non-affective psychosis	Separation Adj. $OR = 2.45 (95\% CI 1.66 - 3.59)$ long-term separation overall Adj. $OR = 2.30(95\% CI 1.38 - 3.86)$ White British Adj. $OR = 2.23 (95\% CI 1.20 - 4.16)$ White British Father only Adj. $OR = 2.71 (95\% CI 1.20 - 4.16)$ White British Father only Adj. $OR = 2.71 (95\% CI 1.20 - 4.16)$ White British Father only Adj. $OR = 2.71 (95\% CI 1.15 - 6.39)$ WB both parents Adj. $OR = 2.92 (95\% CI 1.36 - 6.28)$ Black Caribbean Adj. $OR = 4.73 (95\% CI 1.36 - 6.28)$ Black Caribbean Father only Adj. $OR = 4.73 (95\% CI 0.74 - 7.64)$ Black Caribbean both parents Adj. $OR = 1.47 (95\% CI 0.27 - 7.93)$ Black African Death $OR = 3.06 (95\% CI 1.34 - 7.00)$ overall

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis	Main findings
						Adj. OR = 4.00 (95% CI 1.41 – 11.32) White British Adj. OR = 4.71 (95% CI 0.72 – 30.91) Black Caribbean Adj. OR = 1.62 (95% CI 0.27 – 9.85) Black African
Cooper et al. (2008) (UK)	Incidence and case-control (AESOP)	197 cases 285 controls	MRC Socio- demographic schedule (Mallett 1997) 2001 UK Census ONS classification Self-ascribed Other informants Researcher – ascribed (via case notes)	ICD-10 SCAN SCAN-IGC	Psychosis	Incidence of psychosis OR = 4.7 (95% CI $3.1 - 7.2$) Black ethnic group Adj. OR = 3.0 (95% CI $1.6 - 5.4$) Black ethnic group Case status vs. Black ethnic group Adj. OR = 4.1 (95% CI $2.5 - 6.8$) overall perception of disadvantage Adj. OR = 6.9 (95% CI $3.7 - 13.0$) due to skin colour Adj. OR = 5.0 (95% CI $3.0 - 8.3$) due to culture Adj. OR = 4.6 (95% CI $2.8 - 7.7$) due to social class
Morgan et al. (2008) (UK)	Incidence and case-control (AESOP)	390 cases 391 controls	MRC Socio- demographic schedule (Mallett 1997)	ICD-10 SCAN SCAN-IGC	Affective psychosis Non-affective psychosis	Education – no qualification Adj. OR = 4.59 (95% Cl 1.03 – 20.38) Black Caribbean Adj. OR = 2.95 (95% Cl1.51 – 5.76) white British

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis	Main findings
			Self-ascribed Other informants Researcher – ascribed (via case notes)			Employment – unemployed Adj. OR = 3.64 (95% Cl 1.81 – 9.20) Black Caribbean Adj .OR = 3.13 (95% Cl 1.85 – 5.31) White British Living arrangements – lives alone Adj. OR = 3.27 (95% Cl 1.42 – 7.52) Black Caribbean Adj. OR = 2.70 (95% Cl 1.69 – 4.32) White British
Morgan et al. (2009) (UK)	Cross-sectional survey (sub- sample of AESOP controls)	372 Controls – healthy volunteers	MRC Socio- demographic schedule (Mallett 1997) 2001 UK Census ONS classification Self-ascribed Other informants Researcher – ascribed (via	PSQ	Psychotic symptoms	prevalence of psychosis = 19% OR = 2.08 (95% Cl 1.08 – 3.89) Black Caribbean OR = 4.59 (95% Cl 1.69- 12.46) Black African OR = 3.83 (95% Cl 1.03 – 14.29) Other Age OR = 2.05 (95% Cl 1.08 – 3.89) 16- 15-year-olds Housing status OR = 2.26 (95% Cl 1.25 – 4.07) Rented Living circumstances

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis	Main findings
			case notes)			OR = 2.11 (95% CI 1.17 – 3.79) Lives alone Relationship status OR = 1.85 (95% CI 1.02 – 2.39) Never in long-term (1+year) relationship
Reininghaus et al. (2010) UK	Case control (AESOP)	139 Cases 243 controls	MRC Socio- demographic schedule (Mallett 1997) 2001 UK Census ONS classification	ICD-10 SCAN SCAN-IGC	Non-affective psychosis Affective psychosis	Overall Ethnic identity Adj. OR = 1.08 (95% Cl 0.89 – 1.30) White British Adj. OR = 1.45 (95% Cl 1.08 – 1.96) BME Strong vs. weak ethnic identification Adj. OR = 1.60 (95% 0.85 – 2.99) White British Adj. OR = 2.73 (95% Cl 1.22 – 6.12) BME
Boydell et al. (2012) (UK)	Case register cohort	178 FEP	2001 UK Census classification	ICD-9 OPCRIT RDC	Schizophrenia Psychosis mania Psychotic depression	Standardised incidence ratio for psychosis among unemployed patients SIR = 11.75 (95% CI 6.42 – 19.71) White British SIR = 12.05 (95% CI 7.87 –17.65) Black Caribbean SIR = 6.78 (95% CI 4.30 –10.17) Black African SIR = (95% CI)
Morgan et al. (2014)	Case-control AESOP	390 cases 391	MRC Socio- demographic	ICD-10 SCAN	Affective psychosis	Parental separation effect by ethnicity

Authors (Country)	Study Design	Sample	Ascertainment of ethnicity	Scale for measuring psychosis	Type of Psychosis	Main findings
(UK)		controls	schedule (Mallett 1997) Self-ascribed Other informants Researcher – ascribed (via case notes)	SCAN-IGC	Non-affective psychosis	White British Adj. OR = 2.98 (95% Cl $1.59 - 4.45$) Direct Adj. OR = 1.86 (95% Cl $1.10-3.14$) Total indirect Adj. OR = 5.53 (95% Cl $2.83 - 10.84$) Total Black Caribbean Adj. OR = 2.16 (95% Cl $1.07 - 4.36$) Direct Adj. OR = 2.79 (95% Cl $1.06 - 7.35$) Total indirect Adj. OR = 6.03 (95% Cl $1.87 - 19.43$) Total Black African Adj. OR = 2.14 (95% Cl $0.47 - 9.81$) Direct Adj. OR = 0.92 (95% Cl $0.03 - 27.44$) Total indirect Adj. OR = 1.97 (95% Cl $0.05 - 81.04$) Total

RR, relative risks. OR, odds ratio. Adj. OR, Adjusted odds ratio. MRC Socio-demographic schedule (Mallett, 1997). SIR, standardised incidence ratio. PE, psychotic experience. GAF, global assessment of functioning. CIDI, composite international diagnostic interview. SAPS, scale of assessment for positive symptoms. SANS, scale of assessment for negative symptoms. SCID, structured interview for DSM-IV. SOS, symptom onset in schizophrenia. PSE, present state examination. RDC, research diagnostic criteria. NOS, Nottingham Onset Schedule (Sign et al. 2005). PSQ, psychosis screening questionnaire.

2.3 Synthesis of current evidence

Although the last decade has seen improvements in study designs measuring incidence rates of psychosis, from the dimensions of ethnicity, ethnic density and social disadvantage, there remain some methodological issues for consideration For example, twenty-two population-based studies employed case-control or cohort study designs which have many benefits in their own right such as the measure of exposure preceding the measure of psychosis (cohort design), identifying cases and controls from the same population (casecontrol) and being able to control for a range of possible confounders. Several of the studies included in this review have also employed semi-structured diagnostic interviews to elicit psychotic symptoms and psychotic experiences. There has also been better comparative denominator population data that enables incidence rates to be measured more accurately (e.g. UK Census data ONS 2011 and the Dutch Bureau of Statistics). However, this review has highlighted some study design and methodological issues in studies estimating the incidence of psychosis which have not been fully addressed in previous research on aetiology of psychosis and ethnicity. Three key areas that have impacted on findings from previous research include: a) definition of ethnic groups; b) sampling and design; and c) source used to collect data.

2.3.1 Definition of ethnic groups

Apart from the two large UK epidemiological studies, namely Aetiology and Ethnicity of Schizophrenia and Other Psychoses (AESOP) and the East London First Episode Psychosis (ELFEP) (Fearon et al., 2006b, Coid et al., 2008a), Black African and Black Caribbean patients are often amalgamated into one ethnic group (African-Caribbean) meaning that generalizable conclusions cannot be drawn about the true rates of psychosis among these two very different groups. It is important to treat these groups separately. The classification of ethnicity has evolved over time and it is argued to be a social construct (Mateos et al., 2009). For example, there are immigration status differences between the African and Caribbean groups in the UK. Emigration of Black Africans is more recent compared to Black Caribbeans who have mostly settled in the UK since the post– Second World War period (Mason, 1995). Other key features that differentiate Black African and Black Caribbean groups include diverse histories, culture and self-identity (Kaneshiro et al., 2011, Mason, 1995). This issue of crude classification of ethnic groups is also notable in research studies in

the United States of America, where for example 'Black' people are referred to as 'African-American'. However, this broad categorisation is not limited to research; the United States (US) Office of Management and Budget use the same classification for population data collection where only five categories of ethnicity are available for citizens to self-ascribe (Kaneshiro et al., 2011). The amalgamation of ethnic groups, e.g. under the term 'African-American', makes it difficult to disentangle differences within groups. In reality the Black population in the US have different cultures and social structures. For example, a sizeable proportion of black folks in the US have lived there for many generations whilst others may be recent migrants (Rumbaut and Portes, 2001). In addition, there is a dearth of research examining generational differences in rate of psychosis among the black populations in the US. There have also been calls for a review of the US ethnic categorisation not only for research purposes but also in recognition of social changes in this diverse population (Bhopal, 2004, Mateos et al., 2009).

Furthermore, studies from the rest of Europe have used only country of birth to define ethnicity, as opposed to self-ascription (Veling et al., 2008, Selten et al., 2001, Velthorst et al., 2012). This raises the need for cautious inference from such ethnic classifications. As highlighted by the UK Office of National Statistics and other literature, ethnicity is a multifaceted concept which requires multiple approaches to ascertain, e.g. language spoken, skin colour, geographical region, racial and religious group (ONS, 2003, Mason, 1995).

2.3.2 Study design and sampling

Despite the methodological improvements observed in recent studies, some design issues still remain. For example, thirty-six per cent (n = 17) of studies reviewed above used a crosssectional design, a method which if not well-designed, is open to criticism about the limitations in the inferences that can be made about causal associations. One of the most key criticisms of such studies is the use of retrospective self-report data (Susser et al., 2006). For instance, among studies considered in section 2.2.3, factors such as unemployment may be difficult to disentangle since psychosis, which is linked to behavioural problems and affects relationships and executive functioning, may possibly precede unemployment, and

indeed vice versa. Another methodological consideration is the use of perceived state as a measure of social disadvantage, i.e. how someone views social interaction may be affected by their mental state e.g. 'perceived discrimination'. It is possible that people with psychosis, when making sense of the illness, may consider their social experiences e.g. discrimination, racism or lack of opportunities as possible explanation.

The design limitations in this review are noteworthy but it is also worth acknowledging that in order to disentangle the association and causal relationships between psychosis and risk factors, a prospective cohort sample needs to be recruited prior to the onset of illness. Such a design will be very expensive if not impossible in the case of psychosis, which tends to emerge in adulthood. In the absence of prospective cohort design, psychiatric epidemiology has had to rely on historical cohort study design, which has the ability to take into account the latent period and outcomes using existing clinical records to determine the number of individuals that meet study criteria and outcomes (Mortensen, 1995, Cantor-Graae and Pedersen, 2007, Susser et al., 2006)..

2.3.3 Sources used to collect data

The source used to collect data varies widely between studies. Considering the dimension of social adversity and psychosis, more than half of the studies (n = 6/8) were conducted using samples from the AESOP study. This highlights the need for new research to re-examine these indicators in relation to ethnicity, particularly those of Black African and Black Caribbean ethnicity, who have consistently shown increased risk of these adversities at first episode of psychosis.

Furthermore, overall patient samples have also varied between those including hospital inpatients for first episode psychosis and those including FEP patients presenting to both hospital and community mental health services. With several studies relying on hospital admission samples and clinical diagnosis alone, Kendell and colleagues (1993) pointed out a number of the pitfalls of using hospital first admission to measure the incidence of psychosis, including the potential for underestimation (as not everyone with psychosis gets admitted to hospital), the fact that the diagnosis of psychosis may not be made during first admission and the reality of population movements in study catchment areas (Kendell et al.,

1993). Recent hospital admission studies in this review (Tortelli et al., 2014, Selten et al., 2002, Smith et al., 2006) have also demonstrated the limitations of using such samples as there is a high likelihood of missing cases, since they have not considered those using other services such as community-based health care or outpatients. In addition, it is not possible to be sure of absence of previous psychotic episode in such samples since the capability to carry out routine administrative data may not be available.

Despite the improvements in recent studies' ability to include both hospital and community based patients, it is still possible to miss cases as not all individuals experiencing psychotic symptoms necessarily present to health services to seek help. As alluded to in section 2.3.2, the ultimate solution regarding case identification is prospective population sampling through cohort study design that has the ability to follow up a population possibly from birth and identify exposed and unexposed individuals, but given that psychosis is a rare disorder, this approach may not be feasible as it would be very expensive and time consuming (Susser et al., 2006).

2.4 Summary

The three dimensions discussed above – migration, ethnic density and social disadvantage – are evidently interconnected, and the literature reviewed so far has pointed to differences in rates of psychosis. From ethnic distribution and area level perspective the evidence suggests that ethnic density in a particular population may be a protective factor from the risk of developing of psychosis. Research on social disadvantage has demonstrated that exposure to adversities has deleterious consequences and contributes to increased risk of psychosis. And finally from migration status perspective, the rate of psychosis has been consistently higher among the migrant groups, but the few studies reporting incidence of psychosis for Black African and Black Caribbean groups have painted an inconsistent picture of incidence rates in these two groups, which suggests that the questions considered at the start of this chapter remain unresolved.

3 Chapter 3: Literature Review – Early Intervention for Psychosis, Pathways to Care and Ethnicity

3.1 Synopsis

The discussions in Chapter 2 point to the evidence that heterogeneous incidence of psychosis is a mainstay in the epidemiology of psychosis, with higher rates of incidence and prevalence reported in migrant groups in Western nations. However, understanding the pathways into and through care from the onset of illness until first help-seeking contact with mental health services is important in early recognition and intervention, to thus minimise suffering and improve outcomes (Compton et al., 2006, Birchwood, 2008, McGorry et al., 2008). Additionally, variation in aetiology of psychosis has prompted international initiatives of early intervention services (EIS) for psychosis to promote equity of care, early detection of psychosis and improved outcomes (Mann et al., 2014a). These services are now established in Europe, America and Australia (Marshall and Rathbone, 2011).

In the UK, for example, EIS is recommended in the National Institute of Clinical Excellence (NICE) guidelines on schizophrenia for England and Wales (Excellence and Health, 2002). EIS services aim to identify as soon as possible individuals who already experience psychotic symptoms but have not yet received adequate treatment (Addington, 2007, Marshall and Rathbone, 2011). As such, an early intervention service offers two components: (a) early detection and reduction in delay to first treatment and (b) phase-specific treatment. Phase-specific treatment includes psychosocial or physical treatment which has been developed or modified specifically for use with people at an early stage in the illness (Marshall and Rathbone, 2011, Joseph and Birchwood, 2005, Craig et al., 2004a, McGorry et al., 2009).

Furthermore, much research has been carried out to measure the effectiveness of such services and evidence suggests that early intervention services for psychosis reduces rates of relapse, hospital admissions, improved access to care and increased engagement with services (Garety et al., 2006, Bird et al., 2010). In spite of this encouraging evidence, it remains unclear how patients from the black and minority ethnic groups fare under the care of this specialist intervention service during first contact for episode of psychosis, because many previous studies have largely relied on symptoms and clinical predictors to explain service engagement (Birchwood et al., 2013).

One of the aims of this thesis is to investigate the pattern of engagement with services at first contact and through the treatment period. The aims of this chapter are therefore to highlight the importance of early intervention pathways into care during first episode of psychosis and touch on the concepts and processes involved. This is followed by a review of the literature on referral routes to mental health services and the pattern of engagement with such services by sociodemographic factors, especially ethnicity.

3.2 Examining the Role of Early Intervention Services in the Pathways into Care

The chapter will now outline the literature on individual (especially ethnicity) and service level factors associated with the pathways into care. The discussion will be guided by three main sub-topics: a) duration of untreated psychosis (DUP) and mode of onset (section 3.2.1-3), b) source of referral and mode of contact (section 3.3). Mode of contact in the form of compulsory admission will be reviewed and discussed extensively in section 3.4, specifically addressing the question: are there any differences in pathways to care for first episode psychosis (FEP) cases compared to 15 years ago, which was prior to the introduction of EIS; and is this associated with the introduction of EIS? Subsequently, a discussion surrounding the limitations of the current literature is provided in section 3.5, and finally section 3.6 presents a summary of the literature reviewed.

In reviewing the literature for this chapter, I have chiefly included studies published between 2002 and 2017, which were conducted using populations of first episode psychosis (FEP) patients, in order to give a picture of research efforts and evidence in the last ten years. However, studies investigating DUP are separated into two categories, UK versus international. This is due to the marked differences in the health services structure globally and particularly in Western countries. For example, the UK healthcare service is free at the point of delivery and has a very efficient primary care system, and therefore direct comparison with other healthcare systems may not be helpful. For example, in the USA access and use of healthcare incur costs at the point of use and studies have shown that this influences patients' and families' decisions to seek help for psychosis (Compton et al., 2004, Anderson et al., 2014b) . Furthermore, owing to the fact that factors such as duration of untreated psychosis and mode of contact may be influenced by how health care systems are structured, it is important to consider the evidence separately.

Searches for relevant literature were conducted using the following keywords: 'first episode', 'psychosis', 'duration of untreated', 'early intervention', 'ethnicity', 'African', 'Caribbean', 'pathway to care' and 'compulsory admission'. Bibliographic databases including Embase, Pubmed, PsychoInfo, Web of Science, Ovid and Google Scholar were used to identify relevant literature.

3.2.1 Duration of untreated psychosis, mode of onset and ethnicity – the UK evidence Duration of untreated psychosis (DUP) (i.e. the delay between the onset of first episode psychosis and receiving an effective treatment) has been identified as a major contributor to the variation in outcomes following first episode psychosis (Birchwood et al., 2013, Marshall et al., 2005). Similarly, mode of onset of psychosis (i.e. how quickly psychotic symptoms develop) is regarded as a robust predictor of outcomes (Morgan et al., 2006b, Compton et al., 2011) While it might seem possible to regard DUP and mode of onset as two terms describing the same thing, the measurement of mode of onset is very different from that used in estimating DUP. Compton and colleagues (2011) give a clear distinction between the two and define mode of onset as 'the rapidity of development of psychotic symptoms up to the point of frank psychotic symptoms', whereas DUP is defined as 'beginning at the point of frank psychosis to the point of first contact with services' (Compton et al., 2011). In addition, some individuals may have been experiencing attenuated psychosis symptoms long before first frank symptoms appear (Fusar-Poli et al., 2009). Therefore, reducing DUP has become an international priority and much effort has been spent assessing the impact of early intervention services. As such, in the last decade a number of studies have been carried out evaluating patient-level and service-level factors associated with DUP and the effectiveness of EIS in reducing DUP.

In Table 3.1, studies carried out in the UK specifically examining the relationship between DUP and ethnicity in the last decade are presented (n = 5). From the table, an overall median DUP with interquartile range (IQR) ranged from 9 (2-40) weeks (Morgan et al., 2006b) to 3 (1-95) months (Bhui et al., 2014).

Morgan and colleagues (Morgan et al., 2006b) found no evidence that Black Caribbean patients experienced longer median DUP than White British patients (13 weeks and 10

weeks respectively). However, they reported that Black African patients experienced shorter DUP (8 weeks; log rank test X2 = 9.96, df = 1 P = 0.002) compared to their White British counterparts. This is further substantiated in their adjusted cox regression analysis where they found no evidence that Black Caribbean (adj. HR = 1.04; 95% CI 0.80 - 1.36) patients experienced longer DUP, but there was strong evidence that Black African (Adj. HR = 1.06; 95% CI 1.15 - 2.22) patients had shorter DUP than White British patients.

Similarly, (Ghali et al., 2012) in a study exploring ethnic differences in the nature and duration of pathways into early EIS, found longer delays (median (IOR)) from onset of psychosis to commencement of medication among White British = 113 (345) days compared with other ethnic groups: White Other = 72 (198) days, South Asian = 60 (164) days, Black British = 98 (182) days, Black Caribbean = 55 (236) and Black African = 57 (155) days p< 0.001. It is argued that because Black African and Black Caribbean patients tend to experience acute onset of psychosis leading to more rapid presentation to services this may explain the shorter DUP (Morgan et al., 2006b). Furthermore, Dominguez and colleagues (2013) in a study comparing DUP in adolescent-onset and adult-onset of psychosis analysed 940 FEP cases (136 adolescent vs. 804 adult). They reported greater median DUP (179 days) among the adolescent cases compared to 81 days in the adult sample (p = 0.005). They also found difference in DUP by ethnicity; the median DUP was White = 345 days; Black = 103 days; Asian and mixed = 36.5 days, p = 0.001 among the younger patients aged less than 18 years old (Dominguez et al., 2013). The higher median DUP reported among the White British patients is consistent with other studies (Birchwood et al., 2013) but one that is concerning and above the UK Department of Health target of under three months (Department of Health, 2002).

Kirkbride and colleagues (2010) investigated the role of neighbourhood on DUP and demonstrated no association of neighbourhood effect with DUP but found that Black African patients had a shorter DUP compared with other groups. Similarly, (Bhui et al., 2014) reported ethnic differences in DUP between White and Black patients (median = 4 and 3 months, p<0.001) respectively.

It can be seen from evidence emerging from the UK that there are variations in DUP across ethnic groups. An explanation may be immigration status; for example, first generation

migrants have been reported to be more likely to use emergency, acute or crisis services, which may be a reflection of their limited understanding of healthcare systems and how to utilise them (Kirmayer et al., 2007, Anderson et al., 2015). Further, plausible explanations for the prolonged DUP and treatment delay among White British patients include the tendency to seek help via GP, and living with family, which may help to reduce the burden of the illness and manage symptoms (Compton et al., 2011).

Authors (Country)	Sample/design	Start of DUP	End of DUP	DUP instrument	Age range (years)	Main findings, DUP mean, median
Morgan et al.	Case-control	Onset of psychotic	First contact with	PPHS	16-64	Overall Median DUP
(2006) (UK)	AESOP 414 FEP	symptoms lasting one week or more	mental health services			(IQR) = 9 (2 – 40) weeks
	patients					Southeast London site
						White British – Median
						(IQR) = 10 (4 – 52) weeks
						Black Caribbean Median
						(IQR) = 13 (4 – 87)
						Weeks Dlack African Madian
						(IOP) = 9 (1 - 22) weaks
						(IQR) = 8(1 - 22) weeks
						Nottingham site
						White British – Median
						(IQR) = 7 (1 – 33) weeks
						Black Caribbean –
						Median (IQR) = 5 (2 –
						13) weeks
						Adj. HR = 1.04 (95% Cl
						0.80 – 1.36) – Black
						Caribbean
						Adj. HR = 1.60 (95% Cl
						1.15 – 2.22) – White
						British vs. Black African

 Table 3.1: UK studies of duration of untreated psychosis and ethnicity.

Authors (Country)	Sample/design	Start of DUP	End of DUP	DUP	Age range	Main findings, DUP
				instrument	(years)	mean, median
Kirkbride et al. (2010) (UK)	Case control study 314 FEP patients	Onset of psychotic symptoms lasting one week or more	First contact with mental health services	PPHS	16-64	Association between DUP and ethnicity Reference = White British White other = β -0.13 (- 0.93 - 0.67) Black Caribbean= β 0.21 (-0.33 - 0.77) Black African = β -0.74 (- 1.40.10)*(shorter because CI excludes zero) Asian = β -1.2 (-3.1 - 0.68) Mixed = β -0.83 (-1.8 - 0.17) Other = β -0.43 (-1.6 - 0.76)
Ghali et al. (2012) (UK)	Naturalistic cross-sectional study 775 EIS patients	Onset of psychosis	Commencement of regular antipsychotic medication	NOS	18-35	DUP by ethnic group White British – Median(IQR) = 113 (345) days White Other – Median = 72 (198) days South Asian – Median = 60 (164) days Black British – Median = 98 (182) days Black Caribbean –
Authors (Country)	Sample/design	Start of DUP	End of DUP	DUP	Age range	Main findings, DUP
-------------------	-----------------	--------------------	----------------------	------------	-----------	--------------------------
				instrument	(years)	mean, median
						Median = 55 (236) days
						Black African – Median
						(IQR) = 57 (155) days
Dominguez et al.	Cross-sectional	Date of first	Date of	NOS	14-35	Individual DUP; Median
(2013) (UK)	study	psychotic	commencement of			(IQR), p = 0.005
	940 sample of	symptoms	regular medication			Total sample DUP = 86
	adults (804)		with 75% adherence			(19–282) days
	and					Adolescent – Median
	adolescents					(IQR) = 179 (18 – 514)
	(136)					Adult – Median (IQR)
						=81 (19 – 244) days
						Service DUP; Median
						(IQR), <0.001
						Total sample DUP = 135
						(19.3 – 372) days
						Adolescent = 346 (105.5
						- 721.3)
						Adult = 120 (37 – 311.3)
						days
						Adolescent sample by
						ethnic group, p = 0.03
						Median (IQR)
						White = 345.5 (99.3 –
						985.3) days
						Black = 103.5 (14.8 –
						385.3) days
						Asian & Mixed = 36.5
						(6.8 – 327.8) days
Bhui et al (2014)	ELFEP cohort	Onset of psychotic	Date of first taking	PPSH	18-64	Median DUP (IQR) in

Authors (Country)	Sample/design	Start of DUP	End of DUP	DUP	Age range	Main findings, DUP
				instrument	(years)	mean, median
(UK)	study – 480	symptoms	prescribed			months
	FEP patients		antipsychotic			Overall = 3 (1 – 95)
			medication			Ethnicity p<0.001
						White = 4 (2 – 10)
						Black = 3 (1 – 5)
						Indian sub-continent = 4
						(2 – 12)
						Gender , p = 0.05
						Male = 4 (1 – 10)
						Female = 3 (1 – 8)
						Compulsory admission,
						p = 0.02
						Yes = 3 (1 – 6)
						No = 4 (1 – 12)

NOS, Nottingham onset schedule (Singh et al. 2005). FEP, first episode psychosis. ARMS, at risk mental state. PPHS, personal and psychiatric history schedule (WHO, 1996).

3.2.2 Duration of untreated psychosis, mode of onset and ethnicity – international evidence

Table 3.2 presents international studies that have specifically examined the relationship between DUP and ethnicity during this literature review window. Of the seven studies, median DUP ranged from 22.1 weeks (Archie, 2010) to 74 (1 – 1456) weeks (Addington, 2015).

As can be seen in Table 3.2, many of the international studies on ethnicity and duration of untreated psychosis have originated from Canada and the USA. The two Canadian studies consistently found ethnic differences in help-seeking behaviours among minority ethnic groups, with a higher proportion using the accident and emergency department (Archie et al., 2010, Anderson et al., 2015). However, they found no ethnic variations in the duration of untreated psychosis. Similarly, studies from the USA found no ethnic differences in DUP (Compton et al., 2006, Addington et al., 2015).

In contrast to UK evidence, international evidence points to no ethnic difference in DUP, but variations in pathways to accessing care were reported. A number of explanations have been put forward for this contrast. For example, economic factors are argued to be predictors of differences in accessing care in a privately funded healthcare system (Snowden and Yamada, 2005). In the study by Archie and colleagues (2010) for instance, patients of White ethnic groups were more likely to make initial contact with psychologist services, and psychologists are not widely available on the Canadian state-funded healthcare system (Archie et al., 2010). Further, language barriers, feelings of shame and stigmatisation may be more pronounced among ethnic minority groups, which may hinder help-seeking from healthcare system (Sterk et al., 2010).

Authors (Country)	Sample/desig n	Start of DUP	End of DUP	DUP instrumen t	Age range (years)	Main findings, DUP mean, median
Compton et al. (2004) (USA)	Hospital admission Cross- sectional 10 family members 6 patients	Initial date of onset of symptoms reported by family member	Time of help seeking	Clinician- led questionn aire	18-65	DUP reported by family/parent Mean (sd) 59.5 (69.65) weeks Median DUP = 40.0 (2 - 234) weeks
Compton et al. (2006) (USA)	Hospital admission Cross- sectional 25 patients	Onset of first psychotic symptoms	First hospital admission	SOS	18-39	DUI Median = 128 weeks (0.6 - 467.9) DUP Median (IQR) = 32.9 weeks (0.4 - 337.7) Median DUI (IQR) = 128 (0.6 - 476.9) Pathways to care/source of referral MH professional = 30% Psychiatric Emergency = 28% General emergency department = 12%

Table 3.2: International studies of duration of untreated psychosis and ethnicity

Authors	Sample/desig	Start of DUP	End of DUP	DUP	Age	Main findings, DUP
(Country)	n			instrumen t	range (years)	mean, median
						Primary care = 5% Police = 21%
						Other = 5%
Archie et al. (2010) (Canada)	Cross- sectional 200 patients	Onset of positive symptoms	Date of initiation of treatment	CORS	16 - 50	DUP and ethnicity, p = 0.48 Overall median DUP = 22.1 weeks, Median DUP = 22.4 weeks – white Median DUP = 31.3 weeks – Black Median DUP = 29.4 weeks – Asian Median DUP = 19.5 weeks – Other
Sterk et al. (2010) (The Netherlands)	Cross- sectional 150 patients	Onset of positive symptoms	Start of intensive treatment	NOS	Mean age = 20.9 years (sd = 4.93)	DUP and ethnicity (chi-sq = 6.30 , p = 0.39) Median (IQR) Median = 16.5 (0 – 832) weeks – overall Median = 7.0 weeks (0 – 654) – Dutch Median = 39.0 weeks (0 – 832)- non-Dutch Median = 48.0 weeks

Authors	Sample/desig	Start of DUP	End of DUP	DUP	Age	Main findings, DUP
(Country)	n			instrumen t	range (years)	mean, median
						 Surinamese Median = 29.0 weeks Ghanaian Median = 24.0 weeks Moroccan Median = 16.0 weeks Turkish Median = 4.0 weeks Other European
Anderson et al. (2015) (Canada)	Cross- sectional 208 EIS patients	Onset of positive symptoms	Contact with an EIS	PPHS	14+	ED contact and ethnicity (reference = White European) OR = 3.78 (95% Cl 1.31 – 10.92) – Black African OR = 2.42 (95% Cl 0.85 – 6.89) – Black Caribbean OR = 0.13 (95% Cl 0.05 – 0.33) GP referral DUP and ethnicity Median = 9 (IQR = 3 – 28) months (expB = 1.10; 95% Cl 0.6 –

Authors	Sample/desig	Start of DUP	End of DUP	DUP	Age	Main findings, DUP
(Country)	n			instrumen	range	mean, median
				t	(years)	
						2.03) – Black African
						Median = 16 (IQR = 6
						– 46) months (expB =
						1.49; 95% Cl 0.82 –
						2.69) – Black
						Caribbean
						Median = 7 months
						(IQR = 2 – 34)
						months – White
						European
						Age and DUP
						expB= 1.08, (95% Cl
						1.03 – 1.14) – Older
						age
Addington et al.	Cross-	onset of	Initiation of treatment	SCID	15-40	Overall median DUP
(2015)	sectional	psychotic	with antipsychotic			(IQR) = 74 (1 – 1456)
(USA)	404 patients	symptoms	medication			weeks
						DUP and ethnicity,
						no difference
						Mean DUP (sd) = 49
						(6.5) weeks – White
						(7) iviean DUP (sd) = 63
						(/) weeks –
						Black/African
						American
						Mean DUP (sd) =52

Authors	Sample/desig	Start of DUP	End of DUP	DUP	Age	Main findings, DUP
(Country)	n			instrumen	range	mean, median
				t	(years)	
				t	(years)	 (1) weeks – American Indian Mean DUP (sd) =55 (4.5) weeks – Asian/ Pacific Islander DUP and demographic factors Mean (sd) = 40 (7) weeks, p<0.001 – previous hospitalisation Mean (sd) = 74 (6.5) weeks, p<0.01 –
						substance misuse
				1	1	

CORS, circumstances of onset of symptoms and relapse schedule (Norman et al., 2004); CORS/TOPE, circumstances of onset of symptoms and relapse schedule/topography of psychotic episode (Norman et al., 2004). NOS, Nottingham onset schedule (Singh et al., 2005). SOS, symptoms onset in schizophrenia (Perkins et al., 2000). SCID, structured clinical interview for DSM-IV axis 1 disorders.

3.3 Source of referral and mode of contact with mental health services

Before reviewing the literature on source of referral and mode of contact with mental health services, let us turn to Fig. 3.2, which maps the common route into mental health care in the UK. The map is inspired and guided by the Goldberg-Huxley pathway model, which describes five levels of care pathway routes from the community to hospital admission (Goldberg and Huxley, 2003). The model has proven very useful in understanding how far along the pathway levels an individual has reached and possibly the extent of severity of illness. As shown on the map, people can access secondary mental health services via different routes, but by and large the most common source of referral is via the general practitioner (GP) (Goldberg and Huxley, 2003), who in the UK plays a key role in gatekeeping referrals to secondary services. Once in contact with mental health care providers, patients can access services either by community-based services or hospital inpatient admission. Most hospital admissions are voluntary, but in some cases patients who are assessed as posing risks to themselves or others and do not accept voluntary admission may be admitted compulsorily without their consent, using sections of the Mental Health Act (MHA) 1983 (amended 2007) (Mentalhealthcare, 2015). Other referrals to secondary mental health care tend to operate during crisis and emergency periods, when patients require urgent treatment and intervention. These include the police, criminal justice agencies and accident and emergency departments (Morgan et al., 2004).



Figure 3.1: FEP pathways to care

There is well-established evidence that there are differences in referral routes and modes of contact with mental health services between ethnic groups in the Western nations. In the past two to three decades, there has been a stream of research evidence suggesting that compared to the native population, minority ethnic group patients are more likely to come into contact with mental health services via crisis routes such as emergency departments and police (Morgan et al., 2005b, Bhui and Bhugra, 2002, Anderson et al., 2010, Anderson et al., 2013b), and they are less likely to have general practitioner (GP) involvement in their referral to specialist mental healthcare (Bhui and Bhugra, 2002). For instance, Morgan and colleagues (2005) in a case-control study of 462 FEP patients, reported that compared to White British, Black Caribbean (adj. OR = 1.98; 95% Cl 1.04 – 3.77) patients were more likely to be referred to mental health services via the criminal justice agency but the evidence was weaker for Black African patients (adj. OR = 1.87; 95% CI 0.86 – 4.05). However, both groups were less likely to be referred by their general practitioners: Black Caribbean (adj. OR = 0.48; 95% CI 0.25 – 0.90) and Black African (adj. OR = 0.41; 95% CI 0.18 – 0.95) (Morgan et al., 2005b). This is consistent with Ghali et al. (2012) findings that ethnic minority group patients were less likely than White British patients to make contact with a GP White Other, OR = 0.55 (95% CI 0.34 – 0.91); Black British, OR = 0.62 (95% CI0.40 – 0.96); Black Caribbean OR = 0.30 (95% CI 0.12 -0.75); Black African OR = 0.69 (95% CI 0.44 – 1.07) within their pathway to care. For contact via emergency services and the criminal justice system, the authors reported that compared to White British patients all ethnic minority groups except the South Asian patients were more likely to make contact via emergency services, whereas they found only Black ethnic groups (African, Caribbean and Black British) were more likely to make contact with services via the criminal justice system (Ghali et al., 2012).

Furthermore, Anderson et al. (2013) examined the determinants (prodromal contact, first contact, referral source and total contacts) of pathway to care using a sample of 324 FEP patients from a Canadian EIS program. They found that the likelihood of prodromal contact increased with increasing prodrome length (OR = 1.61; 95%CI 1.33 - 1.94); and that previous contact with primary care was associated with any contact during prodrome (OR = 2.70; 95%CI 1.48 - 4.96). In addition, those whose first contact was with primary care were less likely to have contacts with emergency services (OR = 0.07; 95% CI 0.04 - 0.14), but more

likely to have more than two contacts prior to reaching EIS (OR = 3.50; 95% CI 1.95 – 6.30). They reported a median number of contacts of IQR = 3 (95% CI 2 - 4) prior to contact with EIS. Asian patients were less likely to have more than two contacts (OR = 0.47; 95% Cl 0.22 -0.98). Forty-five per cent of participants came into contact with police or ambulance services and the likelihood of using such service increased as the duration of treatment delays increased (OR = 1.12; 95% CI 1.20 - 1.43) but this likelihood reduced if the initial contact was with primary care (OR = 0.47, 0.27 - 0.82). Patients in contact with primary care experienced delays in referral to EIS (expB = 2.31, 1.36 – 3.92). In terms of service engagement, Anderson and colleagues (2013) found that 28% of the sample dropped out, with median time to dropout reported as five months (IQR = 1 -11). Older age was also reported to be associated with disengagement (HR = 2.10, 1.02 - 1.19) as was Black ethnicity (HR 2.10, 1.19 – 3.70). However, individuals living alone were found to be less likely to disengage (HR = 0.46, 0.21 – 1.00) (Anderson et al., 2013b). A number of systematic reviews have been carried out exploring relationships between pathways to care, mode of contact, source of referrals and ethnicity. Anderson et al. (2014) in a meta-analysis of ethnic difference in pathways to care at FEP, reported that compared with White patients, Black patients are less likely to have GP involvement (OR = 0.70; 0.57 – 0.86) but are more likely to have police involvement (OR = 2.11; 1.67 – 2.66). Such association was not reported among other ethnic groups, e.g. Asian patients were reported to be less likely to have both police involvement (OR = 0.86; 0.57 – 1.30) and GP involvement (OR= 1.23; 0.87 – 1.75) (Anderson et al., 2014a). With only seven papers meeting the inclusion criteria for this review, Anderson and colleagues' findings also hinted at evidence of higher compulsory admission among Black patients, but pooling of such data was limited by effect modification of other sociodemographic factors. They concluded that studies on ethnic difference in pathways to care during FEP are inconsistent compared with other psychiatric disorders (Anderson et al., 2014a). Similarly, Singh et al. (2006) suggest that pathways for FEP are varied and diverse. They reported that FEP patients are more likely to have first contact with mental health professionals, while contact with non-medical or non-statutory services are less common. Methodological variations in the included studies did not allow for meta-analysis of data, but the findings confirm previous evidence that certain ethnic groups in the UK experience more adverse pathways to care; they experience delayed help-seeking which is related to the undetected prodromal and early psychosis symptoms; and family and friends play a vital

role in their help-seeking (Singh and Grange, 2006). Although not specific to FEP samples, Sass et al. (2009) conducted a systematic review of initiatives to improve pathways into mental health care for Black and minority patients. Their findings suggest that ethnically focussed care or services could potentially improve pathways to care, but the majority of the studies in their review were from outside the UK (Sass et al., 2009). However, the idea of developing culturally appropriate treatment during FEP in the UK has been previously mooted (Bhui and Bhugra, 2002, Bhugra et al., 2011a).

3.4 Pathways to Care – Compulsory Admission and Ethnicity

So far, treatment delays, source of referral and mode of contact with mental health services have been considered as key determinants of pathway to care. The association of these factors with ethnicity suggests that ethnic minority patients tend to have shorter DUP, particularly in the UK, due to urgency of symptom presentation compared with the native population. The evidence also points to a less favourable pathway to care, e.g. use of emergency services at first presentation and lesser GP involvement in the referral pathway for ethnic minority patients. Although a few of the studies reviewed above reported correlational associations, causal inference cannot be drawn.

This section now considers the negative pathways into care, i.e. compulsory admission, criminal justice involvement and emergency referral. In keeping with the evidence reviewed in sub-section 3.2.3, it is well documented that there are higher compulsory admission rates among Black and minority ethnic groups in Western nations, and this remains a major concern for patients, mental healthcare providers and policy-makers (Singh et al., 2007). So much so, in the UK for example, the government published a report on reducing the inequality in detention rates in the white paper 'Delivering Race Equality in Mental Health Care' (Department of Health, 2005). Over the past decade a number of primary research and systematic reviews have been conducted to shed light on this clinically and politically contentious issue.

Literature comparing compulsory admission rates in different ethnic groups has highlighted a paucity of evidence on the topic, especially for the Black African and Black Caribbean groups. Table 3.3 shows that 20 primary research studies have been carried out in the last ten years on this topic. However, there are concerns about the quality of studies; for

instance, Mann et al. 2014 in a systematic review found only seven eligible studies, 70% of which reported no differences in rates of compulsory admission, except Morgan et al. (2005), who reported odds ratios of compulsory admission by ethnic group, whereas others reported proportions and chi-squared test (Mann et al., 2014a). Bhui and Bhugra (2002) highlighted variations in access to services; for example, they observed an over-representation of Black patients in inpatient facilities, whereas Asian patients are less likely than White patients to use such facilities. They also reported variations in primary care assessment and GP involvement, which may hint at some of the ethnic differences in pathways to specialist services. Anderson and colleagues (2010) in their systematic review of pathways to care during FEP reported that in the majority of the studies they included, patients' first contact for help-seeking is their GP. However, they found that the greatest proportion of referral source to be via emergency services (Anderson et al., 2010). There was no consistent evidence across studies for demographic factors as predictors of pathways to care, but the authors acknowledged the need for further research to understand help-seeking behaviour during FEP (Anderson et al., 2010).

Studies reporting the odds ratios of compulsory admission compared to native populations have shown variation in the effect of the odds ratios (Table 3.3). Only two studies have examined the association between compulsory admission and specifically Black African and Black Caribbean ethnicity. For Black African patients the effect size ranges from adj. OR of 4.33 (95% CI 1.88 – 9.99) (Morgan et al., 2005a) to adj. OR of 5.4 (95% CI 2.7 – 10.7) (Mann et al., 2014b); for Black Caribbean it ranges from adj. OR of 2.0 (95% CI 0.7 – 5.5) (Mann et al., 2014b) to OR of 2.30 (95% CI 1.23 – 4.32) (Morgan et al., 2005a). Other studies have employed broad groupings of ethnicity such as Black and the effect of association for this group range from OR of 1.35 (95% CI, 1.13 - 1.62) (Singh et al., 2014a) to OR of 4.31 (95% CI 3.33 – 5.58)(Bhui et al., 2003). For Asian patients, the odds ratios for compulsory admission range from OR of 0.3 (95% CI 0.0 – 2.6) (Corrigall and Bhugra, 2013) to OR of 1.1 (95% CI 0.6 – 2.13) (Jarvis et al., 2014). These studies have mainly been conducted in the UK and were usually in large cities like London.

Evidence from the rest of Europe also suggests a similar trend. For example, compared to their Dutch counterparts, Moroccan patients have increased odds of compulsory admission ranging from adj. OR of 1.8 (95% Cl 1.0 - 3.3) to adj. OR of 2.03 (95% Cl 1.34 - 3.06). For the

Turkish patients, the odds of compulsory admission range from adj. OR of 1.3 (95% Cl 0.5 - 3.0) to adj. OR of 3.95 (95% Cl 1.52 - 10.24). Similarly, the Surinamese patients also showed a likelihood of compulsory admission, with a range of adj. OR of 2.4 (95% Cl 1.5 - 3.8) to adj. OR of 1.88 (95% Cl 1.19 - 2.97) (de Wit et al., 2012, van der Post et al., 2012). Furthermore, Berg and Johnsen (2004) in an investigation of whether immigrants were more likely to use emergency departments than native Norwegians, reported no difference (49 in 10,000 vs. 52 in 10,000 respectively, p = 0.72) but they found immigrants using such services were more likely to be men, younger and more likely to be compulsorily detained (Berg and Johnsen, 2004). Of note is a Dutch study by Van der Post et al. (2012); having adjusted for a number of confounders, they also replicated UK findings that sub-Saharan African patients are more likely to be detained compulsorily, with an adj. OR of 3.0 (95% Cl 1.4 - 6.4) (van der Post et al., 2012).

Further, in a meta-analysis, Singh and colleagues (2007) impressively attempted to disentangle possible explanations for the excess rate of compulsory admission among minority groups. Their findings were consistent with those comparing detention rates in native patients with minority ethnic groups e.g. White vs. Black, White vs. Asian, but they went a step further and examined detention rates between ethnic minority groups, e.g. Black vs. Asian, where they found that Black patients remain more likely than Asian patients to be compulsorily detained, with an OR of 2.25 (95% CI 1.72 – 2.94). This difference between minority groups provides an opportunity to explore further the role of cultural, socioeconomic and help-seeking behaviours to improve our understanding on this topic and test the hypothesis that differences in detention rates may not all be down to ethnicity. They explored the association of detention with illness episode i.e. FEP or mixed episode and reported that the minority groups are more likely to be detained during both episodes, except the Asian patients who are less likely to be detained during FEP (Singh et al., 2007). A few years later, Singh and colleagues (2014) in an investigation of effect of ethnicity on detention following a Mental Health Act (MHA) assessment, found that Black patients (OR=1.35; 95% CI 1.37 – 1.62) and 'Other' ethnic groups (OR=1.31; 95% CI, 1.03 – 1.66) were more likely to be detained than White patients. They also found other demographic factors to be associated with MHA, for example female OR = 1.30 (95% CI 1.10 - 1.55), older age OR= 1.35 (95% CI 1.11 – 1.64) and living in supported accommodation OR = 1.80 (95% CI 1.14 –

2.85). Paradoxically, when they took the site of MHA assessment into account they observed that ethnicity was not associated with detention, which may be explained by distribution of ethnic groups since London had the highest proportion of Black patients and Birmingham had the largest number of Asians in the sample (Singh et al., 2014a). However, in a letter to the editor Sasidharan et al. (2014) highlighted key fundamental issues to bear in mind with the Singh et al. (2014) study; for example, the sampling framing they used to define compulsory detention were individuals assessed under the MHA rather than the patient population in contact with mental health services who were therefore be at risk of being detained. It is possible that the sample used in this study was an underestimate since only the patients who were seen by an approved social worker (ASW) or approved mental health professional (AMHP) would have been made prior to the patient being assessed by an ASW.

Authors **Main findings Study Design** Sample Ethnic groups (Country) 5 yr 189 patients White Proportion of schizophrenia diagnosis Law-Min et al. admitted 63.6% Asian (2003) (UK) Retrospective Black under the 47.6% Black study Asian MHA 41.5% White Proportion of compulsory admission 61.2 White 86.7% Black 64.7% Asian Webber & Huxley Cross-sectional 300 patients White British Association between compulsory admission and (2004) (UK) survey admitted Non-White British ethnicity (reference = White British) OR = 2.3 (95% CI 1.40 – 3.79) non-white ethnicity under section 4 of MHA OR = 2.23 (95% CI1.38 – 3.61) Psychosis diagnosis OR = 9.78 (95% CI 5.71 – 16.75) Present risk OR = 1.72 (95% CI 1.05 – 2.79) Insure housing OR = 2.01 (95% CI 1.22 – 3.31) multiple exclusion Cougnard et al. 97 hospital Cross-sectional n/a OR = 2.9 (95% CI 1.2 – 7.4) Male (2004) (France) patients OR = 1.8 (95% CI 0.6 – 5.6) Single survey OR = 0.7 (95% CI 0.3 – 2.4) No children OR = 1.2 (95% CI 0.5 – 2.9) Lives alone OR = 0.7 (95% CI 0.2- 2.6) No friends OR = 1.7 (95% CI 0.7- 4.1) Educational level <12yrs OR = 0.9 (95% CI 0.4 – 2.2) Unemployed OR = 2.4 (95% CI 1.1 – 5.9) Broadly defined SCZ

 Table 3.3: Studies of psychosis compulsory admission and ethnicity

Authors (Country)	Study Design	Sample	Ethnic groups	Main findings
Oluwatayo et al. (2004) (UK)	Cross-sectional survey	382 inpatients	White British African-Caribbean	Factors associated with compulsory admission. White British vs. African-Caribbean X2 = 9.2 df = 3 p = 0.02 (GP involvement) X2 = 40.1 df = 12 p = 0.01 (source of referral) X2 = 24.2 df = 12 p = 0.05 (duration of symptoms)
Wheeler et al. (2005) (New Zealand)	Retrospective case-note survey	932 patient admitted during 1 year period	n/a	62% compulsory admissions 38% schizophrenia – Most common diagnosis
Lay et al. (2005) (Switzerland)	Administrative record survey	23,377 referrals	Country of origin: Switzerland Southern Europe Western and Northern Europe Former Yugoslavia Turkey Eastern Europe Other countries	Compulsory admission by country of origin and gender (male) – Reference = Swiss OR = 1.09 (95% CI 0.93 – 1.26) Southern Europe OR = 0.9 (95% CI 0.73 – 1.12) Western and Northern Europe OR = 1.62 (95% CI 1.36 – 1.93) Former Yugoslavia OR = 1.27 (95% CI 1.00 – 1.62) Turkey OR = 1.70 (95% CI 1.15 – 2.51) Eastern Europe OR = 2.33 (95% CI 1.99 – 2.73) Other countries Female OR = 1.14 (95% CI 0.94 – 1.39) Southern Europe OR = 0.99 (95% CI 0.79 – 1.27) Western and Northern Europe OR = 1.41 (95% CI 1.16 – 1.71) Former Yugoslavia OR = 1.19 (95% CI 0.86 – 1.64) Turkey OR = 2.78 (95% CI 1.73 – 4.45) Eastern Europe

Authors (Country)	Study Design	Sample	Ethnic groups	Main findings
				OR = 2.63 (95% CI 2.08 – 3.33) Other countries
Morgan et al. (2005) (UK)	Incidence and Case-control (AESOP)	462 FEP	White British White other Black African Black Caribbean	Association between source of referral and ethnicity (White British as reference group) GP referral OR = 0.46 (95% CI 0.29 – 0.75) African-Caribbean OR = 0.39 (95% CI 0.20 – 0.73) Black African Adj. OR = 0.48 (95% CI 0.25 – 0.90) African- Caribbean Adj. OR = 0.41 (95% CI 0.18 – 0.95) Black African Criminal Justice Agency OR = 2.52 (95% CI 1.49 – 4.27) African-Caribbean OR = 2.89 (95% CI 1.53 – 5.46) Black African Adj. OR = 1.98 (95% CI 1.04 – 3.77) African- Caribbean Adj. OR = 1.87 (95% CI 0.86 – 4.05) Black African
Morgan et al. (2005) (UK)	Incidence and Case-control (AESOP)	512 FEP	White British White other Black African Black Caribbean	Association between compulsory admission and ethnicity (white British as reference group) Adj. OR = 2.30 (95% Cl 1.23 – 4.32) Black Caribbean Adj. OR = 4.33 (95% Cl 1.88 – 9.99) Black African OR = 2.88 (95% Cl 1.84 – 4.51) Black Caribbean OR = 3.26 (95% Cl 1.85 – 5.77) Black African OR = 1.18 (95% Cl 0.53 – 2.61) Other White

Authors	Study Design	Sample	Ethnic groups	Main findings
(Country)				
Turner et al. (2006)(New Zealand)	Cross-sectional survey	200 first admission patients	Maori Non-Maori	X2 = 2.08, df = 4, p = 0.72 Diagnosis vs. types of help-seeking X2 = 1.82, df = 2, p = 0.40 rate of contact with police X2 = 2.77, df = 2, p = 0.25 outpatient X2 = 0.75, df = 2, p = 0.68 GP X2 = 1.42, df = 2, p = 0.49 rate of admission X2 = 0.49, df = 2, p = 0.78 compulsory admission
Mulder et al. (2006) (the Netherlands)	Case-register survey	720 patients	Dutch native Moroccan Turkish Surinamese Antillean Other Western Other non- Western	Risk of contact with emergency services for psychotic disorder (Dutch as reference group) Adj. RR = 4.2 (95% Cl 2.7 – 6.7) Moroccan Adj. RR = 2.0 (95% Cl 1.2 – 3.3) Turkish Adj. RR = 3.5 (95% Cl 2.4 – 5.1) Surinamese Adj. RR = 3.7 (95% Cl 2.1 – 6.4) Antillean Adj. RR = 0.9 (95% Cl 0.5 – 1.6) Other Western Adj. RR = 3.9 (95% Cl 2.8 – 5.6) Other non- Western Risk of contact followed by compulsory admission (Dutch as reference group) Adj. RR = 2.2 (95% Cl 1.0 – 5.2) Moroccan Adj. RR = 1.4 (95% Cl 0.6 – 3.2) Turks Adj. RR = 3.0 (95% Cl 1.7 – 5.2) Surinamese Adj. RR = 3.6 (95% Cl 1.6 – 7.9) Antillean Adj. RR = 0.7 (95% Cl 0.3 – 1.6) Other Western

Authors (Country)	Study Design	Sample	Ethnic groups	Main findings
				Western
Archie et al. (2010) (Canada)	Cross-sectional survey	200 patients	White Black Asian Other	X2 = 9.500, df = 3, p = 0.23 Compulsory vs. voluntary admission X2 = 2.398, df = 3, p = 0.49 ER vs non-ER X2 = 4.349, df = 3, p = 0.22 Hospitalisation vs. no- hospitalisation X2 = 3.358, df = 3, p = 0.34 Arrested 6 months prior vs. no Arrest Odds of ER use by ethnicity Adj. OR = 1.11 (95% CI 0.37 – 3.27) Black Adj. OR = 3.97 (95% CI 1.39 – 11.34) Asian Adj. OR = 3.27 (95% CI 1.16 – 9.18) Other
Vinkers et al. (2010) (the Netherlands)	Cross-sectional survey	14,540 pre- trial reports	Dutch native Whites Black and minority ethnic group	Association between ethnicity and compulsory admission recommendation among prisoners (Dutch as reference group) OR = 1.38 (95% CI 1.16 – 1.64) BME OR = 1.54 (95% CI 1.05 – 2.27) White
Lawlor et al. (2012) (UK)	Cross-sectional survey	287 women admitted to acute inpatient wards and crisis teams	White British Other White Black African Black Caribbean Other Black	Association between compulsory admission and ethnicity (reference = White British) Adj. OR = 3.53 (95% CI 1.57 – 7.94) Other White Adj. OR = 3.88 (95% CI 1.47 – 10.2) Black Caribbean Adj. OR = 5.80 (95% CI 2.52 – 13.4) Black African Adj. OR = 5.22 (95% CI 2.06 – 13.2) Black Other

Authors	Study Design	Sample	Ethnic groups	Main findings
(country)				
Van der Post et al. (2012) (the Netherlands)	Prospective cohort study	2,245 patients	Country of origin: Netherlands Morocco Turkey Surinam/Dutch Antilles Sub-Saharan Africa Other non- Western Other Western	Compulsory admission (adjusted for age, gender, living situation) (Dutch as reference group) Adj. OR = 1.8 (95% Cl $1.0 - 3.3$) Morocco Adj. OR = 1.3 (95% Cl $0.5 - 3.0$) Turkey Adj. OR = 2.4 (95% Cl $1.5 - 3.8$) Surinam/Dutch Antilles Adj. OR = 3.0 (95% Cl $1.4 - 6.4$) Sub-Saharan Africa Adj. OR = 1.7 (95% Cl $1.0 - 2.9$) Other non- Western Adj. OR = 1.2 (95% Cl $0.7 - 1.9$) Other Western Compulsory admission (adjusted for age, gender, living situation, referral source, diagnosis and psychiatric history) Adj. OR = 1.6 (95% Cl $0.8 - 3.4$) Morocco
				Adj. OR = 1.0 (95% Cl 0.4 – 2.4) Turkey Adj. OR = 1.4 (95% Cl 0.8 – 2.5) Surinam/Dutch Antilles
				Adj. OR = 2.1 (95% Cl 0.9 – 5.1) Sub-Saharan Africa
				Adj. OR = 1.5 (95% Cl 0.8 – 2.8) Other non- Western
				Adj. OR = 1.1 (95% Cl 1.0 – 1.0) Other Western
De Wit et al.	10 year	2,646 patients	Dutch	Association between ethnicity and compulsory
(2012) (the	Longituainai	with first	Sumamese	aumission (Dutch as reference group)

Authors	Study Design	Sample	Ethnic groups	Main findings
(Country)				
Netherlands)	case-note cohort	compulsory	Antillean	Adj. OR = 1.48 (95% Cl 1.14 – 1.92) Surinamese
	study	admission	Moroccan	1 st gen
			Turkish	Adj. OR = 1.88 (95% Cl 1.19 – 2.97) Surinamese
			Other Western	2 nd gen
			Other non-	Adj. OR = 2.02 (95% CI 1.17 – 3.49) Antillean 1 st
			Western	gen
				Adj. OR = 2.18 (95% Cl 0.75 – 6.36) Antillean 2nd
				gen
				Adj. OR = 2.03 (95% Cl 1.34 – 3.06) Moroccan 1st
				gen
				Adj. OR = 1.32 (95% Cl 0.73 – 2.40) Moroccan 2nd
				gen
				Adj. OR = 2.03 (95% Cl 1.22 – 3.38) Turkish 1st
				gen
				Adj. OR = 3.95 (95% Cl 1.52 – 10.24) Turkish 2nd
				gen
				Adj. OR = 0.89 (95% Cl 0.65 – 1.23) Other Western
				1st gen
				Adj. OR = 1.04(95% Cl 0.76 – 1.44) Other Western
				2nd gen
				Adj. OR = 0.97 (95% Cl 0.72 – 1.32) Other non-
				Western 1st gen
				Adj. OR = 0.90 (95% Cl 0.38 – 2.13) Other non-
				Western 2nd gen
Corrigall and	10 yr	435 hospital	White	Association with MHA White British as reference
Bhugra (2013)		admission	Black	group
(UK)	case-note study	adolescent	Asian	OR = 3.0 (95% Cl 1.3 – 6.7) Black

Authors	Study Design	Sample	Ethnic groups	Main findings
(Country)				
		patients	Other	OR = 0.3 (95% CI 0.0 – 2.6) Asian
				OR = 3.1 (95% Cl 1.1 – 8.8) Other
				Associations with admission for psychosis
				OR = 5.7 (95% Cl 3.9 – 8.3) Black
				OR = 2.6 (95% Cl 1.2 – 5.3) Asian
				OR = 3.4 (95% Cl 2.1 – 5.6) Other
Singh et al. (2014) (UK)				Association between MHA and demographic factors OR = 1.30 (95% Cl 1.10 – 1.55) Female OR = 1.35 (95% Cl 1.11 – 1.64) <30yrs (Living with family as reference group) OR = 1.22 (95% Cl 0.84 – 1.78) Homeless OR = 0.96 (95% Cl 0.79 – 1.16) Living alone OR = 1.80 (95% Cl 1.14 – 2.85) Supported accommodation Association between MHA and Ethnicity only (White as reference group) OR = 1.35 (95% Cl 1.137 – 1.62) Black OR = 0.95 (95% Cl 0.77 – 1.18) Asian OR = 1.31 (95% Cl 1.03 – 1.66) Other Association between MHA and Site only (ref. = London)
				London)

Authors (Country)	Study Design	Sample	Ethnic groups	Main findings
				OR = 0.53 (95% CI 0.46 – 0.62) Oxfordshire
				OR = 0.51 (95% CI 0.43 – 0.59) Birmingham
				Association between MHA and Ethnicity and site
				(ref. = White and London)
				OR = 1.19 (95% CI 0.99 – 1.42) Black
				OR = 1.04 (95% CI 0.83 – 1.29) Asian
				OR = 1.07 (95% CI 1.84 – 1.37) Other
Mann et al.	Cross-sectional	674 EIS		Association with compulsory detention by
(2014)	audit – with 12-	patients		ethnicity (White British as reference group)
(UK)	month follow-up	Across 4 El		Adj. OR = 1.8 (95% Cl 0.84 – 4.0) White other
		services in		Adj. OR = 3.4 (95% Cl 0.9 – 12.6) Mixed
		London		Black/White
				Adj. OR = 2.5 (95% Cl 0.9 – 7.0) South Asian
				Adj. OR = 1.6 (95% CI 0.5 – 5.3) Asian other
				Adj. OR = 2.0 (95% CI 0.6 – 7.2) Black British
				Adj. OR = 2.0 (95% Cl 0.7 – 5.5) Black Caribbean
				Adj. OR = 5.4 (95% Cl 2.7 – 10.7) Black African
				Association with hospital admission by ethnicity
				(White British as reference group)
				Adj. OR = 1.8 (95% Cl 0.9 – 3.7) White other
				Adj. OR = 1.4 (95% Cl 0.4 – 5.2) Mixed
				Black/White
				Adj. OR = 1.1(95% CI 0.4 – 3.0) South Asian
				Adj. OR = 1.8 (95% CI 0.6 – 5.8) Asian other
				Adj. OR =1.7 (95% Cl 0.5 – 5.7) Black British
				Adj. OR = 1.6 (95% CI 0.6 – 4.3) Black Caribbean

Authors	Study Design	Sample	Ethnic groups	Main findings
(Country)				
				Adj. OR = 4.9(95% Cl 2.4 – 9.7) Black African
Jarvis et al. (2014)	Cross-sectional	351 hospital	Euro-Canadian	Predictors of police or ambulance referral
(Canada)	survey	patients	Afro-Canadian	(reference = Canadian)
			Asian-Canadian	Exp (B) = 2.9 (95% Cl 1.26 – 6.62) Afro-Canadian
				Exp (B) = 1.1 (95% CI 0.6 – 2.13) Asian-Canadian

Adj. OR, adjusted odds ratio. OR, odds ratio. DUP, duration of untreated psychosis. LOS, length of stay. DUP, Duration of untreated psychosis. RR, relative risks. Adj. RR, adjusted relative risks. X2, chi-sq test. Exp (B), exponential beta coefficient. EIS, early intervention service. BME, black and minority ethnic group. EI, early intervention.

3.5 Methodological consideration and gaps in current evidence

This review has picked up upon some methodological issues in studies on DUP, mode of onset, EIS and pathways into care, which have not been adequately addressed in previous research. Firstly, the challenge of ethnic categories is posed, where for example the Black Caribbean, Black African and other Black patients are often amalgamated into one group – Black, meaning that no conclusion can be drawn about the pathway into care for Black Caribbean and Black African patients. However, it is noteworthy that three studies in this review addressed this issue (Morgan et al., 2005a, Mann et al., 2014b, Ghali et al., 2012), and they had sufficient sample power to analyse data for these two different groups separately.

Secondly, the issue of sampling and design must be addressed. In order to adjust for a range of well-known confounders and to assess the differential effect of pathways to care as well as between ethnicities, large numbers of participants are required to ensure false positive associations are not reported. However, despite a number of studies in this review having sufficient sampling power, quite a few had small sample sizes. For example, in the studies by (Compton et al., 2006), (Goulding et al., 2008), (Chien and Compton, 2008) and (Cheung et al., 2014), only 25, 34 (34 carers) and 50 participants respectively were recruited. Moreover, in the study by (Dominguez et al., 2013), differences found by ethnic groups in the adolescent sample may be subject to selection bias as the sample size was small (73 patients). In addition, age of onset was available for only 69% of the sample, so the findings may not be generalisable. Indeed some studies on compulsory admission and mode of contact have controlled for confounders such as age, gender, socioeconomic status and clinical factors(Morgan et al., 2006a, Morgan et al., 2005a, Morgan et al., 2005b, de Wit et al., 2012, Mulder et al., 2006, Mann et al., 2014b, Lawlor et al., 2012), although Webber and Huxley (2004) only adjusted for age. Some have adjusted for none and only reported unadjusted odds ratios (Corrigall and Bhugra, 2013, Lay et al., 2005), while others have only reported proportional differences and X² tests (Turner et al., 2006, Wheeler et al., 2005, Archie et al., 2010, Oluwatayo and Gater, 2004).

Furthermore, this review picked up some inconsistencies in measuring clinical variables in connection with pathways to care. For instance, variation in duration of untreated psychosis may be due to the way DUP have been measured across studies; some studies calculated

DUP by subtracting the date of first positive psychotic symptom from the date antipsychotic medication started (Ehmann et al., 2014, Ghali et al., 2012, Dominguez et al., 2013, Power et al., 2007, Thomas and Nandhra, 2009), others calculated to the date of first hospital admission (Compton et al., 2011, Chien and Compton, 2008) and a few used date of first contact with any mental health services (Morgan et al., 2006b, Kirkbride et al., 2010). The variation in pathways to care given the differences in healthcare systems in different countries may contribute to the lack of consensus in measuring DUP. For example, outside of the UK, the primary care system (e.g. GP) is widely available and free at point of use in Canada and parts of Europe but not in the USA, where healthcare is insurance- and payment-based.

Thirdly, we consider study design. Similar to issues picked up in the previous chapter, many of the studies included in this review have employed a cross-sectional survey design, which has the potential to limit exploration of other factors that may explain pathways to care. Sources used to draw samples vary across studies, making generalisability of findings difficult. For example, Compton et al. (2011) used hospital admission to define FEP in a cross-sectional design which has its own flaws as previously highlighted in Chapter 2. Moreover, the study sample has a majority of African-American (89.9%), a small sample of White (6.4%) and other (3.6%) subjects.

3.6 Summary

In conclusion, it appears that despite the outcome-driven nature of EIS, which is to demonstrate positive impact on early detection and treatment of psychosis, research studies in the last ten years suggest little improvement in the equality of access to care between ethnic groups.

The evidence so far indicates that only a few sociodemographic and clinical factors are predictive of DUP and pathway into care. But more importantly, evidence from the handful of studies that have examined service-level factors have demonstrated that type of contact and delays in time seeking treatment (e.g. steps taken to reach an EIS) have stronger impact on pathways to care and service engagement patterns.

Minority ethnic group patients continue to be more likely to access mental health care via negative pathways, i.e. compulsory admission and emergency departments, and have police involvement during first episode psychosis.

4 Chapter 4: Literature Review – Ethnicity, Course and Service Use Outcomes Following First Episode Psychosis

4.1 Synopsis

This chapter is anchored in the National Institute for Clinical Excellence (NICE) guidelines for schizophrenia and psychosis in adults. Specifically, it is concerned with NICE quality statement one, which states that "adults with a first episode of psychosis (FEP) start treatment in early intervention in psychosis (EIS) services within 2 weeks of referral"(NICE, 2015). The statement is based on the rationale that EIS can improve clinical and service use outcomes, such as admission rates, symptoms and relapse, for people with a first episode of psychosis. The outcomes set out in the guidelines focus on acute hospital admissions, duration of untreated psychosis and readmission rates, measured by locally collected data. In Chapter 3, I discussed early intervention services and duration of untreated psychosis (DUP) in relation to pathways to care. This chapter now goes on to review and discuss the current evidence of clinical outcomes following FEP both in the context of EIS and generic service use, specifically addressing the questions: a) are there differences by ethnic group in the length of time patients spend in hospital? and b) are there differences in rates of hospital admission and compulsory detention by ethnic groups? In section 4.2, I revisit EIS and examine its role on clinical and service use outcomes and any variations by ethnicity. I review literature on course and outcomes of FEP, examining evidence from longitudinal and follow-up studies, in section 4.3. Finally, section 4.4 includes a discussion of the implications for the findings and methodological considerations.

Although this is not a systematic review, in searching the literature to address the questions proposed above, I used the following keywords: 'early intervention', 'first episode', 'admission', 'psychosis', 'psychotic disorders', 'course', 'outcome', 'follow-up', 'longitudinal', 'cohort' and 'ethnicity', using bibliographical databases including Pubmed, Web of Science, Embase, PsychoInfo and Google Scholar. This chapter reports on the body of work carried out on the topic of course and outcome of first episode psychosis among ethnic minority populations.

4.2 Early intervention for psychosis – the promise of improved pathways to care and better clinical outcomes

As briefly mentioned in Chapter 3, we have observed the establishment of early intervention for psychosis programmes, several of which have been carefully evaluated (Craig et al., 2004a, Birchwood, 1995, Kuipers et al., 2004, McGorry et al., 2009, Bertelsen et al., 2008b). These evaluations show relatively consistent findings of improved clinical and functional outcomes (Bird et al., 2010). But perhaps surprisingly, very few of these studies have explored variation in course and outcome by ethnicity.

The Lambeth Early Onset (LEO) trial was a randomised controlled trial (RCT) comparing an EIS to standard care for people experiencing a first episode of psychosis (Craig et al., 2004b). The clinical trial took place between 2001 and 2003 towards the end of the Aetiology and Ethnicity of Schizophrenia and Other Psychosis (AESOP) study and included as one of its objectives to reduce levels of disengagement and relapse rates among young Black African and Black Caribbean patients. Attention was paid to the ethnic 'mix' of the clinical team and the service included a number of social activities provided out of hours that were expected to be of interest to local young people. There was also an informal drop in to which families were invited and encouraged to attend. The results showed that those in the intervention group were less likely to relapse (β = 0.39; 95% Cl 0.10 – 0.68), and less likely to be readmitted to hospital over an 18-month period (Craig et al., 2004b). Although ethnic differences were not specifically assessed in this study, those in the EI group were also much less likely to drop out of care (β =0.28; 95% Cl 0.12 – 0.73) and less likely to be readmitted $(\beta=0.36; 95\% \text{ CI } 0.04 - 0.66)$ than patients receiving standard care, with this difference being almost entirely accounted for by people of Black African and Caribbean backgrounds. Rates of hospitalisation, access to psychological therapies and medication did not differ across ethnic groups in the intervention arm.

It is worth noting that as LEO became embedded in the wider health system and the numbers of patients and hence work-pressure increased, some of the initial gains in terms of improved sensitivity to the needs of ethnic minorities may have been lost. An audit survey of four inner London EIS teams that included the LEO service about three years after

the conclusion of this clinical trial produced rather less impressive results (Mann et al., 2014b).

More recent surveys in the UK have tended to confirm this view, as for example, the findings of an observational study of 123 (45 White, 35 Black and 43 Asian) EIS patients (Singh et al., 2015a). Black patients were four times more likely than White patients to be compulsorily detained and both Black and Asian patients were more likely to have consulted faith-based institutions prior to accessing services (Singh et al., 2015a).

4.3 Examining course and outcome of psychosis and ethnicity

In the last decade, only a few longitudinal studies have been carried out addressing the course and outcome of psychotic disorder in minority ethnic groups, but the evidence so far points to a mixed picture. Available literature on course and service use outcomes in relations to ethnicity is discussed below.

In a recent systematic review, Chorlton et al. (2011) assessed differences in course and outcome following psychosis for Black Caribbean patients compared with other ethnic groups living in the UK. They included fourteen studies with varying methodology, sample size and follow-up time. In the investigation of ethnic differences in course and outcomes, Chorlton and colleagues considered a broad range of outcomes as reported in the studies included in their review. For one such outcome, service use at follow-up, the majority of studies reported no difference (Mohan et al., 2006, Goater et al., 1999) except (McGovern and Cope, 1991), who found that Black Caribbean patients were less likely to maintain contact with services. For compulsory admission, the majority of studies (Callan, 1996, McKenzie et al., 1995, Takei et al., 1995, Leff, 1988) reported that Black Caribbean patients were more likely to be compulsorily admitted to hospital compared with White British patients, but (Harrison et al., 1999) and (Goater et al., 1999) both found no difference. It is noteworthy that these two studies used crude Black and White ethnic categorisation, which may have influenced the lack of differences being observed. Next, for the frequency and duration of hospital admission, the picture is rather mixed. For example, some studies reported no difference (Callan, 1996, Harrison et al., 1999) while others found that Black Caribbean patients were more likely to experience a higher rates of admissions (McGovern and Cope, 1991) compared with White British patients, and one study (Goater et al., 1999)

reported that White British were more likely to have been admitted than other ethnicities at one-year follow-up, but the effect did not hold at five-year follow-up.

The lack of ethnic differences in Chorlton and colleagues' (2011) review may be largely due to methodological issues sample limitation as well as poorly defined ethnic groups. The limited research on course and outcome of psychotic disorder in minority ethnic groups means that important questions about development of psychotic disorder remain unanswered.

Ethnic differences by hospital admission and diagnosis have been examined in other studies with varying results. For example, Walsh and colleagues (Walsh et al., 2002) compared samples from the UK700 intervention study. They examined ethnic differences in hospitalisation using community and forensic hospital samples (n = 708 vs. 905) respectively, and found no difference between the White and African-Caribbean (adj. OR of 0.86; 95% CI 0.59 – 1.25) groups, but reported that the 'Other' ethnic group were less likely to be admitted to a forensic hospital (adj. OR = 0.23; 95% CI 0.14 – 0.38). By diagnosis they found that those with schizoaffective disorder (adj. OR = 0.46; 95% CI 0.26 – 0.83) and affective disorder (adj. OR = 0.35; 95% CI 0.19 – 0.69) were less likely to be admitted to a forensic.

A few research studies have also been carried out to assess patients' prognoses following first episode psychosis and by ethnicity. Selten and colleagues (2007) conducted a 30-month follow-up of their first episode psychosis incidence cohort in the Netherlands identified in 1997–1999 and assessed the predictors of poor and good outcomes following first contact for psychosis. Using a sample of 139 patients, they reported that the majority of the sample were diagnosed as having schizophrenia (n = 125), followed by mood disorder (n = 17) and other psychotic disorder (n = 25). While they found no difference in poor (Dutch 52% vs. Non-Western 63%) or good (48% Dutch vs. 37% Non-Western, p = 0.21) outcomes by ethnicity, they reported predictors of poor outcomes among those diagnosed with schizophrenia as being male (adj. OR = 3.0; 95%; Cl=1.0 - 8.9), heavy cannabis users and undergoing a long duration of dysfunctioning prior to psychosis. Age at onset, ethnicity, duration of untreated psychosis and socioeconomic status were not associated with poor outcome.

Harrison and colleagues (1999) followed up a sample of first episode patients (n = 166) who were identified in Nottingham between 1992 and 1994 for three years; they assessed the relationship between good prognosis and ethnicity (African-Caribbean and 'Other'). They found no difference between African-Caribbean and 'Other' for mean symptom score (adj. OR = 2.53; 95% CI 0.88 – 7.28). Meanwhile, when they assessed association with disability score in a crude logistic regression, there were no differences between the two ethnic groups (OR = 1.60; 95% CI 0.68 – 4.98) but associations emerged after adjusting for potential confounders e.g. age, gender, social class, DUP and diagnosis (adj. OR = 3.29; 95% Cl 1.03 – 10.57), which suggests some confounding by age, social class and socioeconomic position. Although this study suffered from small sample size, the findings suggest that increased psychosis among African-Caribbean patients is not characterised by differences in symptoms either at onset or over a medium-term course of the disorder. Another UK study by Takei and colleagues (1999) followed patients up for eighteen years and compared outcomes between White and African-Caribbean groups. In this study, a sample of (n = 76) was identified between 1973–1974 using the Camberwell Case Register. They found no ethnic difference by diagnosis but reported that at follow-up, African-Caribbean patients had poor outcomes characterised by longer hospitalisation, higher readmission rates and were more likely to be involuntarily admitted compared with their White British counterparts. This is consistent with findings by (Bhugra et al., 1997), in a London study comparing incidence and outcomes for first psychosis inception in a sample of (38 White, 38 African-Caribbean and 24 Asian) which reported that poor outcomes (defined by relapse, episodic or suicidal ideation) were more common among African-Caribbean (60%) patients than for the other two groups (24% and 17%, G²=15.33, p<0.005).

In a more recent 10-year follow-up of the AESOP sample (n = 532), Morgan and colleagues (in press) found that compared with White British, Black Caribbean patients experienced worse clinical, social and service use outcomes, and Black African patients experienced worse social and service use outcomes. At follow-up, Black Caribbean patients were around 60% less likely than White British patients to have recovered symptomatically (38.7% vs. 55.5%; adj. OR = 0.42; 95% Cl 0.22 – 0.80), and were around two times more likely to have a non-episodic course (adj. OR = 2.27; 95% Cl 1.09 – 4.76). Indicators of social outcomes suggest that only 1 (of 21; 5%) Black African patient, 3 (of 54; 6%) Black Caribbean patients,

and 9 (of 63; 14%) White British patients who were unemployed at baseline were employed at follow-up. In addition, the numbers and rates of admissions were greater in Black Caribbean and Black African patients than in White British, with patients in both groups admitted at a rate of around 0.36 per year (or once every three years) compared with a rate of 0.26 per year (or once every four years) for White British patients, constituting a 20% higher rate than White British patients. Further, admissions tended to be longer. Compared with a median length of 37 days for White British patients, the median inpatient stay for Black Caribbean patients was 62 days (rank sum test z -3.53, p < 0.001) and for Black African patients it was 54 days (rank sum test z -2.12, p < 0.034) (Morgan et al., in press).

4.4 Limitation of current research

While the breadth of research on this topic is limited, the varied nature of the available evidence makes drawing a definitive conclusion on course and outcomes of psychosis among black and minority ethnic groups difficult. This review has picked up some methodological issues that contribute to the variations in the research findings. For example, a number of studies were only able to include small sample size (Goater et al., 1999, Harrison et al., 1999) which may be due to attrition rates in the follow-up studies, hence making the reported estimates prone to imprecision. Furthermore, and as highlighted in the previous chapters, many studies with the exception of Morgan et al. (in press) have focussed on narrow ethnic minority categories, e.g. African-Caribbean, Asian, which may also be due to small sample size, and therefore it is difficult to disentangle differences -- if any -- between White British and Black African and Black Caribbean groups. In addition, the methodological quality of the research to date also varies in design and length of follow-up.

4.5 Summary

To summarise, evidence from evaluation of early intervention for psychosis indicates that such intervention is effective for reducing poor clinical outcomes but research is very limited into whether early intervention service improves outcomes for minority ethnic groups. It is

clear from the literature above, that there is a dearth of research on course and outcome of psychosis among ethnic minority groups, particularly in recent times. Studies carried out pre-early intervention service investigating the course and outcomes of psychosis among minority ethnic group patients present heterogeneous evidence. Therefore, firm conclusions cannot be drawn from the available evidence and so the question of whether ethnic differences operate in the course and outcome of psychosis remains unclear.
5 Chapter 5: At risk population and the mental health provider

5.1 Background

In this chapter, some contextual details about the population under investigation, e.g. socioeconomic factors and the set-up of the local mental health services are considered, before going into the methodology of this thesis. It is useful to consider the socioeconomic contexts of the two catchment areas included in this study through the lens of the social domains that were included in the literature review in Chapter 2. These social domains can be considered as factors that might be contributing to the incidence of psychosis in the geographical areas being investigated.



5.2 Contexts

London borough of Southwark

London borough of Lambeth

Figure 5.1: The study catchment areas (Source: Google Wikipedia, Se16boy)

5.3 The Denominator: population of Lambeth and Southwark – Socioeconomic context

This study was conducted within two inner-city areas served by the South London and Maudsley NHS Foundation Trust (SLaM), these are the London boroughs of Lambeth (population 303,086) and Southwark (population 288,283) (ONS, 2011a). Both boroughs have risen in population since the 2001 census, 11% in Lambeth and 12% in Southwark. There is a high proportion of ethnic minority groups in both boroughs, principally African (11.6% Lambeth; 16.4% Southwark) and Caribbean (9.5% Lambeth; 6.2% Southwark). The 2011 census (ONS, 2011a) reported the overall ethnic composition of the two catchment areas as follows:

Ethnicity	% Lambeth	% Southwark
White British	39	39.7
White Irish	2.5	2.2
White Gypsy	0.1	0.1
Other White	15.5	12.3
Mixed White & Black (African)	1.4	1.3
Mixed White & Black	2.7	2
(Caribbean)		
Mixed White and Asian	1.2	1
Mixed Other	2.3	1.9
Indian	1.6	2
Pakistani	1	0.6
Bangladeshi	0.7	1.4
Chinese	1.5	2.8
Other Asian	2	2.7
Black African	11.6	16.4

Table 5.1: Ethnic breakdown of at risk population 2011 census

Black Caribbean	9.5	6.2
Other Black	4.8	4.2
Arab	0.6	0.8
Any Other Ethnic group	1.9	2.4

5.3.1 London Borough of Lambeth

The London borough of Lambeth is one of the most diverse local authorities in England and Wales. The borough is the third most populated borough in Inner London after Wandsworth and Newham boroughs of London (ONS, 2011a). 66.1% of the Lambeth population were UK born while 33.9% were born outside of the UK. The population has changed due to internal migration across this study period (May 2010 – April 2012), with 33,600 people (22%) moving into the borough and 35,900 (24.5%) moving out (London Borough of Lambeth, 2012).

The borough has a striking cultural and ethnic diversity, with 30% of residents describing themselves as belonging to non-white ethnic groups. Based on the 2011 census, Lambeth has a low percentage of people with an English-only identity and the proportion of White British people has decreased from 50% to 39% in the last 10 years. There are wellestablished Black and Irish communities with newer populations of Portuguese, South American and African residents, as well as a high and changing refugee population. Approximately 150 languages are spoken in the borough. After English the main languages spoken are: Portuguese, Yoruba, French, Spanish and Twi (London Borough of Lambeth, 2012).

A wide range of indicators highlight Lambeth as one of the most deprived areas in England. According to the 2010 indices of multiple deprivation, Lambeth ranked 14th most deprived borough in England and the 5th most deprived area in London, a worsening position since 2007, when the borough ranked 19th most deprived in England (Greater London Authority, 2011). That said, the borough also has some areas of affluence, for example 39.9% of Lambeth households are not deprived in any dimension (ONS, 2011a). Compared to national averages, the area has lower socioeconomic status, higher unemployment, higher death

rates from suicide and higher use of acute inpatient mental health services (London Borough of Lambeth, 2012).

As evidence has shown in the previous chapters, unemployment is a major risk factor for the onset of psychosis. This is an issue in Lambeth, with 7.6% of its working age population out of work according to the ONS 2012; this is marginally above the London average of 7.4%.

Educationally, 37% of Lambeth 16-year-olds did not achieve five GCSEs grade A*-C in 2011–2012, although this figure is an improvement on the previous three years, where 47% did not attain the aforementioned grades compared with the London average of 38% (Department of Education, 2012).

Nearly 6 per 1000 household were accepted as homeless in Lambeth during the periods 2010–2011, worse than the London average of 3.06 per 1000 household (New Policy Institute, 2014).

5.3.2 The London Borough of Southwark

The London borough of Southwark is the 12th most deprived area in London and 41st in England, showing a relative improvement from the 2007 ranking as 9th in London and 26th in England (Greater London Authority, 2011). It has the highest proportion of residents in the country who were born in Africa (12.9%) and over 300 languages are spoken in Southwark, making it an ethnically diverse population (London Borough of Southwark, 2011). Akin to Lambeth, unemployment is also high in Southwark, as 9.5% of its economically active population are unemployed compared to the London average of 7.4%.

Educational attainment at age 16 in Southwark is poor as 41% of 16-year-olds did not achieve five GCSEs grade A*-C in 2011–2012 compared to the London average of 38% (Department of Education, 2012). In addition, 4.04 per 1000 households were accepted as homeless in Southwark in 2010–2011 compared to the London average of 3.06 (New Policy Institute, 2014).

5.4 The Mental Health Provider

The South London and Maudsley (SLaM) NHS Foundation Trust is the largest mental health provider in Europe (Stewart et al., 2009), serving a population over 1.3 million across the London Boroughs of Lambeth, Lewisham, Croydon and Southwark and providing specialist services in Greenwich and Bromley. SLaM provides a wide range of multi-disciplinary integrated and specialist services across Child and Adolescents, Forensic, General Adult, Older Adults Addictions and Learning Disabilities clinical groups, employing approximately 4,500 members of staff. There is at least one hospital per borough providing inpatient services to acutely unwell patients; a large proportion of SLaM services are communitybased. The Maudsley and Lambeth Hospitals serve the populations of the two boroughs under investigation in this study. Other hospitals within the Trust are the Bethlem Royal and Lewisham University hospitals, serving the boroughs of Croydon and Lewisham respectively. Each year, there are over 30,000 patients referred for assessment and treatment in SLaM (SLaM, 2014).

SLaM is one of the pioneering NHS Trusts to implement the use of an electronic medical record system over the last decade. By 2006 the Trust was effectively paperless, having transitioned to the electronic Patient Journey System (PJS). The PJS is a single purpose-built system that integrates all clinical notes/records used across all Trust services. PJS captures information recorded throughout patients' journeys through the clinical services including demographics, dates, contact information, detail of referrals and transfers, detailed clinical assessments, care plans, medication and clinical reviews (Stewart et al., 2009). The system is used and maintained by multi-disciplinary professionals and consists of both structured elements (e.g. checkbox, dates) and unstructured free text (including written assessments, correspondence letters and progress notes). Specific assessments and outcome measures are also available on PJS such as the Mini Mental State Examination and the Health of the Nation Outcome Scales (HONOS).

One of the questions this study sought to examine is whether the introduction of early intervention services has had significant impact on the pathways to care for patients with first episode psychosis. SLaM has well-established early intervention services across its catchment areas, i.e. one in each borough. For example, the Lambeth early onset (LEO) service has been operational since 2004 (Craig et al., 2004a), while the Southwark early

intervention team (STEP) was set up and running in 2007. The South London and Maudsley NHS Trust is a leading NHS institution that actively engages in research through working closely with its academic partner the Institute of Psychiatry, Psychology & Neuroscience (IoPPN), King's College London. The IoPPN is Europe's largest centre for research and post-graduate education in psychiatry, psychology, and neuroscience (SLaM website, 2014). In 2010, SLaM in collaboration with IoPPN launched a new research infrastructure (Clinical Record Interactive Search (CRIS) System) in the form of a case register, bringing it on the same platform as some of the well-known international case registers, such as Danish, Italian, Israeli, Japanese, Australian and American psychiatric case registers (Perera et al., 2009).

5.5 Case registers: History

Case registers have been long running in mental health research. In Europe for example, earlier national registers were created in Norway, followed by Denmark and Iceland in the 1930s. In the 1940s, registers were created in Israel, Japan, New Zealand and many in the USA (Perera et al., 2009). The 1960s saw the creation of regional registers in the UK (Aberdeen and Camberwell), Australia (Victoria) and Germany (Mannheim). The Italian and Dutch registers were introduced in the 1970s.

The Danish psychiatric central register was arguably the only ongoing case register in Europe until recently, as the Norwegian and Icelandic registers were discontinued in the 1980s (Munk-Jørgensen et al., 1993). As a national database of psychiatric records, the Danish register has served and proved its benefit for service planning, epidemiological and longitudinal studies, investigating course and predicting outcomes of illness.

In the two decades leading up to the 2000s, the popularity and use of case registers in mental health research saw a decline (Perera et al., 2009). However, the new technological advancement in computer software and opportunities for health care providers in using electronic clinical records has helped case registers become popular again as a research resource.

Stewart and colleagues (2009) developed arguably the largest regional case register in Europe, at the South London and Maudsley NHS Foundation Trust Biomedical Research Centre (Stewart et al., 2009). The Clinical Record Interactive Search (CRIS) system was created using the electronic clinical records at SLaM.

Case registers for mental health research have been defined as 'patient-centred longitudinal records of contacts with a defined set of psychiatric services originating from a defined population' (Ten Horn et al., 1986). They contribute significantly to epidemiological research, service planning and utilisation, operational and administrative research (Perera et al., 2009). The availability of information on risk factors of rare disorders such as psychosis; demographics; and social, clinical and environmental characteristics about individuals using mental health services with any diagnosis makes it possible to interrogate case registers for conducting studies on topics such as course and outcomes (Stewart et al., 2009).

In the next chapter, the methodological approach employed in this thesis is discussed. I will give a detailed description and careful attention to detail in the use of the CRIS case register for identifying first episode psychosis cases that form the sample in this study.

6 Chapter 6: Study Methodology

6.1 Synopsis

Data for this thesis were collected as part of a larger programme of research on first episode psychosis (the Clinical Record Interactive Search - First Episode Psychosis Study: CRIS-FEP). To begin with, this chapter summarises key findings from the literature reviews from previous chapters. Subsequently, the general aims and justification for the methodology used are outlined (sections 6.2 and 6.3). The CRIS case register as research infrastructure is discussed in section 6.4. Next, the general methodology for the larger study is discussed (section 6.5), including the study design, the screening procedures employed for identifying cases and data extraction. The battery of the measures used within this thesis is described and outlined in section 6.5.5. Section 6.6 describes in more detail the automated data extraction procedures employed in the investigation of two-year follow-up outcomes in this PhD. Finally, the methods of statistical analyses are summarised (Section 6.7).

Key findings in the literature review set out in previous chapters can be summarised as below:

- Over the last decade, studies have repeatedly reported higher incidence rates of psychosis among Black Caribbean and Black African populations in the UK compared with White British populations. The reasons for the higher rates remain unclear, partly due to methodological issues which limit causal inferences that can be drawn.
- 2. Most previous studies have examined pathways to care for broadly defined ethnic groups with little consideration of specific ethnic categories that mirror the base population, e.g. as defined for the UK Census 2001 and 2011. With this caveat, most studies show higher rates of compulsory hospitalisation among people from minority ethnic groups. The explanations for these higher rates remain unclear.
- 3. Early intervention services have shown promising signs in improving outcomes for psychosis, but there remain gaps in understanding their impact on pathways into care, especially by ethnic groups.

4. There has been less research on the course and outcome of psychotic disorder in ethnic minority groups but the available evidence points to a mixed picture, with some studies showing no difference by ethnic group while others suggest Black ethnic groups were at increased risk of hospital admissions and poor clinical outcomes.

It is on these conclusions and the limitations of previous research that the methodology of this thesis is based.

6.2 Aims and justification of study design

The overarching aim of the work presented in this thesis is to investigate patterns of the incidence of psychosis over time, pathways to care and service use among patients of Black African and Black Caribbean ethnicity with first episode psychotic illness. Individuals with first episode psychosis were drawn from both hospital and community settings of the South London and Maudsley NHS Trust (SLaM) General Adult and Forensic services. These services serve patients aged 18 to 64 years. This thesis comprises two phases – baseline and follow-up – and data were collected as follows.

Table 6.1: Study time points

Point of first contact for psychosis (case	Follow-up period (longitudinal data
identification period)	collection)
May 2010 – April 2012	Up to April 2014

At first contact for psychosis (baseline), cross-sectional data on sociodemographic and clinical presentation, compulsory detention and pathways to care at first contact were collected. Two questions were considered: (1) is the discrepancy in incidence rates between ethnic groups any larger (or smaller) now than it was 15 years ago? and (2) are there any differences in pathways to care for FEP cases compared to 15 years ago and is this associated with the introduction of EIS?

The aims of the first phase (baseline) were:

 To estimate the incidence of psychosis by ethnic groups during a two-year period (May 2010 – April 2012) in a well-defined catchment area in south London.

- To compare pathways to care, hospital admission (compulsory vs. noncompulsory) and source of referral (GP, A&E and others) at first contact for psychosis by ethnic groups.
- To compare these pathways to care (compulsory vs. non-compulsory) and source of referral (GP, A&E and others) with those reported in the earlier AESOP sample and between those using early intervention service or not in the current study sample.

Data relating to number of hospitalisations, number of compulsory admissions and total number of days spent in hospital were collected longitudinally throughout the follow-up period. Collection of this data was intended to address two questions: a) are there differences by ethnic group in the length of time patients spend in hospital and compulsory admission during the follow-up period? and b) are there differences in rates of hospital admission and compulsory detention by ethnic groups during the follow-up period?

The follow-up phase aims were:

- 1. To estimate rates and rate ratios of hospital admission and compulsory detention during the two-year follow-up by ethnic group.
- 2. To estimate differences in length of hospital stay by ethnic group during the twoyear follow-up.
- To compare clinical and service use (hospital admission, compulsory admission) between those using early intervention services and standard community care during the two-year follow-up.
- 4. To determine whether there are ethnic differences in clinical and service use outcomes during the two-year follow-up in relation to early intervention service use status.

This study is not dissimilar to previous research in its methodological approach and data collection. It differs from previous research in the following respects:

I. It includes a larger number of first episode psychosis cases, especially from ethnic minority groups, than any previous first episode study in a single site.

- II. It offers larger scope for collecting a wider range of variables, i.e. baseline data on incidence of psychosis, pathways to care and longitudinal follow-up data over a fouryear period within the same study.
- III. Using a mental health case register as the primary sampling source minimises the problem of attrition since requesting cases to consent to follow up is not required; therefore, attrition was only of those who were discharged from SLaM services before the end of the study period.
- IV. Psychosis case ascertainment is more comprehensive, as a three-stage screening procedure is used.
- V. It forms the basis of analysing data relating to the effect of early intervention services on pathways into and through care.

Given the dearth of research studies on course and outcomes by ethnicity, this study advances our understanding in this area, by analysing data on service engagement outcomes over a two-year follow-up period by ethnic groups.

6.3 Justification of administrative incidence with a follow up cohort study design

To assess possible causal relationships between risk factors and onset of disorders, a cohort study design is regarded as the most robust design. This design allows for the assessment of exposures prior to onset of disorder and can therefore potentially disentangle the direction of causal effects; it also allows for investigating exposures and multiple outcomes (Susser et al., 2006). However, it can be a challenging design, practically speaking, for rare disorders such as psychosis, which with a lifetime prevalence of around 1% - 3% and one year incidence of around 30 per 100,000 person years at risk (Van Os et al., 2009), requires an extremely large number of people to be followed up, possibly from birth. Since this would be an expensive endeavour, and disorders such as psychosis are also often diagnosed in adulthood, an alternative design is required.

An example of such an alternative is a case-control study design, which is more efficient for studying rare disorders like psychosis because it assesses both outcome and exposures

following occurrence of disorder (Susser et al., 2006). However, if not well-designed this type of study is subject to a number of flaws and shortcomings. For example, identifying and recruiting controls in a random manner from the same source population may be challenging (Prince et al., 2003, Susser et al., 2006). Furthermore, a case-control design is criticised for information bias of two types: (a) recall bias from the participants since they are being asked information that occurred prior to their experience of the disorder and b) observer bias which may occur when a researcher / interviewer is aware of the participant outcome status (Prince et al., 2003, Susser et al., 2006). Finally, a case-control design is able to assess only a single outcome whereas a cohort design has the ability to assess multiple outcomes (Prince et al., 2003).

With this in mind, this thesis employed an administrative incidence study design with a follow-up cohort study of outcomes of psychoses that 1) collects cross-sectional data for the period following onset of psychosis and 2) allows the investigation of the multiple outcomes of interest prospectively.

Administrative data is data collected routinely for monitoring or regulatory purposes (Ward, 2005). Hospital clinical records are an example of administrative data, which in mental health is often described as a case register. Administrative data are an attractive resource for epidemiological studies.

As such, many high-quality epidemiological studies in mental health have used cohorts constructed using case registers, including (Susser et al., 2010, Smith et al., 2006, Mortensen, 1995). An administrative incidence study design with a follow up cohort study of outcomes permits existing records to be used for determining how many individuals meet entry criteria of a study (Susser et al., 2006) and outcomes. It is useful for case ascertainment, i.e. sorting people into affected and unaffected, since all the events (exposure, latent period and outcomes) have already occurred. For these reasons, administrative incidence study design with follow-up cohort is less expensive and more time saving (Prince et al., 2003). It is also powerful in dealing with multiple outcomes, which is the main focus of the follow-up phase of this thesis and is described in detail below.

As mentioned previously, cohort studies allow us to draw inferences about associations between exposure and outcome, if they are well-designed and meet specific criteria. For

example, Susser and colleagues (2006) outline six key steps in ensuring a sound design of a cohort study: a) articulate questions; b) define source population; c) measure the exposures; d) follow the cohort; e) classify the outcome; and f) analyse the data (Susser et al., 2006). Measures taken to address these steps are described in detail in this chapter (sections 6.2, 6.4, 6.5, 6.6 and 6.7). Measures taken to address and reduce the likelihood of selection bias and confounding are outlined in sections 6.5.3 and 6.7 respectively, and these are discussed further in the Results and Discussion chapters (Chapters 7 - 12).

At this point it is important to mention that the study design employed for the work presented in this thesis, taking into consideration steps a-e above, is solely based on a regional psychiatric case register, namely the Clinical Record Interactive Search (CRIS) System. However, it is useful to put into context some background and description of the CRIS functionality as a historical case register, before discussing the study methodology in more detail.

6.4 CRIS: a case register and research platform

As discussed in the previous chapter, case registers have been pivotal to psychiatric epidemiology over many decades and as part of this tradition, this thesis utilised the CRIS system to define its sample population. CRIS is an innovative technological application built within the National Institute for Health Research (NIHR) - Biomedical Research Centre awarded to the South London & Maudsley NHS Foundation Trust. CRIS transforms clinical information in the electronic health records (locally known as electronic patient journey system, ePJS) (described in Chapter 5) so that it is anonymous. Whilst CRIS ensures that significant epidemiological data are available without compromising anonymity, some personal identifiable data are truncated, e.g. only the first half of the UK postcode and only month and year of birth are included (Stewart et al., 2009). In addition, personal information such as full name, address and NHS number are masked out and replaced with 'zzzzzzz' (see Figure 6.1).

🖉 CRIS - Search Results - Win	dows Internet Explorer	provided by SLaM IC	T Department										_ & ×
🚱 🗢 🙋 http://cris.slar	n.nhs.uk/search/show-result	s.action					- 🖻 •	4 ×	🔎 Live Sea	arch			P -
File Edit View Favorites	Tools Help												
🖕 Favorites 🛛 🚖 🤌 Web	Slice Gallery 🔹												
🔏 CRIS - Search Results								🟠 •	5 - 🗈	🚔 🔹 Page 🕶	Safety 🕶	Tools 🕶	?
× Find: accomodation		Previous N	lext 📝 Options	•									
	gout												<u>•</u>
			Events										
			Event										
Primary Diag (Axis 1a)	Team/ward name	e Spell number	Date	Team/ward name	Notes							Spell n	umb
aDIPr05	aDILo13	aDISp29	eEVSt08	eEVLo17	eEVCo33							eEVSp	49
episode	Ab I - East ream Chaucer	1.0	2011-06-29	Ab 1 - Eds L ream Chaucer	[10177380] ABT Assessment: by Dr Lipinski who omplications. Sh scratching hersel to the extent of 3 stratching hersel to the extent of 3 a funny way whe the signing on to hears noise in her way understandir has not worked s leave to remain a same situation. S she is going now would end her life ZZZZZ was born She reported tha wives. She went sitting her O'level the Nigerian film g she left for the U husband when he visit and while vis make family. She were trving for an	ZZZZZ ZZZZ o was concern ms. Presentin try she has de reported thi f all over her b showing what s made worse the Home Offic ears most of ears most of g that this mi ince the birth nd has been a g that this mi nice the birth nd has been a g that this mi he states thai he states thai he states that he states that the comes fi to school at ti s. She then al oroduction con was visiting f iting she fell p however lost	Z Reasor ed that it as the is soldy three the second second second appearement of the second second second second second second second second second second second second second second second second second second second	n for Re she mig physic itchy a uugh ou d to be . She s n on be repoid t. She s and s ue to s on fours on fours	ferrai: ZZ repo al and ps ZZZ repo al and ps ZZZ repo al and ps the the two spaces of the spaces and the spaces as manual table of the spaces as manual table of the spaces as the space of the spaces as the space of the space o	2222 was reperiencing reted that sin ychological e and was in erview. She u e and was in they back, si they have n rthey have n rthet that she was e reported th e does not h d who is in 1 and thus fe e reported th e does not h d who is in 1 and thus fe e reported the r her son sh ersonal Histo ental milester f ather had d a year bef hool before j ed as an act et her currer ncouraged h ad to stay an ofking while I four, years of	ferred Ce ideed went tating inds it her in aduced e now in a nat she ave the e e the lels like e e two ore g oring oring oring the the the the the the the the the the	1.0	
Done			1		4				😜 Inte	rnet	4	• 🔍 100°	% • //
💐 Start 🗀 2nd review for 2	Working second	🥖 Get DAAT/LA fr	🖉 CRIS - Searct	1 🟉 Employee Online	😡 Inbox - Microsof	🔄 Step%20by%	%20 🤨	Ascertai	ning Res	CRIS screens	shot «	· 🗄 🛛 🗸	14:38

Figure 6.1: CRIS front end

One of the distinctive features of CRIS is its ability to specify populations of given disorders, and to interrogate level of service use, health state, course and outcomes of a given disorder (Stewart et al., 2009). As Stewart and colleagues put it, 'broadly speaking, a fit for purpose case register is one that is a patient-centred longitudinal record of contacts with a defined set of a psychiatric services originating from a defined population' (Stewart et al., 2009). It is this that CRIS offers, as it enables researchers to use the Trust electronic health record system for research, by conducting automated, anonymised searches on structured (date, numerical, demography etc.) and unstructured (free text) information contained in electronic case notes, where much of the clinical record is kept. The system is novel in that it accelerates and expands the scope of mental health research significantly, giving researchers answers in days rather than months. CRIS holds over 250,000 real-time mental health records and is updated daily, making it easier to identify patterns of data, such as populations of patients per disorder and what treatment worked for some and not for others (SLaM, 2014). CRIS also enables output from record searches to be extracted into database format such as STATA, SPSS or Excel, which are commonly used by researchers.

The system exists in two forms – (a) web-based CRIS and (b) Structured Query Language CRIS (SQLCRIS). The web-based CRIS system interface has some properties of a 'flat-file' system in which every piece of data about each patient is stored together in a single row of data, as shown in Figure 2a above. CRIS is originally accessed through the web-based interface, which is also the common way of accessing the register. However, it can also be held and accessed as a relational database, i.e. SQLCRIS (Fig. 2b).



Figure 6.2: SQLCRIS front end

SQLCRIS is a relational database. A relational database is a collection of tables of data items, all which are formally described and organised according to a relational model (Codd, 1998). This makes SQLCRIS extremely flexible and powerful in combining and selecting data by using commands and codes. SQL has been widely used in large government departments, businesses and academia since the 1960s (Tulloch, 2013). The ability to generate an output dataset that contains only the data that is needed, arranged exactly as required for analysis, makes SQLCRIS more desirable compared to the CRIS web-based interface, which is limited in its ability to select, combine and arrange data (Tulloch, 2013). SQLCRIS is particularly useful and appropriate for generating datasets for longitudinal studies such as this thesis. Details of how SQLCRIS was employed in generating follow-up data for this study are discussed extensively in section 6.6. Other advantages of SQLCRIS include: codes and commands, when correctly written, are reproducible and will produce the same results every time they are run, and codes can also be simply referred to by other codes without the need to copy and paste. SQL can also manipulate data, reducing the need for manual data cleaning and consequently minimising human error and increasing confidence in reliability of the data (Tulloch, 2013).



6.5 Methodology of the CRIS-FEP study

Figure 6.3: Structure of data collection

6.5.1 Background and study design

6.5.1.1 Relationship to previous study- AESOP

The CRIS-FEP study methodological approaches are based on those used in the Aetiology and Ethnicity in Schizophrenia and Other Psychosis (AESOP) study (Fearon et al., 2006a). In the last decade, the AESOP study is arguably the gold standard study in the environmental and social aetiology of psychosis and ethnicity in England to date. In summary, the AESOP study was a large prospective study of first contact psychosis cases conducted between 1997 and 1999 in three catchment areas in England, namely, southeast London, Nottingham and Bristol. The study revealed that incidence of psychosis was higher among migrant groups compared to White British patients (Fearon et al., 2006a, Kirkbride et al., 2006). In addition, the study highlighted that crisis pathways to care (such as police involvement, coercive admission) were more common among migrant groups compared with the reference population (Morgan et al., 2005a, Morgan et al., 2005b). Therefore, AESOP provides a methodological template to re-examine the evidence on incidence, pathways to care and course and outcomes for FEP patients.

The current study employs an incidence study design with a follow-up cohort study of outcomes of psychoses using the Clinical Register Interactive Search System (CRIS) to identify all probable cases of psychosis (ICD-10 F20-29 and F30-33) presenting to General Adult and Forensic services within two catchment areas of South London and Maudsley NHS Foundation Trust (i.e. Lambeth and Southwark). Cases were identified and cross-sectional data were collected over two years between 1st May 2010 and 30th April 2012. The cohort is then followed up for a further two years as described in table 6.1 above.

6.5.2 Screening and case ascertainment of psychosis cases

Individuals were eligible for inclusion if they were:

- i. Resident in the London boroughs of Lambeth or Southwark
- ii. Aged 18-64 years (inclusive)
- iii. Experiencing psychotic symptoms during the study periods 1st May 2010 30th April 2012
- iv. Scored at least 2 or more for psychotic symptoms as assessed by the Psychosis Screening Schedule (Jablensky et al., 1992a)

Individuals were excluded if they met any of the following criteria:

- i. Under age 18 or over 64
- ii. Not currently a Lambeth or Southwark resident (according to SLaM records)

- iii. Transient psychotic symptoms resulting from acute intoxication (i.e., psychotic symptoms that occur during or after psychoactive substance use and are explained on the basis of acute intoxication alone)
- iv. Evidence of an organic cause of psychotic symptoms
- v. Evidence of previous contact with secondary mental health services for psychotic symptoms prior to 1st May 2010

In order to define the source population for the study, SQL commands were written to interrogate the SQLCRIS to pull back the clinical records of all patients who met basic inclusion criteria and presented to any adult general psychiatric services. Researchers subsequently screened the clinical records to classify individuals who met the inclusion criteria for the presence of psychosis.

The steps for identifying and including patients as first episode psychosis involved running weekly queries in SQLCRIS in the following order:

- i. Define search week and borough, e.g. '1-May-2010 to 7-May-2010, Lambeth and Southwark'.
- Apply search terms, e.g. postcode, age, symptoms, where the terms 'psychos*', 'psychot*', 'delusion*', 'voices' and 'hallucinat*' may appear in a medical note, correspondence letter or ward progress note.
- Identify all probable psychosis cases that presented to services in the search period.
- iv. Steps (i-iii) return 60-70 patients per borough per week.
- The above queries return individual patient records with unique identification numbers (BRCIDs) including ward/community progress notes, and correspondence attachment.
- vi. The patient records are presented in an Excel file which is then screened and reviewed by researchers to identify first episode psychosis cases. Individuals were included if they scored 2 or more psychotic symptoms on the Psychosis Screening Schedule (PSS) (Jablensky et al., 1992a). This means that the individuals have psychotic symptoms that are persistent or severe and are of

clinical significance as assessed by clinicians. Other inclusion criteria were also verified in the notes before an individual was included as a case.

SQL commands for extracting data at this stage are presented in Appendix 1.

6.5.3 Reliability

At this stage of the case identification and data collection procedure, it is important to ensure reliability of data collection, since data were collected by more than one researcher. Table 6.2 presents the inter-rater reliability tests that were carried out among researchers. For the Psychosis Screening Schedule (PSS) (Jablensky et al., 1992a), i.e. when screening for FEP cases and the Operational Criteria (OPCRIT) checklist in generating diagnostic categories for the cases. Reliability was achieved and tested through training and a formal inter-rater reliability exercise. The CRIS-FEP researchers conducting the clinical record screening were required to undergo an OPCRIT and PSS training course to ensure that symptoms were rated in a standardised way. The successful completion of the training course was a prerequisite for researchers using the OPCRIT and PSS.

Following the identification of FEP cases from the steps above, satisfactory PSS inter-rater reliability was confirmed within the CRIS-FEP study group with me and other researchers conducting the original case register screening to identify FEP cases independently. Each rater then swapped an average of seventy cases and repeated the procedure by reviewing one another's screened and included cases. A Kappa score of 0.78 (p<0.001) was achieved between the raters. For the OPCRIT rating a Cronbach alpha r = 0.95 was also achieved between raters.

Two primary researchers including myself subsequently conducted a secondary review of all the included cases from the first stage screen to ensure cases met all inclusion criteria. An inter-rater reliability test was carried out between the two and a Cohen's Kappa coefficient of 0.77 (p <0.01) was achieved. In addition, there were consensus meetings among research staff and the overall project supervisor to agree consensus diagnosis following detailed discussions for all ambiguous cases after the secondary reviews.

Finally, some postcodes overlap across two or more local authorities. To address this, the patients' addresses (Lambeth or Southwark) were verified using the Public Health England postcode widget (Public Health England, 2004). The widget provides information by percentages in which local authority the postcode lies. A consensus threshold of 60% and above was reached as acceptable within our research team to include the residence area that falls within the study catchment. Furthermore, data for the middle layer super output area (MSOA) were collected and linked to individual postcode. The added advantage of this is that MSOA is a neighbourhood-level piece of information (including address) which fits within the boundaries of a local authority (ONS, 2011c). Using the MSOA information during the secondary reviews, individuals whose addresses did not match MSOA information were excluded.

Instrument	Outcome	Number of	Number of	Reliability scores /
		cases per	raters	statistics
		rater		
Psychosis	Psychotic	95	3	Kappa 0.78, p<0.001
Screening	symptoms			
Schedule				
OPCRIT	Diagnostic	10	4	Cronbach alpha r =
	categories			0.95
MRC	Ethnicity	30	3 plus 1 blind	Kappa 0.87, p<0.001
Sociodemographic			rater	
Schedule				
Secondary	Eligibility checks	40	2	Kappa 0.77, p<0.001
screening reviews				

Table 6.2: Inter-rater reliability for rating scales

6.5.4 Main data collection measures

A comprehensive battery of measurement tools was used to extract data from the CRIS system, a process which took 1.5 hours per patient. Measures pertaining to the aims and hypothesis of this thesis are described below.

Psychosis Screening Schedule (PSS) (Jablensky et al., 1992a)

The Psychosis Screening Schedule is designed to screen and identify functional psychosis in first contact patients. It was developed for use as a screening instrument in the World Health Organisation Ten-Country Study (Jablensky et al., 1992a) and can be used for both face-to-face interviews and case note screenings. The PSS consists of six sections: i) demography, ii) study inclusion/exclusion criteria, iii) list of psychotic symptoms, iv) change in personality and behaviour, v) history of hospitalisation for psychosis and vi) hospital diagnosis. For the purpose of this study, we amended the PSS and restricted it to sections i to iv since diagnosis categories were rated separately using the OPCRIT and history of hospitalisation will be collected longitudinally for the cases included in this study over a four-year period. Researchers rated 'yes' or 'no' on the screening items except on the demographic information. To screen positive and be included, the patient must score at least two or more yeses on sections iii and iv. The PSS has been used widely in many previous studies for which interviews with participants were not possible, and it has been found reliable (Fearon et al., 2006a, Coid et al., 2009, Harrison et al., 1997, Bhugra et al., 1997). The schedule is appropriate for ascertaining positive psychotic symptoms from the clinical records (see Appendix 2).

Medical Research Council (MRC) Socio-demographic Schedule (Mallett, 1997)

Sociodemographic data were collected using the Medical Research Council Sociodemographic Schedule(MRC-SDS) (Mallett, 1997). MRC-SDS captures detailed information on current and past social history. An amended version of the MRC-SDS was used to collect data relating to age, gender, and ethnicity, place of birth, education qualifications, living circumstances, employment history and relationship status.

For analysis, data on education qualifications was collapsed as follows: 'School no GCSE', 'School with GCSE', 'A levels/Further education' (tertiary, vocational) and 'University' (undergraduate and postgraduate).

Data on living circumstances were collapsed as 'Alone' (alone, alone with children), 'Family/friends' (partner, partner with children, parents, other family, friends) and 'Other' (e.g. homeless refuge) or 'Not recorded'.

Similarly, data on relationship status were collapsed for analysis as follows: 'Single', 'Married/Steady relationship' (married, living with someone, in a steady relationship) and 'Divorced/separated' (including widowed) or 'Not recorded'.

Employment status data was grouped as 'Unemployed' (including economically inactive), 'Student' and 'Employed' (part-time employee, full-time employee, self-employed) or 'Not recorded' for analysis.

For housing tenure, data were collapsed as follows for analysis: 'Privately owned' (privately owned – self, privately owned –family), 'Rented' (rented – private, rented – government) and 'Other (e.g. halls of residence, homeless refuge).

Ethnicity

Central to the aims of this thesis is ethnicity. Earlier UK studies (Ghali et al., 2013, Agius and Ward, 2009) suffered methodological limitations specifically around ethnicity ascertainment in the population at risk and in identified cases, and as such many studies crudely defined ethnicity as 'Black, White or Asian'. In this study, populations at risk and case ascertainment were estimated according to the 2011 Census ethnic categories for Lambeth and Southwark as shown in Table 5.1 (section 5.3).

Since ethnicity is a multi-faceted phenomenon, the Office of National Statistics (ONS) recommends that a number of categories be employed in ascertaining ethnicity, including country of birth, nationality, language spoken at home, parents' country of birth in conjunction with country of birth, skin colour, geographical region (e.g. Saharan and sub-Saharan Africa), racial group and religious group (ONS, 2003). The ONS also suggests that a combination of these approaches is necessary to obtain the most accurate picture. In the 2011 Census survey, respondents were asked to self-ascribe to one of 18 ethnic categories, taking into account the definitions aforementioned. As this study is solely based on case register, and although the CRIS system captures ethnicity in its structured field, measures were taken to reduce biases when assigning cases' ethnicities. This involved researchers

reading clinical notes looking for information that matched the aforementioned ONS definitions and subsequently assigning patients to an ethnic group. A high inter-rater reliability was achieved (Kappa 0.87, p<0.001). Previous studies (Bhugra et al., 1997, Fearon et al., 2006a) have employed case note ethnicity ascription where participants were not available for interview.

All ethnicities were included in order to:

a. Compare incidence rates of psychosis across all ethnicities,

b. Compare pathways to care by ethnicity and

c. Examine whether variations in incidence and pathways to care across ethnicities can be explained by social factors alone.

For the analysis, the ethnic categories were grouped as follows: White British, Black Caribbean (Black Caribbean and Other Black), Black African, Asian (Indian, Pakistani, Bangladeshi, Chinese), Other White (White Irish, White Gypsy, White Other), Other (Arab, Any Other Ethnic group), and Mixed (all mixed groups). This created seven ethnic groups: White British, Black Caribbean, Black African, Other White, Asian, Mixed and Other.

Personal and Psychiatric History Schedule (PPHS): *Pathways to care and mode of contact* Data relating to mode of contact were collected using an adapted form of the Personal and Psychiatric History Schedule (PPHS) for the purpose of case note data extraction (WHO, 1996). The PPHS is a structured clinical schedule designed to assess a patient's psychiatric, medical, social and developmental history using interviews and/or case notes (Jablensky et al., 1992b). It consists of sixteen parts. However, an amended version was used in this thesis, which included largely the first part of the PPHS. Those that are relevant to the aim of this study are source of referral, mode of contact, time of contact, mode of onset, duration of untreated psychosis and early intervention services use.

Clinical notes were reviewed and data relating to the source of referral at the point of first contact for psychosis were extracted. For analysis, this was collapsed into four groups: 'GP', 'A&E', 'Police/Criminal Justice system CJA' (police, court, prison) and 'Other' (nurse, social worker, other healthcare professional) or 'Not recorded'. Data on the Mental Health Act

(MHA) were grouped as (voluntary vs. compulsory). Time of contact data was collected as (office hours vs. out of hours). Data relating to mode of onset was collected by extracting the date of first contact with secondary mental health services for psychosis and the date when symptoms of psychosis first occurred. Subsequently, duration of untreated psychosis (DUP) was calculated by subtracting date of onset from date of contact with mental health services for FEP. Categorical variables for mode of onset were then grouped into four categories as follows for analysis: 'Acute' (within hours/days), 'Moderate' (within a month), 'Gradual' (within six months) and 'Insidious' (more than six months). Data on type of service engagement at first contact with services (early intervention service (EIS) vs. other community services (non-EIS)) were also collected.

Sections of the MRC-SDS and PPHS relevant to the data collected for this thesis were collated into an assessment booklet (see Appendix 3).

OPCRIT

The operational criteria (OPCRIT) checklist for psychotic and affective illness is an instrument designed to facilitate a polydiagnostic approach to mental illness (McGuffin, 1991). It is a 90-item checklist of psychotic and affective symptoms which are rated and scored by trained researchers/raters.

In this study OPCRIT items were employed to generate diagnostic categories from the clinical records and validate clinical diagnosis of all included cases.

6.5.5 Ethical approval

The CRIS system was approved as an anonymised dataset for secondary analysis by the Oxfordshire Research Ethics Committee (reference: 08/H0606/71). A local approval for this study was obtained from the CRIS Oversight Committee at the BRC South London and Maudsley NHS Foundation Trust (reference: 09-041).

6.6 Methodology of the follow-up study

Phase two of this thesis is concerned with the service use and clinical outcomes over the two years in the study period. The follow-up data collection was guided by the WHO Life Chart Schedule (Susser et al., 2000). The WHO Life Chart Schedule (LCS) is a semi-structured instrument, designed to assess the long-term course of schizophrenia (Susser et al., 2000). It includes four main areas: symptoms, treatment, residence and work. The LCS was originally designed as an interview schedule; however, aspects of it have been used for case record retrieval in previous studies, for example in (Bertelsen et al., 2008a, Bebbington et al., 2006). The LCS has been adapted for the follow-up phase of this study, specifically to include more information on service use and clinical course.

Using the structured query language (SQL) commands, the structured clinical and administrative fields of CRIS were interrogated to retrieve information on the study cohort. Information relating to hospital admission, community service use, compulsory admission, length of stay (LOS) (dates of admission/discharge), clinical course i.e. diagnosis, and patient destination during four years post-FEP were extracted using the commands presented in Appendix 4.

To obtain detailed information on hospital admission patterns, specific commands (see Appendix 4a) were written to extract data in relation to number of inpatient admissions, admission and discharge dates and length of hospital stay during follow-up post-FEP.

Data on compulsory admission were collected in a similar manner to inpatient admission whether an individual was detained under the MHA on admission. . Commands are listed in Appendix 4b.

For community service use, data relating to type of service (EIS, CMHT, A&T, specialist, forensic and other) and discharge destination (if applicable) were extracted using commands found in Appendix 4c.

Data on clinical course were collected in relation to primary clinical diagnosis and any subsequent diagnosis using the commands listed in Appendix 4d.

All the variables collected for the follow-up phase were used in the analysis.

6.7 Statistical analysis

The following hypotheses were tested in the analyses of data presented in this thesis: Compared with those of White British ethnicity:

Incidence

- The incidence of all psychoses will be higher in those of Black Caribbean and Black African ethnicity.
- 1.2 The magnitude of relative risk will be smaller for the minority ethnic groups (Black African and Black Caribbean) compared with those reported in previous studies (e.g. AESOP study, Fearon et al. 2006).

Pathways to care at first contact for psychosis

- 2.1 Higher rates of hospital admissions will be associated with Black African and Black Caribbean ethnicity.
- 2.2 Increased risk of compulsory admissions will be associated with Black African and Black Caribbean ethnicity.
- 2.3 Higher rates of police involvement will be associated with Black African and Black Caribbean ethnicity.
- 2.4 Lower levels of GP referral will be associated with Black African and Black Caribbean ethnicity.
- 2.5 Higher rates of accident and emergency referral will be associated with Black African and Black Caribbean ethnicity.
- 2.6 Compared with 15 years ago (i.e. AESOP vs. CRIS-FEP samples), among patients aged 18–35 years old, ethnic differences in GP referral and crisis source of referral (criminal justice agency, accident and emergency) and compulsory admission at first contact for psychosis will be smaller for Black African and Black Caribbean patients.

Course and outcome

- 3.3 Over the two-year follow-up period, worse service use outcomes (characterised by increased rates of hospital admissions, compulsory admission and longer duration of hospital stay) will be observed among those of Black Caribbean and Black African ethnicity compared with White British patients.
- 3.4 Over the two-year follow-up period, worse service use outcomes will be associated with non-EIS use compared with EIS use and these will vary by ethnic groups.

Descriptive statistics

To begin with, an account of whole sample sociodemographic characteristics, clinical presentation, pathways to care and follow-up clinical outcomes is presented as frequencies and percentages for categorical variables, and mean and medians (including interquartile range) for continuous variables. Appropriate statistical tests were used to compare differences between ethnic groups and sociodemographic characteristics, pathways to care characteristics and four-year clinical outcome characteristics, i.e. chi-square for categorical, t-test and ANOVA for normally distributed continuous data, and Wilcoxon rank-sum and Kruskal-Wallis tests for non-normally distributed continuous data.

Incidence

Population at risk was estimated according to the 2011 Census and stratified by age (fiveyear age-band, i.e. 18–19; 20–24; 25–29; 30–34; 35–39; 40–44; 45–49; 50–54; 55–59; 60– 64), gender and ethnicity. Age and sex-standardised incidence rates (using ONS Europe agestandardised population) of first psychosis were calculated using direct standardisation (Kirkwood and Sterne, 2003b). Crude incidence rate ratios (IRR) were calculated then adjusted for potential confounders like age and gender using Poisson regression.

Pathways to care

In assessing the risk factors associated with compulsory admission and source of referral, analysis was broken down into two parts, i.e. unadjusted and adjusted analyses using logistic regression (Kohler and Kreuter, 2005). Within the unadjusted regression model, sociodemographic and clinical presentation and pathway to care variable were assessed individually in relation to compulsory admission and source of referral. The adjusted

regression model focussed on the main exposure variable (ethnicity) in relation to compulsory admission and source of referral, and then *a priori* confounders were adjusted (age, gender, employment status, education qualification and EIS service use).

DUP was positively skewed; therefore, non-parametric Kruskal Wallis tests were employed to assess its association with sociodemographic and clinical variables.

DUP was then converted into a categorical variable for subsequent analysis e.g. logistic regression.

Course and outcomes

In this phase of the analyses, outcome data on service use variables collected during the follow-up period between May 2012 and April 2014 (i.e. hospital admission, compulsory admission and length of hospital stay) were included.

The rate of hospital admission and compulsory admission by ethnic groups were estimated per year over the follow-up period using Poisson regression (Kirkwood and Sterne, 2003b). Follow-up time/time at risk was defined as follows:

Time at risk = exit date minus entry date.

Entry date = date of contact with mental health services for FEP or date of discharge for first admission.

Exit date = study end date (30 April 2014) or date the patient left the study (i.e. discharge from SLaM services), whichever happened first.

This took into account the staggered entry into and exit from the study.

Next, for outcome variables which were count data (i.e. number of hospital admissions, number of compulsory admissions, length of hospital stay), the distribution of data were inspected using simple descriptive statistics, e.g. histogram, variance and goodness of fit test (UCLA: Statistical Consulting Group, 2016). Where the histogram distribution was skewed, i.e. the variance was greater than the mean, or where the goodness to fit test was statistically significant at p=0.05, this indicated over-dispersion; therefore, negative binomial regression was employed for adjusted and unadjusted rate ratios (Kirkwood and Sterne,

2003b, UCLA: Statistical Consulting Group, 2015). The *'exposure'* option was used to specify the length of time each patient remained in the study and *'irr'* to obtain rate ratios estimates (Kirkwood and Sterne, 2003).

For all analyses of risk factors, results are presented as follows: effect size, 95% confidence interval and p-values. For analyses of rates, results are presented as rate and 95% confidence interval per 100,000 person-years, unless otherwise stated.

All analyses were conducted using Stata version 12 (StataCorp, 2011).

7 Chapter 7: Results of Identifying First Episode Psychosis, Incidence of Psychosis and Ethnicity

7.1 Synopsis

To begin with this chapter will set the scene with a description of the Aetiology and Ethnicity of Schizophrenia and Other Psychoses (ASEOP) study and the reported findings as compared to the results from this study. Second, I will present results of the screening procedures in identifying FEP cases. Third, I will describe the overall first contact for psychosis (baseline) sample in the first of the four keys areas that data have been collected: sociodemographic characteristics, overall and by ethnic group. A detailed description of the three remaining data domains – pathways to care (at baseline), then stratified by early intervention use – will be presented and discussed in Chapters 8 and 9 respectively. Service use outcomes during the two-year follow-up period are presented and discussed in Chapters 10 and 11 respectively. The final part of this chapter (section 7.3) pertains to the analyses of incidence rates and incidence rate ratios of first episode psychosis, addressing the hypotheses that:

Compared to White British, the incidence of all psychoses will be higher in those of Black Caribbean and Black African ethnicity but lower than that reported in previous studies (e.g. AESOP Fearon et al. 2006).

The magnitude of relative risk will be attenuated for the minority ethnic groups (Black African and Black Caribbean) compared with those reported in previous studies (e.g. AESOP Fearon et al. 2006).

7.2 Background: the AESOP findings

As described in the previous chapter, the Aetiology and Ethnicity in Schizophrenia and Other Psychoses (AESOP) study was completed over ten years ago. In that study, higher incidence rates were reported for people of Black Caribbean (Adj. IR = 140.8; 95% CI 114.4 – 167.2) and Black African (Adj. IR = 80.6; 95% CI 60.0 – 101.2) ethnicity compared to the White British population (Adj. IR = 20.2; 95% CI 17.8 – 22.7). They observed that compared to White British, Black Caribbean were almost seven times (IRR = 6.7; 95% CI 5.4 – 8.3) and

Black African were four times (IRR = 4.1; 95% Cl 3.2 - 5.3) more likely to experience psychosis (Fearon et al., 2006a).

7.3 Results

7.3.1 Screening results

The search strategy described in Chapter 6 produced 9109 unique clinical records during the two-year case identification period (1 May 2010 – 30 April 2012). Figure 7.1 presents a flowchart of screening allocation. From this diagram, it can be seen that a total of 560 individuals screened positive and met the inclusion criteria. 8549 screened negative and these were excluded either for no evidence of psychosis (n = 5234) or they had signs of psychosis but did not meet the inclusion criteria (n = 3315). During the data collection phase, it came to light that two cases did not have sufficient information (missing age, ethnicity) to be included in analyses and therefore they were excluded. This left a total of 558 first episode psychosis cases over the two-year case identification period.



7.3.2 Demographic characteristics of total sample at baseline

A description of the baseline demographic characteristics for FEP cases is shown in Table 7.1. It can be seen from this table that there were more men (52.6%) and the mean age was 33.3 years. Most cases were of Black African ethnicity (26.3%) followed by White British (23.8%) and Black Caribbeans (16.3%). The majority were registered with a general practitioner (GP) (96.2%), were single (62.2%), were unemployed (68.5%), had education qualification up to A level (28.9%), lived with family or friends (60.1%), lived in rented accommodation (63.2%), were born outside the UK (53.7%) and reported to have experienced social isolation (52.6%).

Characteristics	Cases
	n = 558 (%)
Mean age (sd)	33.26 (10.7)
Age-Band	
18–29	242 (43.4)
30–64	316 (56.6)
Gender	
Male	292 (52.3)
Female	266 (47.7)
Ethnicity	
White British	133 (23.8)
Black African	147 (26.3)
Black Caribbean	91 (16.3)
Other White	75 (13.4)
Asian	44 (7.9)
Mixed	27 (4.8)
Other	41 (7.3)
GP	
Yes	537 (96.2)
No	21 (3.8)
Relationship status ¹	
Single	331 (62.2)
Married/Steady relationship	127 (23.8)
Divorced/Separated	74 (13.9)
Employment ²	
Unemployed	346 (68.5)
Student	60 (11.9)

Table 7.1: Overall sociodemographic variables

Employed	99 (19.6)
Education ³	
School, no GCSE	99 (25.2)
School, with GCSE	73 (18.5)
A level/ Further education	114 (28.9)
University	108 (27.4)
Lives with ⁴	
Alone	161 (29.7)
Family/relatives	325 (60.1)
Other	55 (10.2)
Housing Tenure ⁵	
Privately owned	68 (19.4)
Rented	222 (63.2)
Other	61 (17.4)
Country of birth ⁶	
UK born	243 (46.3)
Non-UK born	282 (53.7)
Report of social isolation ⁷	
Yes	248 (52.6)
No	223 (47.4)

1 (26 missing data)

2 (53 missing data)

3 (164 missing data)

4 (17 missing data)

5 (207 missing data)

6 (33 missing data)

7 (87 missing data)

7.3.3 Comparison between sociodemographic variables and ethnic groups

Table 7.2 shows a comparison of sociodemographic variables by ethnic groups. Patients of Black Caribbean and Asian ethnic groups were more likely to be women (57.1% female vs. 42.9% male and 56.8% female vs. 43.2% male respectively, p = 0.02) whereas those of 'Other' and 'Mixed' ethnic groups were more likely to be men (75% male vs. 24.4% female and 55.6% male vs. 44.6% female respectively, p = 0.02). Black Caribbean (9.8%) and 'Other' (8.8%) ethnic groups were least likely to be in employment compared with other ethnic groups (p = 0.06). Similarly, those with educational qualification up to university level were more likely to be White British (32.3%), Asian (34.4%) and White Other (33.9%) compared with other ethnic groups (p = 0.06). Patients born outside the UK were more likely to be Black African (83.8%), White Other (90.7%), Asian (73.2%) and Other (91.1%) (p<0.001). There were no differences observed in other sociodemographic variables by ethnic groups.

	White British n = 133	Black African n =147 (%)	Black Caribbean n = 91 (%)	Other White n = 75 (%)	Asian n = 44 (%)	Mixed n = 27 (%)	Other n = 41 (%)	Chi-sq test	df	p
	(%)									
Gender										
Male	68 (51.1)	79 (53.7)	39 (42.9)	41 (54.7)	19 (43.2)	15 (55.6)	31 (75.6)	14.12	6	0.02
Female	65 (48.9)	68 (46.3)	52 (57.1)	34 (45.3)	25 (56.8)	12 (44.4)	10 (24.4)			
Age-band										
18–29	56 (42.1)	67 (45.6)	38 (41.8)	23 (30.7)	22 (50.0)	17 (63.0)	19 (46.3)	10.55	6	0.10
	77 (57.9)	80 (54.4)	53 (58.2)	52 (69.3)	22 (50.0)	10 (37.0)	22 (53.7)			
30–64										
Relationship status ¹										
Single	84 (66.1)	82 (58.2)	54 (62.8)	45 (60.8)	23 (56.1)	20 (76.9)	23 (62.2)	11.71	12	0.46
Married/Steady relationship	32 (25.2)	32 (22.7)	19 (22.1)	20 (27.0)	13 (31.7)	4 (15.4)	7 (18.9)			
Divorced/Separated	11 (8.7)	27 (19.1)	13 (12.1)	9 (12.2)	5 (12.2)	2 (7.7)	7 (18.9)			
Employment ²										
Unemployed	78 (63.9)	84 (65.1)	66 (80.4)	47 (65.3)	24 (61.5)	18 (66.7)	29 (85.3)	20.05	12	0.06
Student	15 (12.3)	19 (14.7)	8 (9.8)	5 (6.9)	8 (20.5)	3 (11.1)	2 (5.9)			
Employed	29 (23.8)	26 (20.2)	8 (9.8)	20 (27.8)	7 (18.0)	6 (22.2)	3 (8.8)			
Education ³										
School, no GCSE	27 (28.1)	17 (16.5)	22 (36.1)	13 (24.5)	6 (18.7)	6 (28.6)	8 (28.6)	28.04	18	0.06
School with GCSE	18 (18.7)	29 (28.2)	8 (13.1)	4 (7.6)	5 (15.6)	5 (23.8)	4 (14.3)			

 Table 7.2: Comparison between sociodemographic variables and ethnic groups
A level/ Further	20 (20.8)	29 (28.2)	22 (36.1)	18 (34.0)	10 (31.3)	5 (23.8)	10 (35.7)			
education										
University	31 (32.3)	28 (27.1)	9 (14.7)	18 (33.9)	11 (34.4)	5 (23.8)	6 (21.4)			
Lives with ⁴										
Alone	41 (32.0)	39 (27.1)	31 (35.2)	21 (28.8)	12 (27.3)	7 (26.9)	10 (26.3)	6.60	12	0.88
Family/relatives	78 (61.0)	88 (61.1)	49 (55.7)	42 (57.5)	27 (61.4)	18 (69.2)	23 (60.5)			
Other	9 (7.0)	17 (11.8)	8 (9.1)	10 (13.7)	5 (11.4)	1 (3.9)	5 (13.2)			
Housing Tenure ⁵										
Privately owned	21 (25.0)	14 (16.1)	9 (18.0)	9 (16.4)	10 (32.3)	3 (13.6)	2 (9.1)	15.89	12	0.19
Rented	54 (64.3)	59 (67.8)	32 (64.0)	33 (60.0)	17 (54.8)	15 (68.2)	12 (54.6)			
Other	9 (10.7)	14 (16.1)	9 (18.0)	13 (23.6)	4 (12.9)	4 (18.2)	8 (36.4)			
Country of birth ⁶										
UK born	122 (96.8)	23 (16.2)	52 (64.2)	7 (9.3)	11 (26.8)	21 (91.3)	7 (18.9)	268	6	<0.001
Non-UK born	4 (3.2)	119 (83.8)	29 (35.8)	68 (90.7)	30 (73.2)	2 (8.7)	30 (81.1)			
Report of social										
isolation ⁷										
Yes	49 (44.9)	66 (53.2)	44 (59.5)	32 (47.8)	22 (55.0)	14 (60.9)	21 (61.8)	6.46	6	0.37
No	60 (55.1)	58 (46.8)	30 (40.5)	35 (52.2)	18 (45.0)	9 (39.1)	13 (38.2)			

1 – 7 (Missing values as indicated in 7.2.2)

7.4 Estimating incidence rate of first episode psychosis

In this section, I will present analyses of crude and age-gender-standardised incidence rates for the overall sample, followed by analyses of incidence rate ratios by age, gender and ethnic groups using White British as the reference ethnic group.

Table 7.3 shows the comparison between cases and the population at risk. It can be seen that there were no differences by gender, but cases were more likely to be younger, aged 20–24 years (cases 20.6% vs. population at risk 12.6%, p<0.001) and of Black African ethnicity (cases 26.3% vs. 12.6%), Black Caribbean ethnicity (cases 16.3% vs. population at risk 10.3%), 'Other' ethnicity (cases7.3% vs. population at risk 2.9%) or White British ethnicity (cases 23.8% vs. 41.2%)(p<0.001).

	Cases	Population at	Chi-sa test (df)	P-value
	cases			1 Value
		risk		
	N = 558	N = 852920		
	(%)	(%)		
Gender			4.95 (3)	0.17
Male	292 (52.3)	425968 (49.9)		
Female	266 (47.7)	426952 (50.1)		
Age-band			169.0 (27)	<0.001
18–19	29 (5.2)	25706 (3.0)		
20–24	115 (20.6)	107400 (12.6)		
25–29	106 (19.0)	157722 (18.5)		
30–34	79 (14.2)	138604 (16.3)		
35–39	76 (13.6)	104994 (12.3)		
40–44	60 (10.8)	92030 (10.8)		
45–49	35 (6.3)	82040 (9.6)		
50-54	38 (6.8)	63342 (7.4)		
50 51	14 (2.5)	44914 (5.3)		
55–59	6 (1.1)	36198 (4.2)		
60–64	· · /			

rable / is summary of overall stady population and population at his
--

Ethnicity			725.3 (18)	<0.001
White British	133 (23.8)	351412 (41.2)		
Black African	147 (26.3)	107670 (12.6)		
Black Caribbean	91 (16.3)	87788 (10.3)		
Other White	75 (13.4)	160918 (18.9)		
Asian	44 (7.9)	74432 (8.7)		
Mixed	27 (4.8)	45354 (5.3)		
Other	41 (7.3)	25346 (2.9)		

In estimating the incidence rates for the overall sample, direct age-standardisation was employed using the ONS Europe standard population. Age-gender-standardisation is widely used and recommended (Kirkwood and Sterne, 2003a) to ensure that rates are standardised and comparable to the standard population age and gender. An overall age-gender standardised incidence rate for all psychoses was 59.7 (95% CI 54.4 – 65.0) per 100,000 person-years at risk (PPY).

Table 7.4 displays the crude and standardised rates by age, gender and ethnic groups. When comparing incidence rates by gender, these appear to be equal between the two groups. Compared by age-bands, incidence rates were particularly raised for those 18–19 years (113.3 per 100,000 PPY) and 20–24 years (108.0/100,000 PPY) compared with older age groups, e.g. 60–64 years (16.4 per 100,000 PPY). This would be expected since the onset of psychosis peaks during adolescence and early adulthood (McGrath et al., 2016, Drake et al., 2016, Kirkbride et al., 2012a). This is also illustrated in Figure 7.2.

With regard to ethnicity, there was evidence of elevated rates across ethnic groups compared to White British (39.4 per 100,000 PPY). But these were markedly raised for Black Africans (120.8 per 100,000 PPY), Black Caribbean (93.6 per 100,000 PPY) and 'Other' ethnic groups (126.4 per 100,000 PPY). The Mixed groups (47.5 per 100,000 PPY) had a more modest increased rate compared to White British.

 Table 7.4: 4 Crude and age-standardised incidence per 100,000 person-years with 95% CI for first episode psychosis by demographic factors

	Cases	Population at	Crude	Standardised
		risk	incidence rate	incidence rate*
	N = 558	N = 852920		(95% CI)
	(%)	(%)		
Gender				
Male	292 (52.3)	425968 (49.9)	68.5	59.0 (51.8 – 66.3)
Female	266 (47.7)	426952 (50.1)	62.3	60.3 (52.5 - 68.1)
Age-band				
18-19	29 (5.2)	25706 (3.0)	112.8	113.3 (72.3 – 155.2)
20-24	115 (20.6)	107400 (12.6)	107.1	108.0 (88.2 – 127.7)
25-29	106 (19.0)	157722 (18.5)	67.2	67.2 (54.4 - 80.0)
30-34	79 (14.2)	138604 (16.3)	57.0	57.0 (44.4 – 69.5)
35-39	76 (13.6)	104994 (12.3)	72.4	72.4 (56.1 – 88.6)
40-44	60 (10.8)	92030 (10.8)	65.2	65.4 (48.8 – 81.9)
45-49	35 (6.3)	82040 (9.6)	42.7	42.6 (28.5 – 56.7)
50-54	38 (6.8)	63342 (7.4)	60.0	59.8 (40.8 – 78.8)
55-59	14 (2.5)	44914 (5.3)	31.2	31.1 (14.8 – 47.3)
60-64	6 (1.1)	36198 (4.2)	16.6	16.4 (3.3 – 29.5)
Ethnicity				
White British	133 (23.8)	351412 (41.2)	37.8	39.4 (32.2 – 46.6)
Black African	147 (26.3)	107670 (12.6)	136.5	120.8 (100.0 – 141.5)
Black Caribbean	91 (16.3)	87788 (10.3)	103.7	93.6 (74.0 – 113.2)
Other White	75 (13.4)	160918 (18.9)	46.6	49.1 (36.0 - 62.2)
Asian	44 (7.9)	74432 (8.7)	59.1	51.9 (33.6 – 70.2)
Mixed	27 (4.8)	45354 (5.3)	59.5	47.5 (26.2 – 68.9)
Other	41 (7.3)	25346 (2.9)	161.8	126.4 (85.5 – 167.2)

*age-gender standardised.





7.4.1 Unadjusted and adjusted incidence rate ratios for psychosis by ethnic groups Analysis of unadjusted and adjusted incidence rate ratios (IRR) by ethnic groups for psychosis is presented in Table 7.5 along with 95% confidence intervals (CI). Using Poisson regression, the model in this table is adjusted for age and gender. From the unadjusted model, there was evidence to suggest that compared to the White British group, Black African, Black Caribbean, Asian, Mixed and the 'Other' ethnic groups were all at increased risk of psychosis. After adjusting for age and gender, this evidence remained robust and showed that Black Africans were almost four times more likely (IRR = 3.59; 95% CI 2.8 -4.55), Black Caribbeans were almost three times more likely (IRR = 2.81; 95% CI 2.15 - 3.68), Asians were almost one and a half times more likely (IRR = 1.43; 95% CI 1.02 - 2.02) and the 'Other' group four times more likely (IRR = 4.14; 95% CI 2.91 - 5.88) to be at risk of psychosis. However, this evidence did not hold for the 'Mixed' group after adjusting for potential confounders. There was no evidence to suggest that the Other White group were different from the White British patients.

When the risk of psychosis by ethnic groups stratified by gender was assessed (Table 7.6), the following observations were made. Increased risks were present for both men and

women, particularly among three ethnic minority groups: Black African, Black Caribbean and Other, independent of age. Men of Black African and Other ethnic groups were generally more at risk than their women counterparts. In contrast, the risk of psychosis appears to be similar in the Black Caribbean men and women. In the unadjusted model, it can be seen that increased risk among men of the 'Mixed' (IRR = 1.86; 95% CI 1.06 - 3.25) and women of Asian (IRR = 1.63; 95% CI 1.03 - 2.59) ethnic groups was evident, but the strength of evidence became weaker when age was adjusted for. This would suggest that there may be age interaction effect in these two groups, which is explored further in Table 7.6.

Table 7.5: Incidence rate ratios (IRR) with 95% CI in ethnic minority groups for first episode psychosis

	Case n = 558 (%)	Unadjusted IRR (95% CI), p-	Adjusted IRR** (95% CI),
		value	p-value
All			
White British	133 (23.8)	1.00	1.00
Black African	147 (26.3)	3.60 (2.85 – 4.56), <0.01	3.59 (2.8 – 4.55) <0.001
Black Caribbean	91 (16.3)	2.73 (2.09 – 3.57), <0.01	2.81 (2.15 - 3.68) <0.001
Other White	75 (13.4)	1.23 (0.92 – 1.63), 0.14	1.20 (0.9 – 1.59) 0.20
Asian	44 (7.9)	1.56 (1.11 – 2.19, 0.03	1.43 (1.02 – 2.02) 0.03
Mixed	27 (4.8)	1.57 (1.04 – 2.37), 0.03	1.41 (0.93 – 2.14) 0.09
Other	41 (7.3)	4.27 (3.01 – 6.06), <0.01	4.14 (2.91 – 5.88) <0.001

**adjusted for age and gender

Table 7.6: Incidence rate ratios (IRR) with 95% CI in ethnic minority groups for first episode psychosis by gender

Men	N = 292 (%)	Unadjusted IRR	Adjusted IRR*
White British	68 (23.3)	1.00	1.00
Black African	79 (27.1)	4.39 (3.17 – 6.08), <0.001	4.35 (3.13 – 6.02), <0.001
Black Caribbean	39 (13.4)	2.74 (1.84 – 4.06), <0.001	2.76 (1.85 – 4.10), <0.001
Other White	41 (14.0)	1.42 (0.96 – 2.09), 0.07	1.34 (0.91 – 1.98), 0.13
Asian	19 (6.5)	1.46 (0.87 – 2.43), 0.14	1.26 (0.75 – 2.10), 0.36
Mixed	15 (5.1)	1.86 (1.06 – 3.25), 0.03	1.57 (0.90 – 2.76), 0.11

Other	31 (10.6)	6.30 (4.12 – 9.64), <0.001	5.98 (3.90 – 9.15), <0.001
Women	N = 266 (%)		
White British	65 (24.4)	1.00	1.00
Black African	68 (25.5)	2.95 (2.10 – 4.14), <0.001	2.87 (2.04 – 4.04), <0.001
Black Caribbean	52 (19.6)	2.70 (1.87 – 3.89), <0.001	2.68 (1.86 – 3.88), <0.001
Other White	34 (12.8)	1.05 (0.69 – 1.59), 0.81	1.05 (0.69 – 1.59), 0.81
Asian	25 (9.4)	1.63 (1.03 -2.59), 0.03	1.56 (0.98 – 2.48), 0.06
Mixed	12 (4.5)	1.30 (0.70 – 2.42), 0.39	1.23 (0.66 – 2.48), 0.66
Other	10 (3.8)	2.13 (1.09 – 4.16), <0.001	2.09 (1.07 – 4.08), 0.02

*adjusted for age

7.4.2 Unadjusted and adjusted age-specific incidence rate ratios by ethnic groups

Table 7.7 shows age-specific incidence rate ratios by ethnic groups using White British as reference group. The observed results suggest elevated risk in the young and middle age-bands (20–44 years) among the Black African, Black Caribbean and Other groups. Moreover, patients aged 25–29 and of Asian or Mixed ethnicities were also twice to three times more likely to experience psychosis (IRR = 2.37; 95% Cl 1.10 - 5.10), (IRR = 3.36; 95% Cl 1.52 - 7.44) respectively. In addition, older (55–59 years) Asian patients had four-fold increased risk of psychosis (IRR = 4.67; 95% Cl 1.04 - 20.95), but no other ethnic minority group showed increased risk in this particular age group. It is noteworthy that the point estimate of risk among the Asian and 'Other' ethnic groups appears imprecise due to wide confidence intervals and this could be due to the small sample size in these groups. The observed difference in risk among Asians and the Mixed group by age-band in the adjusted IRR model may explain the gender difference shown in Table 6.5 above, meaning that it is the older women of the Asian ethnic group and men aged 22–29 of Mixed ethnicity that are most at risk. This is consistent with previously reported findings (Coid et al., 2008b, Bhugra et al., 1997).

Age-band	Black African	Black Caribbean	Other White	Asian	Mixed	Other
(years)						
18–19	0.79 (0.299 – 2.12),	0.64 (0.26 – 1.98),	0.79 (0.22 – 2.80),	0.38 (0.08 – 1.74),	0.49 (0.11 – 2.12),	No cases
	0.65	0.44	0.71	0.21	0.35	
20–24	4.37 (2.61 – 7.31),	3.26 (1.82 – 5.84),	1.06 (0.51 – 2.12),	1.25 (0.60 – 2.59),	1.45 (0.63 – 3.36),	4.98 (2.33 – 10.63),
	>0.001	<0.001	0.85	0.54	0.37	<0.001
25–29	8.10 (4.57 – 14.34),	4.64 (2.33 – 9.27),	1.09 (0.52 – 2.29),	2.37 (1.10 – 5.10),	3.36 (1.52 – 7.44),	9.84 (4.77 – 20.27),
	<0.001	<0.001	0.81	0.02	<0.01	<0.001
30–34	4.83 (2.59 – 9.03),	4.37 (2.13 – 8.93),	1.11 (0.54 – 2.26),	1.94 (0.84 – 4.47),	0.41 (0.05 – 3.09),	2.77 (0.93 – 8.20),
	<0.001	<0.001	0.78	0.11	0.38	0.06
35–39	3.76 (1.92 – 7.37),	3.55 (1.66 – 7.61),	1.59 (0.77 – 3.30),	1.18 (0.39 – 3.56),	1.53 (0.44 – 5.30),	5.78 (2.45 – 13.64),
	<0.001	<0.001	0.20	0.76	0.49	<0.001
40–44	4.24 (1.90 – 9.47),	3.49 (1.44 – 8.45),	2.86 (1.20 – 6.80),	3.06 (1.02 – 9.16),	0.86 (0.11 – 6.82),	5.06 (1.56 – 16.45),
	<0.001	<0.01	0.01	0.04	0.83	<0.01
45–49	1.84 (0.72 – 4.68),	1.55 (0.58 – 4.09),	1.40 (0.48 – 4.11),	0.68 (0.08 – 5.35),	1.70 (0.37 – 7.77),	2.58 (0.56 – 11.80),
	0.19	0.37	0.53	0.71	0.49	0.22
50–54	0.86 (0.31 -2.36),	1.17 (0.48 – 2.86),	1.01 (0.37 – 2.78),	0.44 (0.05 – 3.33),	1.40 (0.32 – 6.12),	2.07 (0.47 – 9.03),
	0.78	0.72	0.97	0.42	0.65	0.32
55–59	2.10 (0.38 – 11.51),	2.82 (0.62 – 12.65),	1.92 (0.35 – 10.54),	4.67 (1.04 – 20.95),	No cases	No cases

Table 7.7: Adjusted age-specific incidence rate ratios (IRR) with 95% CI and p-value in ethnic minority groups for first episode all psychosis

	0.38	0.17	0.44	0.04		
60–64	2.98 (0.54 – 16.37),	No cases				
	0.20					

Reference = White British. Adjusted for gender.

Table 7.8: Comparison of findings from AESOP and current study by ethnicity

	Standardised IR (95% CI)	Standardised IR (95% CI)	Adjusted IRR (95% CI)	Adjusted IRR (95% CI)
	AESOP	Current study	AESOP	Current study
All				
White British	20.2 (17.8 – 22.7)	39.4 (32.2 – 46.6)	1.00	1.00
Black African	80.6 (60.0 - 101.2)	120.8 (100.0 – 141.5)	4.1 (3.2 – 5.3)	3.59 (2.8 – 4.55)
Black Caribbean	140.8 (114.4 – 167.2)	93.6 (74.0 - 113.2)	6.7 (5.4 – 8.3)	2.81 (2.15 – 3.68)
Other White	33.1 (22.0 – 44.2)	49.1 (36.0 - 62.2)	1.6 (1.1 – 2.2)	1.20 (0.9 – 1.59)
Asian	31.6 (16.7 – 46.5)	51.9 (33.6 – 70.2)	1.5 (0.9 – 2.4)	1.43 (1.02 – 2.02)
Mixed	45.9 (26.4 – 65.5)	47.5 (26.2 – 68.9)	2.7 (1.8 – 4.2)	1.41 (0.93 – 2.14)
Other	55.0 (30.9 – 79.1)	126.4 (85.5 – 167.2)	2.6 (1.7 – 3.9)	4.14 (2.91 – 5.88)

7.5 Summary

To summarise, incidence rates of psychosis were higher among Black African, Black Caribbean and 'Other' ethnic groups than among White British. Rates were also highest among those aged 18–19 and 20–24 and least among those aged 60 and above. However, given the small number of people aged 18–19 years in the sample and the wide confidence intervals, the incidence rates for this group are likely to be less precise and therefore the interpretations for age band 18–19 years are tentative. Analyses of incidence rate ratios by age and ethnic group also confirmed higher risk of psychosis across ages 20–44 years if patients were of Black African, Black Caribbean or 'Other' ethnic groups. There were higher risks among Asian patients aged 25–29 years and 40–44 years, and this is the only ethnic group with increased risk at age 55–59 years. For patients of 'Mixed' ethnicity, the risk was only significant at 25–29 years.

In terms of incidence rates by ethnic group, these were markedly higher among Black African, Black Caribbean and 'Other' ethnic groups compared with White British. The magnitude of the risk varied across ethnic groups, but there was strong evidence that four main groups were at risk: starting with the group most at risk, there was evidence for fourfold risk among 'Other', and three-fold risk among Black African. Black Caribbean patients were almost three times at risk and Asian patients had almost one and a half increased risk compared with White British. There was tentative evidence for increased risk among those of 'Mixed' ethnicity, but this did not hold after controlling for confounders.

In relation to gender-specific rate ratios, there was evidence that men of Black African, Black Caribbean and 'Other' ethnic groups were at particularly increased risk. The increased risk for men was consistent with overall rate ratios but was stronger. For example, men of 'Other' ethnic group were around six times, Black Africans were four times and Black Caribbeans were almost three times more at risk compared with White British.

Again as before, there was tentative evidence that men of 'Mixed' ethnic groups were almost twice at risk of psychosis compared with White British, but this was no longer significant at traditional level (i.e., p<0.05) after adjusting for age.

For women, the picture was similar to men, particularly for the same three ethnic groups most at risk, namely Black African, Black Caribbean and 'Other'. However, the magnitude of the effect was somewhat reduced for women of Black African (almost three times increased risk) and 'Other' (twice at risk), but the effect for men and women of Black Caribbean ethnic group appears to be similar.

Asian women showed almost twice the risk of psychosis compared with White British women, but after controlling for age, the effect weakened slightly and therefore this is only a tentative interpretation.

In conclusion, the evidence in this chapter supports the hypotheses that incidence rates were higher among Black African and Black Caribbean compared with White British patients and that the magnitude of risk was attenuated. In fact magnitude of risk among Black Caribbeans appears to have halved compared to AESOP findings (Table 7.8). However, it is noteworthy that compared to the AESOP study, a higher incidence rate (almost doubled) was observed among the White British population in this study, while the rate was lower for the Black Caribbean population (Table 7.8). These two findings may explain the reduced rate ratio in the Black Caribbean ethnic group.

For the Black African group, the incidence rate remained higher and the rate ratio showed little or no difference compared to the AESOP study (Table 7.8).

7.6 Limitations

While this study has several methodological strengths which are discussed in Chapter 12, the findings in this chapter need to be interpreted with a number of key methodological limitations in mind.

Diagnostic accuracy

First, first episode psychosis cases in this study were identified using routinely collected clinical material. Although the study used standardised approach to diagnosis applied by the research team (ie. the PSS), these diagnoses are necessarily vulnerable to the accuracy and rigor with which individual symptoms are assessed. Clinicians are likely to vary in their competence at assessing and recording the key symptoms that comprise a diagnosis. Few will routinely use standardised clinical measures and some records will have been made by less experienced junior psychiatrists. However, researchers in this study have clinical expertise and where possible took into account verbatim quotes of patients that were often recorded as justification by the clinicians for the presence of symptom.

Further, it is acknowledged that excluding those with transient psychotic symptoms resulting from acute intoxication may lead to selection bias, as brief psychoses can turn out to be first signs of more enduring illness. However, given that we were able to observe the clinical records of individuals across the entire screening period (two years) and the three-stage screening process (see page 126), the risk of missing those who initially presented with transient psychotic features but later met the inclusion criteria of a case, were minimal.

Misclassification of ethnic groups

As described in Chapter one, self-ascription to an ethnic group is largely influenced by the cultural context of the individual. Participants in this study were grouped according to the ONS classifications. However, the assumption of cultural uniformity in large clusters such as African and Caribbean is crude and likely to be flawed. For example, the Caribbean people came from countries that are geographically as much as 2000 kilometres from each other and have very different cultures and 'identities'. Another example is diversity within the African group. For instance, in an African country like Nigeria there are many tribes and

ethnicities with different cultures, identities and languages. Furthermore, some second and third generation offspring whose grand- and great-grand parents migrated to the UK may be assimilated well enough to think of themselves as a culturally integrated (Singh, 1997) and consider themselves simply as British. Therefore misclassification is likely to be intrinsic in both the denominator and numerator populations. Another complication is that ethnic classifications in health records may not be taken at face value since it is difficult to be certain whether patients self-ascribe their ethnicity and where this is clinician ascribed, heterogeneity in the recording may be inevitable. This is because clinician's judgement of patient's ethnicity is inevitably subjective. While steps were taken by researchers in this study to carefully check patients' ethnicity (see Chapter 6, p 128), given the potential misclassifications in both denominator and numerator populations, these caveats will affect the findings in this study. This inevitably can only lead to imperfect findings but is an issue of population research in general, not just a concern of this thesis.

Migration status

Finally, recent migrants of refugee status may be even more at risk of mental disorder than economic migrants or those whose parents and grand-parents were the original migrants. This cannot be addressed by present study as migration status was not adjusted for due to this data not being available for the denominator population at the time of the study. Previous studies that were able to adjust for generation status and migration status had shown that higher incidence rate ratios were observed particularly in the black ethnic groups and their children (Coid et al., 2008b, Kirkbride et al., 2017). In addition, a recent study from Sweden investigated the rates of schizophrenia and non-affective psychosis in refugee and non-refugee migrants compared with native Swedish populations. They found that refugee migrants had higher rates of psychosis than non-refugee migrants, but the risk was even greater for those from sub-Saharan Africa irrespective of refugee status (Hollander et al., 2016). In addition, the increased risk of psychosis among the Asian women in this study could be due to some other social difficulties that have been highlighted as pertinent to this population. These include social isolation, quality of relationship with male partner and lack of English language proficiency (Talbani and Hasanali, 2000, Nilaweera et al., 2014).

8 Chapter 8: Results of Pathways to Care, Compulsory Admission at First Episode Psychosis and Ethnicity

8.1 Synopsis

In this chapter, I will first describe the overall sample characteristics for pathways to care at first contact for psychosis (baseline). Secondly, in section 8.3, six unadjusted logistic regression models are presented assessing the prevalence and relationships between hospital admission, compulsory admission, pathways to care/source of referral (namely: police/criminal justice referral, GP referral, A&E referral and other source of referral), sociodemographic and clinical factors. Thirdly, adjusted logistic regression models are presented assessing the main predictor variable (i.e. ethnicity) and the outcome variables considered in Section 8.3, to address the following Hypotheses: compared to White British ethnicity,

- Higher rates of hospital admissions will be associated with Black African and Black Caribbean ethnicity.
- 2. Increased risk of compulsory admissions will be associated with Black African and Black Caribbean ethnicity.
- 3. Higher rates of police involvement will be associated with Black African and Black Caribbean ethnicity.
- 4. Lower levels of GP referral will be associated with Black African and Black Caribbean ethnicity.
- Higher rates of accident and emergency referral will be associated with Black African and Black Caribbean ethnicity.

On the basis of previous literature, there is no evidence that Black African and Black Caribbean ethnic groups had longer duration of untreated psychosis (DUP) compared with the White British ethnic group. Therefore, I did not formally test the hypothesis that DUP will be longer. Nonetheless for completeness, I did an analysis of the associations between DUP and ethnicity.

8.2 Pathway to care variables at first contact for psychosis (baseline)

To begin with, as noted in Chapter 6, data for this stage of the analysis were collected at the point of first contact for psychosis in seven key domains: hospital admission, compulsory admission, source of referral, time of contact, family involvement in help-seeking, mode of onset and duration of untreated psychosis. Data relating to hospital admission and compulsory admission were assessed as whether present or not (yes or no) on the day of first contact for psychosis. It is on these cross-sectional data that the analyses of outcomes in this chapter are based.

Of the total number of 558 patients, data were complete for all pathways to care variables except cannabis use (122 missing data) and early intervention service use (2 missing data), which were included in the analyses. It is worth noting that some information was not recorded in the clinical records and therefore these items were coded as missing, which then means that total frequencies vary for these items.

Table 8.1 describes the basic characteristics of pathways to care of the total sample at first contact (baseline). From this table it can be seen that a median of 93 days of duration of untreated psychosis was observed among the total sample. Onset of psychosis in the majority of the sample was insidious (n = 209, 37.5%). Around half of the sample had ever smoked cannabis (52.5%). Almost a quarter of the sample had used mental health services previously for a diagnosis other than psychosis. A majority of individuals made first contact with secondary mental health services for psychosis as in-patients (42.7%) while the remainder came in via community services (57.3%). Many were referred by the accident and emergency services (39.4%), followed by GP referral (35.1%). 77 (13.8%) people were referred by the police/criminal justice agency and the remainder, 68 (12.2%), were referred by 'Other' (i.e. healthcare professionals, nurse, doctor in non-mental health care setting). Only around 40% of the sample used an early intervention service during first contact for psychosis, while the rest used other types of mental health services. The use of the Mental Health Act was observed in (24.2%) of the sample at first contact for psychosis and around a third of the sample made contact outside of normal working hours. Most of the sample did not (64.9%) have family involvement in their help-seeking for psychosis.

Table 8.1:	Basic characteri	istics of path	ways to care	variables
10010 011	. Dusie characteri	stics of putil	ways to care	, vanasies

Characteristics	Cases			
	n = 558 (%)			
Median DUP in days(IQR)	93 (19 – 447)			
Mean age at FEP in years(sd)	33.2 (10.7)			
Mode of onset	N (%)			
Acute (within a week)	116 (20.7)			
Moderate (within a month)	111 (19.9)			
Gradual (up to 6months)	122 (21.9)			
Insidious (more than 6 months)	209 (37.5)			
Pervious Service Use				
No	426 (76.3)			
Yes	132 (23.7)			
Hospital admission at FEP				
contact				
No	320 (57.3)			
Yes	238 (42.7)			
Source of referral				
GP/Primary care	196 (35.1)			
A&E	217 (38.9)			
Police/Criminal Justice agency	77 (13.8)			
Other (non-mental health	68 (12.2)			
professionals)				
MHA at FEP				
Yes	135 (24.2)			
No	423 (75.8)			
Time of FEP contact				
Office hours	363 (65.1)			
Out of hours	195 (34.9)			
Family involvement in help- seeking				
Yes	196 (32.1)			
No	362 (64.9)			
Ever smoked cannabis ¹				
Yes	229 (52.5)			
No	207 (47.5)			
Early intervention service use ²				
Yes	222 (39.9)			
No	334 (60.1)			

1 (122 missing data)

2 (2missing data)

8.2.1 Comparison between ethnicity and pathways to care variables

In order to examine whether association of source of referral varied by ethnic group, each source of referral variable was dichotomised to explore each as an individual outcome variable. This resulted in four different outcomes for source of referral variables. Further, hospital admission and compulsory admission were assessed separately as outcomes in the analyses.

Table 8.2 displays comparisons between pathways to care and ethnic groups. The results show that 'Other' ethnic groups had the shortest median duration of untreated psychosis (60 days) and Black Caribbean patients the longest (126 days). In line with previous studies there was no evidence that DUP was longer in Black African and Black Caribbean ethnic groups compared with the White British group, therefore I did not proceed to do further analysis. By age, patients of 'Mixed' ethnicity were younger (mean age = 29.7, sd = 9.6) compared with other ethnic groups (X2 = 12.5 df 6, p = 0.05). Those who had used services prior to onset of psychosis were more likely to be White British (32.3%, p = 0.02) compared with other groups. Differences were also observed in those who had used cannabis and these were more likely to be White British (63.2%), Black Caribbean (65.2%) and Mixed (73.9%) ethnic groups, whist individuals of the Asian ethnic group were least likely to use cannabis (28.1%) (X2 = 29.3,df 6, p<0.001).

Black Africans were more likely to be detained under the Mental Health Act, i.e. compulsory admission, compared with White British (17.3% vs. 34.0% respectively, p = 0.01). First contact for psychosis via the community services was the most common route of access to mental health services among all patients (48.2% to 68.3%). A higher proportion of White British (41.4%) and Black African (47.6%) patients used early intervention services, but no statistical difference was detected between these groups. Although no strong statistical ethnic differences were observed for hospital admission at first contact, among those admitted, a higher proportion were of Black African (49%), Mixed (51.9%), White Other (48%) and Asian (47.7%) ethnic groups.

In terms of source of referral, most patients were referred by the Accident and Emergency department compared with GP referral. For example, 42.9% Black African, 41.3% White Other and 47.7% Asian patients were referred by A&E compared with 30.6%, 33.3% and 34.1% referred by GP respectively, but this did not reach statistical significance which may be explained by sampling error or the smaller sample size among the Asian and 'Other' groups.

Referral via police and criminal justice agency was more common among the Black African (16.3%) and 'Mixed' (22.2%) ethnic groups and the majority of the Black Caribbean group were referred by Other healthcare professionals, but no statistical differences were detected across the ethnic groups on source of referrals.

	White British n = 133 (%)	Black African n = 147 (%)	Black Caribbean n = 91 (%)	White Other n = 75 (%)	Asian n = 44 (%)	Mixed n = 27 (%)	Other n = 41 (%)	ANOVA/ Kwallis Chi-sq test	df	p
Median DUP in days (IQR)	105 (22 – 514)	88 (17 – 447)	126 (28 – 449)	86 (14 – 408)	76.5 (8.5 – 243)	92 (23 – 361)	60 (12 – 560)	2.40	6	0.87
Mean Age at FEP in years (SD)	34.5 (12.5)	32.4 (9.9)	33.4 (10.6)	34.9 (9.6)	32.4 (10.3)	29.7 (9.6)	31.9 (9.1)	12.56	6	0.05
Median LOS (IQR) at FEP	15 (4 – 33)	29 (14 – 61)	29.5 (11 – 75)	21 (8.5 – 45.5)	14 (6 – 63)	13 (3 – 34)	28 (11 – 88)	13.06	6	0.04
	White British n = 133 (%)	Black African n = 147 (%)	Black Caribbean n = 91 (%)	Other White n = 75 (%)	Asian n = 44 (%)	Mixed n = 27 (%)	Other n = 41 (%)	Chi-sq test	df	p
Previous service use										
No	90 (67.7)	122 (83.0)	65 (71.4)	58 (77.3)	38 (86.4)	19 (70.4)	34 (82.9)	14.36	6	0.02
Yes	43 (32.3)	25 (17.0)	26 (28.6)	17 (26.7)	6 (13.6)	8 (29.6)	7 (17.1)			
Hospital admission at FEP										
No	83 (62.4)	75 (51.0)	59 (64.8)	39 (52.0)	23 (52.3)	13 (48.1)	28 (68.3)	10.16	6	0.11
Yes	50 (37.6)	72 (49.0)	32 (35.2)	36 (48.0)	21 (47.7)	14 (51.9)	13 (31.7)			ļ
Mode of onset										

 Table 8.2: Comparison between clinical, pathways to care variables and ethnic groups

Acute (within a	25 (18.8)	33 (22.6)	18 (19.8)	15 (20.0)	11 (25.0)	4 (14.8)	10 (24.4)	11.68	18	0.86
week)										
Moderate (within	26 (19.6)	29 (19.8)	16 (17.6)	18 (24.0)	7 (15.9)	6 (22.2)	9 (21.9)			
a month)										
Gradual (up to	25 (18.8)	36 (24.5)	22 (24.2)	13 (17.3)	12 (27.3)	9 (33.3)	5 (12.2)			
6months)										
Insidious (more	57 (42.8)	49 (33.3)	35 (38.4)	29 (38.7)	14 (31.8)	8 (29.6)	17 (41.5)			
than 6 months)										
Ever smoked										
cannabis										
Yes	72 (63.2)	44 (40.0)	45 (65.2)	28 (46.7)	9 (28.1)	17 (73.9)	14 (50.0)	29.30	6	0.000
No	42 (36.8)	66 (60.0)	24 (34.8)	32 (53.3)	23 (71.9)	6 (26.1)	14 (50.0)			
Compulsory										
admission (MHA)										
at FEP										
Yes	23 (17.3)	50 (34.0)	19 (20.9)	14 (18.7)	13 (29.9)	9 (33.3)	7 (17.1)	16.02	6	0.01
No	110 (82.7)	97 (66.0)	72 (79.1)	61(81.3)	31 (70.4)	18 (66.7)	34 (82.9)			
Time of FEP										
contact										
Office hours	88 (66.2)	86 (58.5)	62 (68.1)	50 (66.7)	27 (61.4)	19 (70.4)	31 (75.6)	5.92	6	0.43
Out of hours	45 (33.8)	61 (41.5)	29 (31.9)	25 (33.3)	17 (38.6)	8 (29.6)	10 (24.4)			
Early intervention										
service use										
No	78 (85.6)	77 (52.4)	55 (61.1)	51 (68.0)	29 (67.4)	15 (55.6)	29 (70.7)	8.88	6	0.18
Yes	55 (41.4)	70 (47.6)	35 (38.9)	24 (32.0)	14 (32.6)	12 (44.4)	12 (29.3)			
Family										
involvement										
Yes	44 (33.1)	48 (32.7)	39 (42.9)	29 (38.7)	17 (38.6)	8 (29.6)	11 (26.8)	5.27	6	0.50

No	89 (66.9)	99 (67.4)	52 (57.1)	46 (61.3)	27 (61.4)	19 (70.4)	30 (73.2)			
GP referral										
No	83 (62.4)	102 (69.4)	60 (65.9)	50 (66.7)	29 (65.9)	17 (63.0)	21 (51.2)	5.23	6	0.51
Yes	50 (37.6)	45 (30.6)	31 (34.1)	25 (33.3)	15 (34.1)	10 (37.0)	20 (48.8)			
A&E referral										
No	81 (60.9)	85 (57.8)	61 (67.0)	45 (60.0)	23 (52.3)	18 (66.7)	28 (68.3)	4.73	6	0.57
Yes	52 (39.1)	62 (42.2)	30 (33.0)	30 (40.0)	21 (47.7)	9 (33.3)	13 (31.7)			
Police / Criminal										
Justice system referral										
No	117 (88.0)	123 (83.7)	80 (87.9)	64 (85.3)	39 (88.6)	21 (77.8)	37 (90.2)	3.80	6	0.72
Yes	16 (12.0)	24 (16.3)	11 (12.1)	11 (14.7)	5 (11.4)	6 (22.2)	4 (9.8)			
Other referral										
No	118 (88.7)	131 (89.1)	72 (79.1)	66 (88.0)	41 (93.2)	25 (92.6)	37 (90.2)	8.75	6	0.27
Yes	15 (11.3)	16 (10.9)	19 (20.9)	9 (12.0)	3 (6.8)	2 (7.4)	4 (9.8)			

8.3 Associations between sociodemographic, pathways to care variables and compulsory admission

This section explores the prevalence of and relationship between hospital admission, compulsory admission and sociodemographic and pathways to care factors at first contact for psychosis, using univariable logistic regression.

8.3.1 Unadjusted odds ratios for hospital admission at first contact

Table 8.3 displays the prevalence and unadjusted odds ratios of hospital admission by sociodemographic, clinical and pathways to care variables. It can be seen in this table that four sociodemographic variables were strongly associated with hospital admission. Black African (OR = 1.59; 95% CI 0.98 – 2.56) patients were almost two times more likely to be admitted. Patients with the housing tenure (hostel/refuge) were also twice as likely to be admitted to hospital (OR = 2.16; 95% CI 1.06 – 4.38). Patients who were registered with the GP (OR = 0.22; 95% CI 0.07 – 0.61) and those divorced or separated (OR = 0.48; 95% CI 0.28 – 0.83) were less likely to be hospitalised. Meanwhile, there was weak evidence of reduced likelihood of hospital admission and the following demographic factors: age, living with relatives/friends and born outside of the UK. In terms of pathways to care variables, strong evidence of associations were observed between out of hours contact (OR = 4.89; 95% CI 3.36 – 7.12), accident and emergency referral (OR = 12.65; 95% CI 7.34 – 21.81), police/CJA referral (OR = 134; 95% CI 48.25 – 372.95), Other referral (OR = 4.45; 95% CI 2.22 – 8.92) and hospital admission. Longer (≥one year) duration of untreated psychosis and mode of onset had reduced likelihood of being hospitalised (Table 8.3).

	No-hospital admission n = (%)	Hospital admission n = (%)	Unadjuste d OR	95% confidence interval	P = value
Gender					
Male	161 (55.1)	131 (44.9)	1.00		
Female	159 (59.8)	107 (40.2)	0.82	0.59 – 1.15	0.26
Age-band					
18–29	129 (53.3)	113 (46.7)	1.00		
30–64	191 (60.4)	125 (39.6)	0.74	0.53 – 1.04	0.09
Ethnicity					
White British	83 (62.4)	50 (37.6)	1.00		
Black African	75 (51.0)	72 (49.0)	1.59	0.98 – 2.56	0.05

Table 8.3: Unadjusted odds ratios of hospital admission by sociodemographic and pathways to care variables at first contact

Black Caribbean	59 (64.8)	32 (35.2)	0.90	0.51 – 1.56	0.71
White Other	36 (52.0)	36 (50.0)	1.53	0.86 – 2.71	0.14
Asian	23 (52.3)	21 (47.7)	1.51	0.76 - 3.01	0.23
Mixed	13 (48.2)	14 (51.8)	1.78	0.77 – 4.10	0.17
Other	28 (68.3)	13 (31.7)	0.77	0.36 - 1.62	0.49
GP Registered					
No	5 (23.8)	16 (76.2)	1.00		
Yes	315 (58.7)	222 (41.3)	0.22	0.07 - 0.61	0.004
Ever employed					
No	27 (64.3)	15 (35.7)			
Yes	230 (56.9)	174 (43.1)	1.36	0.70 - 2.63	0.36
Relationship status			1.00		
Single	1// (53.5)	154 (46.5)	1.00	0.40 1.12	0.10
rolationship	77 (60.6)	50 (39.4)	0.74	0.49 - 1.13	0.16
Divorcod/Soparatod	52 (70 2)	22 (20 7)	0.48	0.28 - 0.82	0.000
Employment	52 (70.3)	22 (29.7)	0.48	0.28 - 0.85	0.009
Unemployed	205 (59 2)	141 (40.8)	1.00		
Student	29 (48 3)	31 (51 7)	1.00	0 89 - 2 69	0.11
Employed	57 (57.6)	42 (42.4)	1.07	0.68 - 1.68	0.76
Education		(,	1.07		
School, no GCSE	55 (55.6)	44 (44.4)	1.00		
School with GCSE	44 (60.3)	29 (39.7)	0.82	0.44 - 1.52	0.53
A level/ Further	67 (58.8)	47 (41.2)	0.87	0.50 - 1.51	0.63
education					
University	59 (54.6)	49 (45.4)	1.03	0.59 – 1.79	0.89
Lives with					
Alone	85 (52.8)	76 (47.2)	1.00		
Family/relatives	200 (61.5)	125 (38.5)	0.69	0.47 – 1.02	0.06
Other	26 (47.3)	29 (52.7)	1.24	0.67 – 2.30	0.48
Housing Tenure					
Privately owned	43 (63.2)	25 (36.8)	1.00		
Rented	127 (57.2)	95 (42.8)	1.28	0.73 – 2.25	0.37
Other (e.g. Hostel)	27 (44.3)	34 (55.7)	2.16	1.06 - 4.38	0.03
Place of birth					
UK born	149 (61.3)	94 (38.7)	1.00		
Non-UK born	152 (53.9)	130 (46.1)	1.35	0.96 – 1.92	0.08
Report of social isolation					
No	122 (54.7)	101 (45.3)	1.00		1
Yes	147 (59.3)	101 (40.7)	0.82	0.57 – 1.19	0.31
Ever smoked					

cannabis					
No	118 (57.0)	89 (43.0)	1.00		
Yes	128 (55.9)	101 (44.1)	1.04	0.71 – 1.52	0.81
Time of FEP contact					
Office hours	256 (70.5)	107 (29.5)	1.00		
Out of hours	64 (32.8)	131 (67.2)	4.89	3.36 - 7.12	<0.001
DUP					
Short (< 1year)	208 (52.0)	192 (48.0)	1.00		
Long (>1 year)	112 (70.9)	46 (29.1)	0.44	0.29 - 0.66	<0.001
Previous service use					
No	239 (56.1)	187 (43.9)	1.00		
Yes	81 (61.4)	51 (38.6)	0.80	0.53 – 1.19	0.28
Mode of onset					
Acute (within a week)	43 (37.1)	73 (62.9)	1.00		
Moderate (within a month)	60 (54.0)	51 (46.0)	0.50	0.29 - 0.85	0.01
Gradual (up to 6months)	71 (58.2)	51 (41.8)	0.42	0.25 – 0.71	0.001
Insidious (more than 6 months)	146 (69.9)	63 (30.1)	0.25	0.15 - 0.41	<0.001
Source of referral					
GP	177 (90.3)	19 (9.7)	1.00		
A&E	92 (42.4)	125 (57.6)	12.65	7.34 – 21.81	<0.001
Police/CJA	5 (6.5)	72 (93.5)	134.14	48.25 - 372.95	<0.001
Other	46 (67.7)	22 (32.3)	4.45	2.22 – 8.92	<0.001
Family involvement					
No	214 (59.1)	148 (40.9)			
Yes	106 (54.1)	90 (45.9)	1.22	0.86 - 1.74	0.25
Early intervention service					
No	194 (58.1)	140 (41.9)	1.00		
Yes	125 (56.3)	97 (43.7)	1.07	0.76 - 1.51	0.67

8.3.2 Unadjusted odds ratios for compulsory admission at first contact

Table 8.4 presents proportions and univariable logistic regression analyses of compulsory admission by sociodemographic, clinical and pathways to care variables. From this table, the following observations were made: Black African patients were almost two and a half times more likely to be compulsorily admitted (OR = 2.46; 95% Cl 1.40 - 4.33) than were the White

British, there was a weak evidence of association between compulsory admission and Mixed (OR = 2.39; 95% CI 0.95 – 5.98) and Asian (OR = 2.00; 95% CI 0.91 – 4.41) ethnic groups. There was no difference between White British and all other ethnic groups. Being married or in a steady relationship appears to be a protective factor for compulsory admission (OR = 0.60; 95% CI 0.36 – 1.00). Patients born outside the UK were almost twice as likely to be compulsorily admitted (OR = 1.72; 95% CI 1.14 – 2.60).

Furthermore, there was evidence that those detained compulsorily had the most adverse pathways to care (Table 8.3) The results also suggest that those with longer DUP were less likely to be detained (OR = 0.38; 95% CI 0.23 - 0.63), so were those with previous contact with mental health services (OR = 0.78; 95% CI 0.50 - 1.21), but this was a weak association and did not reach a statistical significance level.

In terms of mode of onset, the evidence suggests that people with longer and insidious onset of psychosis were less likely to be compulsorily detained compared with those with acute onset (Table 8.4).

There was insufficient evidence to suggest that sociodemographic and other pathways to care factors such as gender, age, employment status, educational qualification, living circumstances, report of social isolation, cannabis use, type of service use (i.e. EIS or previous service), family involvement in help-seeking or housing tenure were associated with compulsory admission.

	Non- compulsory n=423 (%)	Compulsory admission n= 135(%)	Unadjusted OR	95% confidence interval	P=value
Gender					
Male	214 (73.3)	78 (26.7)	1.00		
Female	209 (78.6)	57 (21.4)	0.74	0.50-1.10	0.14
Age-band					
18-29	182 (75.2)	60 (24.8)	1.00		
30-64	241 (76.3)	75 (23.7)	0.94	0.63 – 1.39	0.77
Ethnicity					

 Table 8.4: Unadjusted odds ratios of compulsory admission by socio-demographic and pathways to care variables at first contact

White British	110 (82.7)	23 (17.3)	1.00		
Black African	97 (66.0)	50 (34.0)	2.46	1.40 - 4.33	0.002
Black Caribbean	72 (79.1)	19 (20.9)	1.26	0.64 – 2.48	0.50
White Other	61 (81.3)	14 (18.7)	1.09	0.52 – 2.28	0.80
Asian	31 (70.4)	13 (29.6)	2.00	0.91 - 4.41	0.08
Mixed	18 (66.7)	13 (29.3)	2.39	0.95 – 5.98	0.06
Other	34 (82.9)	7 (17.1)	0.98	0.38 – 2.49	0.97
GP Registered					
No	13 (61.9)	8 (38.1)	1.00		
Yes	410 (76.3)	127 (23.7)	0.50	0.20 – 1.24	0.13
Ever employed					
No	33 (78.6)	9 (21.4)	1.00		
Yes	305 (75.5)	99 (24.5)	1.19	0.55 – 2.57	0.65
Relationship status					
Single	242 (43.1)	89 (26.9)	1.00		
Married / Steady	104 (81.9)	23 (18.1)	0.60	0.36 - 1.00	0.05
relationship					
Divorced/Separated	60 (81.1)	14 (18.9)	0.63	0.33 – 1.19	0.15
Employment					
Unemployed	267 (77.2)	79 (22.8)	1.00		
Student	42 (70.0)	18 (30.0)	1.44	0.78 – 2.65	0.23
Employed	73 (73.7)	26 (26.3)	1.20	0.72 – 2.01	0.47
Education	(>				
School, no GCSE	75 (75.8)	24 (24.2)	1.00		
School with GCSE	54 (74.0)	19 (26.0)	1.09	0.54 – 2.20	0.78
A level/ Further	87 (76.3)	27 (23.7)	0.96	0.51 – 1.82	0.92
University	77 (71.3)	31 (28.7)	1.25	0.67 – 2.34	0.46
Lives with					
Alone	119 (73 9)	42 (26 1)	1.00		
Family/relatives	253 (77.8)	72 (22.2)	0.80	0 52 - 1 25	0 33
Other	293 (77.8)	17 (22.2)	0.00	0.52 1.25	0.55
Other	38 (69.1)	17 (30.9)	1.20	0.64 - 2.48	0.49
Housing Tenure					
Privately owned	51 (75.0)	17 (25.0)	1.00		
Rented	168 (75.7)	54 (24.3)	0.96	0.51 – 1.80	0.91
Other (e.g. Hostel)	43 (70.5)	18 (29.5)	1.25	0.57 – 2.73	0.56
Place of birth					
UK born	197 (81.1)	46 (18.9)	1.00		
Non-UK born	201 (71.3)	81 (28.7)	1.72	1.14 - 2.60	0.009
Report of social					

isolation					
No	165 (74.0)	58 (26.0)	1.00		
Yes	194 (78.2)	54 (21.8)	0.79	0.51 – 1.21	0.28
Ever smoked cannabis					
No	158 (76.3)	49 (23.7)	1.00		
Yes	171 (74.7)	58 (25.3)	1.09	0.70 – 1.69	0.68
Time of FEP contact					
Office hours	308 (84.8)	55 (15.2)	1.00		
Out of hours	115 (59.0)	80 (41.0)	3.89	2.59 – 5.83	<0.001
DUP					
Short (< 1year)	286 (71.5)	114 (28.5)	1.00		
Long (>1 year)	137 (86.7)	21 (13.3)	0.38	0.23 - 0.63	<0.001
Previous service use					
No	318 (74.6)	108 (25.4)	1.00		
Yes	105 (79.6)	27 (20.4)	0.75	0.47 – 1.21	0.25
Mode of onset					
Acute (within a week)	68 (58.6)	48 (41.4)	1.00		
Moderate (within a month)	84 (75.7)	27 (24.3)	0.45	0.25 – 0.80	0.007
Gradual (up to 6months)	95 (77.9)	27 (22.1)	0.40	0.22 - 0.70	0.002
Insidious (more than 6 months)	176 (84.2)	33 (15.8)	0.26	0.15 - 0.44	<0.001
Source of referral					
GP	192 (98.0)	4 (2.0)	1.00		
A&E	156 (71.9)	61 (28.1)	18.76	6.67 - 52.75	<0.001
Police/CJA	17 (22.1)	60 (77.9)	169.41	54.88 – 522.92	<0.001
Other	58 (85.3)	10 (14.7)	8.27	2.50 - 27.37	0.001
Family involvement					
No	274 (75.7)	88 (24.3)	1.00		
Yes	149 (76.0)	47 (24.0)	0.98	0.65 – 1.47	0.93
Early intervention					
service			1.00		
INO Mar	257 (76.9)	77 (23.1)	1.00	0.70 4.74	0.40
Yes	164 (73.9)	58 (26.1)	1.18	0.79 - 1.74	0.40

8.3.3 Unadjusted odds ratios for 'GP' referral at first contact

In Table 8.5 below, associations between GP referral and sociodemographic and pathways to care variables are presented. It can be seen from the table that there were no associations between GP referral and gender, age, ethnicity, relationship status, place of birth, cannabis use or previous contact with mental health services.

However, with regard to employment, those with student status (OR = 0.53; 95% Cl 0.28 – 1.00) were less likely to be referred by GP compared with those who were unemployed. For housing, those with 'Other' tenure such as hostel or refuge (OR = 0.43; 95% Cl 0.19 – 0.95) and 'Other' living arrangement (OR = 0.28;, 95% Cl 0.12 – 0.64) were less likely to be referred by GP compared to those living in privately owned homes.. Among those reporting social isolation, GP referral was almost twice more likely (OR = 1.57; 95% Cl 1.07 – 2.30) compared to those who did not report social isolation. There was a weak association with divorce/widowed status (OR = 1.61; 95% Cl 0.96 – 2.70) compared with those who were single p = 0.06.

As would be expected, GP referral was less common during out of office hours (OR = 0.10; 95% CI 0.06 – 0.17) compared with office hours and family involvement in help-seeking (OR = 0.63; 95% CI 0.43 – 0.92) compared with those who did not have family input. Patients with longer DUP were nearly twice as likely to be referred by GP (OR = 1.79; 95% CI 1.23 – 2.62) compared to those with short DUP. In addition, GP referral was associated with moderate (OR = 4.26; 95% CI 2.16 – 8.41), gradual (OR = 5.41; 95% CI 2.78 – 10.51), and insidious (OR = 5.40; 95% CI 2.90 – 10.06) mode of onset compared with those with acute onset.

	GP referral No n=362 (%)	GP referral Yes n=196 (%)	Unadjusted OR	95% confidence interval	P=value
Gender					
Male	193 (66.1)	99 (33.9)	1.00		

Table 8.5: Unadjusted odds ratios of GP referral by socio-demographic and pathways to care variables at first contact

Female	169 (62 F)	97 (36.5)	1.11	0.79 – 1.58	0.57
Ago band	(03.5)				
19 20	166	76 (21 4)	1.00		
10-29	100	70 (51.4)	1.00		
20.64	(00.0)	120	1 22	0.02 1.00	0.10
50-04	190	(28 0)	1.55	0.95 -1.90	0.10
Ethnicity	(02.0)	(38.0)			
	02 (62 4)	EO (27 6)	1.00		
Disck African	05 (02.4)	50 (57.0) 4E (20.6)	1.00	0.44 1.20	0.21
DIdCK AITICAT	102	45 (50.0)	0.75	0.44 - 1.20	0.21
Plack Caribboan	(09.4) 60 (65 0)	21 /24 1)	0.95	0.40 1.40	0.59
Mite Other	50 (65.9)	2E(22.2)	0.05	0.49 - 1.49	0.50
Acian	20 (00.7)	25(55.5)	0.65	0.43 - 1.50	0.55
Asidn	29 (05.9)	15 (34.1)	0.85	0.41 - 1.75	0.67
Nixeu Othar	17 (03.)	10(37.0)	0.97	0.41 - 2.29	0.95
Other Beletienskin status	21 (51.2)	20 (48.8)	1.58	0.78 - 3.20	0.20
Relationship status	225	100	1.00		
Single	225	106	1.00		
Maniael / Changler		(32.0)	1.22	0.07 0.04	0.10
Married / Steady	78 (61.4)	49 (38.6)	1.33	0.87 - 2.04	0.18
	42 (56.0)	22 (42 2)	1.61	0.00 0.70	0.00
Divorced/Separated	42 (56.8)	32 (43.2)	1.61	0.96 - 2.70	0.06
Employment	220	420	1.00		
Unemployed	220 (63.6)	(36.4)	1.00		
Student	46 (76.7)	14 (23.3)	0.53	0.28 - 1.00	0.05
Employed	63 (63.6)	36 (36.4)	0.99	0.62 – 1.58	0.99
Education					
School, no GCSE	58 (58.6)	41 (41.4)	1		
School with GCSE	47 (64.4)	26 (35.6)	0.78	0.41 - 1.46	0.44
A level/ Further	73 (64.0)	41 (36.0)	0.79	0.45 – 1.38	0.41
education					
University	77 (71.3)	31 (28.7)	0.56	0.31 - 1.01	0.05
Lives with					
Alone	101	60 (37.3)	1.00		
	(62.7)				
Family/relatives	202	123	1.02	0.69 - 1.51	0.90
	(62.1)	(37.9)			
Other	47 (85.4)	8 (14.6)	0.28	0.12 - 0.64	0.003
(Hostel/Refuge)					
Housing Tenure					
Privately owned	42 (61.8)	26 (38.2)	1.00		
Rented	141	81 (36.5)	0.97	0.52 – 1.62	0.79
	(63.5)				

Other	48 (78.7)	13 (21.3)	0.43	0.19 - 0.95	0.03
Place of birth					
Non -UK born	182	100	1.00		
	(64.5)	(35.5)			
UK born	157	86 (35.4)	0.99	0.69 – 1.42	0.98
	(64.6)				
Report of social					
No	156	67 (30 0)	1.00		
	(70.0)	07 (30.0)	1.00		
Yes	148	100	1.57	1.07 – 2.30	0.02
	(59.7)	(40.3)			
Ever smoked					
cannabis					
No	138	69 (33.3)	1.00		
Maa	(66.7)		0.00	0.00 1.40	0.07
Yes	153	76 (33.2)	0.99	0.66 - 1.48	0.97
Time of FEP contact	(00.8)				
	195	170	1.00		
Office flours	(51.0)	(49.0)	1.00		
Out of hours	177	18 (9.2)	0.10	0.06 - 0.17	<0.0001
	(90.8)				
DUP					
Short	275	125	1.00		
	(68.7)	(31.3)			
Long	87 (55.1)	71 (44.9)	1.79	1.23 – 2.62	0.002
Previous service use					
No	284	142	1.00		
	(66.7)	(33.3)			
Yes	78 (59.1)	54 (40.9)	1.38	0.92 – 1.06	0.11
Mode of onset					
Acute (within a	102	14 (12.1)	1.00		
week)	(87.9)				
Moderate (within a	70 (63.1)	41 (36.9)	4.26	2.16 - 8.41	<0.0001
month) Gradual (up to	70 (57 4)	52 (12 6)	5 <i>4</i> 1	2 78 - 10 51	<0.0001
6months)	70 (37.4)	52 (42.0)	3.71	2.70 - 10.51	~0.0001
Insidious (more than	120	89 (42.6)	5.40	2.90 - 10.06	<0.0001
6 months)	(57.4)				
Family involvement					
No	222	140	1.00		
	(61.3)	(38.7)			
Yes	140	56 (28.6)	0.63	0.43 - 0.92	0.01

	(71.4)				
Early intervention					
service use					
No	214	120	1.00		
	(64.1)	(35.9)			
Yes	147	75 (33.8)	0.90	0.63 – 1.29	0.60
	(66.2)				

8.3.3.1 Associations between ethnicity and DUP, stratified by GP referral

Table 8.5.1 shows the stratified analysis of ethnic differences in DUP by GP referral. As it can be seen from the table, there were no evidence of differences between those with short or long DUP in their referral GP status.

	GP referral (Yes)		<i>X</i> ² , (df), p	X ² , (df), p GP referral (No)		
	N=196			N= 362		
Ethnicity	DUP short	DUP long		DUP short	DUP long	
-	n=125 (%)	n=71 (%)		n=275 (%)	n=87 (%)	
White British	33 (26.4)	17 (24.0)	4.16 (6),	62 (22.5)	21 (24.1)	1.96 (6),
			0.65			0.95
Black African	28 (22.4)	17 (24.0)		77 (28.0)	25 (28.7)	
Black	20 (16.0)	11 (15.5)		45 (16.4)	15 (17.2	
Caribbean						
White Other	18 (14.4)	7 (9.9)		36 (13.1)	14 (16.1)	
Asian	10 (8.0)	5 (7.0)		24 (8.7)	5 (5.8)	
Mixed	7 (5.6)	3 (4.2)		14 (5.1)	3 (3.5)	
Other	9 (7.2)	11 (15.4)		17 (6.2)	4 (4.6)	

Table 8.5.1: Ethnic differences in DUP stratified by GP referral

DUP short: <1 year

DUP long: > 1 year

8.3.4 Unadjusted odds ratios for A&E referral at first contact

Table 8.6 displays the unadjusted logistic regression of A&E source of referral by sociodemographic and pathways to care variables. The table shows that there were no differences by ethnicity among those referred via A&E. Although a higher proportion of the older age group presented to A&E, this did not reach the traditional (p<0.05) statistical significance. Students and those employed were two times more likely to be come into contact via A&E (OR = 2.26; 95% CI 1.30 – 3.94 and OR = 2.10; 95% CI 1.33 – 3.31)

respectively compared to those who were unemployed. Similarly, those who were educated above GCSE (A 'level OR = 1.77; 95% Cl 1.00 - 3.16 and university OR = 1.88; 95% Cl 1.05 - 3.36) were almost twice likely to present to A&E compared with those without GCSE qualification.

With regard to living circumstances, those who lived with relatives and friends were almost twice more likely to present to A&E (OR = 1.70; 95% CI 1.14 – 2.55) compared with those that lived alone. Consequently, the data shows that family involvement in help-seeking was almost four times more common in A&E referral (OR = 3.74; 95% CI 2.59 – 5.40). Conversely, patients who reported social isolation were less likely to present to A&E (OR = 0.55; 95% CI 0.38 – 0.80). Unsurprisingly, those who present to A&E tend to do so out of office hours, and in fact are almost seven times more likely to do so (OR = 6.59; 95% CI 4.48 – 9.68).

In terms of mode of onset, patients with non-acute onset (moderate onset OR = 0.35; 95% CI 0.20 - 0.61; gradual onset OR = 0.27; 95% CI 0.16 - 0.47 and insidious onset OR = 0.25; 95% CI 0.15 - 0.41) were less likely to present to A&E as well as those with longer DUP (OR = 0.62; 95% CI 0.41 - 0.91). In addition, those who have used mental health services in the past were also less likely to present to A&E (OR = 0.50; 95% CI 0.33 - 0.77), compared to those who had not.

The data did not show evidence of association between A&E referral and sociodemographic factors such as gender, relationship status, registered with GP, housing tenure or cannabis use.

	A&E referral No n= 341 (%)	A&E referral Yes n=217 (%)	Unadjusted OR	95% confidence interval	P=value
Gender					
Male	180	112	1.00		
	(61.6)	(38.4)			
Female	161	105	1.04	0.74 – 1.47	0.78
	(60.5)	(39.5)			

 Table 8.6: Unadjusted odds ratios of A&E referral by socio-demographic and pathways to care

 variables at first contact

Age-band					
18-29	138	104	1.00		
	(57.0)	(43.0)			
30-64	203	113	0.73	0.52 – 1.04	0.08
	(64.2)	(35.8)			
Ethnicity					
White British	81 (60.9)	52 (39.1)	1.00		
Black African	85 (57.8)	62 (42.2)	1.13	0.70 - 1.83	0.60
Black Caribbean	61 (67.0)	30 (33.0)	0.76	0.43 – 1.33	0.35
White Other	45 (60.0)	30 (40.0)	1.03	0.58 – 1.85	0.89
Asian	23 (52.3)	21 (47.7)	1.42	0.71 – 2.82	0.31
Mixed	18 (66.7)	9 (33.3)	0.77	0.32 – 1.86	0.57
Other	28 (68.3)	13 (31.7)	0.72	0.34 – 1.52	0.39
GP Registered					
No	11 (52.4)	10 (47.6)	1.00		
Yes	330	207	0.69	0.28 – 1.65	0.40
	(61.4)	(38.6)			
Relationship status					
Single	198	133	1.00		
	(59.8)	(40.2)			
Married / Steady	73(57.5)	54 (42.5)	1.10	0.72 – 1.66	0.64
relationship					
Divorced/Separated	49 (66.2)	25 (33.8)	0.75	0.44 - 1.28	0.30
Employment					
Unemployed	230	116	1.00		
	(66.5)	(33.5)			
Student	28 (46.7)	32 (53.3)	2.26	1.30 - 3.94	0.004
Employed	48 (48.5)	51 (51.5)	2.10	1.33 – 3.31	0.001
Education					
School, no GCSE	71 (71.7)	28 (28.3)	1.00		
School with GCSE	43 (58.9)	30 (41.1)	1.76	0.93 – 3.35	0.08
A level/ Further	67 (58.8)	47 (41.2)	1.77	1.00 - 3.16	0.05
education	CO (57 A)		4.00	4.05.0.00	
University	62 (57.4)	46 (42.6)	1.88	1.05 - 3.36	0.03
Lives with					
Alone	112	49 (30.4)	1.00		
	(69.6)				
Family/relatives	186	139	1.70	1.14 – 2.55	0.009
	(57.2)	(42.8)			
Other	32 (58.4)	23 (41.8)	1.64	0.87 - 3.09	0.12
(Hostel/Refuge)					
Housing Tenure					
Privately owned	40 (58.8)	28 (41.2)	1.00		
Rented	135	87 (39.2)	0.92	0.52 - 1.60	0.76
	(60.8)				

Other	38 (62.3)	23 (37.7)	0.86	0.42 -1.75	0.68
Place of birth					
Non-UK born	171	111	1		
	(60.6)	(39.4)			
UK born	151	92 (37.9)	0.93	0.65 – 1.33	0.72
	(62.1)				
Report of social					
No	115	100	1.00		
INO	(51.6)	108 (48 4)	1.00		
Yes	163	85 (34.3)	0.55	0.38 - 0.80	0.002
	(65.7)	00 (0 110)	0.00		
Ever smoked	,				
cannabis					
No	118	89 (43.0)	1.00		
	(57.0)				
Yes	137	92 (40.2)	0.89	0.60 -1.30	0.55
	(59.8)				
Time of FEP contact					
Office hours	277	86 (23.7)	1.00		
	(76.3)	4.2.4	6.50		
Out of hours	64 (32.8)	131 (67.2)	6.59	4.48 – 9.68	<0.0001
DUP					
Short	232	168	1.00		
	(58.0)	(42.0)			
Long	109 (69.0)	49(31.0)	0.62	0.41 – 0.91	0.01
Previous service use					
No	245	181	1.00		
	(57.5)	(42.5)			
Yes	96 72.7()	36 (27.3)	0.50	0.33 – 0.77	0.002
Mode of onset					
Acute (within a	43 (37.1)	73 (62.9)	1.00		
week)					
Moderate (within a month)	69 (62.2)	42 (37.8)	0.35	0.20 - 0.61	<0.0001
Gradual (up to 6months)	83 (68.0)	39 (32.0)	0.27	0.16 - 0.47	<0.0001
Insidious (more than 6 months)	146 (69.9)	63 (30.1)	0.25	0.15 - 0.41	<0.0001
Family involvement	(
No	261	101	1.00		
	(72.1)	(27.9)			

Yes	80 (40.8)	116	3.74	2.59 – 5.40	<0.0001
		(59.2)			
Early intervention					
service use					
No	207	127	1.00		
	(62.0)	(38.0)			
Yes	132	90 (40.5)	1.11	0.78 – 1.57	0.55
	(59.5)				

A&E, Accident and Emergency.

8.3.5 Unadjusted odds ratios for Police referral at first contact

Associations between police/CJA referral by sociodemographic and pathways to care are presented in Table 8.7. From this table, it can be seen that there were no differences in police referral by gender, age, education level, employment status, housing tenure, place of birth, previous service use, mode of onset, cannabis use or ethnicity.

However, four key variables were associated with police referral. Those registered with GP were less likely to be referred by police/CJA (OR = 0.29; 95% CI 0.11 – 0.76) compared to those who were not. Being married or in a steady relationship was associated with a reduced likelihood of police/CJA referral (OR = 0.41; 95% CI 0.19 – 0.87). Similarly, those living with family and friends (OR = 0.46; 95%CI 0.26 – 0.79) and those who had family involvement in their help-seeking (OR = 0.20; 95% CI 0.10 – 0.42, p<0.001) were less likely to be referred by police/CJA compared with those who lived alone. In addition, police/CJA referral was almost three times more likely to occur during out of office hours (OR = 2.57; 95% CI 1.57 – 4.19) compared to normal working hours.

Table 8.7: Unadjusted o care variables at first co	dds ratios of p ntact	oolice/CJA refe	rral by socio-den	nographic and pa	thways to
	Police/CJA	Police/CJA	Unadiusted	95%	P=value

	Police/CJA referred No n=481 (%)	Police/CJA referred Yes n=77 (%)	Unadjusted OR	95% confidence interval	P=value
Gender					
Male	248 (84.9)	44 (15.1)	1.00		
Female	233 (87.6)	33 (12.4)	0.79	0.49 - 1.29	0.36
Age-band					
18-29	208 (85.9)	34 (14.1)	1.00		
----------------------------	-------------	-----------	------	-------------	-------
30-64	273 (86.4)	43 (13.6)	0.96	0.59 – 1.56	0.88
Ethnicity					
White British	117 (88.0)	16 (12.0)	1.00		
Black African	123 (83.7)	24 (16.3)	1.42	0.72 – 2.82	0.30
Black Caribbean	80 (87.9)	11 (12.1)	1.00	0.44 – 2.27	0.99
White Other	64 (85.3)	11 (14.7)	1.25	0.55 – 2.87	0.57
Asian	39 (88.6)	5 (11.4)	0.93	0.73 – 2,15	0.90
Mixed	21 (77.8)	6 (22.2)	2.08	0.73 – 2.51	0.16
Other	37 (90.2)	4 (9.8)	2.08	0.24 – 2.51	0.69
GP Registered					
No	14 (66.7)	7 (33.3)	1.00		
Yes	467 (87.0)	70 (13.0)	0.29	0.11 - 0.76	0.01
Relationship status					
Single	280 (84.6)	51 (15.4)	1.00		
Married / Steady	118 (92.9)	9 (7.1)	0.41	0.19 – 0.87	0.02
relationship					
Divorced/Separated			0.66	0.30 - 1.46	0.31
Employment					
Unemployed	298 (86.1)	48 (13.9)	1.00		
Student	51 (85.0)	9 (15.0)	1.09	0.50 – 2.36	0.81
Employed	90 (90.9)	9 (9.1)	0.62	0.29 – 1.31	0.21
Education					
School, no GCSE	82 (82.8)	17 (17.2)	1.00		
School with GCSE	67 (91.8)	6 (8.2)	0.43	0.16 - 1.15	0.09
A level/ Further education	99 (86.8)	15 (13.6)	0.73	0.34 – 1.55	0.41
University	88 (81.5)	20 (18.5)	1.09	0.53 – 2.23	0.80
Lives with					
Alone	131 (81.4)	30 (18.6)	1.00		
Family/relatives	294 (90.5)	31 (9.5)	0.46	0.26 - 0.79	0.005
Other	42 (76.4)	13 (23.6)	1.35	0.64 – 2.82	0.42
(Hostel/Refuge)					
Housing Tenure					
Privately owned	58 (85.3)	10 (14.7)	1.00		
Rented	192 (86.5)	30 (13.5)	0.90	0.41 - 1.96	0.80
Other	49 (80.3)	12.19.7()	1.42	0.56 – 3.56	0.45
Place of birth					
No-UK born	243 (86.2)	39 (13.8)	1.00		
UK born	20.7 (85.2)	36 (14.8)	1.08	0.66 - 1.76	0.78
Report of social isolation					
No	198 (88.8)	25 (11.2)	1.00		

Yes	211 (85.1)	37 (14.9)	1.38	0.80 - 2.39	0.23
Ever smoked					
cannabis					
Yes	179 (86.5)	28 (13.5)	1.00		
No	198 (86.5)	31 (13.5)	1.00	0.57 – 1.73	0.99
Time of FEP contact					
Office hours	328 (90.4)	35 (9.6)	1.00		
Out of hours	153 (78.5)	42 (21.5)	2.57	1.57 – 4.19	<0.0001
DUP					
Short	338 (84.5)	62 (15.5)	1.00		
Long	143 (90.5)	15 (9.5)	0.57	0.31 - 1.08	0.06
Previous service use					
No	364 (85.4)	62 (14.6)	1.00		
Yes	117 (88.6)	15 (11.4)	0.75	0.41 - 1.37	0.35
Mode of onset					
Acute (within a week)	97 (83.6)	19 (16.4)	1.00		
Moderate (within a month)	91 (82.0)	20 (18.0)	1.12	0.56 – 2.23	0.74
Gradual (up to 6months)	103 (84.4)	19 .15.6()	0.94	0.47 – 1.88	0.86
Insidious (more than 6 months)	190 (90.9)	19 (9.1)	0.51	0.25 - 1.00	0.05
Family involvement					
No	294 (81.2)	68 (18.8)	1.00		
Yes	187 (95.4)	9 (4.6)	0.20	0.10 -0.42	<0.0001
Early intervention					
service use					
No	290 (86.8)	44 (13.2)	1.00		
Yes	190 (85.6)	32 (14.4)	1.11	0.67 – 1.81	0.67

8.3.6 Unadjusted odds ratios for 'Other' referral at first contact

Table 8.8 displays associations between 'Other' source of referral (i.e. other healthcare worker, social services) and sociodemographic and pathways to care variables. The table shows that seven variables were associated with being referred via an 'Other' route. Black Caribbean patients (OR = 2.07; 95% CI 0.99 - 4.34) were twice more likely to be referred via 'Other' routes compared with White British. Compared to those unemployed, those in

employment were less likely to be referred by 'Other' sources (OR = 0.16; 95% Cl 0.04 - 0.52).

In terms of housing tenure, those who lived in an 'Other' housing arrangement (e.g. hostel, refuge) were four times more likely (OR = 4.33; 95% Cl 1.32 - 14.12) to be referred through an 'Other' route compared to those living in a privately owned home.

Furthermore, those contacting services via an 'Other' route were less likely to do so during out of office hours (OR = 0.09; 95%Cl 0.03 - 0.27). Patients with insidious onset (OR = 2.35; 95% Cl 1.12 - 4.92) were twice more likely to be referred via an 'Other' route and they were two times more likely to have used mental health services previously (OR = 2.41; 95% Cl 1.41 - 4.10). Family involvement (OR = 0.48; 95% Cl 0.26 - 0.88) was less common among patients referred through 'Other' routes.

There were no significant associations with the remaining variables in the table.

	'Other' source referred Yes n=490 (%)	'Other source' referred No n=68 (%)	Unadjusted OR	95% confidence interval	P=value
Gender					
Male	255 (87.3)	37 (12.7)	1		
Female	235 (88.4)	31 (11.6)	0.90	0.54 – 1.51	0.71
Age-band					
18-29	214 (88.4)	28 (11.6)	1.00		
30-64	276 (87.3)	40 (12.7)	1.10	0.66 – 1.85	0.69
Ethnicity	()	()			
White British	118 (88.7)	15 (11.3)	1.00		
Black African	131	16 (10.9)	0.96	0.45 - 2.02	0.91

Table 8.8: Unadjusted odds ratios of 'Other	' source of referral by socio-demographic and pathways
to care variables at first contact	

	(89.1)				
Black Caribbean	72 (79.1)	19 (20.9)	2.07	0.99 - 4.34	0.05
White Other	66 (88.0)	9 (12.0)	1.07	0.44 – 2.58	0.87
Asian	41 (93.2)	3 (6.8)	0.57	0.15 - 2.09	0.40
Mixed	25 (92.6)	2 (7.4)	0.62	0.13 – 2.92	0.55
Other	37 (90.2)	4 (9.8)	0.85	0.26 – 2.72	0.78
GP Registered					
No	18 (85.7)	3 (14.3)	1.00		
Yes	472	65 (12.1)	0.82	0.23 – 2.88	0.76
	(87.9)				
Relationship status					
Single	290	41 (12.4)	1.00		
	(87.6)				
Married / Steady	112	15 (11.8)	0.94	0.50 – 1.77	0.86
relationship	(88.2)				
Divorced/Separated	65 (87.8)	9 (12.2)	0.97	0.45 – 2.11	0.95
Employment					
Unemployed	290	56 (16.2)	1		
	(83.8)				
Student	55 (91.5)	5 (8.3)	0.47	0.18 – 1.22	0.12
Employed	96 (97.0)	3 (3.0)	0.16	0.04 - 0.52	0.003
Education					
School, no GCSE	86 (86.9)	13 (13.1)	1.00		
School with GCSE	62 (84.9)	11 (15.1)	1.17	0.49 – 2.79	0.71
A level/ Further	103	11 (9.7)	0.70	0.30 - 1.65	0.42
education	(90.3)				
University	97 (89.8)	11 (10.2)	0.75	0.32 – 1.76	0.50
Lives with					
Alone	139	22 (13.7)	1.00		
	(86.3)				
Family/relatives	293	32 (9.8)	0.69	0.38 – 1.23	0.20
	(90.2)				
Other	44 (80.)	11 (20.0)	1.57	0.71 – 3.51	0.26
(Hostel/Refuge)					
Housing Tenure					
Privately owned	64 (94.1)	4 (5.9)	1.00		
Rented	198	24 (10.8)	1.93	0.64 – 5.79	0.23
	(89.2)				
Other	48 (78.7)	13 (21.3)	4.33	1.32 – 14.12	0.01
Place of birth					
Non- UK born	250	32 (11.3)	1.00		
	(88.7)				
UK born	214	29 (11.9)	1.05	0.62 – 1.80	0.83

	(88.1)				
Report of social					
isolation					
No	200 (89.7)	23 (10.3)	1.00		
Yes	222 (89.5)	26 (10.5)	1.01	0.56 - 1.84	0.95
Ever smoked cannabis					
No	186 (89.9)	21 (10.4)	1.00		
Yes	199 (86.9)	30 (13.1)	1.33	0.73 – 2.41	0.33
Time of FEP contact					
Office hours	299 (82.4)	64 (17.6)	1.00		
Out of hours	191 (97.9)	4 (2.1)	0.09	0.03 - 0.27	<0.001
DUP					
Short	355 (88.7)	45 (11.3)	1.00		
Long	135 (85.4)	23 (14.6)	1.34	0.78 – 2.30	0.28
Previous service use					
No	385 (90.4)	41 (9.6)	1.00		
Yes	105 (79.6)	27 (20.5)	2.41	1.41 - 4.10	0.001
Mode of onset					
Acute (within a week)	106 (91.4)	10 (8.6)	1.00		
, Moderate (within a month)	103 (92.8)	8 (7.2)	0.82	0.31 – 2.16	0.69
Gradual (up to	110	12 (9.8)	1.15	0.47 – 2.78	0.74
6months)	(90.2)				
Insidious (more than	171	38 (18.2)	2.35	1.12 – 4.92	0.02
Family involvement	(01.0)				
No	300	53 (1/ 6)	1.00		
	(85.4)	55 (14.0)	1.00		
Yes	181	15 (7.6)	0.48	0.26 - 0.88	0.01

	(92.4)				
Early intervention					
service use					
No	291	43 (12.9)	1.00		
	(87.1)				
Yes	197	25 (11.3)	0.85	0.50 – 1.45	0.57
	(88.7)				

8.4 Adjusted Odds of admission (hospital and compulsory) and source of referral by ethnicity at first contact

Results from the unadjusted regression models above revealed that only three of the outcomes variables were associated with ethnicity, namely hospital admission, compulsory admission and 'Other' referral source. It is still important, however, to assess association of ethnicity with all pathways to care outcomes and therefore, all five outcomes and ethnicity were assessed further and entered into fully adjusted logistic regression analyses.

Adjusted logistic regression models were fitted for each of the outcome variables with ethnicity as predictor variable. Three logistic regression models were fitted, unadjusted and then adjusted for *a priori* confounders as follows:

Model 1 – unadjusted

Model 2 – adjusted for age, gender

Model 3 – all variables in model 2, plus education level and employment status

Two further independent variables were adjusted for in the final model, namely duration of untreated psychosis and family involvement as they showed associations with the outcome variables under consideration. Therefore, a fourth logistic regression model was fitted as follows:

Model 4 – all variables in model 3, plus duration of untreated psychosis and family involvement

All analyses henceforth were carried out using complete case sample n = 367 (66% of whole sample) for whom data were complete for the main exposure variable, and each of the confounding variables (e.g. information on education level and employment status was not

available for 191 patients) were included in the regression model. This will allow comparison of associations across the models (i.e. unadjusted and adjusted).

8.4.1 Adjusted odds ratios for hospital admission by ethnicity at first contact

Table 8.9 shows the associations between ethnicity and hospital admission at first contact for psychosis. The results in this table show that three main ethnic groups (Black African, White Other and Mixed) showed consistent and strong associations with hospital admission throughout all the regression models. Interestingly, for the three groups, the effect sizes of association attenuated from the univariable analysis when age and gender were controlled for in model 2. Black African (OR = 2.31; 95% Cl 1.27 – 4.21 to adj. OR = 2.24; 95% Cl 1.23 – 4.11) White Other (OR = 2.03; 95% Cl 1.01 – 4.08 to adj. OR = 1.96; 95% Cl 0.97 – 3.97) and Mixed (OR = 2.71; 95% Cl 1.02 – 7.41 to adj. OR = 2.64; 95% Cl 0.99 – 7.00). However, these became stronger after adjusting for socioeconomic and pathways to care variables in models 3 and 4 (Table 8.9).

	Model 1	95% confidence interval (p-value)	Model 2	95% confidence interval (p-value)	Model 3	95% confidence interval (p-value)	Model 4	95% confidence interval (p-value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	2.31	1.27 -4.21 (<0.01)	2.24	1.23 -4.11 (<0.01)	2.27	1.23 -4.18 (<0.01)	2.38	1.27 -4.42 (<0.01)
Black Caribbean	1.12	0.56 -2.27 (0.73)	1.14	0.56 -2.31 (0.71)	1.18	0.57 -2.42 (0.64)	1.20	0.57 -2.49 (0.62)
White Other	2.03	1.01 -4.08 (0.04)	1.96	0.97 -3.97 (0.05)	2.02	0.99 -4.11 (0.05)	2.08	1.01 -4.27 (0.04)
Asian	1.89	0.81 -4.43 (0.13)	1.97	0.83 -4.64 (0.12)	1.94	0.81 -4.61 (0.13)	2.10	0.86 -5.11 (0.10)
Mixed	2.71	1.02 – 7.14 (0.04)	2.64	0.99 -7.00 (0.05)	2.71	1.02 -7.22 (0.04)	2.65	0.99 -7.09 (0.05)
Other	0.61	0.31 -0.76 (<0.01)	0.54	0.19 -1.51 (0.24)	0.56	0.20 -1.59 (0.28)	0.58	0.20 -1.65 (0.31)

Table 8.9: Adjusted odds ratios of associations between hospital admission and ethnicity at first contact (n=367)

Model 1 – unadjusted; Model 2 – adjusted for age and gender; Model 3 – adjusted for age, gender, level of education, employment status; Model 4 – all variables in model 3 plus duration of untreated psychosis and family involvement

8.4.2 Adjusted odds ratios for compulsory admission by ethnicity at first contact

To begin with, a Mantel Haenzel test of homogeneity was carried out to test for effect modification for gender and the following results were observed: X^2 = 4.29, df=1, p=0.03. With the Mantel Haenzel significant test result showing that association for men and women differs, stratified analyses by gender were necessary to further investigate differences in associations for compulsory admission and ethnicity by gender, whilst adjusting for the remaining *a priori* confounders.

8.4.3 Adjusted odds ratios for compulsory admission by ethnicity and gender at first contact

Table 8.10 shows the results for men and reveals that, in all four logistic regression models, only Black African men had strong association with compulsory admission. It is worth highlighting the change in effect sizes between models 2 (adj. OR = 2.54; 95% CI 1.02 – 6.29, p = 0.04) and 3 (adj. OR = 2.68; 95% CI 1.06 – 6.74, p = 0.03), which also held in model 4 (adj. OR = 3.25; 95% CI 1.24 – 8.52, p = 0.01). There were no further ethnic differences observed for compulsory admission compared with White British patients.

For women, the results were somewhat different. In Table 8.11, models 1–4 showed that women of 'Mixed', Asian and Black African ethnic groups were more likely to be compulsorily admitted at first contact for psychosis and these were independent of potential confounders. The wide confidence intervals in the adjusted and unadjusted models are noteworthy due to the smaller sample size for women.

	Model 1	95% confidence interval (p- value)	Model 2	95% confidence interval (p- value)	Model 3	95% confidence interval (p- value)	Model 4	95% confidence interval (p- value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	2.55	1.03 -6.29 (0.04)	2.54	1.02 -6.29 (0.04)	2.68	1.06 -6.74 (0.03)	3.25	1.24 -8.52 (0.01)
Black Caribbean	1.32	0.43 -4.06 (0.62)	1.32	0.42 -4.05 (0.62)	1.49	0.46 -4.79 (0.49)	1.65	0.50 -5.44 (0.40)
White Other	0.48	0.13 -1.71 (0.26)	0.48	0.13 -1.72 (0.26)	0.51	0.14 -1.85 (0.31)	0.54	0.15 -1.98 (0.35)
Asian	1.48	0.31 -6.69 (0.62)	1.45	0.31 -6.70 (0.62)	1.51	0.31 -7.25 (0.60)	1.44	0.30 -6.98 (0.64)
Mixed	1.27	0.28 -5.72 (0.75)	1.27	0.28 -5.72 (0.75)	1.28	0.28 -5.87 (0.75)	1.42	0.30 -6.55 (0.65)
Other	0.60	0.14 -2.47 (0.49)	0.60	0.14 -2.47 (0.48)	0.64	0.15 -2.71 (0.55)	0.70	0.16 -3.02 (0.63)

 Table 8.10: Adjusted odds ratios of associations between compulsory admission and ethnicity for men at first contact (n=191)

Model 1 – unadjusted; Model 2 – adjusted for age; Model 3 – adjusted for age, gender, level of education, employment status; Model 4 – all variables in model 3 plus duration of untreated psychosis and family involvement

Table 8.11: Adjusted odds ratios of associations between compulsory admission and ethnicity for women at first contact (n=176)

Model	95% confidence	Model	95% confidence	Model 3	95% confidence	Model 4	95% confidence
1	interval (p-value)	2	interval (p-value)		interval (p-value)		interval (p-
							value)

Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	4.05	1.31 -12.49 (0.01)	4.26	1.37 -13.29 (0.01)	3.84	1.21 -12.17 (0.02)	3.88	1.19 -12.57 (0.02)
Black Caribbean	2.01	0.55 -7.29 (0.25)	2.03	0.56 -7.40 (0.27)	2.21	0.59 -8.25 (0.23)	2.43	0.62 -9.45 (0.20)
White Other	2.80	0.70 -11.04 (0.14)	2.82	0.71 -11.16 (0.14)	2.71	0.65 -11.20 (0.16)	2.72	0.64 -11.55 (0.17)
Asian	3.87	1.01 -14.80 (0.04)	4.14	1.07 -16.05 (0.03)	3.96	1.00-15.56 (0.04)	4.18	1.02 -17.15 (0.04)
Mixed	8.40	1.71 -39.49	8.76	1.84 -41.59	10.01	2.02 -49.66 (<0.01)	8.36	1.67 - 41.85
		(<0.01)		(<0.01)				0.01)
Other	1.68	0.16 -17.41 (0.66)	1.81	0.17 -18.96 (0.61)	2.23	0.20 -24.92 (0.51)	2.10	0.18 -23-34 (0.54)

Model 1- unadjusted ; Model 2 – adjusted for age; Model 3- adjusted for age, gender, level of education, employment status; Model 4- all variables in model 3 plus duration of untreated psychosis and family involvement

The relationships between compulsory admission and ethnicity for the overall sample are shown in Table 8.12, along with 95% confidence intervals (CI) for association with being compulsorily admitted.

The results from models 1 and 2 (Table 8.12) show that Black African and 'Mixed' ethnic group patients were more likely to be detained compared with White British. Meanwhile, a weak association was observed in the Asian group (adj. OR = 2.39; 95% Cl 0.90 – 6.30). There were no associations between compulsory admission and Black Caribbean or any other ethnic groups.

After adjusting for socioeconomic factors in model 3, (Table 8.12), strong associations remained for Black African and 'Mixed' patients (adj. OR = 3.11; 95% Cl 1.53 – 6.33 and adj. OR = 3.29; 95% Cl 1.14 – 9.43 respectively). A weak association between compulsory admission and Asian ethnic group also remained (adj. OR = 2.35; 95% Cl 0.88 – 6.28, p = 0.08). No associations were observed among the Black Caribbean or any other ethnic groups compared with White British patients.

In the final model where duration of untreated psychosis and family involvement in helpseeking were taken into account (Table 8.12), the strength of association with Black African, 'Mixed' and Asian ethnic groups remained relatively the same for compulsory admission.

	Model 1	95% confidence interval (p- value)	Model 2	95% confidence interval (p- value)	Model 3	95% confidence interval (p- value)	Model 4	95% confidence interval (p- value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	3.11	1.55-6.23 (<0.01)	3.13	1.56 -6.37 (<0.01)	3.11	1.53 -6.33 (<0.01)	3.23	1.57 -6.63 (<0.01)
Black Caribbean	1.53	0.66 -3.51 (0.31)	1.55	0.67 -3.57 (0.30)	1.72	0.73 -4.04 (0.20)	1.78	0.75 -4.24 (0.18)
White Other	1.06	0.42 -2.62 (0.89)	1.02	0.41 -2.55 (0.95)	1.06	0.42- 2.67 (0.88)	1.08	0.43 -2.73 (0.86)
Asian	2.28	0.87 -5.96 (0.09)	2.39	0.90 -6.30 (0.07)	2.35	0.88 -6.28 (0.08)	2.53	0.93 -6.87 (0.06)
Mixed	3.11	1.10 -8.82 (0.03)	3.15	1.11 -8.96 (0.03)	3.29	1.14 -9.43 (0.02)	3.20	1.10 -9.26 (0.03)
Other	0.92	0.27 -3.06 (0.89)	0.87	0.48 -1.30 (0.82)	0.94	0.27 -3.20 (0.92)	0.94	0.27 -3.25 (0.93)

Table 8.12: Adjusted odds ratios of associations between ethnicity and compulsory admission at first contact

Model 1- unadjusted ; Model 2 – adjusted for age; Model 3- adjusted for age, gender, level of education, employment status; Model 4- all

variables in model 3 plus duration of untreated psychosis and family involvement

8.4.4 Adjusted odds ratios for 'Other' referral by ethnicity at first contact

To begin with, there were no Asian patients referred by 'Other' source of referral, and therefore the number of patients included in this analysis was 338. In assessing the association of 'Other' source of referral and ethnicity, results from models 1–4 (Table 8.13) suggest that there were no ethnic differences in being referred to mental health services via 'Other' (i.e. healthcare professionals, nurse, doctor in non-mental health care setting).

	Model 1	95% confidence interval (p- value)	Model 2	95% confidence interval (p- value)	Model 3	95% confidence interval (p- value)	Model 4	95% confidence interval (p- value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	0.78	0.31 -2.00 (0.61)	0.82	0.32 -2.10 (0.68)	0.76	0.29 -2.00 (0.58)	0.73	0.27 -1.92 (0.52)
Black Caribbean	1.98	0.80 -4.86 (0.13)	2.06	0.83 -5.10 (0.11)	1.85	0.73 -4.72 (0.19)	1.96	0.75 -5.07 (0.16)
White Other	0.94	0.32 -2.73 (0.92)	0.90	0.31 -2.64 (0.86)	0.90	0,30-2.69 (0.82)	0.89	0.29 -2.70 (0.84)
Asian	No cases	-	No cases	-	No cases	-	No cases	-
Mixed	0.75	0.15 -3.74 (0.74)	0.80	0.16 -3.96 (0.78)	0.78	0.15 -3.94 (0.76)	0.78	0.15 -3.96 (0.76)
Other	0.94	0.24 -3.68 (0.93)	0.88	0.22 -3.52 (0.30)	0.77	0.19 -3.15 (0.72)	0.73	0.17 -3.01 (0.67)

 Table 8.13: Adjusted odds ratios of associations between ethnicity and 'Other' referral at first contact (n=338)

Model 1- unadjusted ; Model 2 – adjusted for age and gender; Model 3- adjusted for age, gender, level of education, employment status;

Model 4- all variables in model 3 plus duration of untreated psychosis and family involvement

8.4.5 Adjusted odds ratios for GP referral by ethnicity at first contact

There were no differences in GP referral by ethnicity regardless of all potential confounders (see Table 8.14).

8.4.6 Adjusted odds ratios for A&E referral by ethnicity at first contact

There were no associations observed between A&E referral and ethnicity both in the unadjusted and adjusted regression models (see Table 8.15).

8.4.7 Adjusted odds ratios for Police/CJA referral by ethnicity at first contact

In investigating association of police/criminal justice agency referral and ethnicity, the results shown in Table 8.16 reveal that there was no association noted in the univariable or fully adjusted models.

	Model 1	95% confidence interval (p- value)	Model 2	95% confidence interval (p- value)	Model 3	95% confidence interval (p- value)	Model 4	95% confidence interval (p- value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	0.60	0.32 -1.11 (0.11)	0.62	0.33 -1.16 (0.13)	0.66	0.35 -1.25 (0.21)	0.63	0.33 -1.19 (0.15)
Black Caribbean	0.58	0.28 -1.19 (0.14)	0.58	0.28 - 1.19 (0.13)	0.54	0.26 -1.14 (0.10)	0.54	0.35 -1.15 (0.11)
White Other	0.64	0.31 -1.33 (0.24)	0.65	0.31 -1.36 (0.25)	0.65	0.31 -1.36 (0.25)	0.63	0.29 -1.34 (0.23)
Asian	1.03	0.44 -2.40 (0.94)	1.03	0.43 -2.42 (0.94)	1.13	0.47 -2.73 (0.77)	1.06	0.43 -2.62 (0.89)
Mixed	0.72	0.26 -1.98 (0.53)	0.75	0.27 -2.06 (0.58)	0.72	0.26 -1.99 (0.53)	0.73	0.26 -2.03 (0.55)
Other	1.70	0.70 -4.09 (0.23)	1.82	0.74 -4.43 (0.18)	1.81	0.72 -4.50 (0.20)	1.74	0.68 -4.40 (0.24)

Table 8.14: Adjusted odds ratios of associations between ethnicity and GP referral at first contact

Model 1- unadjusted ; Model 2 – adjusted for age and gender; Model 3- adjusted for age, gender, level of education, employment status;

Model 4- all variables in model 3 plus duration of untreated psychosis and family involvement

	Model 1	95% confidence interval (p- value)	Model 2	95% confidence interval (p- value)	Model 3	95% confidence interval (p- value)	Model 4	95% confidence interval (p- value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	1.62	0.89 -2.95 (0.11)	1.56	085 -2.85 (0.14)	1.50	0.81 -2.78 (0.19)	1.70	0.89 -3.26 (0.10)
Black Caribbean	1.16	0.58 -2.32 (0.62)	1.15	0.57 -2.33 (0.68)	1.29	0.62 -2.65 (0.48)	1.07	0.50 -2.30 (0.85)
White Other	1.31	0.64 -2.64 (0.45)	1.34	0.66 -2.74 (0.41)	1.41	0.68 -2.92 (0.35)	1.39	0.65 -2.99 (0.39)
Asian	2.07	0.88 -4.84 (0.09)	2.01	0.85 -4.73 (0.11)	1.97	0.82 -4.71 (0.34)	2.13	0.85 -5.33 (0.10)
Mixed	0.77	0.27 -2.19 (0.63)	0.73	0.25 -2.09 (0.56)	0.75	0.25 -2.09 (0.55)	0.75	0.24 -2.31 (0.62)
Other	0.71	0.27 -1.87 (0.49)	0.69	0.26 -1.86 (0.47)	0.77	0.28 -2.09 (0.61)	0.85	0.29 -2.46 (0.77)

Table 8.15: Adjusted odds ratios of associations between ethnicity and A&E referral at first contact

Model 1- unadjusted ; Model 2 – adjusted for age and gender; Model 3- adjusted for age, gender, level of education, employment status;

Model 4- all variables in model 3 plus duration of untreated psychosis and family involvement.

	Model 1	95% confidence interval (p- value)	Model 2	95% confidence interval (p- value)	Model 3	95% confidence interval (p- value)	Model 4	95% confidence interval (p- value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	1.18	0.51 -2.71 (0.69)	1.18	0.51 -2.71 (0.69)	1.19	0.50 -2.79 (0.68)	1.05	0.43 -2.57 (0.90)
Black Caribbean	0.94	0.34 -2.55 (0.90)	094	0.34 -2.55 (0.90)	0.94	0.34 -2.64 (0.92)	1.28	0.43 -3.77 (0.64)
White Other	1.37	0.53 -3.52 (0.50)	1.36	0.53 -3.50 (0.51)	1.28	0.49 -3.34 (0.32)	1.48	0.53 -3.99 (0.45)
Asian	0.48	0.10 -2.31 (0.36)	0.49	0.10 -2.35 (0.37)	0.45	0.94 -9.33 (0.06)	0.38	0.07 -1.99 (0.25)
Mixed	2.63	0.85 -8.10 (0.09)	2.63	0.85 -8.12 (0.09)	2.97	0.10 -2.51 (0.40)	2.45	0.73 -8.14 (0.14)
Other	0.54	0.11 -2.62 (0.45)	0.53	0.11 -2.60 (0.44)	0.51	0.51 -1.77 (0.88)	0.42	0.08 -2.20 (0.31)

Table 8.16: Adjusted odds ratios of associations between police/CJA referral and ethnicity at first contact

Model 1- unadjusted ; Model 2 – adjusted for age and gender; Model 3- adjusted for age, gender, level of education, employment status;

Model 4- all variables in model 3 plus duration of untreated psychosis and family involvement

8.5 Summary

In relation to odds of compulsory admission, there was strong evidence that Black African and 'Mixed' ethnic groups were associated with increased risk of compulsory admission, independent of potential confounders. Stratification by gender revealed that only Black African men had increased risk of compulsory admission. However, the stratified analysis showed strong and independent associations between women of Black African, Asian and 'Mixed' ethnicity and the risk of being detained compulsorily. The evidence of the excess rate of compulsory admissions among the Black African ethnic group suggest that the risk is present in both men and women, which is in contrast to evidence for the other minority ethnic groups.

The odds ratios for hospital admission also indicated that compared with White British patients, Black African, White Other and 'Mixed' ethnic group patients were more likely to be admitted to hospital at first contact for psychosis. These findings were independent of all the potential confounders.

With regard to associations between source of referral and ethnicity, the key findings were that Black Caribbean individuals were more likely to follow alternative routes to care, but this evidence did not hold in the adjusted analyses.

There was no evidence of associations between any other source of referral (i.e. criminal justice agency, A&E or GP) and ethnicity.

In conclusion, the results in this chapter partly support the hypotheses that pathways to care are problematic for the Black African patients in the domains of hospital admission and compulsory admission at first contact for psychosis. However, the results here did not support the same hypothesis for the Black Caribbean ethnic group, with the evidence suggesting that there were no differences in pathways to care compared with the White British ethnic group. Overall, the evidence showed that there were no differences in source of referrals (GP, criminal justice agency or A&E) between Black African, Black Caribbean and White British patients. Detailed discussion of possible explanations for these findings will be discussed in Chapter 12.

199

8.6 Limitations

This phase of the study used cross-sectional sociodemographic and pathways to care data and the design does not allow causality conclusion of the relationships between ethnicity, compulsory admission and source of referrals. In addition to the limitations highlighted in Chapter 7, two important methodological shortcomings are also noteworthy in interpreting the results in this chapter.

Classification of variables

First, epidemiological studies often use crude dichotomies e.g. employed versus unemployed to explore relationships between exposure and outcome. This means that "the precise meaning of observed associations is unclear, and the social experiences that these variables may index remain unknown" (Morgan et al., 2008). This is true in this chapter with dichotomies such as police/CJA referral, which involved collapsing down referrals via the police and criminal justice systems into a single variable. This could potentially result in the inability to discern which of the variables are salient to the findings. However, given the small number of patients that were referred via these routes, separating them may produce a less meaningful result. Therefore, based on previous studies (Morgan et al., 2005, Ghali et al., 2012) referrals via police and CJA were collapsed into one as they are considered to involve some element of coercion (Morgan et al., 2005).

Missing data

Second, the issue of missing values. While this thesis is reasonably well-powered, the lack of complete data on some of the covariates is an important limitation to be acknowledged. As highlighted in Chapter 3, (section 3.5) it is important to adjust for the known confounders in order to assess differential effect of pathways to care and between ethnic groups. This study attempted to achieve this by controlling for a range of confounders such as socioeconomic status, demographic and clinical characteristics. However, the socioeconomic variables suffered significant missing values, particularly 'educational qualification', which meant a loss of statistical power. To this end, a two-stage regression analysis was employed to estimate crude odds ratios, (a) on the whole sample (n=558) and (b) on the people for whom complete data was available (n=367). This helped in assessing the impact missing data may or may not have on the observed associations. There were no notable differences

200

in the odds ratio estimates between analysis (a) and (b), particularly for associations between compulsory admission and ethnicity.

9 Chapter 9: Results of the Impact of Early Intervention Services on Pathways to Care for Minority Ethnic Groups during First Episode Psychosis: comparing data from the AESOP and present (CRIS-FEP) studies

9.1 Synopsis

In Chapter 8, relationships between ethnicity and hospital admission, compulsory admission and source of referrals were presented and discussed using full samples from the study for this thesis. The findings suggest that there were ethnic differences in relation to hospital admissions and compulsory admission but somewhat weaker associations with source of referral. This chapter builds on the work in Chapter 8 and examines the question: Are there any differences in pathways to care for FEP cases compared to 15 years ago and did the introduction of early intervention service (EIS) explain the difference?? The chapter addresses the following aim and hypothesis:

1. To estimate and compare pathways to care (compulsory vs. non-compulsory) and source of referral (general practitioner (GP), accident and emergency (A&E) and criminal justice agency (CJA)) at first contact for psychosis by ethnic groups in the AESOP sample and those using early intervention services or not in the current study sample.

Hypothesis

Compared with 15 years ago (i.e. AESOP vs. CRIS-FEP samples), among patients aged 18–35 years old ethnic differences in GP referral and crisis source of referral (criminal justice agency, accident and emergency) and compulsory admission at first contact for psychosis will be smaller for Black African and Black Caribbean patients.

To begin with, a sample from the Aetiology and Ethnicity in Schizophrenia and Other Psychoses (ASEOP) study was used as baseline for comparison with the current study. As mentioned in Chapter 7, the AESOP study was completed over a decade ago, before the introduction of early intervention services in SLaM, and so it provides a good template against which to test differences and changes in compulsory admission, pathways to care and help-seeking behaviours among people with FEP since the introduction of EIS.

At the time of the study for this thesis (CRIS-FEP), the eligibility criteria for early intervention services (EIS) in SLaM were that the patient be 18–35 years old. Therefore, data from the CRIS-FEP and the AESOP studies for this chapter were restricted to patients aged 18–35 years. In addition, ethnic groups White British, Black African, Black Caribbean and White Other have been reported to experience more problematic pathways to care. To that end, for the purpose of analysis, data from both CRIS-FEP and AESOP studies were restricted to the aforementioned ethnic groups. This enabled direct comparisons between the two study samples.

9.2 Analysis

A secondary analysis on the AESOP (London site) data was performed to compare results with the CRIS-FEP study stratified by EIS use status. First, unadjusted logistic regression analyses are presented assessing the crude associations between sociodemographic and pathways to care variables and each source of referral outcome variables (i.e. GP, police or criminal justice agency and emergency referrals) and compulsory admission. Second, multivariable logistic regression models are presented assessing associations between the main exposure variable (ethnicity) and each outcome variable, using White British as the reference ethnic group. In the multivariable regression model, I began by adjusting for *a priori* potential confounders (age, gender, education level and employment status) to control for demographic and socioeconomic status. Subsequently, other variables (covariates) that were crudely associated with the outcome variables from the unadjusted analyses were added to the model; covariates were included if the p-value was ≤0.05. Complete data were used in the multivariable analyses, i.e. only patients who had complete data for all the covariates were included (i.e. 201 for CRIS-FEP sample and 183 for the AESOP sample).

9.3 Results

9.3.1 Sample characteristics

A total of 278 first episode psychosis patients at the London site were included in the AESOP study, of whom 193 were aged 18–35 years and so were included for the current analysis. From the CRIS-FEP study, a total of 558 first episode patients were identified, of whom 265 were 18–35 years and thus would have been eligible for EIS services. Of these, 184 received EIS and 81 patients were eligible for EIS care but did not receive it (non-EIS).

Table 9.1 shows the demographic and pathways to care variables for AESOP and CRIS-FEP, with the latter reflecting whether or not the patient received care from a specialist EIS team. Compared to AESOP, the Black Caribbean ethnic group comprises a smaller, and the Black African group a larger fraction of the total sample, and there was a somewhat greater proportion of people with higher or university level education. The proportion of patients accessing services via police and criminal justice agencies was also substantially lower than observed in AESOP. Further, the prevalence of compulsory admission was lower in the CRIS-FEP sample compared with the AESOP sample. More differences emerge when the CRIS-FEP sample was broken down by whether or not they received EIS services. It is apparent that the patients who were seen by EIS were somewhat younger, more likely to be male, educated to university level, live with family and to have family involvement in their pathway to care than either patients in AESOP or those who were eligible for EIS but did not receive it (non-EIS). Interestingly, a greater proportion of those compulsorily detained in the CRIS-FEP were in the EIS group.

	AESOP		CRIS-FEP	
	AESOP	Whole sample	Non-EIS	EIS
	N=193 (%)	N =265 (%)	N=81 (%)	N= 184 (%)
DEMOGRAPIC VARIABLES				
Mean age (sd)years	26.7 (4.9)	26.2 (5.1)	27.5 (5.0)	25.5 (5.1)
Gender				
Men	110 (57.0)	154 (58.1)	38 (46.9)	116 (63.0)
Women	83 (43.0)	111 (41.9)	43 (53.1)	68 (37.0)
Ethnicity				
White British	60 (31.1)	78 (29.4)	23 (28.4)	55 (29.9)
Black African	50 (25.9)	95 (35.9)	25 (30.9)	70 (38.1)
Black Caribbean	65 (33.7)	52 (19.6)	17 (21.0)	35 (19.0)
White Other	18 (9.3)	40 (15.1)	16 (19.7)	24 (13.0)
Education ¹				
School, no GCSE	51 (27.6)	50 (24.0)	23 (35.4)	27 (18.9)
School, GCSE	46 (24.8)	39 (18.7)	11 (16.9)	28 (19.6)
Further	57 (30.8)	60 (28.9)	15 (23.1)	45 (31.4)
Higher / University	31 (16.8)	59 (28.4)	16 (24.6)	43 (30.1)
Employment ²				
Unemployed	112 (59.0)	157 (63.0)	56 (73.7)	101 (58.4)
Student/Other	24 (12.6)	46 (18.5)	7 (9.2)	39 (22.5)
Employed	54 (28.4)	46 (18.5)	13 (17.1)	33 (19.1)
Relationship status ³				
Single	124 (70.1)	178 (69.8)	55 (71.4)	123 (69.1)
Married/in steady	42 (23.7)	58 (22.7)	17 (22.1)	41 (23.0)
relationship				
Divorced/ widowed	11 (6.2)	19 (7.5)	5 (6.5)	14 (7.9)
Living arrangements ⁴				

 Table 9.1: Demographic and pathways to care characteristics of AESOP and CRIS – FEP samples

Alone	86 (45.0)	59 (22.7)	25 (31.6)	34 (18.8)
Family/ relatives	90 (47.1)	168 (64.6)	42 (53.2)	126 (69.6)
Other	15 (7.9)	33 (12.7)	12 (15.2)	21 (11.6)
PATHWAYS TO CARE				
VARIABLES				
DUP⁵				
Short (≤ 1year)	157 (81.4)	190 (71.7)	61 (75.3)	129 (70.1)
Long (> 1year)	36 (18.6)	75 (28.3)	20 (24.7)	55 (29.9)
Family involvement in				
help-seeking ⁶				
Yes	65 (35.5)	154 (58.1)	25 (30.9)	86 (46.7)
No	118 (64.5)	111 (41.9)	56 (69.1)	93 (53.3)
GP referral ⁷				
No	129 (67.2)	184 (69.4)	59 (72.8)	125 (67.9)
Yes	63 (32.8)	81 (30.6)	22 (27.2)	59 (32.1)
Emergency referral ⁸				
No	127 (66.1)	154 (58.1)	46 (56.8)	108 (58.7)
Yes	65 (33.9)	111 (41.9)	35 (43.2)	76 (41.3)
Police/CJA referral ⁹				
No	141 (73.4)	227 (85.7.1)	71 (87.7)	156 (84.8)
Yes	51 (26.6)	38 (14.3)	10 (12.3)	28 (15.2)
Compulsory admission				
No	110 (57.0)	197 (74.3)	63 (77.8)	134 (72.8)
Yes	83 (43.0)	68 (25.7)	18 (22.2)	50 (27.2)

Missing data

1 AESOP, 8; CRIS – FEP, 57 (EIS= 41; non-EIS= 16)

2 AESOP, 3; CRIS – FEP, 16 (EIS= 11; non-EIS= 5)

3 AESOP, 16; CRIS – FEP, 10 (EIS= 6; non-EIS= 4)

4 AESOP, 2; CRIS – FEP, 5 (EIS= 3; non-EIS= 2)

6 AESOP, 10; CRIS –FEP, none

7-9 AESOP, 1; CRIS – FEP, none

In the next stage of the analyses, first the prevalence of each outcome variable (compulsory admission, GP, emergency and criminal justice referral) by sociodemographic and pathways to care characteristics was estimated. Second, univariable logistic regression was used to assess crude associations between sociodemographic and pathways to care characteristics and each outcome variable on the AESOP and CRIS-FEP samples; the latter was stratified by EIS status.

9.3.2 Unadjusted associations between sociodemographic/pathways to care variables and GP referral by use of EIS

Table 9.2 shows relationships between sociodemographic and pathways to care characteristics and GP referral. It can be seen from this table that there was notable evidence in the AESOP sample that Black African (OR = 0.29; 95% CI 0.12 – 0.70) and Black Caribbean (OR = 0.45; 95% CI 0.21 – 0.95) patients were less likely to be referred to mental health services by their GP. There was weak evidence that those employed and students in the AESOP sample were more likely to be referred by their GP. In the non-EIS group of the CRIS-FEP sample, there was weak evidence that patients of Black African (OR = 0.32; 95% CI 0.09 – 1.16) ethnic group were less likely to be referred by GP but the evidence was stronger for the Black Caribbean group (OR = 0.17; 95% CI 0.03 – 0.93). There was no evidence of association between sociodemographic variables and GP referral in the EIS group, although those with longer duration of untreated psychosis were twice more likely to be referred by GP.

	AESOP N=193	3		CRIS-FEP STUDY: Non- EIS group CRIS-FEP STUDY: E				FEP STUDY: EIS	group
					N= 81			N= 184	
DEMOGRAPIC	GP referral	GP	OR	GP referral	GP referral	OR	GP referral	GP referral	OR
VARIABLES	(No) n=129	referral	(95% CI)	(No) n=59	(Yes) n=22	(95% CI)	(No) n=125	(Yes) n=59	(95% CI)
	(%)	(Yes) n=53		(%)	(%)		(%)	(%)	
		(%)							
Age (sd) years	27.1 (4.8)	26.0 (4.9)	0.95	28.0 (5.1)	25.9 (4.5)	0.91	25.5 (5.0)	25.8 (5.0)	1.01
			(0.89-1.01)			(0.82-1.00)~			(0.95 – 1.07)
Gender									
Men	74 (67.9)	35 (32.1)	1	28 (73.7)	10 (26.3)	1	80 (69.0)	36 (31.0)	1
Women	55 (66.3)	28 (33.7)	1.07	31 (72.1)	12 (27.9)	1.08	45 (66.2)	23 (33.8)	1.13
			(0.58 – 1.97)			(0.40-2.89)			(0.60-2.14)
Ethnicity									
White British	32 (54.2)	27 (45.8)	1	13 (56.5)	10 (43.5)	1	35 (63.6)	20 (36.4)	1
Black African	40 (80.0)	10 (20.0)	0.29	20 (80.0)	5 (20.0)	0.32	53 (75.7)	17 (24.3)	0.56
			(0.12-0.70) 🕇			(0.09-1.16)~			(0.25-1.21)
Black Caribbean	47 (72.3)	18 (27.7)	0.45	15 (88.2)	2 (11.8)	0.17	24 (68.6)	11 (31.4)	0.80
			(0.21-0.95)*			(0.03-0.93)*			(0.32-1.97)
White Other	10 (55.6)	8 (44.4)	0.94	11 (68.7)	5 (31.3)	0.59	13 (54.2)	11 (45.8)	1.48
			(0.32-2.74)			(0.15-2.25)			(0.55-3.91)
Education ¹									
School, no GCSE	33 (66.0)	17 (34.0)	1	16 (69.6)	7 (30.4)	1	15 (55.6)	12 (44.4)	1
School, GCSE	32 (69.6)	14 (30.4)	0.84	8 (72.7)	3 (27.3)	0.85	19 (67.9)	9 (32.1)	0.59
			(0.35-2.00)			(0.17-4.23)			(0.19-1.77)
Further	40 (70.2)	17 (29.8)	0.85	10 (66.7)	5 (33.3)	1.14	32 (71.1)	13 (28.9)	0.50
			(0.36-1.86)			(0.28-4.60)			(0.18-1.37)
Higher /	19 (61.3)	12 (38.7)	1.22	12 (75.0)	4 (25.0)	0.76	32 (74.4)	11 (25.6)	0.42
University			(0.48-3.10)			(0.18-3.21)			(0.15-1.19)

Table 9.2: Unadjusted odds ratios of associations between sociodemographic/pathways to care variables and GP referral by use of EIS

Employment ²									
Unemployed	81 (73.0)	30 (27.0)	1	39 (69.6)	17 (30.4)	1	66 (65.3)	35 (34.7)	1
Student/Other	13 (54.2)	11 (45.8)	1.85	6 (85.7)	1 (14.3)	0.38	30 (76.9)	9 (23.1)	0.56
			(0.93-3.68)~			(0.04-3.42)			(0.24-1.32)
Employed	32 (59.3)	22 (40.7)	2.28	10 (76.9)	3 (23.1)	0.68	21 (63.6)	12 (36.4)	1.07
			(0.92-5.65)~			(0.16-2.81)			(0.47-2.44)
Relationship									
status ³									
Single	83 (67.5)	40 (32.5)	1	38 (69.1)	17 (30.9)	1	87 (70.7)	36 (29.3)	1
In steady	28 (66.7)	14 (33.3)	1.03	14 (82.3)	3 (17.7)	0.47	25 (61.0)	16 (39.0)	1.54
relationship			(0.49-2.18)			(0.12-1.88)			(0.73-3.23)
Divorced/	9 (81.8)	2 (18.2)	0.46	4 (80.0)	1 (20.0)	0.55	8 (57.1)	6 (42.9)	1.81
widowed			(0.09-2.23)			(0.05-5.38)			(0.58-5.59)
Living									
arrangements ⁴									
Alone	63 (73.3)	23 (26.7)	1	18 (72.0)	7 (28.0)	1	23 (67.7)	11 (32.3)	1
Family/ relatives	50 (56.2)	39 (43.8)	2.13	29 (69.0)	13 (31.0)	1.15	82 (65.1)	44 (34.9)	1.12
			(1.13-4.03)			(0.38-3.43)			(0.50-2.51)
Other	14 (93.3)	1 (6.7)	0.19	10 (83.3)	2 (16.7)	0.51	18 (85.7)	3 (14.3)	0.34
			(0.02-1.57)			(0.08-2.96)			(0.08-1.43)
PATHWAYS TO									
CARE									
VARIABLES									
DUP									
Short (≤ 1year)	105 (67.3)	51 (32.7)	1	42 (68.8)	19 (31.2)	1	96 (74.4)	33 (25.6)	1
Long (> 1year)	24 (66.7)	12 (33.3)	1.02	17 (85.0)	3 (15.0)	0.39	29 (52.7)	26 (47.3)	2.60
			(0.47-2.22)			(0.10-1.49)			(1.34-5.05) †
Family									
involvement in									
help-seeking ⁵									

No	87 (73.7)	31 (26.3)	1	38 (67.9)	18 (32.1)	1	64 (65.3)	34 (34.7)	1
Yes	42 (65.6)	22 (34.4)	1.47	21 (84.0)	4 (16.0)	0.40	61 (70.9)	25 (29.1)	0.77
			(0.76 -2.84)			(0.12 -1.34)			(0.41 -1.44)

~P=0.1; *p≤0.05; †p≤0.01

9.3.3 Unadjusted associations between sociodemographic/pathways to care variables and accident and emergency referral by use of EIS

Table 9.3 shows associations between sociodemographic, pathways to care variables and accident and emergency referral. There was strong evidence that those who had family involvement in their pathways to care were more likely to seek help via the A&E department; this was notable in both the AESOP (25.4%, OR = 3.64; 95% CI 1.91 – 6.91, p<0.01) and CRIS-FEP (non-EIS 32.1%, OR = 4.48; 95% CI 1.63 – 12.32, p<0.04 and EIS 23.5%, OR = 5.23; 95% CI 2.72 – 9.91, p<0.01) samples. This suggests that family play a crucial role in help-seeking, particularly during an acute onset of psychosis when help may be required urgently. There was no evidence of associations between other independent variables and A&E referral in the AESOP sample. In the non-EIS group, Black Caribbean patients were four times more likely to come into contact with mental health services via emergency route (58.8%, OR = 4.04; 95% CI 1.05 – 15.47, p = 0.04). In the EIS group, compared with those educated to university level, patients without school qualifications were less likely to access services via the accident and emergency route (OR = 0.29; 95% CI 0.10 – 0.88, p = 0.03).

	AESOP N=193	3		CRIS-FE	P STUDY: Non-	EIS group	CRIS	-FEP STUDY: EI	S group
					N= 81			N= 184	
DEMOGRAPIC	A&E	A&E	OR	A&E	A&E	OR	A&E	A&E	OR
VARIABLES	referral	referral	(95% CI)	referral	referral	(95% CI)	referral	referral	(95% CI)
	(No)	(Yes) n=65		(No) n=46	(Yes) n= 35		(No) n=108	(Yes) n=76	
	n=127 (%)	(%)		(%)	(%)		(%)	(%)	
Age (sd)	27.0 (4.9)	26.1 (4.8)	0.96	28.1 (4.8)	26.8 (5.6)	0.95	25.6 (4.9)	25.4 (5.2)	0.99
			(0.90-1.02)			(0.87-1.04)			(0.93-1.05)
Gender									
Men	73 (67.0)	36 (33.0)	1	25 (65.8)	13 (34.2)	1	68 (58.6)	48 (41.4)	1
Women	54 (65.1)	29 (34.9)	1.08	21 (48.8)	22 (51.2)	2.01	40 (58.8)	28 (41.2)	0.99
			(0.59-1.98)			(0.82-4.94)			(0.53-1.82)
Ethnicity									
White British	37 (62.7)	22 (37.3)	1	17 (73.9)	6 (26.1)	1	35 (63.6)	20 (36.4)	1
Black African	31 (62.0)	19 (38.0)	1.03	14 (56.0)	11 (44.0)	2.22	37 (52.9)	33 (47.1)	1.56
			(0.47-2.24)			(0.65-7.54)			(0.75-3.21)
Black Caribbean	47 (72.3)	18 (27.7)	0.64	7 (41.2)	10 (58.8)	4.04	22 (62.9)	13 (37.1)	1.03
			(0.30-1.37)			(1.05-15.47)*			(0.42-2.48)
White Other	12 (66.7)	6 (33.3)	0.84	8 (50.0)	8 (50.0)	2.83	14 (58.3)	10 (41.7)	1.25
			(0.27-2.55)			(0.76-10.94)			(0.46-3.30)
Education ¹									
School, no GCSE	36 (72.0)	14 (28.0)	1	12 (52.2)	11 (47.8)	1	21 (77.8)	6 (22.2)	1
School, GCSE	28 (60.9)	18 (39.1)	1.65	6 (54.6)	4 (45.4)	0.90	17 (60.7)	11 (39.3)	2.26
			(0.70-3.88)			(0.21-3.84)			(0.69-7.38)
Further	36 (63.2)	21 (36.8)	1.50	10 (66.7)	5 (33.3)	0.54	23 (51.1)	22 (48.9)	3.34
			(0.66-3.40)			(0.14-2.10)			(1.13-9.85)*

Table 9.3: Unadjusted odds ratios of associations between sociodemographic/pathways to care variables and A&E referral by use of EIS

Higher /	24 (77.4)	7 (22.6)	0.75	9 (56.2)	7 (43.8)	0.84	22 (51.2)	21 (48.8)	3.34
University			(0.26-2.13)			(0.23-3.05)			(1.12-9.90)*
Employment ²									
Unemployed	75 (67.6)	36 (32.4)	1	34 (60.7)	22 (39.3)	1	66 (65.4)	35 (34.6)	1
Student/Other	17 (70.8)	7 (29.2)	0.85	2 (28.6)	5 (71.4)	3.86	19 (48.7)	20 (51.3)	1.98
			(0.32-2.25)			(0.68-21.69)			(0.94-4.20)~
Employed	33 (61.1)	21 (38.9)	1.32	6 (46.1)	7 (53.9)	1.80	16 (48.5)	17 (51.5)	2.00
			(0.67-2.60)			(0.53-6.07)			(0.90-4.44)~
Relationship									
status ³									
Single	85 (69.1)	38 (30.9)	1	30 (54.6)	25 (45.4)	1	70 (56.9)	53 (43.1)	1
In steady	26 (61.9)	16 (38.1)	1.37	9 (52.9)	8 (47.1)	1.06	23 (56.1)	18 (43.9)	1.03
relationship			(0.66-2.85)			(0.35-3.17)			(0.50-2.10)
Divorced/	6 (54.6)	5 (45.4)	1.86	4 (80.0)	1 (20.0)	0.30	10 (71.4)	4 (28.6)	0.52
widowed			(0.53-6.48)			(0.03-2.85)			(0.15-1.77)
Living									
arrangements ⁴									
Alone	60 (69.8)	26 (30.2)	1	17 (68.0)	8 (32.0)	1	24 (70.6)	10 (29.4)	1
Family/ relatives	58 (60.2)	31 (34.8)	1.23	21 (50.0)	21 (50.0)	2.15	70 (55.6)	56 (44.4)	1.92
			(0.65-2.32)			(0.75-5.98)			(0.84-4.34)
Other	8 (53.3)	7 (46.7)	2.01	7 (58.3)	5 (41.7)	1.51	13 (61.9)	8 (38.1)	1.47
			(0.66-6.15)			(0.36-6.29)			(0.46-4.65)
PATHWAYS TO									
CARE									
VARIABLES									
DUP ⁵									
Short (≤ 1year)	101 (64.7)	55 (35.3)	1	36 (59.0)	25 (41.0)	1	71 (55.0)	58 (45.0)	1
Long (> 1year)	26 (72.2)	10 (27.8)	0.70	10 (50.0)	10 (50.0)	1.44	37 (67.3)	18 (32.7)	0.59
			(0.31-1.57)			(0.55-3.97)			(0.30-1.15)
Family									

involvement in									
help-seeking ⁶									
No	88 (74.6)	30 (25.4)	1	38 (67.9)	18 (32.1)	1	75 (76.5)	23 (23.5)	1
Yes	29 (45.3)	35 (54.7)	3.64	8 (32.0)	17 (68.0)	4.48	33 (38.4)	53 (61.6)	5.23
			(1.91 -6.91)			(1.63 -12.32)			(2.76 -9.91) 🕇
			+			+			

~P=0.1; *p≤0.05; †p≤0.01

A&E, accident and emergency
9.3.4 Unadjusted associations between sociodemographic/pathways to care variables and CJA referral by use of EIS

Table 9.4 displays the relationship between sociodemographic and pathways to care variables and criminal justice agency referral. In the AESOP sample, three independent variables, namely mean age (28 years old, OR = 1.11; 95% Cl 1.03 - 1.19, p<0.01), ethnicity (Black African: 34%, OR = 4.55; 95% Cl 1.62 - 12.71 and Black Caribbean: 38.5%, OR = 5.52; 95% Cl 2.06 - 14.72, p<0.01) were strongly associated with CJA referral. As would be expected, those who had family involvement in their help-seeking were less likely to be referred by the criminal justice agency (35.5%, OR = 0.20; 95% Cl 0.08 - 0.49, p<0.01). This was echoed in the EIS group showing that having family involvement was associated with reduced likelihood of criminal justice agency referral (58.1%, OR = 0.15; 95% Cl 0.04 - 0.45, p<0.01). There was weak evidence suggestive of some associations between relationships status, DUP, family involvement and CJA referral in the EIS sample. By contrast, only age was weakly associated with CJA referral in the non-EIS group.

	AESOP N=193	3		CRIS-FE	P STUDY: Noi	n- EIS group	CRIS	-FEP STUDY: EI	S group
					N= 81			N= 184	
DEMOGRAPIC	CJA referral	CJA	OR	CJA	CJA	OR	CJA referral	CJA referral	OR
VARIABLES	(No) n=131	referral	(95% CI)	referral	referral	(95% CI)	(No) n=156	(Yes) n=28	(95% CI)
	(%)	(Yes) n=51		(No) n=71	(Yes) n=		(%)	(%)	
		(%)		(%)	10 (%)				
Age (sd) years	26.08 (4.9)	28 (4.3)	1.11	27.2 (5.0)	30.2 (4.4)	1.14	25.5 (5.1)	25.6 (4.8)	1.00
			(1.03-1.19) †			(0.9-1.35)~			(0.92-1.08)
Gender									
Men	77 (70.6)	32 (29.4)	1	32 (84.2)	6 (15.8)	1	98 (84.5)	18 (15.5)	1
Women	64 (77.1)	19 (22.9)	0.71	39 (90.7)	4 (9.3)	0.54	58 (85.3)	10 (14.7)	0.93
			(0.37-1.37)			(0.14-2.10)			(0.40-2.17)
Ethnicity									
White British	53 (89.8)	6 (10.2)	1	19 (82.6)	4 (17.4)	1	47 (85.5)	8 (14.5)	1
Black African	33 (66.0)	17 (34.0)	4.55	22 (88.0)	3 (12.0)	0.64	57 (81.4)	13 (18.6)	1.33
			(1.62-12.71) †			(0.12-3.26)			(0.51-3.50)
Black Caribbean	40 (61.5)	25 (38.5)	5.52	15 (88.2)	2 (11.8)	0.63	30 (85.7)	5 (14.3)	0.97
			(2.06-14.72) 🕇			(0.10-3.93)			(0.29-3.27)
White Other	15 (83.3)	3 (16.7)	1.76	15 (93.8)	1 (6.2)	0.31	22 (91.7)	2 (8.3)	0.53
			(0.39-7.91)			(0.03-3.13)			(0.10-2.72)
Education ¹									
School, no GCSE	36 (72.0)	14 (28.0)	1	20 (87.0)	3 (13.0)	1	21 (77.8)	6 (22.2)	1
School, GCSE	34 (73.9)	12 (26.1)	0.90	10 (90.9)	1 (9.1)	0.66	24 (85.7)	4 (14.3)	0.58
			(0.36-2.23)			(0.06-7.25)			(0.14-2.35)
Further	42 (73.9)	15 (26.3)	0.91	14 (93.3)	1 (6.7)	047	40 (88.9)	5 (11.1)	0.43
			(0.39-2.15)			(0.04-5.06)			(0.11-1.60)

Table 9.4: Unadjusted odds ratios of associations between sociodemographic/pathways to care variables and CJA referral by use of EIS

Higher /	21 (67.7)	10 (32.3)	1.22	14 (87.5)	2 (12.5)	0.95	35 (81.4)	8 (18.6)	0.79
University			(0.46-3.24)			(0.14-6.46)			(0.24-2.62)
Employment ²									
Unemployed	76 (68.5)	35 (31.5)	1	50 (89.3)	6 (10.7)	1	85 (84.2)	16 (15.8)	1
Student/Other	20 (83.3)	4 (16.7)	0.55	7 (100)	0	No cases	32 (82.0)	7 (18.0)	1.16
			(0.25-1.20)						(0.43-3.06)
Employed	43 (79.6)	11 (20.4)	0.43	12 (92.3)	1 (7.7)	0.69	30 (90.9)	3 (9.1)	0.53
			(0.13-1.36)			(0.07-6.32)			(0.14-1.95)
Relationship									
status ³									
Single	87 (70.7)	36 (29.3)	1	48 (87.3)	7 (12.7)	1	102 (82.9)	21 (17.1)	1
In steady	33 (78.6)	9 (21.4)	0.65	15 (88.2)	2 (11.8)	0.91	39 (95.1)	2 (4.9)	0.24
relationship			(0.28-1.51)			(0.17-4.88)			(0.05-1.11)~
Divorced/	7 (63.6)	4 (36.4)	1.38	5 (100)	0	No cases	11 (78.6)	3 (21.4)	1.32
widowed			(0.38-5.00)						(0.33-5.16)
Living									
arrangements ⁴									
Alone	59 (68.6)	27 (31.4)	1	20 (80.0)	5 (20.0)	1	27 (79.4)	7 (20.6)	1
Family/ relatives	72 (80.9)	17 (19.1)	0.51	39 (92.9)	3 (7.1)	0.30	113 (89.7)	13 (10.3)	0.44
			(0.25-1.03)~			(0.06-1.42)			(0.16-121)
Other	9 (60.0)	6 (40.0)	1.43	10 (83.3)	2 (16.7)	0.80	13 (61.9)	8 (38.1)	2.37
			(0.47-4.50)			(0.13-4.87)			(0.70-7.96)
PATHWAYS TO									
CARE									
VARIABLES									
DUP ⁵									
Short (≤ 1year)	115 (73.7)	41 (26.3)	1	53 (86.9)	8 (13.1)	1	105 (81.4)	24 (18.6)	1
Long (> 1year)	26 (72.2)	10 (27.8)	1.07	18 (90.0)	2 (10.0)	0.73	51 (92.7)	4 (7.3)	0.34
			(0.74-2.42)			(0.14-3.79)			(0.11-1.04)~
Family									

involvement in									
help-seeking ⁶									
No	74 (62.7)	44 (37.3)	1	47 (83.9)	9 (16.1)	1	74 (75.5)	24 (24.5)	1
Yes	57 (89.1)	7 (10.9)	0.20	24 (96.0)	1 (4.0)	0.21	82 (95.3)	4 (4.7)	0.15
			(0.08 -0.49) †			(0.02 -1.81)			(0.49 -0.45) †

9.3.5 Unadjusted associations between sociodemographic/pathways to care variables and compulsory admission by use of EIS

Finally, the crude associations between sociodemographic and pathways to care variables and compulsory admission at first contact for psychosis are presented in Table 9.5. The AESOP sample shows that there were strong associations between compulsory admission and Black African and Black Caribbean ethnic groups. The evidence suggests that both groups were five and four times (respectively) more likely to be compulsorily detained. Being older was also associated with increased odds of compulsory admission in both the AESOP and non-EIS samples. By contrast, among those who received EIS, Black African patients were twice more likely to be detained, whereas longer DUP was associated with less likelihood of compulsory admission. This suggests perhaps an acute onset of psychosis increases the odds of compulsory detention.

		AESOP		CRIS-FE	P STUDY: Noi	n- EIS group	CRIS	-FEP STUDY: EI	S group
		N=193			N= 81			N= 184	
DEMOGRAPIC	Non-	Compulsor	OR	Non-	Compulsor	OR	Non-	Compulsory	OR
VARIABLES	Compulsory	У	(95% CI)	Compulsor	У	(95% CI)	Compulsory	admission	(95% CI)
	admission	admission		у	admission		admission	n= (%)	
	n= (%)	n= (%)		admission	n= (%)		n= (%)		
				n= (%)					
Age (sd) years	26.1 (4.9)	27.5 (4.8)	1.06	26.9 (5.0)	29.8 (4.5)	1.13 (1.00 -	25.5 (2.5)	25.8 (5.5)	1.01
			(1.00 -1.12)*			1.28)*			(0.94 -1.07)
Gender									
Men	67 (60.9)	43 (39.1)	1	28 (73.7)	10 (26.3)	1	81(69.8)	35 (30.2)	1
Women	43 (51.8)	40 (48.2)	1.44	35 (81.4)	8 (18.6)	0.64	53 (77.9)	15 (22.1)	0.65
			(0.81 -2.57)			(0.22 -1.83)			(0.32 -1.31)
Ethnicity									
White British	47 (78.3)	13 (21.7)	1	19 (82.6)	4 (17.4)	1	43 (78.2)	12 (21.8)	1
Black African	21 (42.0)	29 (58.0)	4.99	17 (68.0)	8 (32.0)	2.23	43 (61.4)	27 (38.6)	2.25
			(2.17-11.47) 🕇			(0.56 -8.77)			(1.01 -5.01)*
Black Caribbean	30 (46.1)	35 (53.9)	4.21	14 (82.3)	3 (17.7)	1.01	25 (71.4)	10 (28.6)	1.43
			(1.92 -9.24) †			(0.19 -5.29)			(0.54 -3.79)
White Other	12 (66.7)	6 (33.3)	1.84	13 (81.2)	3 (18.8)	1.09	23 (95.8)	1 (4.2)	0.15
			(0.56 -5.74)			(0.20 -5.73)			(0.01 -1.27)
Education ¹									
School, no GCSE	27 (52.9)	24 (47.1)	1	17 (73.9)	6 (26.1)	1	20 (74.1)	7 (25.9)	1
School, GCSE	25 (54.3)	21 (45.7)	0.94	9 (81.8)	2 (18.2)	0.62	18 (64.3)	10 (35.7)	1.58
			(0.42 -2.10)			(0.10 -378)			(0.49-5.04)
Further	34 (59.7)	23 (40.3)	0.76	12 (80.0)	3 (20.0)	0.70	36 (80.0)	9 (20.0)	0.71

Table 9.5: Unadjusted odds ratios of associations between sociodemographic/pathways to care variables and compulsory admission by use of EIS

			(0.35 -1.63)			(0.14 -3.40)			(0.23 -2.20)
Higher /	17 (54.8)	14 (45.2)	0.92	13 (81.2)	3 (18.8)	0.65	29 (67.4)	14 (32.6)	1.37
University			(0.37 -2.26)			(0.13 -3.12)			(0.47 -4.02)
Employment ²									
Unemployed	55 (49.1)	57 (50.9)	1	44 (78.6)	12 (21.4)	1	74 (73.3)	27 (26.7)	1
Student/Other	16 (66.7)	8 (33.3)	0.48	6 (85.7)	1 (14.3)	0.61	27 (69.2)	12 (30.8)	1.21
			(0.19 -1.21)			(0.06 -5.57)			(0.54 -2.73)
Employed	38 (70.4)	16 (29.6)	0.40	10 (76.9)	3 (23.1)	1.10	25 (75.8)	8 (24.2)	0.87
			(0.20-0.81) †			(0.26 -4.64)			(0.35 -2.17)
Relationship									
status ³									
Single	74 (59.7)	50 (40.3)	1	41 (74.6)	14 (25.4)	1	86 (69.9)	37 (30.1)	1
In steady	21 (50.0)	21 (50.0)	1.48	14 (82.3)	3 (17.7)	0.62	33 (80.5)	8 (19.5)	0.56
relationship			(0.73 -2.99)			(0.15 -2.51)			(0.23 -1.33)
Divorced/	5 (45.4)	6 (54.6)	1.77	5 (100)	0	No cases	10 (71.4)	4 (28.6)	0.92
widowed			(0.51 -6.13)						(0.27 -3.15)
Living									
arrangements ⁴									
Alone	59 (68.6)	27 (31.4)	1	19 (76.0)	6 (24.0)	1	25 (73.5)	9 (26.5)	1
Family/ relatives	72 (80.9)	17 (19.1)	0.43	35 (83.3)	7 (16.7)	0.63	93 (73.8)	33 (26.2)	0.98
			(0.23 -0.79) 🕇			(0.18 -2.15)			(0.41 -2.23)
Other	9 (60.0)	6 (40.0)	0.79	8 (66.7)	4 (33.3)	1.58	14 (66.7)	7 (33.3)	1.38
			(0.26 -2.39)			(0.34 -7.17)			(0.42 -4.54)
PATHWAYS TO									
CARE									
VARIABLES									
DUP⁵									
Short (≤ 1year)	87 (55.4)	70 (44.6)	1	45 (73.8)	16 (26.2)	1	85 (65.9)	44 (34.1)	1
Long (> 1year)	23 (63.9)	13 (36.1)	0.70	18 (90.0)	2 (10.0)	0.31	49 (89.1)	6 (10.9)	0.23
			(0.33 -1.48)			(0.06 -1.49)			(0.09 -0.59) 🕇

Family									
involvement in									
help-seeking ⁶									
No	67 (56.8)	51 (43.2)	1	42 (75.0)	14 (25.0)	1	68 (69.4)	30 (30.6)	1
Yes	39 (60.0)	26 (40.0)	0.87	21 (84.0)	4 (16.0)	0.57	66 (76.7)	20 (23.3)	0.68
			(0.47 -1.62)			(0.16 -1.95)			(0.35 -1.32)

9.4 Multivariable logistic regression models

In the next stage of the analyses, multivariable logistic regressions are presented assessing the relationships between ethnicity and each outcome variable (compulsory admission, GP referral, A&E referral and CJA referral) in both the AESOP and CRIS-FEP samples. For these set of analyses, patients who had complete data for all the covariates were included (i.e. 201 (EIS = 137, non-EIS = 64) for the CRIS-FEP sample and 183 for the AESOP sample). From this, four logistic regression models were fitted as follows:

Model 1: unadjusted (reported in the text)

Model 2: age and gender

Model 3: age, gender, employment status and education level

Model 4: age, gender, employment status, education level, family involvement and duration of untreated psychosis

9.4.1 Associations between ethnicity and GP referral by use of EIS

Unadjusted odds ratios for associations between ethnicity and GP referral in the AESOP sample showed that Black Caribbean (OR = 0.44; 95% CI 0.21 – 0.95, p = 0.03) and Black African (OR = 0.28; 95% CI 0.12 – 0.68, p<0.01) were less likely to be referred by GP. There was no difference between White Other (OR = 1.02; 95% CI 0.34 – 3.01, p = 0.97) and White British. Unadjusted OR in the EIS group suggest that there were no differences between White British patients and Black African (OR = 0.59; 95% CI 0.24 – 1.45, p = 0.25), Black Caribbean (OR = 0.91; 95% CI 0.32 – 2.56, p = 0.87) and White Other (OR = 1.01; 95% CI 0.32 – 2.12, p = 0.95) patients for GP referral. Meanwhile, the univariable analysis in the non-EIS group showed a weak association of reduced likelihood of GP referral among Black Caribbean patients (OR = 0.12; 95% CI 0.01 – 1.14, p = 0.06). But no difference was observed in Black African (OR = 0.32; 95% CI 0.07 – 1.33, p = 0.12) or White Other (OR = 0.67; 95% CI 0.16 – 2.76, p = 0.58) compared with White British patients.

In the adjusted regression analyses, Table 9.6 results from the AESOP analysis remained consistent, showing strong evidence that Black African and Black Caribbean patients were less likely to be referred by GP, and this was independent of all potential confounders. For the non-EIS group, there was evidence that Black Caribbean patients were less likely to be

224

referred by GP and this was independent of demographic and socioeconomic status (Model 2). However, the strength of association became attenuated after adjusting for DUP and family involvement in the pathways to care. There were no ethnic differences in GP referral in the EIS group.

		AESOP		CRIS-FE	P STUDY: Non- E	IS group	CRIS-	FEP STUDY: EIS	group
	Adjusted odds ratios (95% CI)		Adjusted odds ratios (95% CI)			Adjusted odds ratios (95% CI)			
	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3
Ethnicity									
White British	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
Black African	0.29	0.24	0.25	0.36	0.27	0.37	0.59	0.64	0.57
	(0.12 -0.69)*	(0.09-0.64) †	(0.08-0.70)*	(0.08 -1.55)	(0.05 -1.33)	(0.07 -2.00)	(0.24 -1.46)	(0.24 -1.68)	(0.20 -1.62)
Black	0.42	0.45	0.37	0.13	0.08	0.18	0.90	0.96	1.20
Caribbean	(0.19-0.92) †	(0.19 -1.04)*	(0.15-0.92) †	(0.01 -1.32)	(0.01 -0.93)*	(0.01 -2.47)	(0.32 -2.53)	(0.32 -2.87)	(0.37 -3.87)
White Other	1.18	1.31	1.12	0.83	0.70	0.97	1.06	1.10	1.14
	(0.39 -3.57)	(0.42-4.09)	(0.32 -3.90)	(0.19 -3.69)	(0.12 -4.16)	(0.14-6.62)		(0.34 -3.54)	(0.32 -4.04)
							(0.34-3.33)		

Table 9.6: Adjusted odds ratios of associations between ethnicity and GP referral by use of EIS

Model 1: age, gender; Model 2: age, gender, employment status, education level; Model 3: age, gender, employment status, education level, family involvement and duration of untreated psychosis

9.4.2 Associations between ethnicity and A&E referral by use of EIS

The following observations were made in the unadjusted regression model for associations between ethnicity and A&E referral. In the AESOP sample, Black Caribbean (OR = 0.64; 95% CI 0.29 – 1.36, p = 0.25), Black African (OR = 1.00; 95% CI 0.46 – 2.18, p = 0.99), White Other (OR = 0.68; 95% CI 0.21 – 2.19, p = 0.52). For unadjusted OR in the EIS group, Black African (OR = 0.59; 95% CI 0.24 – 2.45, p = 0.25), Black Caribbean (OR = 0.91; 95% CI 0.32 – 2.56, p = 0.87), White Other (OR = 1.01; 95% CI 0.32 – 3.12, p = 0.95). For unadjusted in the non-EIS group, Black African (OR = 0.32; 95% CI 0.07 – 1.33, p = 0.12), Black Caribbean (OR = 0.12; 95% CI 0.01 – 1.14, p = 0.06), White Other (OR = 0.67; 95% CI 0.16 – 2.76, p = 0.58).

Table 9.7 displays results from the adjusted regression model assessing the relationships between ethnicity and A&E referral. It can be seen from the table that there were no associations between ethnicity and A&E referral in the AESOP and EIS samples. However, there was strong evidence in the non-EIS group that Black African and Black Caribbean patients were seven and forty-eight times respectively more likely to be referred to mental health services via the accident and emergency department, and this was consistent across the regression models, hence the evidence was independent of all potential confounders. There was also some weak evidence suggestive of association with the White Other group being seven times more likely to be referred by GP after controlling for potential confounders.

		AESOP		CRIS-FE	P STUDY: Non- I	EIS group	CRIS-F	EP STUDY: EIS	group	
	Adjuste	d odds ratios (9	95% CI)	Adjus	Adjusted odds ratios (95% CI)			Adjusted odds ratios (95% CI)		
	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	
Ethnicity										
White British	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	
Black African	1.03	1.00	1.21	4.74	7.87	7.34	1.42	1.36	1.98	
	(0.47 -2.26)	(0.44 -2.50)	(0.47 -3.10)	(1.00 -	(1.34 -	(1.15 -46.74)*	(0.61 -3.31)	(0.54 -3.30)	(0.71 -5.48)	
				22.29)*	46.16)*					
Black	0.61	0.50	0.56	15.73	45.28	48.89	1.09	1.09	0.75	
Caribbean	(0.23 -1.33)	(0.21 -1.19)	(0.22 -1.38)	(2.20 -	(4.05 -	(3.49 -684.71)	(0.34 -3.03)	(0.36 -3.26)	(0.22 -2.55)	
				109.56) 🕇	505.86) 🕇	+				
White Other	0.76	0.82	0.90	4.04	7.11	7.03	1.95	2.01	1.74	
	(0.23 -2.51)	(0.23 -2.83)	(0.23-3.49)	(0.76 -21.50)	(0.93 -54.41)	(0.78 -62.76) ~	(0.63 -5.95)	(0.63 -6.40)	(0.48 -6.32)	
					~					

Table 9.7: Adjusted odds ratios of associations between ethnicity and A&E referral by use of EIS

Model 1: age, gender; Model 2: age, gender, employment status, education level; Model 3: age, gender, employment status, education level, family involvement and duration of untreated psychosis

9.4.3 Associations between ethnicity and criminal justice agency referral by use of EIS In AESOP, unadjusted analyses showed that Black Caribbeans (OR = 5.20; 95% CI 1.94 – 13.92, P<0.01) and Black Africans (OR = 4.46; 95% CI 1.59 – 12.59, p<0.01) were more likely to be referred by a criminal justice agency. White Other (OR = 1.85; 95% CI 0.41 – 8.37, p = 0.42) were less likely. Unadjusted OR in the EIS group showed that Black Africans (OR = 1.42; 95% CI 0.46 – 4.30, p = 0.53), Black Caribbeans (OR = 0.76; 95% CI 0.17 – 3.35, p = 0.71), White Other (OR = 0.68; 95% CI 0.12 – 3.74, p = 0.65) were not associated with criminal justice agency referral. Meanwhile, unadjusted OR in the non-EIS group showed no ethnic differences in criminal justice agency referral: Black African (OR = 0.66; 95% CI 0.09 – 4.50, p = 0.67), Black Caribbean (OR = no cases), White Other (OR = 0.43; 95% CI 0.04 – 4.68, p = 0.49).

Table 9.8 shows the associations between criminal justice agency referral and ethnicity. Consistent with the univariable odds ratios, Black African and Black Caribbean ethnic groups in the AESOP sample were up to four and eight times respectively more likely to be referred by a criminal justice agency, and this evidence was independent of all potential confounders. In the CRIS-FEP sample, there was no evidence of ethnic differences for criminal justice agency referral in both the EIS and non-EIS groups.

		AESOP		CRIS-FEP	STUDY: Non-	EIS group	CRIS-	FEP STUDY: EIS	group	
	Adjusted odds ratios (95% CI)			Adjuste	Adjusted odds ratios (95% CI)			Adjusted odds ratios (95% CI)		
	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	
Ethnicity										
White	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	
British										
Black	4.45	5.30	4.13	0.41	0.20	0.26	1.45	1.47	1.35	
African	(1.55-12.48) 🕇	(1.67-16.84)	(1.24-13.75)*	(0.05-3.34)	(0.15 -2.84)	(0.01 -4.09)	(0.47 -4.42)	(0.45 -4.75)	(0.35 -5.20)	
		+								
Black	6.29	7.66	8.29	No cases	No cases	No cases	0.77	0.91	1.06	
Caribbean	(2.23 -17.67) †	(2.50-23.44)	(2.59-26.54) 🕇							
		+					(0.17 -3.44)	(0.19 -4.34)	(0.16 -7.01)	
White	1.42	1.27	1.31	0.22	0.05	0.62	0.65	0.74	1.14	
Other	(0.30 -6.63)	(0.64 -6.12)	(0.25 -6.84)	(0.01-2.94)	(0.01 -2.82)	(0.01 -3.82)	(0.11 -3.65)	(0.13 -4.24)	(0.14 -8.73)	

Table 9.8: Adjusted odds ratios of associations between ethnicity and CJA referral by use of EIS

Model 1: age, gender; Model 2: age, gender, employment status, education level; Model 3: age, gender, employment status, education level, family involvement and duration of untreated psychosis

9.4.4 Associations between ethnicity and compulsory admissions by use of EIS

Unadjusted odds ratios for compulsory admission in the AESOP sample showed the Black Caribbean (OR = 4.01; 95% CI 1.82 – 8.81, P<0.01) and Black African (OR = 4.88; 95% CI 2.12 – 11.24, p<0.01) ethnic groups were at increased odds compared with the White British group. But White Other ethnic group (OR = 1.47; 95% CI 0.43 – 4.95, p = 0.53) did not differ from the White British. Unadjusted odds ratios in the EIS group showed that Black African (OR = 2.69; 95% CI 1.07 – 6.79, p = 0.03) patients were more likely to be detained compulsorily. However, no differences were observed between Black Caribbean (OR = 1.30; 95% CI 0.42 – 4.09, p = 0.64), and White British. Unadjusted odds ratios in the non-EIS group showed no ethnic difference for compulsory admission: Black African (OR = 1.84; 95% CI 0.41 - 7.96, p = 0.41), Black Caribbean (OR = 0.40; 95% CI 0.03 – 4.10, p = 0.44), White Other (OR = 0.66; 95% CI 0.10 – 4.26, p = 0.66).

Table 9.9 presents associations between compulsory admission and ethnicity; it can be seen from this table that Black African and Black Caribbean patients consistently showed increased odds of compulsory admission in the AESOP sample regardless of potential confounders. The evidence was similar for Black African patients in the EIS sample, showing that they were up to three and a half times more likely to be detained compulsorily independent of potential confounders. Meanwhile, no differences were observed among patients in the non-EIS group.

	A 11	AESOP		CRIS-FEP	STUDY: Non- I	EIS group	CRIS	-FEP STUDY: EI	S group
	Adjusted odds ratios (95% CI)		Adjusted odds ratios (95% CI)			Adjusted odds ratios (95% CI)			
	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3
Ethnicity									
White British	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
Black African	4.84	5.80	6.15	1.35	1.14	1.29	2.81	2.78	3.49
	(2.08-11.25)	(2.27-14.80) †	(2.29-16.54) †	(0.28 -6.43)	(0.20 -6.22)	(0.22 -7.57)	(1.09-7.21)*	(1.02-7.57)*	(1.12 -10.86)*
	+								
Black	3.85	4.15	4.79	0.25	0.16	0.24	1.36	2.04	1.95
Caribbean	(1.72 -8.63)	(1.72-10.03) †	(1.89-12.15) †	(0.02 -2.90)	(0.01-2.63)	(0.01 -5.00)	(0.43 -4.34)	(0.57 -7.20)	(0.49 -7.75)
	+								
White Other	1.25	1.03	0.95	0.40	0.32	0.37	No cases	No cases	No cases
	(0.36 -4.32)	(0.28 -3.73)	(0.23 -3.92)	(0.06 -3.00)	(0.03 -3.01)	(0.03 -3.73)			

Table 9.9: Adjusted odds ratios of associations between ethnicity and compulsory admission by use of EIS

Model 1: age, gender; Model 2: age, gender, employment status, education level; Model 3: age, gender, employment status, education level, family involvement and duration of untreated psychosis

9.5 Summary

To summarise, the results from both AESOP and present study data suggest that there have been some changes in the pathways to care and compulsory admission during first episode psychosis in the intervening 15 or so years. It is intriguing to speculate that this may be due to the introduction of EI services. At the same time, it could be due to other factors. In contrast to AESOP findings, individuals who received EIS in the present study were to some degree less likely to experience crisis pathways to care. For example, there were no ethnic differences observed in GP, CJA/police and A&E referral among those who received EIS. It was observed that Black Caribbean patients in the EIS group also differ from those in the AESOP study for compulsory admission as there were no differences between them and White British patients in the present study. However, patients of the Black African ethnic group remain more likely to be compulsorily detained during their first contact for psychosis despite accessing the early intervention services.

Considering those who did not receive EIS in the CRIS-FEP sample, the results point to a mixed outcome. While there were no increased odds of compulsory admission among the minority ethnic groups, individuals of the Black Caribbean ethnic group were less likely to be referred by their GP, although this finding did not hold in the fully adjusted analysis. Another key shift since the AESOP study appears to be that ethnic minority patients not in contact with EIS were more likely to come into contact with mental health services via the A&E department (these groups are Black African, Black Caribbean and White Other), a finding not seen in the AESOP sample.

9.6 Limitations

The limitations and how they were mitigated, discussed in Chapter 8, also apply in this chapter. However, specific issues related to the findings in this chapter are discussed.

The cross-sectional nature of our data means that the differences in associations between ethnicity and pathways to care in the AESOP and CRIS-FEP samples may not be explained by the introduction of early intervention services alone; it could be due to other factors. While we adjusted for demographic, socioeconomic variables and pathways to care variables, the results here may still be confounded by unmeasured variables. Further, the loss of power in the adjusted analyses means that the observed positive associations may be due to chance. For example, the small sample size in the non-EIS group and wide confidence interval around the elevated odds ratios for minority ethnic status and accident and emergency referral are interpreted tentatively.

10 Chapter 10: Results of Associations between Ethnicity and Service Use Outcomes at Follow-up

10.1 Synopsis

The results presented in Chapters 7, 8 and 9 were related to the cross-sectional data collected at first contact for psychosis (i.e. sociodemographic, socioeconomic, pathways to care, and duration of untreated psychosis). In this chapter, data collected longitudinally across the follow-up period are analysed and reported. These include data on service use outcomes, i.e. number of compulsory detentions, number of hospital admissions, total number of days in hospital and diagnosis. Subsequently, this chapter addresses the following aims:

- 5. To estimate rates and rate ratios of hospital admission and compulsory detention during the two-year follow-up by ethnic group.
- 6. To estimate differences in length of hospital stay by ethnic group during the twoyear follow-up.

This chapter tests the hypothesis that:

Over the two-year follow-up period, worse service use outcomes (characterised by increased rates of hospital admissions, compulsory admission and longer duration of hospital stay) will be observed among those of Black Caribbean and Black African ethnicity compared with White British patients.

Analysis

As mentioned previously in Chapter 6, the analysis set out in this chapter forms the basis for detailed enquiry of the potential factors associated with patterns of service use. The scope of data collected for this stage, and the analytic methods used, are designed to move beyond previous research. The collection of data relating to different outcomes on service use (i.e. hospital admission, compulsory detention, length of hospital stay) over the followup period allows for each of these outcomes and factors associated with them to be examined. A number of previous studies have investigated these outcomes but most considered them as static model of service use, e.g. using cross-sectional data (see Chapters 3 and 4). When considered as a process, earlier outcomes and experiences become important in understanding subsequent service use and engagement. Therefore, this stage of the study attempts to construct the service use process by focussing on both point of first contact with services for psychosis and subsequent outcomes, such as number of compulsory detentions, number of hospitalisations and total length of stay in hospital over the course of the follow-up period.

This chapter now describes service use outcomes at follow-up, and then a comparison of the outcomes between ethnic groups is presented (section 10.3). Secondly, rates of hospital admission and compulsory detention per year during the follow-up period are presented in sections 10.4 and 10.5. Thirdly, univariable and multivariable rate ratios were estimated for number of hospital admissions and number of compulsory admissions using negative binomial regression to assess differences by ethnic groups (sections 10.6 and 10.7). Finally, unadjusted and adjusted analyses of associations between total number of days spent in hospital during the follow-up period and ethnicity are discussed in Section 10.8.

10.2 Sample characteristics at follow-up

Characteristics	Cases
	n=558 (%)
Overall length of follow up	
Mean (SD) years	1.84 (1.14)
Median (IQR) years	2.07 (0.75 – 2.72)
Hospital admission at first contact	
Yes	238 (42.6)
No	320 (57.4)
Hospital admission at follow up	
Yes	145 (26.0)
No	413 (74.0)
Hospital admission, ever	
Once or more	357 (64.0)
None	201 (36.0)
Total number of days hospitalised	

Table 10.1: Service Use and Clinical characteristics during the follow up period

Mean (SD) days	56.82 (124.06)
Median (IQR) days	12 (0 - 68)
Compulsory admission at first contact	
Yes	135 (24.2)
No	423 (75.8)
Compulsory admission at follow up	
Yes	81 (14.5)
No	477 (85.5)
Compulsory admission, ever	
Once or more	268 (48.0)
None	290 (52.0)
OPCRIT ICD10- diagnosis (n=267)	
Non-affective psychosis	119 (44.6)
Affective psychosis	15 (5.6)
Other (Psychosis NOS)	133 (49.8)
ICD-10 Clinical Diagnosis ¹	
Non-affective psychosis	269 (48.7)
Affective psychosis	135 (24.5)
Other (Psychosis NOS)	148 (26.8)
Type of service use ²	
Early intervention service	222 (39.9)
СМНТ	81 (14.6)
Assessment & Treatment	126 (22.7)
Specialist	56 (10.1)
Forensic	7 (1.3)
Other	64 (11.5)
Outcomes	
Active patient in SLaM	222 (39.8)
Discharged to GP	232 (41.7)
Moved out of area/Transferred to	93 (16.7)
other MH provider	7 (1.2)
Noved abroad	/ (1.2)
Prison	2 (0.3)
Died	2 (0.3)

Missing data

1 (5 missing values)

2 (2 missing values)

Table 10.1 shows that during the follow-up period, individuals included in the study were followed up for an average of two years (sd = 1.14) and median 2.07 years (IQR = 0.75 - 1002.72). At first contact 238 (42.6%) patients were admitted to hospital. By the end of the study, 357 (64%) patients experienced at least one hospital admission. The median length of hospital stay was 12 days, interquartile range 0–68. At first contact, 135 (24.2%) patients were compulsorily detained. The number of patients ever compulsorily admitted had doubled during the course of the study from 24.2% to 48%. However, at follow-up, 145 patients were readmitted, of whom 56% were compulsorily detained. In terms of service use, around 40% of the sample entered into contact with early intervention service, followed by 22.7% receiving assessment and treatment service (A&T), 14.6% seeing community mental health teams (CMHT), 10.1% were seen in specialist teams (such as perinatal, HIV, social inclusion or culture-specific), 11.5% were seen in other services (e.g. outpatients, supported housing, psychology and rehabilitation) and only seven patients (1.3%) were with forensic services. Clinical diagnosis were categorised as follows: 48.7% non-affective (F10-19; F20-29), 24.5% affective (F30-33) and 26.8% other/Psychosis-NOS (Z71, F99). OPCRIT ICD-10 diagnosis data were available for 267 of the 588 sample and showed that the proportion of patients with non-affective psychosis (44.6%) was similar to those that were clinically recorded. There were differences in the two diagnostic groups for affective and psychosis NOS (Table 10.1). It is noteworthy that OPCRIT data were collected on cases identified during the first year of case screening, but not the second year, hence around half of the sample had data on this variable. By the end of the study (30 April 2014), only 40% remained active patients of SLaM. Around 42% were discharged to their GP and 17% either moved out of area or transferred to other mental health service providers. Seven individuals moved abroad, two were sentenced to prison and two died. There was a total 1030 persons-years at risk during the follow-up period.

10.3 Comparison between ethnic groups and clinical characteristics at follow-

up

Table 10.2 presents comparisons between follow-up service use characteristics and ethnicity. Chi-square, ANOVA and Kruskal Wallis analyses showed that there was strong evidence of differences between ethnicity and five main variables (i.e. age, total number of days in hospital, compulsory admission and hospital admission and diagnosis). Individuals of

238

the 'Mixed' ethnic group were more likely to be younger (29.7, sd = 9.6 years, p = 0.05) compared with other ethnic groups.

Length of hospital stay was not normally distributed, with some individuals experiencing extremely long hospital stay as illustrated in Figure 10.1; therefore median and interquartile ranges have been used to describe the data. Kruskal Wallis tests showed that during the study period, in total Black African (median = 30; IQR = 0 - 94) and Black Caribbean (median = 16; IQR = 0 - 90) patients spent longer days in hospital during the follow-up period compared with any other ethnic group (Table 10.2).



WB, White British, BA, Black African, BC, Black Caribbean, WO, White Other, AS, Asian, MX, Mixed, OTH, Other

Figure 10.1: Length of hospital stay by ethnicity

Across each of the service use outcome indicators, i.e. ever compulsorily admitted, followup hospital admission and number of days spent in hospital, the Black African ethnic group had the poorest outcome (Table 10.2).

In terms of clinical diagnosis, there was strong evidence of ethnic differences ($X^2 = 24.7$, df = 12, p = 0.01). Non-affective psychosis was common among all minority ethnic group patients compared with White British, while a higher proportion of White British and 'Other' patients

were diagnosed with psychosis NOS (not otherwise stated). Affective psychosis was less common among 'Mixed' ethnic group patients (Table 10.2). There were no ethnic differences in community service use.

By the end of the study, there was weak evidence that people who had left the study were more likely to have been discharged to GP (White Other = 48%; Other = 58.5%) or moved out of area (White Other =24%; White British = 19.5%). Of those who moved abroad , three were Black African (2%) and White Other (4%), $X^2 = 40.9$, df = 30, p = 0.08. There were no differences observed in ethnicity and DUP, and length of follow-up. This may suggest that some of this study sample may be transient population.

	White British	Black African	Black Caribbean	Other White	Asian n= 44 (%)	Mixed n= 27 (%)	Other n= 41 (%)	ANOVA/ Kruskal-	df	р
	n= 133 (%)	n=147 (%)	n = 91 (%)	n =75 (%)				Wallis X ²		
								test		
Mean Age at FEP in	34.5 (12.5)	32.4 (9.9)	33.4 (10.6)	34.9 (9.6)	32.4 (10.3)	29.7 (9.6)	31.9 (9.1)	12.56	6	0.05
years (SD)										
Mean follow up time	2.02 (1.2)	1.98 (1.1)	2.22 (1.2)	1.65 (1.0)	1.75 (1.1)	1.96 (1.2)	1.67 (1.2)	3.90	6	0.68
(SD) years										
Median DUP in days	105 (22 -	88 (17 -	126 (28 -	86 (14 -	76.5 (8.5 -	92 (23 - 361)	60 (12 - 560)	2.40	6	0.87
(IQR)	514)	447)	449)	408)	243)					
Median LOS (IQR)	6 (0 - 50)	30 (0 - 94)	16 (0-90)	13 (0 - 47)	10 (0 -	8 (0 - 45)	0 (0 - 21)	13.47	6	0.03
days- total number					32.5)					
of days hospitalised										
during follow up										
period										
								Chi-sq	df	р
								(exact)		
								test		
Hospital admission										
at follow up										
No	99 (74.4)	95 (64.6)	67 (73.6)	59 (78.7)	38 (86.4)	18 (66.7)	37 (90.2)	17.46	6	0.008
Yes	34 (25.6)	52 (35.4)	24 (26.4)	16 (21.3)	6 (13.6)	9 (33.3)	4 (9.8)			
Hospital admission,										
ever										
None	47 (35.3)	49 (33.3)	35 (38.5)	27 (36.0)	12 (27.3)	8 (29.6)	23 (56.1)	9.83	12	0.13
Once or more	86 (64.7)	98 (66.7)	56 (61.5)	48 (64.0)	32 (72.7)	19 (70.4)	18 (43.9)			
Compulsory										
admission at follow										

Table 10.2: Comparisons between clinical and service use characteristics at follow up and ethnic groups

up										
No	119 (89.5)	111 (75.5)	74 (81.3)	67 (89.3)	42 (95.4)	24 (88.9)	40 (97.6)	24.25	6	<0.001
Yes	14 (10.5)	36 (24.5)	17 (18.7)	8 (10.7)	2 (4.6)	3 (11.1)	1 (2.4)			
Compulsory, ever										
None	79 (59.4)	59 (40.1)	47 (51.6)	40 (23.3)	24 (54.6)	13 (48.1)	28 (68.3)	15.89	6	0.01
Once or more	54 (40.6)	88 (59.9)	44 (48.4)	35 (46.7)	20 (45.4)	14 (51.9)	13 (31.7)			
ICD-10 Clinical										
Diagnosis										
Non-affective	46 (34.6)	80 (54.8)	45 (50.0)	45 (60.8)	22 (52.4)	16 (59.3)	15 (37.5)	24.78	12	0.01
psychosis										
Affective psychosis	36 (27.1)	37 (25.3)	21 (23.3)	16 (21.6)	9 (21.4)	5 (18.5)	11 (27.5)			
Other (Psychosis	51 (38.3)	29 (19.9)	29 (26.7)	13 (17.6)	11 (26.2)	6 (22.2)	14 (35.0)			
NOS)										
Type of service use										
Early intervention	55 (41.3)	70 (47.6)	35 (38.9)	24 (32.0)	14 (32.6)	12 (44.4)	12 (29.3)	35.64	30	0.22
service										
СМНТ	15 (11.3)	24 (16.3)	17 (18.9)	13 (17.3)	6 (13.9)	0	6 (14.6)			
Assessment &	32 (24.1)	25 (17.0)	20 (22.2)	18 (24.0)	7 (16.3)	9 (33.3)	15 (36.6)			
Treatment										
Specialist	17 (12.8)	11 (7.5)	8 (8.9)	8 (10.7)	7 (16.3)	1 (3.7)	4 (9.8)			
Forensic	1 (0.7)	2 (1.4)	2 (2.2)	0	0	1 (3.7)	1 (2.4)			
Other	13 (9.8)	15 (10.2)	8 (8.9)	12 (16.0)	9 (20.9)	4 (4.9)	3 (7.3)			
Administrative										
outcomes										
Active patient in	54 (40.6)	63 (42.9)	45 (49.5)	17 (22.7)	18 (40.9)	14 (51.8)	11 (26.8)	40.99	30	0.08
SLaM										
Discharged to GP	51 (38.3)	54 (36.7)	38 (41.8)	36 (48.0)	18 (40.9)	11 (40.7)	24 (58.5)			
Moved out of	26 (19.5)	27 (18.4)	7 (7.7)	18 (24.0)	7 (15.9)	2 (7.4)	6 (14.6)			
other MH provider										

Moved abroad	0	3 (2.0)	0	3 (4.0)	1 (2.3)	0	0		
Prison	1 (0.8)	0	1 (1.1)	0	0	0	0		
Died	1 (0.8)	0	0	1 (1.3)	0	0	0		

10.4 Rate of hospital admission per year by ethnic group during the follow-up period

In order to address the objective of estimating rates of hospital admission within the followup period by ethnic groups, Poisson regression analysis was employed using the option *exposure (follow-up time)*. It was important to include and specify the time at risk (followup time) using the 'exposure' option since the Poisson command assumes all participants had the same time at risk.

Time at risk was defined as: exit date minus entry date.

Entry date = date of contact with mental health services for FEP or date of discharge for first admission, whichever happened first.

Exit date = study end date (30 April 2014) or date the patient left the study (i.e. discharge from SLaM services) whichever happened first.

Table 10.3: shows the rate of hospital admissions per year with 95% confidence intervals by ethnic groups. From the table, the following observations were made: the rate of admission was highest among Black African patients (0.43 per year or once every two years), followed by the 'Mixed', White British and Black Caribbean ethnic groups (0.34, 0.31 and 0.29 per year or once every three years respectively). The remaining ethnic groups had an admission around once every five years (0.12 to 0.23 per year).

Ethnicity	Number in	Rate of admission	95% confidence
	sample (%)		interval (CI)
White British	133 (23.8)	0.31	0.25 – 0.39
Black African	147 (26.3)	0.43	0.35 – 0.51
Black Caribbean	91 (13.4)	0.29	0.22 – 0.38
White Other	75 (13.4)	0.23	0.16 - 0.34
Asian	44 (7.9)	0.19	0.11 -0.32
Mixed	27 (4.8)	0.34	0.21-0.54
Other	41 (7.3)	0.12	0.06 -0.24

Table 10.3: Rate of hospital admission per year with 95% confidence intervals by ethnic groupduring the follow up period

10.5 Rate of compulsory admission per year by ethnic group during the follow-up period

In this section, I have used the same statistical approach described in section 10.4 to estimate the rate of compulsory admission. Table 10.4 presents the rate of compulsory admission per year with 95% confidence intervals by ethnicity. From the table, a markedly elevated rate of compulsory admissions was observed among patients of Black African ethnicity (0.27 per year or once every three years), and the Black Caribbean group were compulsorily admitted once every five years (0.18 per year). White British, White Other and 'Mixed' ethnic groups experienced compulsory admissions around once every ten years. Patients of Asian and 'Other' ethnicity experienced lower rates of compulsory admissions. Table 10.4: Rate of compulsory admission per year with 95% confidence intervals by ethnic groupduring the follow up period

Ethnicity	Number in	Rate of admission	95% confidence
	sample (%)	per year	interval (CI)
White British	133 (23.8)	0.11	0.06 -0.14
Black African	147 (26.3)	0.27	0.22 -0.34
Black Caribbean	91 (13.4)	0.18	0.13 -0.25
White Other	75 (13.4)	0.10	0.05 -0.16
Asian	44 (7.9)	0.03	0.01 -0.11
Mixed	27 (4.8)	0.10	0.03 -0.22
Other	41 (7.3)	0.03	0.01 -0.12

10.6 Associations between ethnicity and number of hospital admissions

during the follow-up

In exploring the relationship between ethnicity and numbers of hospital admissions during the follow-up period, Poisson regression analysis was initially employed to estimate the rate ratios of hospital admission, as it is count data. This was followed by a test of fit using the goodness of fit test which revealed that over dispersion of zero was present and not normally distributed (Pearson goodness-of-fit $X^2 = 3331.6$, df = 550, p<0.001). Therefore negative binomial regression was employed for the analysis, whilst taking into account the follow-up period using the **exposure** option in Stata. The decision to use negative binomial regression was based on the distribution of the data as shown in the test statistic and histogram displayed in Figure 10.2 (UCLA: Statistical Consulting Group, 2016).



Figure 10.2: Histogram of number of hospital admissions

Unadjusted and multivariable regression analyses were fitted as below. Potential confounders were controlled for as follows:

Model 1 – unadjusted

Model 2 – adjusted for age, gender

Model 3 – adjusted for age, gender, level of education and employment status Model 4 – adjusted for age, gender, level of education, employment status and diagnosis

A final regression model was fitted, extending the analyses to adjust for baseline hospital admission (i.e. model 4 plus baseline hospital admission). This was to test whether baseline hospital admission predicts subsequent hospital admissions. The results are reported in the text.

All analyses in the unadjusted and adjusted models here were carried out using complete case sample, i.e. only 367 for whom data were complete for each of the confounding variables. This will allow comparison of associations across the models.

Table 10.5 displays results of univariable and adjusted association between number of hospital admissions along with 95% CI and ethnicity. From this table, strong evidence can be seen that Black African patients had higher rates of hospital admission over the follow-up period and this was independent of all potential confounders. Despite the small attenuation in the effect size when age and gender were controlled for (i.e. from IRR = 1.98 to 1.81), the strength of association remained robust and became stronger in subsequent models when socioeconomic status and diagnosis were adjusted for. There was no evidence of associations between increased number of admissions and any other ethnic groups.

In the final analyses while adjusting for the *a priori* confounders in model 4 as well as baseline hospital admission, differences in hospital admission at follow-up held, with rate ratios of almost two for Black African patients. The rate ratios along with 95% CI by ethnic group are reported in text, below Table 10.5.

249

	Model 1	95% confidence interval (p-value)	Model 2	95% confidence interval (p-value)	Model 3	95% confidence interval (p-value)	Model 4	95% confidence interval (p-value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	1.98	1.14 - 3.44 (0.01)	1.81	1.05 - 3.13 (0.03)	2.12	1.21 - 3.68 (<0.01)	1.98	1.12 - 3.48 (0.01)
Black Caribbean	1.35	0.72 – 2.54 (0.34)	1.37	0.74 – 2.56 (0.31)	1.25	0.66 – 2.36 (0.48)	1.13	0.58 – 2.19 (0.71)
White Other	1.26	0.62 – 2.55 (0.50)	1.22	0.60 – 2.46 (0.56)	1.23	0.60 – 2.50 (0.56)	1.23	0.60 – 2.49 (0.56)
Asian	0.93	0.36 – 2.38 (0.89)	0.93	0.37 – 2.33 (0.88)	0.96	0.38 – 2.43 (0.94)	0.93	0.36 – 2.40 (0.88)
Mixed	2.00	0.82 - 4.86 (0.12)	1.91	0.79 – 4.58 (0.14)	1.89	0.77 – 4.58 (0.15)	1.83	0.76 - 4.41 (0.17)
Other	0.68	0.24 – 1.97 (0.48)	0.64	0.22 – 1.85 (0.41)	0.65	0.22 – 1.89 (0.43)	0.61	0.21 – 1.78 (0.37)

Table 10.5: Adjusted rate ratios of associations between ethnicity and number of hospital admissions during the follow up period (n=367)

Model 1- unadjusted; Model 2- adjusted for age, gender; Model 3- adjusted for age, gender, level of education and employment status; Model 4 - adjusted for age, gender, level of education, employment status, diagnosis

Final Model: adjusting for model 4 plus baseline hospital admission

White British: Reference group

Black African: adj. IRR = 1.71; 95% CI 0.98 - 2.99 (p = 0.05)

Black Caribbean: adj. IRR = 1.07; 95% CI 0.55 – 2.07 (p = 0.83)

White Other: adj. IRR = 1.04; 95% CI 0.51 – 2.11 (p = 0.89)

Asian: adj. IRR = 0.91; 95% CI 0.33 – 2.34 (p = 0.85)

Mixed: adj. IRR = 1.54; 95% CI 0.65 – 3.64 (p = 0.31)

Other: adj. IRR = 0.63; 95% CI 0.22 –1.81 (p = 0.40)
10.7 Associations between ethnicity and the number of compulsory admissions during the follow-up period

Figure 10.3 displays the distribution of number of compulsory admissions showing the over dispersion of zero. Therefore, the same statistical method (i.e. negative binomial) used in section 10.6 above was employed to explore the relationship between ethnicity and the number of compulsory admissions.



Figure 10.3: Histogram of number of compulsory admissions

Adjustments for the *a priori* confounders were undertaken. Therefore, four regression models were fitted as follows:

Model 1 – unadjusted

- Model 2 adjusted for age, gender
- Model 3 adjusted for age, gender, level of education and employment status
- Model 4 adjusted for age, gender, level of education, employment status and diagnosis

Again here, a final regression model was fitted, extending the analyses to adjust for baseline compulsory admission (i.e. model 4 plus baseline compulsory admission). This was to test whether baseline compulsory admission predicts subsequent compulsory admissions. The results are reported in the text below.

Table 10.6 shows results of rate ratios of number of compulsory admissions. From this table, there was strong evidence of three-fold elevated rate of compulsory admissions among people of Black African ethnicity throughout the unadjusted and adjusted models. In the unadjusted model, there was also evidence that patients of Black Caribbean ethnic group were more likely to experience an increased rate of compulsory admission (IRR = 2.46; 95% Cl 1.03 – 6.01, p = 0.04). This association was attenuated in the fully adjusted models. However, the evidence of associations in models 2 and 3 for the Black Caribbean ethnic group (albeit weak) may suggest that socioeconomic position may play an important role in the relationship between compulsory admission and ethnicity. There were no cases from the 'Other' ethnic group in the multivariable analysis.

In the repeated analyses while adjusting for the *a priori* confounders in model 4 as well as baseline compulsory admission, differences in compulsory admission at follow-up held, with rate ratios of around two and a half for Black Africans. However, the observed associations for the Black Caribbean group did not hold and showed no difference compared with the White British ethnic group. The rate ratios along with 95% CI by ethnic group are reported in text, below Table 10.6.

	Model 1	95% confidence interval (p-value)	Model 2	95% confidence interval (p-value)	Model 3	95% confidence interval (p-value)	Model 4	95% confidence interval (p-value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	3.50	1.59 -7.71 (<0.01)	3.09	1.39 -6.88 (<0.01)	3.46	1.52 -7.83 (<0.01)	3.01	1.33 -6.80 (<0.01)
Black Caribbean	2.46	1.03 -6.01 (0.04)	2.27	0.93 -5.51 (0.07)	2.17	0.89 -5.26 (0.08)	1.62	0.65 -4.03 (0.29)
White Other	1.93	0.47 -4.09 (0.54)	1.28	0.43 -3.73 (0.65)	1.52	0.51 -4.51 (0.44)	1.38	0.46 -4.07 (0.55)
Asian	0.54	0.10 -2.88 (0.47)	0.52	0.09 -2.79 (0.45)	0.56	0.10 -3.01 (0.50)	0.43	0.08 -2.36 (0.33)
Mixed	1.51	0.39 – 5.86 (0.54)	1.54	0.40 -5.91 (0.52)	1.30	0.32 -5.18 (0.70)	1.25	033 -4.77 (0.73)
Other	-	No cases						

Table 10.6: Adjusted rate ratios of associations between ethnicity and number of compulsory admissions during the follow up period (n=367)

Model 1- unadjusted; Model 2- adjusted for age, gender; Model 3- adjusted for age, gender, level of education and employment status; Model 4 - adjusted for age, gender, level of education, employment status, diagnosis

White British: Reference group

Black African: adj. IRR = 2.58; 95% Cl 1.15 – 5.79 (p = 0.02)

Black Caribbean: adj. IRR = 1.49; 95% CI 0.61 – 3.68 (p = 0.37)

White Other: adj. IRR = 1.35; 95% CI 0.47 – 3.88 (p = 0.57)

Asian: adj. IRR = 0.42; 95% CI 0.07 – 2.29 (p = 0.32)

Mixed: adj. IRR = 0.98; 95% CI 0.25 – 3.74 (p = 0.98)

Other: No cases

10.8 Associations between length of hospital stay and ethnicity during the follow-up period

Figure 10.4 displays the distribution of length of hospital stay, indicating over dispersion of zero. Therefore, in keeping with the analyses carried out in Sections 10.6 and 10.7, negative binomial regression was employed here to estimate the rate ratios for length of hospital stay by ethnic group.



Figure 10.4: Histogram of length of hospital stay

Table 10.7 shows results of unadjusted and adjusted regression analyses along with 95% confidence intervals. It can be seen from this table that there was strong evidence that compared with White British patients, Asian patients were more likely to spend fewer days in hospital, and the strength of association was independent of all potential confounders. There were no further differences observed in the length of hospital stay by ethnic group.

	Model 1	95% confidence	Model 2	95% confidence	Model 3	95% confidence	Model 4	95% confidence
		interval (p-value)		interval (p-value)		interval (p-value)		interval (p-value)
Ethnicity								
White British	1.00		1.00		1.00		1.00	
Black African	1.82	0.41 -8.02 (0.43)	1.88	0.42 -8.42 (0.40)	2.17	0.42 -11.28 (0.35)	1.67	0.29 -9.52 (0.56)
Black Caribbean	0.93	0.16 -5.13 (0.93)	1.18	0.20 -6.67 (0.85)	1.08	0.16 -7.06 (0.93)	0.49	0.04 -4.98 (0.54)
White Other	0.33	0.05 -1.96 (0.22)	0.35	0.06 -2.06 (0.24)	0.39	0.05 -2.90 (0.36)	0.36	0.04 -2.82 (0.33)
Asian	0.10	0.01 -0.97 (0.04)	0.08	0.01 -0.79 (0.03)	0.05	0.0 -0.86 (0.03)	0.02	0.01 -0.62 (0.02)
Mixed	4.49	0.30 -65.65 (0.27)	4.50	0.34 -58.68 (0.25)	6.96	0.43 -110.57 (0.16)	4.69	0.30 (71.61)0.26
Other	0.50	0.05 -4.86 (0.55)	0.69	0.07 -6.84 (0.75)	0.67	0.04 -9.53 (0.77)	0.36	0.02 -5.62 (0.47)

Table 10.7: Adjusted rate ratios of associations between ethnicity and length of hospital stay during the follow up period (n=367)

Model 1- unadjusted; Model 2- adjusted for age, gender; Model 3- adjusted for age, gender, level of education and employment status; Model 4 - adjusted for age, gender,

level of education, employment status, diagnosis

10.9 Summary

To summarise, the majority of the sample experienced at least one hospital admission and one compulsory admission during the follow-up period, but the rate of hospital admissions and compulsory detentions was markedly higher among patients of Black African ethnicity.

There was some evidence suggestive of some associations between compulsory admission and the Black Caribbean ethnic group but this only went as far as adjusting for socioeconomic status.

In exploring the associations between length of hospital stay and ethnicity, the results suggest that there was strong evidence that only patients of the Asian ethnic group were more likely to spend fewer days in hospital over the follow-up period compared with White British patients. There were no differences in the other ethnic groups.

To conclude, in support of the hypothesis set out in this chapter, Black African patients experienced worse service use outcomes. However, the results did not support the hypothesis for the Black Caribbean group, who did not show any significant difference compared with White British patients. The implications and possible interpretation of these findings will be addressed in detail in the Discussion chapter (Chapter 12).

10.10 Limitations

In addition to the methodological limitations discussed in the preceding chapters, fundamental to longitudinal studies is the issue of attrition. This is reflected in the findings here, as around sixty per cent of the participants had left SLaM services by the end of the study. However, a key advantage of using a case register (CRIS) for this study is the ability to determine date of discharge for each individual and therefore the amount of data each participant contributed to the follow-up period were included in the analyses. Further, the longitudinal analyses were based on the assumption that participants left the study at random (Susser et al., 2006). For example, there were no statistical significant differences in the administrative outcomes (Table 10.3) by ethnic groups, so the impact of attrition on the findings here were minimal.

11 Chapter 11: Results: Ethnicity and Service Use Outcomes by Early Intervention Service Status during the Follow-up Period

11.1 Synopsis

In Chapter 10, data analyses that addressed the relationships between service use outcomes and ethnicity were presented and the results indicated that there were some ethnic variations in the outcomes that were considered. This chapter expands on the work in Chapter 10, by examining whether the differences observed in the service use outcomes may be explained by early intervention service status. Using in part the same statistical approach described in Chapter 10, and non-parametric tests, the following aims and hypothesis were addressed:

- To compare clinical and service use outcomes (hospital admission, compulsory admission and length of hospital stay) between those using early intervention services (EIS) and non-EIS users and non-eligible patients.
- To determine whether there are ethnic differences in clinical and service use outcomes at the two-year follow-up in relation to early intervention service use status.

This chapter tests the hypothesis that:

Over the two-year follow-up period, poorer clinical and service use outcomes will be associated with non-EIS use compared with EIS status and these will vary by ethnic groups.

As described in Chapter 9, before assessing associations between clinical and service use outcomes and early intervention service use (EIS), it was important to arrange the total sample according to EIS eligibility. With this in mind, since the age eligibility for EIS at the time of this study was 18–35 years, the sample was re-categorised according to patients' age and whether they received EIS or not as follows:

18–35 years – eligible and received (EIS group)

18-35 years - eligible but did not receive (non-EIS group)

From this, comparisons of sociodemographic factors and clinical outcomes by EIS use are described in section 11.1. Next, unadjusted and multivariable regression analyses were fitted to assess association between EIS status and number of hospital admissions, compulsory admissions and length of hospital stay, presented in sections 11.2, 11.3 and 11.4. Subsequently, the associations between ethnicity and service use outcomes by EIS status are described and reported in sections 11.5 to 11.8.

11.2 Comparisons between EIS use, sociodemographic and follow-up clinical characteristics

In Table 11.1, test statistics i.e. Chi-square, ANOVA and Krusal Wallis analyses were employed to explore differences in EIS use by sociodemographic, pathways to care and clinical characteristics during follow-up.

11.2.1 EIS eligible versus Non-eligible comparisons

As shown in Table 11.1, those not eligible for early intervention services are by definition, older than those eligible for EIS. Perhaps reflecting their older age, far fewer were still in education, were less likely to be single and more likely to be separated or divorced, living alone and women (56.3%). Further, the older age of the non-eligible patients for EIS may also explain the higher proportion of this group to receive a diagnosis of affective psychosis (34.1%). In terms of pathways to care, patients who were non-eligible for EIS were more likely to experience an insidious onset of psychosis, less likely to have family involvement in help-seeking and were more likely to be discharged to GP compared with those who received EIS.

11.2.2 EIS versus non-EIS comparisons

Table 11.1 also displays comparisons between the EIS and non-EIS groups. As can be seen from the table, patients in the non-EIS group appear to differ from their EIS counterparts on characteristics that indicate social isolation and crisis pathways to care. For example, the non-EIS patients were more likely to be unemployed, living alone, have housing tenure such

as 'Other' (hostel), make contact with service out of hours and less likely to have family involvement in help-seeking. For service use outcomes, non-EIS patients more likely to have frequent hospital admissions, to receive a diagnosis of Psychosis not otherwise stated, to use specialists (such as perinatal, HIV, social inclusion or culture-specific) and to move out of study area.

	EIS n= 222 (%)	Non-EIS n=119 (%)	Non- eligible n = 215 (%)	ANOVA/ Kruskal- Wallis X ² test	df	p
Mean follow up time (SD) years	2.10 (1.06)	1.69 (1.24)	1.92 (1.21)	5.09	2	0.07
Median DUP in days (IQR)	91.5 (10-447)	82 (17 - 360)	107 (22 - 482)	2.63	2	0.26
Median LOS (IQR) days- total number of days hospitalised during follow up period	22 (0 - 77)	9 (0 - 66)	6 (0 - 50)	5.39	2	0.06
Socio-demographic variables	EIS n= 222 (%)	Non-EIS n=119 (%)	Non- eligible n = 215 (%)	X ² test	df	р
Gender						
Male	137 (61.7)	59 (49.6)	94 (43.7)	14.57	2	0.001
Female	85 (38.3)	60 (50.4)	121 (56.3)			
Ethnicity						
White British	55 (24.8)	23 (19.3)	55 (25.6)	20.86	12	0.05
Black African	70 (31.5)	25 (21.0)	52 (24.2)			
Black Caribbean	35 (15.8)	17 (14.3)	38 (17.7)			
Other White	24 (10.8)	16 (13.5)	35 (16.3)			
Asian	14 (6.3)	15 (12.6)	14 (6.5)			
Mixed	12 (5.4)	8 (6.7)	7 (3.2)			
Other	12 (5.4)	15 (12.6)	14 (6.5)			
Employment						
Unemployed	121 (58.8)	81 (71.7)	143 (77.3)	41.27	4	<0.001
Student	45 (21.8)	13 (11.5)	2 (1.1)			
Employed	40 (19.4)	19 (16.8)	40 (21.6)			
Education						
School, no GCSE	34 (19.6)	30 (32.6)	34 (27.0)	7.05	6	0.31

Table 11.1: Comparisons between baseline socio-demographic, pathways to care and follow up clinical characteristics by EIS status

School with GCSE	32 (18.4)	14 (15.2)	26 (20.6)			
A level/ Further	54 (31.0)	25 (27.2)	35 (27.8)			
education						
University	54 (31.0)	23 (25.0)	31 (24.6)			
Relationship status						
Single	151 (70.2)	78 (70.2)	100 (49.0)	37.12	4	<0.001
Married / Steady	46 (21.4)	27 (24.3)	54 (26.5)			
relationship						
Divorced/Separated	18 (8.4)	6 (5.4)	50 (24.5)			
Living circumstances						
Alone	42 (19.3)	32 (27.8)	86 (41.5)	30.78	4	<0.001
Family/relatives	151 (69.3)	65 (56.5)	109 (52.7)			
	25 (11.4)	18 (15.7)	12 (5.8)			
Housing Tenure	22 (22 7)	11 (12 0)	25 (10.2)	0.52	_	0.04
Privately owned	32 (23.7)	11 (12.9)	25 (19.2)	9.53	4	0.04
Rented Other	80 (59.2)	52 (61.2)	89 (68.5)			
Bathways to care	23 (17.0)	22 (25.9)	10 (12.3)			
variables						
Source of referral						
GP/Primary care	75 (33.8)	32 (26.9)	88 (40.9)	7.58	6	0.27
	90 (40 5)	52 (43 7)	75 (34 9)	7.00		0.27
Rolico / Criminal	32 (14 4)	17 (1/13)	27 (12 6)			
	52 (14.4)	17 (14.5)	27 (12.0)			
Othor	25 (11 2)	10 (15 1)	25 (11 6)			
Nada of exact	23 (11.3)	10(13.1)	25 (11.0)			
Node of onset		24 (20.2)	22 (15 2)	14.04	-	0.02
Acute (within a	59 (20.0)	24 (20.2)	33 (15.3)	14.04	0	0.02
week)	22 (4 4 4)	20 (24 4)	50 (22.2)			
Moderate (within a	32 (14.4)	29 (24.4)	50 (23.3)			
month)	50 (00 1)					
Gradual (up to	52 (23.4)	25 (21.0)	44 (20.5)			
6months)						
Insidious (more	79 (35.6)	41 (34.4)	88 (40.9)			
than 6 months)						
Time of FEP						
contact						
Office hours	141 (63.5)	69 (58.0)	151 (70.2)	5.37	2	0.06
Out of hours	81 (36.5)	50 (42.0)	64 (29.8)			
Family						
involvement						
No	125 (56.3)	82 (68.9)	153 (71.2)	11.70	2	0.003
Yes	97 (43.7)	37 (31.1)	62 (28.8)			
Mode of contact						
Non-compulsory	164 (43.9)	88 (73.9)	169 (78.6)	1.58	2	0.45
compulsory	58 (26.1)	31 (26.1)	46 (21.4)			
Follow up clinical						•

variables				_		
Any hospital						
admission						
No	74 (33.3)	38 (31.9)	88 (40.9)	3.80	2	0.14
Yes Number of bespital	148 (66.7)	81 (68.1)	127 (59.1)			
admission						
No admission	74 (33.3)	38 (31.9)	88 (40.9)	9.32	4	0.05
1-2	177 (52.7)	58 (48.7)	108 (50.2)			
≥3	31 (14.0)	23 (19.3)	19 (8.9)			
Number of MHA						
No MHA	107 (48.2)	60 (50.4)	122 (56.7)	7.11	4	0.13
1-2	64 (28.8)	41 (34.4)	60 (27.9)			
≥3	51 (23.0)	18 (15.1)	33 (15.4)			
ICD-10 Clinical Diagnosis						
Non-affective psychosis	135 (61.6)	47 (39.5)	87 (40.6)	35.10	4	<0.001
Affective psychosis	28 (12.8)	34 (28.6)	73 (34.1)			
Other (Psychosis NOS)	56 (25.6)	38 (31.9)	54 (25.2)			
Type of service use				26.20	4	<0.001
Type of service use CMHT	0	16 (13.5)	65 (30.2)	26.20	4	<0.001
Type of service use CMHT Assessment & Treatment	0	16 (13.5) 39 (32.8)	65 (30.2) 87 (40.5)	26.20	4	<0.001
Type of service use CMHT Assessment & Treatment Specialist	0 0 0	16 (13.5) 39 (32.8) 27 (22.7)	65 (30.2) 87 (40.5) 29 (13.5)	26.20	4	<0.001
Type of service useCMHTAssessment &TreatmentSpecialistForensic	0 0 0 0	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5)	26.20	4	<0.001
Type of service use CMHT Assessment & Treatment Specialist Forensic Other	0 0 0 0 0 0	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3)	26.20	4	<0.001
Type of service use CMHT Assessment & Treatment Specialist Forensic Other	0 0 0 0 0 0	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3)	26.20	4	<0.001
Type of service useCMHTAssessment &TreatmentSpecialistForensicOtherAdministrativeoutcomes	0 0 0 0 0 0	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3)	26.20	4	<0.001
Type of service useCMHTAssessment &TreatmentSpecialistForensicOtherAdministrativeoutcomesActive patient inSLaM	0 0 0 0 0 0 0 102 (45.9)	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0) 37 (31.1)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3) 81 (37.7)	26.20	4	<0.001
Type of service useCMHTAssessment &TreatmentSpecialistForensicOtherAdministrativeoutcomesActive patient inSLaMDischarged to GP	0 0 0 0 0 0 102 (45.9) 77 (34.7)	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0) 37 (31.1) 56 (47.1)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3) 81 (37.7) 99 (46.1)	26.20	4	<0.001
Type of service useCMHTAssessment &TreatmentSpecialistForensicOtherAdministrativeoutcomesActive patient inSLaMDischarged to GPMoved out ofarea/Transferred toother MH provider	0 0 0 0 0 0 102 (45.9) 77 (34.7) 38 (17.1)	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0) 37 (31.1) 56 (47.1) 24 (20.2)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3) 81 (37.7) 99 (46.1) 31 (14.4)	26.20	4	<0.001
Type of service useCMHTAssessment &TreatmentSpecialistForensicOtherAdministrativeoutcomesActive patient inSLaMDischarged to GPMoved out ofarea/Transferred toother MH providerMoved abroad	0 0 0 0 0 0 102 (45.9) 77 (34.7) 38 (17.1) 4 (1.8)	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0) 37 (31.1) 56 (47.1) 24 (20.2) 1 (0.8)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3) 81 (37.7) 99 (46.1) 31 (14.4) 2 (0.9)	26.20	4	<0.001 0.09
Type of service useCMHTAssessment &TreatmentSpecialistForensicOtherAdministrative outcomesActive patient in SLaMDischarged to GPMoved out of area/Transferred to other MH providerMoved abroadPrison	0 0 0 0 0 0 0 102 (45.9) 77 (34.7) 38 (17.1) 4 (1.8) 1 (0.5)	16 (13.5) 39 (32.8) 27 (22.7) 6 (5.0) 31 (26.0) 37 (31.1) 56 (47.1) 24 (20.2) 1 (0.8) 1 (0.8)	65 (30.2) 87 (40.5) 29 (13.5) 1 (0.5) 33 (15.3) 81 (37.7) 99 (46.1) 31 (14.4) 2 (0.9) 0	26.20	4	<0.001 0.09

11.3 Associations between EIS status and number of hospital admissions during follow-up

At this stage of the analysis, I have used the same statistical procedures employed in Chapter 10, section 10.6 (i.e. negative binomial regression) to assess the association between number of hospital admissions and early intervention service use. Unadjusted and adjusted negative binomial regression and potential confounders were adjusted for as follows:

Model 1 – unadjusted

Model 2 – adjusted for age, gender and ethnicity

Model 3 – adjusted for age, gender, ethnicity, level of education and employment status Model 4 – adjusted for age, gender, ethnicity, level of education, employment status, and diagnosis

All analyses in the unadjusted and adjusted models here were carried out using complete case sample, i.e. only 366 (66% of whole sample) for whom data were complete for the main exposure variable and each of the confounding variables (e.g. information on education level and employment status was missing for 191 patients and information on EIS was not available for one patient). This will allow comparison of associations across the models.

From Table 11.2, strong and increasing associations between number of hospital admissions and patients in the non-EIS group were observed. Unadjusted analysis showed that the non-EIS group (IRR = 2.35; 95% CI 1.47 – 3.73) experienced more hospital admissions compared with patients who received EIS. This evidence remained robust and became stronger when age, gender and ethnicity were controlled for in model 2 (adj. IRR = 3.32; 95% CI 2.05 – 5.37). The effect size attenuated slightly in models 3 and 4 but remained strong and robust (Table 11.2). However, there was no evidence of associations between number of hospital admissions and being non-eligible for EIS in models 1–3, but a weak evidence of association

emerged when demographic, socioeconomic status and diagnosis were controlled for in model 4 (adj. IRR = 2.09; 95% CI 0.87 - 5.04, p = 0.09).

	Model 1	95% confidence interval (p-value)	Model 2	95% confidence interval (p-value)	Model 3	95% confidence interval (p-value)	Model 4	95% confidence interval (p-value)
EIS use								
EIS	1.00		1.00		1.00		1.00	
Non-EIS	2.35	1.47 – 3.73 (<0.01)	3.32	2.05 – 5.37 (<0.01)	3.04	1.85 – 4.99 (<0.01)	3.30	1.98 – 5.49 (<0.01)
Non-eligible	0.69	0.42 – 1.13 (0.14)	1.96	0.83 – 4.61 (0.12)	1.77	0.75 – 4.20 (0.19)	2.09	0.87 – 5.04 (0.09)

Table 11.2: Unadjusted and adjusted rate ratios for number of hospital admission and EIS use (n=366)

Model 1 – unadjusted; Model 2- adjusted for ethnicity, age, gender; Model 3- adjusted for ethnicity, age, gender, level of education and

employment status; Model 4- adjusted for ethnicity, age, gender, level of education, employment status and diagnosis

11.4 Associations between number of compulsory admissions and EIS use

during follow-up

In examining associations between number of compulsory admissions and early intervention service use, negative binomial regression analysis was performed. Univariate and adjustment for potential confounders were carried out as follows:

Model 1 – unadjusted

Model 2 – adjusted for age, gender and ethnicity

Model 3 - adjusted for age, gender, ethnicity, level of education and employment status

Model 4 – adjusted for age, gender, ethnicity, level of education, employment status and diagnosis

Table 11.3 shows the results for each model. In the univariable model, there was a weak evidence of association with increased compulsory admission and the non-EIS group. However, in the subsequent multivariable models, compared with those who received EIS, there was strong evidence of associations between number of compulsory admissions and individuals in the non-EIS group with increasing effect size particularly from (adj. IRR = 1.87; 95% CI 1.02 - 3.44) in model 3 to (adj. IRR = 2.22; 95% CI 1.18 - 4.18) in model 4. There was no evidence of association between compulsory admission and individuals who were not eligible for EIS.

	Model 1	95% confidence interval (p-value)	Model 2	95% confidence interval (p-value)	Model 3	95% confidence interval (p-value)	Model 4	95% confidence interval (p-value)
EIS use								
EIS	1.00		1.00		1.00		1.00	
Non-EIS	1.75	0.94 -3.26 (0.07)	1.89	1.04 - 3.44 (0.03)	1.87	1.02 – 3.44 (0.04)	2.22	1.18 - 4.18 (0.01)
Non-eligible	0.93	0.51 – 1.69 (0.81)	1.12	0.40 - 3.11 (0.82)	1.08	0.39 – 2.93 (0.87)	1.23	0.44 – 3.76 (0.68)

Table 11.3: Unadjusted and adjusted rate ratios for number of compulsory admission and EIS use (n=366)

Model 1- unadjusted; Model 2- adjusted for ethnicity, age, gender; Model 3- adjusted for ethnicity, age, gender, level of education and

employment status; Model 4- adjusted for ethnicity, age, gender, level of education, employment status and diagnosis

11.5 Associations between length of hospital stay and EIS use during followup

The next stage of analysis sought to assess whether the use of an early intervention service was independently associated with length of hospital stay after controlling for potential confounders. In keeping with the analytical methods used in section 10.8 (chapter 10), negative binomial regression was employed here to estimate the rate ratios for length of hospital stay by EIS status.

Four unadjusted and adjusted regression models were fitted as follows:

Model 1 – unadjusted

Model 2 – adjusted for age, gender and ethnicity

Model 3 – adjusted for age, gender, ethnicity, level of education and employment status

Model 4 – adjusted for age, gender, ethnicity, level of education, employment status, and diagnosis

In Table 11.4, both the univariable and multivariable analyses showed that there were no associations between length of hospital stay and non-EIS nor non-eligible groups compared with those who used early intervention service (EIS group).

	Model 1	95% confidence interval (p-value)	Model 2	95% confidence interval (p-value)	Model 3	95% confidence interval (p-value)	Model 4	95% confidence interval (p-value)
EIS use								
EIS	1.00		1.00		1.00		1.00	
Non-EIS	1.37	0.44 – 4.19 (0.57)	1.59	0.53 – 4.72 (0.25)	1.56	0.52 – 4.65 (0.41)	1.66	0.61 - 4.48 (0.31)
Non-eligible	0.50	0.24 – 1.05 (0.07)	0.38	0.05 – 2.68 (0.33)	0.28	0.02 – 2.69 (0.27)	0.31	0.03 – 2.64 (0.28)

Table 11.4: Unadjusted and adjusted rate ratios of length of stay during hospital admission and EIS use (n=366)

Model 1 – unadjusted; Model 2- adjusted for ethnicity, age, gender; Model 3- adjusted for ethnicity, age, gender, level of education and

employment status; Model 4- adjusted for ethnicity, age, gender, level of education, employment status and diagnosis

11.6 Associations between ethnicity and service use outcomes by early intervention service status

This stage of the analyses addressed the objective of whether there are ethnic differences in service use outcomes at follow-up in relation to early intervention service use status. Up until this stage, the data have been stretched and manipulated to consider predictors of outcomes and in particular factors contributing to ethnic differences in all the outcomes considered thus far. As the number of statistical tests increases, the possibility of observing associations that are due to chance also increases. This caveat is already apparent in relation to the analysis of factors associated with ethnic variations in the pathways to care stratified by early intervention service use status (Chapter 9). On the basis of this limitation, the analyses for this final stage employ a straightforward test of association between ethnicity and service use outcomes by EIS status using Kruskal Wallis tests of association. The intention is to explore whether and where there are ethnic differences in the three subgroups of EIS status.

11.7 Associations between ethnicity and number of hospital admissions by early intervention service status

Table 11.5 shows there was strong evidence that in the EIS group, Black African and 'Mixed' ethnic group patients were more likely to experience multiple hospitalisations, compared with White British patients (IQR (0-2); (0-1.5) and (0-1) respectively $X^2 = 13.71$, df = 6, p = 0.03). However, Asian patients were less likely to experience hospital admission in the EIS group. In the non-eligible group, there was weak evidence of more hospital admissions among White British and 'Mixed' ethnic group patients and no hospital admissions among the 'Other' ethnic group (Table 11.5). Meanwhile, ethnic variations for hospital admissions were not evident in the non-EIS group.

	Non-	ELIGIBLE			Non- EIS		EIS		
	N	=215			N= 119				
								N=222	
	Number in	Median (IQR)	Kruskal	Number in	Median (IQR)	Kruskal	Number in	Median (IQR)	Kruskal Wallis
	sample (%)		Wallis test;	sample (%)		Wallis test;	sample (%)		test; df (p-
			df (p-value)			df (p-value)			value)
White	55 (25.6)	0 (0 -1)	10.7; 6 (0.09)	23 (19.3)	0 (0 -1)	7.22; 6 (0.30)	55 (24.8)	0 (0 -1)	13.71; 6 (0.03)
British									
Black	52 (24.2)	0 (0)		25 (21.0)	0 (0 -2)		70 (31.5)	0 (0 -2)	
African									
Black	38 (17.7)	0 (0)		17 (14.3)	0 (0 -1)		35 (15.8)	0 (0 -1)	
Caribbean									
White	35 (16.3)	0 (0)		16 (13.4)	0 (0-1)		24 (10.8)	0 (0 -0.5)	
Other									
Asian	14 (6.5)	0 (0)		15 (12.6)	0 (0 -1)		14 (6.3)	0 (0-0)	
Mixed	7 (3.3)	0 (0 -1)		8 (6.7)	1 (0 -1)		12 (5.4)	0 (0 -1.5)	
Other	14 (6.5)	0 (0)		15 (12.6)	0 (0)		12 (5.4)	0 (0 -0.5)	

Table 11.5: Associations between ethnicity and number of hospital admissions by use of El services

11.8 Associations between ethnicity and number of compulsory admissions by early intervention service status

Table 11.6 presents associations between ethnicity and number of compulsory admissions by EIS status. There was strong evidence that elevated risk of compulsory admissions was present for Black African patients compared with White British patients in the EIS group (IQR (0 -3) vs. (0 -1), $X^2 = 15.03$ df = 6 p = 0.01). Asian patients in the EIS group were less likely to be compulsorily detained. In the non-EIS group, the evidence (although weak) also suggests that Black African patients were more likely to experience multiple compulsory admissions compared with White British patients (IQR (0-4) vs. (0-1) $X^2 = 11.50$, df = 6 p = 0.07). Meanwhile, 'Other' ethnic group patients in the non-EIS group did not experience compulsory admission. There was no evidence of ethnic differences for compulsory admission among patients in the non-eligible group.

	No	n- ELIGIBLE			Non- EIS		EIS		
		N=215			N= 119				
							N= 222		
	Number	Median (IQR)	Kruskal Wallis	Number in	Median (IQR)	Kruskal	Number in	Median (IQR)	Kruskal Wallis
	in sample		test; df (p-	sample (%)		Wallis test;	sample (%)		test; df (p-
	(%)		value)			df (p-value)			value)
White	55 (25.6)	0 (0 -1)	1.79; 6 (0.93)	23 (19.3)	0 (0 -1)	11.50; 6	55 (24.8)	0 (0 -1)	15.30; 6 (0.01)
British						(0.07)			
Black	52 (24.2)	0 (0 -1)		25 (21.0)	1 (0 -4)		70 (31.5)	1 (0 -3)	
African									
Black	38 (17.7)	0 (0- 2)		17 (14.3)	0 (0 -1)		35 (15.8)	0 (0 -2)	
Caribbean									
White	35 (16.3)	0 (0 -1)		16 (13.4)	0 (0 -1.5)		24 (10.8)	0 (0 -2)	
Other									
Asian	14 (6.5)	0 (0 -1)		15 (12.6)	0 (0 -1)		14 (6.3)	0 (0)	
Mixed	7 (3.3)	0 (0 -1)		8 (6.7)	0 (0 -2)		12 (5.4)	0.5 (0 -1.5)	
Other	14 (6.5)	0 (0)		15 (12.6)	0 (0)		12 (5.4)	0 (0- 1.5)	

Table 11.6: Associations between ethnicity and number of compulsory admissions by use of EI services

11.9 Associations between ethnicity and length of hospital stay by early intervention service status

Table 11.7 shows the associations between ethnicity and length of hospital stay by EIS status. In the EIS group, there was strong evidence that compared with White British, Black African and Black Caribbean patients experienced longer duration of hospital stay (IQR (0 - 14) vs. (1 -97), (0 -98) respectively, $X^2 = 17.83$ df = 6 p<0.01). In both non-EIS and non-eligible groups, ethnic differences in length of hospital stay were not evident (Table 11.7).

	Non- I	ELIGIBLE			Non- EIS		EIS		
	N=	=215			N= 119				
							N= 222		
	Number in	Median (IQR)	Kruskal	Number in	Median (IQR)	Kruskal	Number in	Median (IQR)	Kruskal Wallis
	sample (%)		Wallis test;	sample (%)		Wallis test;	sample (%)		test; df (p-
			df (p-value)			df (p-value)			value)
White	55 (25.6)	5 (0 -50)	6.82; 6 (0.33)	23 (19.3)	7 (0 -60)	4.48; 6 (0.59)	55 (24.8)	14 (0-14)	17.83; 6
British									(<0.01)
Black	52 (24.2)	2 (0 -53.5)		25 (21.0)	29 (2 -157)		70 (31.5)	46.5 (1 -97)	
African									
Black	38 (17.7)	0 (0 -50)		17 (14.3)	29 (0 -106)		35 (15.8)	28 (0-98)	
Caribbean									
White	35 (16.3)	15 (0 -32)		16 (13.4)	7 (0-51.5)		24 (10.8)	8.5 (0 -38.5)	
Other									
Asian	14 (6.5)	24 (1 -63)		15 (12.6)	12 (2 -29)		14 (6.3)	4 (0 -22)	
Mixed	7 (3.3)	19 (0-137)		8 (6.7)	17.5 (3-39.5)		12 (5.4)	6 (0 -95)	
Other	14 (6.5)	0 (0-11)		15 (12.6)	8 (0 -28)		12 (5.4)	0 (0 -38.5)	

Table 11.7: Associations between ethnicity and length of hospital stay by use of EI services

11.10 Summary

In summary, the analyses carried out in the first part of this chapter revealed that independent of potential confounders, patients who were eligible for EI services but who did not access them (non-EIS), were at increased risk of multiple hospital admissions and compulsory detentions compared with the EIS group. There were no differences in length of hospital stay by EIS status.

The stratified analysis by EIS status for ethnic differences shed some light on where associations between ethnicity and service use outcomes lie within the EIS strata. Compared with White British patients, Black African patients who accessed EI services experienced poor outcomes in the three domains examined (hospital admission, compulsory admission and length of stay). There was also strong evidence that Black Caribbean patients in the EIS group experienced longer duration of hospital stay. Meanwhile, 'Mixed' ethnic group patients were also more likely to be compulsorily detained in the EIS group.

Interestingly, ethnic differences for service use outcomes were not evident in the non-EIS users or among those who were not eligible for EIS.

In conclusion, the evidence shown in this chapter partly supports the hypothesis that patients in the non-EIS group would have poorer service use outcome. There was also a robust evidence of ethnic variations in the service use outcomes for the EIS group, which appears to suggest that EIS may actually not have influenced patient outcomes by ethnicity. The implications of these findings will be discussed in more detail in Chapter 12.

11.11 Limitations

As well as the issue of attrition, another key limitation of the follow up results is multiple testing. As discussed in section 11.6, increasing numbers of statistical tests increase the likelihood of finding a statistically significant difference due to chance. Within this chapter, multiple tests were conducted to assess associations between service outcomes and EIS membership, but a simple descriptive analysis was employed to explore the ethnic differences in service use outcomes. This means that only the p value threshold was used to assess likelihood of associations between the outcomes and ethnicity. Therefore, firm conclusions cannot be drawn from the findings without formally testing the hypothesis that outcome would differ by ethnic group and by EIS membership.

12 Chapter 12: General Discussion and Methodological Consideration

12.1 Synopsis

In this chapter, after summarising the key findings in relation to the hypotheses that were set out in Chapter 1, I will address the methodological strengths and limitations of the study. This will be followed by methodological considerations put in place to mitigate potential shortcomings. I will then attempt to infer possible explanations, draw together the key findings and offer a broad interpretation of the results in relation to previous studies. Finally, a discussion of some possible implications and future research on ethnicity, service use, course and outcome and psychosis will follow.

12.2 Summary of findings

This thesis tested a number of hypotheses about the incidence of psychosis, pathways to care, service use outcome and the role of early intervention services and association with ethnicity. The following hypotheses were supported by the results:

Incidence of psychosis

- Incidence of psychosis was higher among Black African Africans (SIR = 120.8 per 100,000 PPY) and Black Caribbean (SIR = 93.6 per 100,000 PPY) ethnic groups compared with the White British (SIR = 39.4 per 100,000 PPY) population.
- The magnitude of risk of psychosis was lower for Black Caribbeans (adj. IRR = 2.81; 95% CI 2.15 3.63) than those reported in the AESOP study (adj. IRR = 6.7; 95% CI 5.4 8.3), but little or no change for the Black African population (adj. IRR = 3.59; 95% CI 2.80 4.55) compared with the AESOP study (adj. IRR = 4.1; 95% CI 3.2 5.3).

Pathways to care characteristics at first contact for psychosis

- Increased risk of hospital admission at first contact for psychosis was associated with Black African ethnicity (adj. OR = 2.38; 95% CI 1.27 – 4.42).
- Increased risk of compulsory admission at first contact for psychosis was associated with Black African ethnicity (adj. OR = 3.23; 95% CI 1.57 – 6.63).

- Ethnic differences in GP referral were smaller for Black African (non-EIS users: adj. OR = 0.37; 95% CI 0.07 2.00; EIS users: adj. OR = 0.57; 95% CI 0.20 1.62) and Black Caribbean (non-EIS users: adj. OR = 0.18; 95% CI 0.01 2.47; EIS users: adj. OR = 1.20; 95% CI 0.37 3.87) ethnic groups compared with findings from 15 years ago (the AESOP study).
- Ethnic differences in criminal justice referral were smaller for Black African (non-EIS users: adj. OR = 0.26; 95% CI 0.01 4.09; EIS users: adj. OR = 1.35; 95% CI 0.35 5.20) and Black Caribbean (non-EIS users: no cases; EIS users: adj. OR = 1.06; 95% CI 0.16 7.01) ethnic groups compared with findings from 15 years ago (the AESOP study).

Service use outcomes at two-year follow-up

- Higher rates of hospital admission were associated with Black African (adj. IRR = 1.98; 95% Cl1.12 3.48) ethnicity compared with White British at the two-year follow-up.
- Higher rates of compulsory admission were associated with Black African (adj. IRR = 2.27; 95% Cl 1.18 4.34) ethnicity compared with White British at the two-year follow-up.
- Worse service use outcomes were associated with non-EIS status, hospital admission (adj. IRR = 3.30; 95% CI 1.98 5.49) and compulsory admission (adj. IRR = 2.22; 95% CI 1.18 4.18) compared with the EIS group at the two-year follow-up.

The following hypotheses were not supported by the results:

- Increased risk of hospital admission at first contact for psychosis was not associated with Black Caribbean ethnicity (adj. OR = 1.20; 95% CI 0.57 – 2.49).
- Increased risk of compulsory admission at first contact for psychosis was not associated with Black Caribbean ethnicity (adj. OR = 1.78; 95% CI 0.75 – 4.24).
- Increased risk of police/criminal justice referral was not associated with Black African (adj. OR = 1.05; 95% CI 0.43 2.57) or Black Caribbean (adj. OR = 1.28; 95% CI 0.43 3.77) ethnicity.

- Reduced likelihood of GP referral was not associated with Black African (adj. OR = 0.63; 95% CI 0.33 1.19) or Black Caribbean (adj. OR = 0.54; 95% CI 0.35 1.15) ethnicity.
- Ethnic differences in accident and emergency referral were actually evident and larger for Black African (adj. OR = 7.34; 95% Cl 1.15 – 46.74) and Black Caribbean (adj. OR = 48.89; 95% Cl 3.49 – 684.71) ethnic groups compared with findings from 15 years ago (the AESOP study).
- Higher rates of hospital admission at follow-up were not associated with Black Caribbean (adj. IRR = 1.13; 95% CI 0.58 – 2.19) ethnicity compared with White British ethnicity.
- Higher rates of compulsory admission at follow-up were not associated with Black Caribbean (adj. IRR = 1.27; 95% CI 0.59 – 2.74) ethnicity compared with White British ethnicity.
- Longer length of hospital stay at follow-up was not associated with Black African (adj. IRR = 1.67; 95% CI 0.29 – 9.52) or Black Caribbean (adj. IRR = 0.0.49; 95% CI 0.04 – 4.98) ethnicity.

12.3 Methodological Strengths and Limitations

12.3.1 Strengths

In identifying first episode psychosis cases using the Clinical Record Interactive Search database (CRIS) and collecting a wide range of data relating to risk factors for psychosis and service use, a range of important research questions were addressed. Since the completion of the AESOP study, this study (CRIS-FEP) is one of the largest studies of first episode psychosis.

The key strengths of this study are its large sample size, comprehensive case identification strategy, the involvement of clinicians in screening clinical records using a standardised screening schedule (PSS, Jablensky et al. 1992), inter-rater reliability tests across the measurements and 'real time' information from the case register. Each of these will be discussed in more detail below. Furthermore, the availability of AESOP data on pathways to care allowed for comparison of two datasets collected in the same study area at different

time points to assess change over time in pathways to care and service use characteristics by ethnic groups.

• Comprehensive case identification

To our best knowledge, this is one of few studies to manually screen clinical records for symptoms of psychosis, carefully elicit date of onset and match cases' resident address to the study catchment area. The three-stage screening and monthly consensus meetings with the principal investigator ensured that only cases that met the inclusion and exclusion criteria were included in the sample. In this study, more people with clear psychotic symptoms were identified; we found a higher incidence rate of psychosis compared with two previous studies that have estimated incidence of psychosis among ethnic minority groups (Fearon et al., 2006a, Coid et al., 2008b). Importantly, this thesis presents a marked improvement on other case ascertainment, e.g. face-to-face first contact methods as used in previous studies (Coid et al., 2008b, Cheng et al., 2011b).

• Large sample size

This study was conducted using the same inclusion/exclusion criteria and within the same catchment area of south London as the AESOP study over a decade ago. We had a large sample that is representative of the catchment area population of patients that present to mainstream mental health services. Data on the 18 ethnic group categories of the 2011 Census were collapsed into seven groups for clarity of presentation. Doing so provides an improvement on previous studies which tend to merge ethnic groups crudely, e.g. Black, White, Asian etc. The size of the sample in this study enabled a more fine-grain classification of ethnic groups, e.g. separating out Black African and Black Caribbean. In addition, White British instead of an all-White group was used as the reference group in the multivariable analysis. By using the 2011 Census data, which was closer to the time of this study and which accounted for under-enumeration in certain minority ethnic groups, it is believed that the most accurate denominator was used, particularly in estimating the incidence rates of psychosis.

Human resources and hours spent by experts in identifying first episode psychosis cases

In order to undertake an endeavour such as screening clinical records to identify patients with psychotic features who met the well-defined inclusion criteria in this study, some clinical expertise is required for accuracy. The screening exercise was undertaken by clinically trained personnel: me – a psychiatric nurse, another PhD candidate – a psychiatrist, a foundation medical doctor, an experienced clinical researcher and the principal investigator. In total, four and a half years were spent screening, extracting and validating demographic, socioeconomic, pathways to care, service use and clinical data. While this is a resource-intensive and time-consuming approach, it offers confidence in data validity and integrity as will be discussed next. In addition, the manual annotation approach in this thesis is an improvement on other case register studies that tend to rely on clinically recorded diagnoses or other clinical features which may vary according who is doing the recording and may not be recorded at all.

• Reliability scores

Researchers undertaking the screening exercise in this study received training on all the screening and data collection measures. Inter-rater reliability was tested between researchers at each stage of the data collection and high reliability scores were achieved as discussed and shown in section 6.5.4.

• Real-time information

Since the CRIS database pulls data daily and directly from the SLaM electronic health record, it provides valuable real-time information on routine mental healthcare. This automatically generates and accumulates large volumes of data ready for research interrogation.

12.3.2 Overall limitations

As discussed in section 7.6, inaccurate ethnic classification will lead to inaccurate findings. The current ethnic classification is based on an imperfect categorisation that combines people who may have very different cultural backgrounds (e.g 'black Africans come from several African countries and indeed, tribes within countries each with different social and

cultural traditions). This inevitably can only lead to imperfect findings but is an issue of the wider society in general, not just a concern of this thesis.

Specific to the incidence investigation, is an issue surrounding diagnostic accuracy, particularly within a case register. Several studies have used case registers to estimate the incidence of psychosis, but many have done so based on the recorded clinical diagnosis with no attempt to check the likely accuracy of the diagnosis in terms of a detailed examination of the constituent symptoms and signs in the case notes (Smith et al., 2006, Kennedy et al., 2004, Boydell et al., 2012). In the present study, diagnosis was based on manually screening the detailed clinical records for symptoms of psychosis, carefully eliciting the date of onset and matching each case's resident address to the study catchment area. Of course even this is not perfect as the symptoms and signs of illness that go to make up the diagnosis are not assessed using standardised rating instruments as used in epidemiological surveys, but are instead based on the appraisal of different clinicians of varying levels of expertise. Further, if information is not recorded, it is not known whether the symptoms was genuinely absent or was missed by the clinician.

Despite the large sample size and denominator of 852,920 person-years at risk, the high rate ratios observed in some smaller ethnic groups may be due to chance, given the small sample size and wide confidence intervals in those groups. This also applies to the associations seen in these groups and pathways to care and service use outcomes.

In addition, in spite of the comprehensive case finding in this study, it is possible that some cases may be missed as we have only included people who sought help from publicly funded mental health services within the catchment area and have not accounted for those who may have accessed privately funded services. It is also possible that our search criteria were not exhaustive and the screening was not 100% watertight. However, the possibility of missing cases owing to our inception cohort design was minimal because people presenting with major mental health issues such as psychosis tend to present to statutory specialist mental health services directly or are referred by their general practitioners or emergency departments to specialist mental health services. This means that although we did not identify cases through other sources, we are much more confident about the quality of our inclusion since SLaM is the only mental healthcare provider for our study catchment area

and CRIS holds multi-disciplinary clinical information for services both in the community and hospitals (Perera et al., 2016).

As with pathways to care and compulsory admission investigation, the analyses were based on cross-sectional data, therefore direction of causality cannot be inferred. Further within the pathways to care investigation, crude dichotomies have been used in some variables to simplify analyses. These include criminal justice agency (yes) – 'police/courts/prisons' versus none. These dichotomies are likely to over simplify the true nature of criminal justice involvement in care pathway by ethnicity. However, simplifications at this stage are likely to clarify the areas on which future research are focussed.

Specific to the outcomes investigation, is an issue surrounding attrition. Within this thesis a sizeable number of participants had left SLaM services by the end of the study. Had this study been a face-to-face survey, where participants had to be re-contacted for follow up data, the impact of the attrition would have been greater in this study. However, owing to the ability to take full advantage of the clinical records (Morrison et al., 1997), particularly information on dates of discharge for each participant, the whole sample was included in the analyses.

12.4 Methodological considerations

In all research studies inference is a key activity, which is a process of passing from observations to generalisations (Prince et al., 2003). In epidemiology, an observed association between an exposure and outcome does not mean one caused the other. However, before the direction of causality can be considered the role of bias, chance and confounding must be assessed. Each will be considered in turn below in relation to this study.

12.4.1 The role of bias

In order to ensure a sample is representative, it is important the selection process is random, so that the sample does not encounter systematic errors that vary between exposure and outcome variables. Selection bias (occurring at study design phase) and information bias (occurring at data collection phase) are the two main sources of nonrandom bias.

12.4.2 Selection bias

Selection bias occurs when a sample includes or excludes certain people systematically and the differences in the characteristics of those selected are related to the exposure and outcome of interest (Susser et al., 2006).

In this study, efforts were made to reduce selection bias by using the Clinical Record Interactive Search (CRIS) system, which allowed us to comprehensively interrogate clinical records to screen and identify first episode psychosis cases presenting to any adult mental health service both in hospital inpatients and community-based services, including those presenting to psychiatric liaison services in accident and emergency departments. Our included cases had to exhibit evidence of psychotic symptoms, and scored at least 2 or more for psychotic symptoms as assessed by the Psychosis Screening Schedule (Jablensky et al., 1992a). The screening procedure was conducted on a weekly basis which provides a prospective observation of point of contact of new episode of psychosis cases. This also reduces the possibility of missing cases or of differences between ethnic groups occurring since our screening procedures were based on psychopathology over the course of the study period; therefore, it is unlikely that clear psychotic symptoms in any ethnic group would not have been recorded or missed in our search strategy. Further, our screening approach ensured that all psychosis diagnostic spectrums were included rather than focussing on specific diagnoses since there is evidence that psychiatric diagnoses are unstable over time (Heslin et al., 2015). Another form of selection bias is attrition. The completeness of data is important to the success of a cohort study, but inherent in any kind of cohort study is some degree of attrition of participants. In this study, around 60% of the sample had left by the end of the study. However, we were able to determine date of discharge for each individual and therefore the amount of data each participant contributed to the follow-up period were included in the analyses.

12.4.3 Information bias (observer)

In cohort studies information bias may arise due to random error in information. To reduce information bias 'blind' ratings are considered most desirable (Lee et al., 2007). However, this is not a plausible method for examining the influence of ethnicity on the risk of psychosis, pathways to care and outcome using clinical records. Careful considerations were

taken in this study to ensure misclassification of exposure and outcome were kept to a minimum. To minimise misclassification in the outcome data, researchers underwent intensive training for screening manually for psychosis as well as other data collection schedule at the outset of the study. A three-stage manual screening procedure was employed in identifying first episode psychosis cases as described in Chapter 6 (section 6.5.2). Further, to reduce information bias in the exposure data, careful steps were taken to ensure consistency in the ascription of ethnic groups as discussed in the Methods chapter 6 (section 6.5.4).

12.4.4 Chance

In this thesis, despite the relatively large sample size of 558 patients but given the multiple testing of associations between ethnicity and a number of outcomes (i.e. incidence of psychosis, compulsory admissions at first episode, as well as clinical and service use outcome hospital admission, length of stay), it is still possible that some positive associations (p<0.05) observed may have arisen due to chance. For instance, in estimating the age-specific incidence rate ratios by ethnic groups, the small numbers of patient of Asian, Mixed and Other ethnic groups may have influenced the observed elevated risk as the range of confidence intervals were very wide for these groups. However, evidence of elevated risk of psychosis among Black African and Black Caribbean is less likely to be due to chance given the narrow confidence intervals. Further, some findings in this study are consistent with the wider literature. However, some of the associations observed may be due to chance. For example, data was complete for 367 patients, therefore for some analyses statistical power was slightly reduced and raises the question of biases in the estimates of associations. However, where necessary, data were analysed in two ways a) on the whole sample and b) on the 367 people for whom data on all variables were complete. This helped in assessing the impact missing data may or may not have on the observed associations. The associations we observed between female patients of Asian and 'Mixed' ethnic groups and compulsory admissions at FEP were imprecise due to the small sample size in these groups. Therefore, only tentative conclusions can be drawn from the results for these groups. More research with larger sample sizes particularly for Asian, Mixed and Other ethnic group is needed for replication of these results.
12.4.5 Confounding

A-priori sociodemographic and socioeconomic confounders (age, gender, and employment status and education level) were adjusted for in this study with the exception of incidence of psychosis in which only age and gender were controlled for. In line with previous studies, associations between ethnicity, incidence of psychosis, compulsory admission, hospital admission and length of hospital stay were investigated with and without adjustment for the *a priori* confounders. Furthermore, other variables identified in the literature as potential confounders in relationship between ethnicity and outcomes of interest namely, diagnosis, previous service use and duration of untreated psychosis were selected in this study as potential confounders

Meanwhile it is possible that we did not identify or measure other potential confounders in this study which may distort the validity of positive associations observed. For example, a key limitation of the incidence analysis was in the lack of data on socioeconomic status for our denominator sample and therefore did not adjust for them (such as employment status, education level, income level etc.) which may possibly explain differences in incidence rates. Previously, where investigators have adjusted for socioeconomic factors, this made little difference to the estimates. For example in the East London First Episode study, the authors adjusted for socioeconomic variables but these did not explain the elevated rate of psychosis observed in the minority ethnic group patients (Kirkbride et al., 2008). Similarly in a Swedish study of all persons admitted for schizophrenia, where socioeconomic status was controlled for, the elevated risk of schizophrenia remained significant among immigrants (Hjern et al., 2004).

12.5 Explaining the findings and relationship with previous studies

The results from this thesis suggest that changes have occurred in the last two decades in the relationship between first episode psychosis and minority ethnic status, albeit for one group in particular – Black Caribbean. Evidence from the incidence of psychosis estimates in this study appears to suggest that while rates of psychosis were high among Black Africans, a notable reduction in magnitude of risk (rate ratios) than has previously been reported was observed among Black Caribbeans. The remainder of the results consistently show that only the Black African group were more likely to be detained compulsorily at first contact for psychosis and were more likely to have poorer outcomes at follow-up, i.e. higher numbers of hospital admissions and compulsory admissions, and they were also more likely to experience these poor outcomes despite the use of early intervention services compared with White British patients.

However, the findings from this study highlight the complexities around ethnicity and mental disorder. In interpreting the findings, the scope and depth devoted to explaining differences in the ethnic groups is uneven. This is because, although this thesis has included and reported on all ethnic groups, it is concerned with the outcomes of three key ethnic groups, i.e. White British, Black African and Black Caribbean. Therefore, it seems important to give much more in-depth explanations to these ethnic groups and so the discussion surrounding the findings is guided by the hypotheses under investigation and each will be addressed in turn.

12.5.1 Interpreting the Incidence findings and relationship to previous studies Main findings

The findings of higher incidence of first episode psychosis among Black African and Black Caribbean populations are consistent with a number of previous research studies. In this study, standardised incidence rates (SIR) of 120.8/100,000 and 93.6/100,000 persons-years (PPY) were observed among the Black African and Black Caribbean ethnic groups compared with White British (39.4/100,000). This is comparable with the incidence rates of FEP found by Coid et al. (2008) (Black Caribbean: 90.8/100,000). But in contrast to findings by both (Fearon et al., 2006a, Coid et al., 2008b), incidence rates of psychosis seem to have increased for the Black African ethnic group in this study (80.6 PPY and 73.6 PPY vs. 120.8 PPY respectively).

However there was evidence in this study that the risk of psychosis has attenuated among Black Caribbean groups but there is little or no change in the Black African groups compared with 15 years ago – but the difference was more striking for the Black Caribbean group. For example, in contrast to the AESOP and other previous studies, this study showed a lower differential rate ratio for psychosis (adj. IRR = 2.81; 95% Cl 2.15 – 3.68) for the Black

Caribbean ethnic group compared with the rate ratio of (adj. IRR = 4.2; 95% CI 3.0 - 5.8) reported by (Coid et al., 2008b); and (adj. IRR = 6.7; 95% CI 5.7 - 8.3) reported by (Fearon et al., 2006a).

Among the other ethnic minority groups included in this study, the Asian ethnic group had a modest rate of psychosis (51.9 PPY) and incidence rate ratio of 1.5. This is slightly different from two previous studies that investigated this group; Bhugra et al. (1997) reported (IR = 37.0 and IRR = 1.3895% CI 0.81 - 2.33) and Coid and colleagues who found (IR = 37.1 and IRR = 1.795% CI 1.2 - 2.3). However, the findings are similar to that reported by (Fearon et al., 2006a) (IRR = 1.5; 95% CI 0.9 - 2.4).

The 'Other' ethnic group also showed elevated rates of psychosis (126.4 PPY) compared with the White British population. Whilst this is higher than those reported in previous studies (Fearon et al. 2006), the estimate needs to be considered with caution. This is because the 'Other' group in this study comprised of any other ethnic group and some newly added ethnic groups in the 2011 Census, e.g. Arab, and so it is possible that the denominator data may not be accurate; therefore this finding may be artificially inflated if under enumeration is present in the denominator population.

Regarding age and gender, the present findings were consistent with previous studies (Cheng et al., 2011b, Coid et al., 2008b). It was observed that rates were at their peak among men between the ages 18–35 years which is in keeping with findings by (Fearon et al., 2006a, Kirkbride et al., 2009). Elevated rates were also observed among women over the age of 35 years, and late onset of psychosis has been widely reported among females, particularly those of Asian ethnic group (Bhugra et al., 1997).

Explaining the difference in incidence rates: Acculturation, generational status and ethnic density

The reduced discrepancies between Black Caribbean and White British ethnic groups may be explained by possible post-migration experiences and material position among the Black Caribbean group, e.g. acculturation, support networks, aspiration and achievement. In this study, data on sociodemographic factors showed that there were no differences between Black Caribbean and White British patients on relationship status, housing tenure, living

arrangements and social isolation variables. Several studies have shown that social networks and ethnic density are important factors in understanding elevated rates of psychosis (Boydell et al., 2001, Mallett et al., 2002). While unemployment was ubiquitous in this thesis sample, which is consistent with previous studies, it is possible that other social disadvantages experienced by minority ethnic groups (e.g. living alone, poor housing, poor education and social isolation), particularly in the post-migration phase, may have improved for the Black Caribbean population. For example, given that this study was conducted in the same catchment area as the AESOP London site, comparing sociodemographic characteristics between this study and those reported for the AESOP London site (Morgan et al. 2006) would shed some light on the changes that may have occurred. First, the proportion of Black Caribbean patients living alone have reduced from (AESOP: 56.3% to CRIS-FEP: 35.2%); being educated to a higher level have improved (AESOP: 4.9% vs CRIS-FEP: 14.7%) and being in a stable relationship also showed some improvement (AESOP: 19.4% vs. CRIS-FEP: 22.1%). In addition, there is a stream of evidence that suggests in areas where Black and minority populations make up a greater proportion of the population, their risk of psychosis was no greater than that of the majority population (Schofield et al., 2011, Das-Munshi et al., 2012, Veling et al., 2008). As previously described (Chapter 5), the Black Caribbean ethnic group (15.2%) was one of the larger groups of the minority ethnic population in Lambeth and Southwark. Therefore, it is also plausible that the Black Caribbean group may have developed community social networks that encourage mutual support, a sense of belonging and enriched social relationships through social capital and acculturation (Sharpley et al., 2001, Bhugra and Becker, 2005). The argument of acculturation, defined as 'a phenomena which results when groups of individuals from different cultures come into continuous first hand contact with subsequent changes in the original culture pattern of either one or both groups' (Bhugra 2004; Berry 1976), seems quite relevant to the Black Caribbean sample in this study. For instance, in addition to the sociodemographic changes described above, generational acculturation may have occurred in the Black Caribbean group. The data in this study showed that the majority of the Black Caribbean group were born in the UK (64.2%). Therefore, although this was not measured specifically here, it is possible that there are up to second or third generations of the Black Caribbean group in the study population and the wider UK, since migration into the UK

among the Black Caribbean population was highest post-World War II, mostly from the 1950s onwards (Coid et al., 2008b, Chamberlain, 2002).

Inversely, the above argument could not be said for the Black African ethnic group in this study. A higher rate of psychosis was found among the Black African group (SIR = 120/100,000) than was reported in some recent previous studies, such as SIR = 80.6/100,000 (Fearon et al., 2006a) and SIR = 73.6/100,000 (Fearon et al., 2006a, Coid et al., 2008b). There may be three possible explanations for this. First, it is possible that the increased rates among Black African ethnic groups may be attributable to stressors related to migration. According to the Office of National Statistics (2005), there has been an increase of 2.4% in the denominator population for Black African ethnic groups since the 2001 census (ONS, 2011a). This is an indication that a proportion of these groups may be recent migrants, and this was also reflected in this study sample, as the majority of the Black African (83.8%) group were non-UK born.

Second, indicators of social disadvantages and post-migration stresses which are associated with psychosis and commonly reported among migrant cases of psychosis may be important in understanding the elevated incidence of psychosis among the Black African ethnic group (Reininghaus et al., 2010a, Tarricone et al., 2015). Previous research suggests that recent migrants may experience more social disadvantage on arrival in the new environment since there may be tension with regard to cultural bereavement and culture shock, i.e. a discrepancy between expectations and achievement in the host country (Bhugra et al., 2011b). For instance, in this study a sizable proportion of Black African patients were educated to higher (university) level (27.1%) or had student status (14.7%), which indicates a sense of aspiration, yet the level of unemployment was high for this group (65.1%) as it was for other minority groups. This suggests a high level of underemployment and possible status loss. In addition, the gap between aspiration and achievement is further widened when the high expectations about life in the UK are not met given the limited opportunities for social participation. It is also possible that disadvantage and psychosis are symbiotic, such that disadvantage exacerbates symptoms and vice versa in a vicious cycle, the consequence of which for many Black African patients, is deep-seated psychosis and enduring social exclusion.

Third, it is possible that perhaps the level of integration into the UK culture among the Black African group is not the same as the Black Caribbean group, since there is continuous flow of migration within the Black African group, particularly to the London boroughs of Lambeth and Southwark. So this group may occupy the position of an outsider compared with the Black Caribbean group, who seem to be more established for many generations. In a study of a prospective multi-ethnic cohort of adolescents in east London, Bhui et al. (2012) suggested that cultural integration (defined by friendship with own or other ethnic groups) was associated with better mental health. Furthermore, the four-fold increase of psychosis among Black Africans in this study may be explained by their greater differences in physical or behavioural characteristics. These visible differences in the Black African ethnic group may make them more vulnerable to feeling different or excluded in the host society (Cantor-Graae and Pedersen, 2007). This argument was also demonstrated in a systematic review by (Bourque et al., 2011), which showed a higher relative risk of 4.8 of schizophrenia among migrants from countries where the majority is Black compared with the reference population.

Interestingly, an almost double rate of psychosis was observed among White British (37.8 PPY) than that reported in the AESOP study (20.2 PPY). This may be due to methodological issues in relations to accessibility to patients during the case identification for the AESOP study. For example, it may be possible that cases were more accessible in crisis or acute services than via the community services. In the AESOP study, the Black African and Black Caribbean experience included shorter duration of untreated psychosis and these patients tended to experience acute onset of psychotic episodes, hence they were often admitted to hospital or crisis resolution services compared to White British patients (Morgan et al., 2006, Ghali et al., 2012). Therefore, the number of cases identified among the White British group in the AESOP study may have been underestimated. By contrast, the case identification employed in this study points to a more comprehensive search for psychosis cases since by screening the multidisciplinary healthcare professional documented clinical records we were able to identify more persons with clear psychotic symptoms within the study catchment area.

12.5.2 Interpreting the Pathways to care characteristics at first contact for psychosis and relationship to previous studies Main findings

Findings on pathways to care at first contact for psychosis (using the whole sample) in this study showed that independent of confounders, Black African patients were two times more likely to be hospitalised and three times more likely to be compulsorily detained. Further, the increased risk of compulsory admission was present in both men and women of Black African ethnicity. Meanwhile, early intervention-specific ethnic differences in pathways to care at first contact showed that Black African patients in the EIS group were more likely to be compulsorily admitted (adj. OR = 2.25; 95% CI 1.01 – 5.01). These findings are consistent with previous studies. For example, Mann and colleagues (2014) found that in an early intervention sample of FEP, Black Africans had a five-fold increased risk of compulsory admission and four-fold risk of hospital admission at first contact. Other studies that have used broadly defined Black ethnic groups also showed increased risk of compulsory admission with effect size ranging from OR = 1.35; 95% Cl 1.13 – 1.62 (Singh et al., 2014b) to OR = 4.31; 95% CI 3.33 – 5.58 (Reininghaus et al., 2010b). Evidence from the rest of Europe also suggests a similar trend. For example, in a Dutch study Van der Post et al. (2012), having adjusted for a number of confounders, also replicated UK findings that sub-Saharan African patients are more likely to be detained compulsorily (adj. OR = 3.0; 95% CI 1.4 – 6.4) (van der Post et al., 2012).

Meanwhile, in contrast to previous studies (Morgan et al., 2005a, Lawlor et al., 2012, Corrigall and Bhugra, 2013), this study did not find associations between Black Caribbean ethnicity and compulsory admission.

The results also indicated that among other minority ethnic groups, the Mixed ethnic group were also at increased risk of hospital admission (adj. OR = 2.65; 95% Cl 0.99 – 7.09) and compulsory admission (adj. OR = 3.20; 95% Cl 1.10 – 9.26) at first contact for psychosis. Meanwhile, gender specific ethnic differences in compulsory admission showed that Asian (adj. OR = 4.18; 95% Cl 1.02 – 17.15) and 'Mixed' (adj. OR = 8.36; 95% Cl 1.67 – 41.85) women were particularly at risk of detention at first contact.

12.5.2.1 Expression of distress and help-seeking – ethnic and cultural differences

To begin with, the cultural difference in relation to the experience of mental disorder and the way in which distress is expressed varies across different cultures and ethnic groups. A vast body of literature exists on the cross-cultural theory of the conception of mental illness, particularly from the anthropological perspective. For example, in some African societies, it has been reported that people believed that illness could be caused either by diseasebearing spirits that struck suddenly and independent of an individual's behaviour or as a result of external forces entering the body, e.g. spirit, germ or disease-carrying object (Lewis, 1971). Furthermore, a review of 11 sub-Saharan African countries showed crosscultural similarities, particularly in relation to the universal belief in supernatural forces as the primary cause of mental illness (Patel, 1995). Patel's (1995) review also highlights the types of experiences and behaviours believed to characterise madness across African societies, centred on outward disturbed behaviour and on difficulty in social relationships rather than on the inner psychological and emotional dysfunction as is the case in psychiatric classifications. For the Caribbean societies, literature around beliefs about mental illness reflects African traditions as well as religious influence. For example, the African-Caribbean folk beliefs were highlighted by (Laguerre, 1987), who documented that individuals with supernatural powers (e.g. voodoo men, obeah men) are believed to be capable of working magic to cause and cure illness, including mental illness.

Kleinman (1980) proposed that taken together, illness-related beliefs form culturally influenced 'explanatory models' and that help-seeking behaviours logically follow (e.g. a person's deep sense of disturbance is interpreted as a sign of spiritual crisis, in response to which he or she prays and consults a respected religious figure). So it would seem reasonable to acknowledge that pathways to care and help-seeking behaviour are influenced by an individual's cultural appropriateness, previous experiences and attitudes towards services. Although this study did not explore the expression of distress specifically, there is substantial evidence that the process of help-seeking and pathways to care is influenced by the patient and their relatives/friends and how the illness is conceptualised (Bhui and Bhugra, 2002, Anderson et al., 2015). In this study sample, more than a third of the Black African patients were detained compulsorily at first contact for psychosis. It would appear that their model and expression of mental (psychotic) illness differ on a number of

counts. Firstly, it is possible that some behaviours and symptoms are perceived as normal or as having spiritual meanings, more than in other ethnic groups. Evidence from work carried out on mental health, culture and beliefs suggest that ethnic minority group patients were more likely to report supernatural attributes and seek help from spiritual healers (Singh et al., 2015b, Leff, 1988). This 'spiritual' perception of abnormal behaviour is likely to contribute to the delay in medical help-seeking, and it is also likely that a religious and spiritual understanding of the patient's symptoms is being sought in the intervening period until the behaviour becomes unmanageable and urgent help is required. The urgency of help required highlights the crisis state of the illness and health services are more likely to respond with coercive intervention as a way of upholding safety and offering effective treatment. Both patient and relative may become resentful of this experience and therefore this may affect future engagement with mental health services. This possibility appears to be confirmed in the follow-up findings in this study where Black African patients consistently experienced more compulsory admissions during the follow-up period of the study.

12.5.2.2 The pathways to care

In this study, data on source of referral and pathways to care suggest that the way people make contact with and seek help from mental health services may have changed compared with some previous studies. Most previous studies have reported that Black African and Black Caribbean patients were more likely to come in contact with services via the criminal justice system (Ghali et al., 2012, Mann et al., 2014b) and were less likely to be referred by their GP (Anderson et al. 2015; Ghali et al. 2014). This study is one of the few to find no ethnic differences in the criminal justice agency referral or GP referral at first contact with mental health services (Burnett et al., 1999). When the data was stratified by early intervention use, in both EIS and non-EIS groups the proportion of Black Caribbean (14.3% and 11.7% respectively) and Black African (18.6% and 12.0% respectively) patients referred via the criminal justice system were similar to those for White British (14.5% and 17.4% respectively). These proportions are much lower than those we found in the AESOP sample here, and even lower than those reported in previous studies. Evidence of lower levels of GP involvement among the Black Caribbean group was partly consistent with previous research (Bhui and Bhugra, 2002, Anderson et al., 2015). There were no associations between ethnic minority status and accident and emergency referral among the AESOP sample. However,

the findings of increased risk for criminal justice agency referral among Black African and Black Caribbean patients in the AESOP sample analysed in this thesis are consistent with other studies (Mann et al., 2014b, Singh and Grange, 2006), and so is the finding of reduced likelihood of GP referral among the same ethnic groups (Ghali et al., 2012). The findings on accident and emergency referral that showed Black African and Black Caribbean patients were more likely to be referred to mental health services via the emergency route are also consistent with previous research (Mulder et al., 2006, Anderson et al., 2010).

Interestingly, socioeconomic position and involvement of family in pathways to care appear to play a key role in explaining the increased likelihood of minority ethnic group and accident and emergency referral. Several authors have argued that careful consideration must be given to socioeconomic status in studies investigating the relationship between ethnicity and mental health (Das-Munshi et al., 2016, Nazroo, 1997, Kirkbride et al., 2008). Considering the non-EIS group in this study, unemployment and poor education were higher compared with EIS users and AESOP sample. It is possible that the previously suggested explanation of poor understanding of help-seeking among minority ethnic groups may also explain this difference (Anderson et al., 2014a, Bhui et al., 2003, Sass et al., 2009). But this may be difficult to disentangle since psychosis is linked to behavioural problems and affects relationships, and executive functioning may possibly precede unemployment or lack of education qualification, and indeed vice versa.

The widely reported reduced likelihood of GP involvement in pathway to care for Black African and Black Caribbean patients was not evident in this study either in the analyses that considered the whole sample (n = 558) or in the EIS-stratified data (n = 201). The evidence here suggests that both Black Caribbean and Black African patients are now comparable to White British patients in the involvement of their general practitioner on the pathway to care. Two explanations may be possible. First, compared with fifteen years ago, there were no early intervention services in SLaM. The introduction of EIS was anticipated to make a significant contribution to earlier, less coercive care and so improve outcomes. This is demonstrated in an RCT study by Power and colleagues (2007) in which they rolled out a GP education programme and direct access to an early intervention team (the Lambeth Early Onset Crisis Assessment Team) versus usual care. They found that GPs in the intervention group were more likely to refer patients directly to mental health services and fewer patients experienced long delays in receiving treatment. A number of other endeavours have also been implemented in raising awareness, knowledge and attitudes about mental health problems in the wider community for both medical and mon-medical professionals as part of early detection and intervention (Lloyd et al. 2015). Second, as noted before, some social changes have occurred within the population under investigation in this study and therefore some of the documented factors associated with lower levels of GP involvement in pathways to care for minority ethnic groups – e.g. lack of family involvement in help-seeking or lack of GP knowledge to recognise early psychosis (Ghali et al. 2012) – may no longer be relevant in explaining the difference in pathways to care by ethnic group.

A key source of concern is the fact that 119 patients (21.3%) in this study who were eligible for EIS did not receive it. This violates the National Institute for Clinical Excellence (NICE) (2014) recommendation that adults with a first episode of psychosis (FEP) start treatment in early intervention in psychosis (EIS) services within two weeks of referral. A key consideration is the difference in the proportions of patients with duration of untreated psychosis of more than a year, particularly among the non-EIS group, which may shed some light on possible explanations for this group not reaching early intervention services. In comparison to the AESOP sample, a long DUP in our non-EIS group was evident for 25% compared with 18%. In addition, data on type of service use for the non-EIS group showed that the majority of patients were receiving care from specialist mental health services (i.e. perinatal, mother and baby unit). This may suggest that these patients may have experienced other disorders prior to the manifestation of psychosis and therefore these services continue to treat them for initial presenting disorder as well as psychosis. This is a plausible explanation since there are mostly women in our non-EIS group. However, delays in reaching early intervention services have been previously reported. Birchwood and colleagues (Birchwood et al., 2013) in a cross-sectional study of DUP and care pathways sought to investigate DUP and their link with delays in accessing specialist EIS. In a sample of 343 patients recruited from an EIS, they found an overall mean DUP of 260 days. However, a third of the sample had a DUP greater than six months; these were patients whose first

contact with mental health services was generic care, e.g. a community mental health team. They found that delay in reaching EIS was strongly correlated with longer DUP.

12.5.2.3 Compulsory admission

Ethnicity was associated with compulsory admission both during first episode psychosis and at follow-up, independent of age, gender, education and employment status. This was partly consistent with previous research as only the Black African ethnicity was consistently associated with this outcome (Anderson et al., 2015, Bhui and Bhugra, 2002, Morgan et al., 2005a, Burnett et al., 1999). Further, the evidence in this study showed that regardless of early intervention service use, Black African ethnic group patients remained at increased risk of compulsory admissions at first contact.

It is noteworthy that the data here suggest that the disparity between White British and Black Caribbean patients seems to be narrowing as the proportions of compulsory detention between the two groups are now relatively similar (20.5% vs. 24.5% respectively). This may suggest that the factors that contributed to differences in previously reported findings such as lack of understanding of health services, distrust in mental health services and reduced family involvement may no longer be relevant to the Black Caribbean groups. For example, it is possible that Black Caribbean patients have better understanding and trust for the healthcare professionals and services and so may be persuaded to come in voluntarily. Inversely, given that the majority of Black African patients in this study sample were born outside of the UK, it could be argued that perhaps they perceive seeking help from mental health services stigmatising, unfair and discriminatory (Anderson et al., 2013a, Henderson et al., 2015, Gabbidon et al., 2014) and so they may be reluctant to come in voluntarily.

12.5.2.4 Social isolation / clinical presentation

Findings here showed that shorter DUPs go along with acute onset of psychosis, crisis source of referral and out of hours contact. In turn, this is associated with greater likelihood that the person will be detained compulsorily and in this case it is the Black African ethnic group that were more likely to be compulsorily admitted to hospital. This supports the argument about cultural beliefs about mental illness and expression of distress discussed above, indicating that Black African held beliefs differ markedly from those rooted in mental health services. That is to say that transition from folk culture into the professional health sector may be challenging for Black Africans. On the other hand, a longer duration of untreated psychosis is a reflection of an onset that goes unnoticed because the person has a limited social network.

12.5.3 Interpreting the Service Use Outcomes at follow-up and relationship to previous studies Main findings

Findings from the second stage of this study suggest that the course of first episode psychosis is characterised for Black African patients by higher rates of admissions (adj. IRR = 1.98; 95% CI 1.12 – 3.48) involving detention (adj. IRR = 2.27; 95% CI 1.18 – 4.34). This is in keeping with previous research that specifically considered Black African and Black Caribbean as separate groups (Morgan, in press) and those that crudely defined a Black group (Commander et al., 1999, Maden et al., 1999, Goater et al., 1999). By contrast, the evidence in this thesis suggests that service use outcomes for Black Caribbean patients were more benign, which goes against the original hypothesis that higher rates of hospital admissions and compulsory admission would be associated with Black Caribbean ethnicity, a hypothesis supported by only a few previous studies (McKenzie and et al., 2001, Walsh et al., 2002, Harrison et al., 1999). In addition, the hypothesis that over the two-year follow-up period, poorer clinical and service use outcomes would be associated with non-EIS (compulsory admission: adj. IRR = 2.22; 95% CI 1.18 – 4.18 and hospital admission: adj. IRR = 3.30; 95% CI 1.98 – 5.49) compared with EIS status and these will vary by ethnic groups was supported. This is also in keeping with previous studies that showed the effectiveness of EI services compared with treatment as usual (Craig et al., 2004a, Bertelsen et al., 2008a). Meanwhile, the EI-specific analysis of service use outcomes by ethnic group also revealed that for Black Africans, worse outcomes persisted, as they were more likely to have higher rates of hospital admissions (interquartile range = 0-2), compulsory admission (interquartile range = 0-3) and longer stay in hospital (median = 46.5; interquartile range = 1-97) compared with White British patients (Tables 11.7 to 11.9). For the Black Caribbean group, the findings only point to the evidence that they were more likely to have a longer hospital stay compared with White British patients.

12.5.3.1 Explaining the difference in service use outcomes: social disadvantages and changes in mental health services.

Making sense of the striking differences in the course and outcomes of psychotic disorder by ethnicity in this study may be considered in two ways. First, over a few decades, the way UK mental health services are structured and provided has changed considerably. Hospital beds have been reduced significantly and there has been a shift towards community-based treatment (Johnson et al., 2001, Tyrer, 2011, Priebe et al., 2008). A number of communitybased crisis resolution initiatives such as Home Treatment and Outreach services were introduced in the early 2000s with a view of seeing patients that would normally require acute hospital admission (Johnson et al., 2001). These services are responsible for assessing, treating and supporting people in psychiatric emergencies in their own homes and the services are available 24 hours a day. Consequently, there is focus on hospital admission prevention and speedy discharge. What this means is that for a patient to be admitted to hospital, they have to be extremely unwell, possibly posing danger to themselves and others, and therefore requiring compulsory admission. In addition, the findings in this study are consistent with findings of wider inequalities in many marginalised and disadvantage groups (Marmot, 2005). This draws attention to the social structures and processes, including institutional discrimination that shape access to material resources and health services, which in turn contribute to health outcomes. Further, the findings here on the persistent use of compulsion in Black African patients highlight the disparities in outcomes; health systems as well as social policy can play a vital role in reducing these disparities. Second, the social circumstances surrounding the individual's admission is noteworthy. As has been alluded to in the preceding sections, social network, significant others and accommodation are important in the mental health of an individual. In this study, the Black African patients (16.1%) were among those with high proportion of living circumstances that involved homeless/refuge hostel. Such accommodation can be precarious as places may be given to someone else while the patient is in hospital. In addition, reports of social isolation were common among Black African patients (53.2%). Taken together, this means that patients may spend longer in hospital while accommodation and community support are being sought.

12.6 Alternative explanations

There are further nuances in this thesis that are noteworthy. First, the reduction in the incidence rate and incidence rate ratios for the Black Caribbean ethnic group in this study (IR= 93.6 (74.0 – 113.2); IRR=2.81 (2.15 – 3.68)) compared with those reported in previous studies e.g. AESOP (IR=140.8 (114.4 – 167.2); IRR=6.7 (5.4 - 8.3)) could be explained by considering other factors pertinent to minority groups. For example, cumulative social disadvantage are reported to be associated with increased risk of risk of psychosis for both White British and ethnic minority groups but the impact of such events appears to be more pervasive among some minority groups (Morgan et al., 2008, Morgan et al., 2007). If as speculated previously that the Black Caribbean group may now have multiple generations in the UK, it is possible that, in addition to acculturation, social disadvantage may have attenuated in this group. For example, fifty-five percent of the Black Caribbean patients in this study lived with family/relatives, and living alone is one of the domains of social disadvantage (Morgan et al., 2008).

Further, data in this study suggest that there were no differences between the Black Caribbean ethnic group compared with the White British group on socio-environmental exposures (e.g. social isolation, p=0.37). This could potentially explain the attenuation in elevated incidence rates as well as better pathways to care and outcomes at follow up. For instance, it is plausible that the Black Caribbean group may have developed social networks that encourage a sense of belonging and enriched social relationships through social capital and acculturation (Sharpley et al.2001). This in turn could mean that some of the previously reported beliefs and attitudes about mental illness that are divergent from those held by professionals in mental health services (Morgan et al 2005) may actually be diminishing. This means that, within the Black Caribbean community, there may be more acceptance and recognition of mental illness as a clinical issue that need to be treated medically. Therefore, the initial engagement with mental health services may be seen as a collaboration with healthcare professionals, which then informs subsequent service engagements.

Second, the contrast in the incidence rates in the White British group IR= 39.4 (95% CI=32.2 - 46.6) compared with the earlier studies e.g. AESOP, IR= 20.2 (95% CI=17.8 - 22.7) may be explained by methodological issues regarding the case identification. Given the evidence from previous studies which suggest that compared with White British, Black patients are over-represented in psychiatric inpatients units (Tolmac and Hodes, 2004, Koffman et al., 1997) and are more likely to experience short onset of illness and in turn more likely to present to emergency room or crisis services (McKenzie et al., 2001, Morgan et al., 2006), it can be argued that relying on sampling patients from service contact alone may overestimate the incidence rate in the Black ethnic groups. Further, researchers conducting surveys (in both inpatient and community settings) rely on the goodwill of clinicians to help identify new patients presenting to services for psychosis, which inevitably influences accurate sampling and indeed estimation of incidence rate ratios. Therefore, studies relying on these methods may have inadvertently underestimated the incidence rate of psychosis in the White British group, given they are less represented in inpatient units, where such method is often used (Tolmac and Hodes, 2004). If true, this means that being the reference group, a rise or drop in the rate of incidence for the White British groups will affect the incidence rate ratios in the other ethnic groups. Also if the white group have been under identified in earlier research this could partially account for the attenuated ethnic differences in this study because of the more comprehensive case identification. There is some evidence that this might be the case. A recent study illuminated this issue, where the incidence of psychosis was estimated using data of people aged 16-35 years presenting to six early intervention services based in rural and urban areas of East Anglia, UK (Kirkbride et al., 2017). They reported an incidence rate of psychosis among White British population in urban area as 38.4 (95% CI= 32.8–44.9). This is consistent with the findings in this thesis where the White British group had an incidence rate of 39.4 (95% CI= 32.2 - 46.6).

Third, migration is arguably the single defining marker of health inequalities (Kirkbride, 2017) in epidemiological studies. Despite not controlling for migration status in this thesis, the findings highlight the vulnerability of migrants from sub-Saharan Africa, for instance they perhaps face more stressful psychosocial adversities before emigration compared with those from other regions, irrespective of refugee status (Hollander et al., 2016). Added to

this, negative post migratory experiences such as chronic social defeat and poor quality of life (Karlsen and Nazroo, 2002, Bhui et al., 2003) would also increase the risk of developing psychosis. For example, interpersonal and institutional racism experienced by migrants is deleterious to health outcomes, and previous studies have shown the interconnectedness between racism, discrimination and health inequalities (Karlsen and Nazroo, 2002, Morgan and Fearon, 2007, Chakraborty et al., 2009). Although, more research is needed on this issue. The findings here showed that the majority of Black African patients and Asian women were born outside of the UK and it is within these ethnic groups that the rates of psychosis were highest. So it is intriguing to speculate that the implication of risk factors from migration itself, such as leaving a native country and family or resettlement-related stress (Tarricone et al., 2014, Reininghaus et al., 2008, Bhugra, 2002) may be operating in these groups. Further, the higher incidence rates observed among women is consistent with previous studies (Kirkbride et al., 2016) where the median age at onset typically occurs a few years later for women as a result of a secondary peak of psychosis close to the time of menopause (Grigoriadis and Seeman, 2002). In addition, the increased risk of psychosis among the Asian women in this study could be due to some other social difficulties that have been highlighted as pertinent to this population. These include social isolation, quality of relationship with male partner and lack of English language proficiency (Talbani and Hasanali, 2000, Nilaweera et al., 2014).

Fourth, we found no evidence of an association between compulsory admissions and White Other patients in both AESOP and CRIS-FEP samples. While this finding is consistent with previous studies (Mann et al., 2014, de Wit et al., 2012, van der Post et al., 2012), it is intriguing, given the arrival of a large number of White migrants from Eastern Europe into the UK since the expansion of the EU in 2004. During the same periods Black African people were also arriving in the UK, resulting to around 2.4% increase in the Black African population in London (ONS, 2011). Yet, we found the most striking odds of being compulsorily detained in the Black African ethnic group. This is important, Black African patients may perceive seeking help from mental health services stigmatising, unfair and discriminatory as has been reported in previous studies (Anderson et al., 2013, Gabbidon et al., 2014, Henderson et al., 2015) and so they may be reluctant to come in voluntarily.

However, the association of compulsory admission with Black African ethnicity also draws attention to the social structures and processes, including institutional discrimination that shape access to material resources and health services, which in turn contribute to differences in health outcomes. Our findings of involuntary admission among Black African patients are consistent with findings of wider inequalities in many marginalised and disadvantage groups (Marmot, 2005).

Fifth, the finding that increased odds of compulsory admission were present in those accessing early intervention service suggests that unmet needs still remain in the delivery of service. In principle, early intervention service would take referrals from GPs and also reach out to groups of patients known to have difficulty accessing services, perhaps outreaching to emergency rooms in major hospitals or developing contacts with criminal justice agencies including prison services to detect and divert patients at an early stage of disorder (Jarrett et al., 2012, Fusar-Poli et al., 2013). In addition, some EIS specifically targeted the ethnic minority groups, for example one of the Lambeth Early Onset project's objectives was to reduce levels of disengagement and relapse rates among young black African and Caribbean patients (Craig et al., 2004). It is worth noting that many of these services which started as part of research projects became embedded in the wider health system and hence work-pressure and number of patients on caseloads increased. Therefore, some of the initial gains in terms of improved sensitivity to the needs of ethnic minorities may have been lost. This is reflected in the results of this study where some but not all ethnic minority groups experienced compulsory pathways to care.

Although, there was insufficient evidence of ethnic differences by DUP for the sample overall or when stratified by GP and A&E referral, people who were referred by the GP had longer DUPs, but those seeking help via A&E had shorter DUP and they were more likely to be students and employed. There are a number of possible explanations for this observation. First, it reflects the long wait to see a GP in London that is resulting in higher consultations in A&E for all conditions (Kaffash, 2017). People who are working may well choose the local A&E rather than wait for a GP appointment. An A&E consultation is also the

chosen route taken by many students. The findings might also reflect choice of recent migrants to the UK. Finally it also possibly reflects acuity itself so that those with abrupt onset and/or with socially disruptive behaviours are taken to A&E by alarmed family members.

In addition, the implication of the GP waiting times in recent years, means that the duration of untreated psychosis becomes prolonged. As has been shown previous studies, longer DUP is associated with poorer outcomes (Drake et al, 2000; Bhui et al. 2014; Addington et al. 2015). Data in this thesis supports this argument where we observed that long DUP (OR=2.60; 95%CI=1.34-5.05) was associated with compulsory admission, even among those accessing early intervention services (EIS). However, it is important to note that the majority of the non-EIS patients were already receiving care from specialist mental health services (i.e. perinatal, mother and baby unit), as shown in Chapter 11 (section 11.2.2). This suggest that by virtue of being within the mental health system, these patients will receive treatment for psychosis as soon as symptoms emerge.

Finally, as mentioned in section 10.1, earlier experiences become important in understanding subsequent service use and engagement. This is reflected in the findings for the Black African patients, whose experience of compulsory admission at first contact was sustained two years later. Some of the social factors discussed above could also explain the poor follow up outcomes in this ethnic group.

12.7 Implications for research, policy and clinical agenda

The complexity of the questions that this thesis has attempted to address and the methodology that was employed means that the work has raised many questions to be answered. However, important matters for both the development of future research and policy arise from the work.

One of the most surprising findings in this thesis related to the Black Caribbean ethnic group. First, contrary to previous research on rates of psychosis, the magnitude of risk of

psychosis among the Black Caribbean ethnic group were attenuated than previously reported (Coid et al., 2008b, Fearon et al., 2006a). Second, there was no evidence of association for problematic pathways to care at first contact and poor service use outcome at follow-up for the Black Caribbean group. This raises a concern to explore whether these findings were genuine or were as a consequence of methodological artefact. The similarities in the inclusion/exclusion criteria, ethnicity classification between this study and the AESOP study as well as the inter-rater reliability checks for ethnicity in this study (Kappa score of 0.87, p<0.001) would suggest that the finding is genuine. The next concern is to explore the reasons for over-representation of Black Caribbean patients in previous studies. One possibility is that this is a consequence of case identification (i.e. face-to-face vs. case register), which has already been discussed in section 12.5.1. A second possibility is that perhaps the sociodemographic and socioeconomic factors (such as support network, relationship, education attainment, employment opportunity) that influenced the discrepancies in pathways to care and follow-up outcomes between Black Caribbean and White British may be improving as highlighted in section 12.5.1 above.

Another set of findings in this study that were striking was the consistent poor outcome for the Black African ethnic group patients from first contact for psychosis through to two-year service use outcomes. When considered as a process, the evidence showed that experiences and outcomes at earlier stages on the pathway to care influenced outcomes at later stages. So, crisis services involvement (e.g. police /criminal justice involvement, A&E referral), acute onset of psychosis, shorter DUP as well as the presence of social isolation increase the risk of compulsory admission. The second stage of this study, in focussing on pattern of service use following first episode psychosis revealed that Black African patients often followed pathways into and through care that at each point increased the likelihood that hospital admission and compulsory admission would be necessary and that length of hospital stay would be longer.

Some changes in incidence rates may be expected at different time points due to changes in methodological approaches, healthcare delivery and population. The fact that there is evidence suggestive that incidence rates have changed across ethnic groups indicates that change may have also occurred in the association between ethnicity and the associated risks of psychosis. This raises the question about what socio-environmental exposures may have

contributed to the observed differences between Black African patients and their Black Caribbean counterparts.

Furthermore, the shifting socio-political landscape may have consequences for the health inequalities in migrant and minority ethnic groups. For example, as (Bhui, 2016) highlighted, the recent referendum election in which the UK voted to leave the European Union appears to have contributed to an increase in discrimination, prejudice and stigma with some overt expressions of hostility toward migrants and ethnic minority groups. Such experiences are well-known contributors to poor mental health.

12.8 Future research

Further research is needed to explore and examine at community level the extent of mental health awareness and what happens when seeking help. The research would need to consider the potential facilitators and barriers to social capital and cohesion, since it is at the community level that prevention of psychosis and problematic pathways to care may be avoided. Such research needs to reach out beyond the clinic to tackle fears and stigma of minority groups. For example, working with police, prisons, social and primary care sectors to shorten pathways to care and provide social engagement opportunities for families and communities is crucial.

Furthermore, the UK government has pledged to reform and improve mental health care by investing an extra £1 billion per year to put mental and physical health on equal footing, including: helping people with mental health issues stay in work, reducing mortality, enhancing quality of life, monitoring the Mental Health Act etc. These areas of reform will require continuous monitoring to assess the impact and effectiveness of the policies as well as investigate inequalities in outcomes.

The recent advancement within the CRIS database such as data linkage and natural language processing provide opportunities to harness these types of research. This would make it possible to link the cohort of individuals in this study sample to other large databases and infrastructure such as those available from the UK Data Service (UK Data Service, 2017) to investigate social outcomes. Another pioneering work that is ongoing in SLaM and linked to the CRIS system is the Consent for Contact programme (C4C). C4C was developed to facilitate recruitment of patients into mental health research studies (Callard et al., 2014,

Patel et al., 2017). Through this mechanism, it would be possible to identify and invite a proportion of this study cohort to conduct face-to-face interviews for the proposed studies.

The following areas form the possible avenues for future research:

- Social outcomes: a future study could explore whether there have been improvements in the social and economic position of this study sample, e.g. change in employment, marital status, housing/living circumstances and education.
- 2. Clinical outcomes: the poor outcomes observed in Black Africans in this study may be a consequence of poor availability of services. It is possible that social factors such as poor housing, social isolation, unemployment, racism and discrimination may well continue to affect outcomes. It is also possible that poor access to treatments such as psychological intervention may also contribute to continuous poor outcomes. Therefore, it is important to continuously monitor the course and outcomes of psychotic disorders, particularly examining ethnic differences in access to and the uptake of therapeutic services, e.g. psychological interventions and service engagement.
- 3. Physical health outcomes: a future study could investigate the effect of psychosis on physical health (e.g. diabetes, heart disease, cancers) risk factors, looking at the role of psychotropic medication and variations by ethnic group. The linkage between CRIS and the Hospital Episode Statistics (HES) will provide robust data to attempt to address this question.
- 4. A mixed methods study could be used to describe the views and experiences of people of ethnic minority group and their families in attempting to obtain assistance or treatment for psychosis symptoms. Such study could also explore patients' and families' perception of their community using social capital measures such as trust, attachment, reliance, membership of organisations and volunteering. Such study may also explore the affected individuals' and their families' experience and beliefs about mental illness and expression of distress.
- Neighbourhood effects on outcomes: evidence has accumulated that implicates socio-environmental risk factors such as ethnic density, isolation and deprivation at neighbourhood level in the aetiology of psychosis and service engagement (Kirkbride et al., 2016, Omer et al., 2016, Schofield et al., 2017, Stain et al., 2017). However,

these studies tend to be cross-sectional in nature; therefore longer-term socioenvironmental impact is lacking. Further research is needed to explore the socioenvironmental predictors of the course and outcomes of psychosis. Large-scale longitudinal follow up studies can describe service utilisation and the influence of socio-environmental factors both overall and in specific subgroups of FEP populations. They can be used to establish effectiveness of mental health policy in at least two ways: first, by demonstrating increased delivery of evidence-based intervention, and second, by demonstrating whether this results in measurable health gain.

6.

12.9 Conclusions

This thesis was able to confirm that incidence rates of first episode psychosis in the Black African and Black Caribbean populations in two catchment areas in south London are higher than those of the White British population. It also confirms previously reported findings of increased risk of compulsory admission during first contact and poor follow-up outcomes for psychosis among the Black African ethnic group. The risk of psychosis was attenuated for the Black Caribbean group compared with that reported in previous research. This is one of very few studies to have sufficient sample to investigate outcomes in these two Black ethnic groups separately.

There were some interesting similarities and differences that emerged across the three ethnic groups under investigation. More similarities were seen between White British and Black Caribbean groups in the pathways to care and service use outcomes. The main difference for the Black Caribbean group was that they were more likely to have a longer duration of hospital stay, particularly for those accessing the early intervention service. By and large, there were more differences for the Black African group, from inception through to two-year follow-up outcomes in the three key outcome domains that were investigated. The risk of psychosis, hospitalisation/compulsory admission and longer duration of hospital stay were shown to be higher among the Black Africans.

The high incidence rates in the Black Caribbean group cannot be wholly explained by migration, as a majority in this group were born in the UK, although the Black African group were more likely to have been born outside the UK. It is therefore plausible that the Black African group are perhaps subject to risk-increasing factors in the socio-environment of the host country (UK) which may be less common in their home country. So factors like social isolation, unemployment and poor housing become important possible contributory factors in the aetiology of psychosis.

It is clear from this study that ethnic minority groups as well as the majority population are heterogeneous. As Morgan & Hutchinson (2009,) asserted, 'the problem is not "in" migrant populations, it is not "in" psychiatry. The problem is "in" society.' The findings in this thesis have to a great extent supported and demonstrated this statement. It is therefore imperative that the place of social factors should be at the heart of healthcare, whether considered as contributing to aetiology and therefore a target for prevention, as indicators of population health inequalities or as healthcare themselves missing out on key outcomes for treatment.

Whilst the UK society is generally tolerant and welcoming of migrants (Holmes, 2015), some migrant groups experience social difficulties more than others, which in turn have significant impact on their mental wellbeing. Concerted efforts are needed to ameliorate health outcomes for our minority populations. The recent humanitarian crisis in Europe, the Middle East, African and parts of Asia which have contributed to an increase in the number of displaced people, asylum seekers (United Nations High Commissioner for Refugees (UNHCR), 2015), as well as the potential consequences of Brexit (Bhui, 2016) could only herald public mental health issues, which will require urgent attention. Healthcare providers need to move beyond generic health education programmes but work closely with community groups to improve pathways to care for ethnic minority patients. There are opportunities to engage with community leaders and religious groups, who have been found to be significant in how people from some minority ethnic groups seek help for mental health distress (Singh et al., 2015a). Public health programmes that raise public awareness specifically on ethnic minority issues need to be developed and evaluated empirically. We also need to harness opportunities of cross-disciplines to explore the implications of migratory adversities associated with psychosis. For example, integration of

observational data with sociological and ethnographic research would be useful to shed light on the possible pre-and-post-migratory factors that increase psychosis risk among minority ethnic populations.

REFERENCES

ADDINGTON, J. 2007. The promise of early intervention. *Early Interv Psychiatry*, 1, 294-307.

ADDINGTON, J., HEINSSEN, R. K., ROBINSON, D. G., SCHOOLER, N. R., MARCY, P., BRUNETTE, M. F., CORRELL, C. U., ESTROFF, S., MUESER, K. T. & PENN, D. 2015. Duration of untreated psychosis in community treatment settings in the United States. *Psychiatric Services*.

- AGIUS, M. & WARD, C. 2009. The epidemiology of psychosis in Luton. *Psychiatria Danubina*, 21, 508-13.
- AMERICAN PSYCHIATRIC ASSOCIATION, A. & ASSOCIATION, A. P. 1980. Diagnostic and statistical manual of mental disorders.

ANDERSON, K., FLORA, N., ARCHIE, S., MORGAN, C. & MCKENZIE, K. 2014a. A meta-analysis of ethnic differences in pathways to care at the first episode of psychosis. *Acta psychiatrica Scandinavica*.

- ANDERSON, K., FUHRER, R. & MALLA, A. 2010. The pathways to mental health care of first-episode psychosis patients: a systematic review. *Psychological medicine*, 40, 1585-1597.
- ANDERSON, K. K., FLORA, N., ARCHIE, S., MORGAN, C. & MCKENZIE, K. 2014b. Race, ethnicity, and the duration of untreated psychosis: a systematic review. *Social psychiatry and psychiatric epidemiology*, 49, 1161-1174.
- ANDERSON, K. K., FLORA, N., FERRARI, M., TUCK, A., ARCHIE, S., KIDD, S., TANG, T., KIRMAYER, L. J. & MCKENZIE, K. 2015. Pathways to First-Episode Care for Psychosis in African-, Caribbean-, and European-Origin Groups in Ontario. *Can J Psychiatry*, 60, 223-31.
- ANDERSON, K. K., FUHRER, R. & MALLA, A. K. 2013a. "There are too many steps before you get to where you need to be": help-seeking by patients with first-episode psychosis. *J Ment Health*, 22, 384-95.
- ANDERSON, K. K., FUHRER, R., SCHMITZ, N. & MALLA, A. K. 2013b. Determinants of negative pathways to care and their impact on service disengagement in first-episode psychosis. *Social psychiatry and psychiatric epidemiology*, 48, 125-136.
- ARCHIE, S., AKHTAR-DANESH, N., NORMAN, R., MALLA, A., ROY, P. & ZIPURSKY, R. 2010. Ethnic diversity and pathways to care for a first episode of psychosis in Ontario. *Schizophrenia bulletin*, 36, 688-701.
- BEBBINGTON, P. E., CRAIG, T., GARETY, P., FOWLER, D., DUNN, G., COLBERT, S., FORNELLS-AMBROJO, M. & KUIPERS, E. 2006. Remission and relapse in psychosis: operational definitions based on case-note data. *Psychol Med*, 36, 1551-62.
- BERG, J. E. & JOHNSEN, E. 2004. [Are immigrants admitted to emergency psychiatric departments more often than ethnic Norwegians?]. *Tidsskr Nor Laegeforen*, 124, 634-6.
- BERTELSEN, M., JEPPESEN, P., PETERSEN, L., THORUP, A., OHLENSCHLAEGER, J., LE QUACH, P., CHRISTENSEN, T. O., KRARUP, G., JORGENSEN, P. & NORDENTOFT, M. 2008a. Five-year follow-up of a randomized multicenter trial of intensive early intervention vs standard treatment for patients with a first episode of psychotic illness: the OPUS trial. *Archives of General Psychiatry*, 65, 762.
- BERTELSEN, M., JEPPESEN, P., PETERSEN, L., THORUP, A., ØHLENSCHLÆGER, J., LE QUACH, P., CHRISTENSEN, T. Ø., KRARUP, G., JØRGENSEN, P. & NORDENTOFT, M. 2008b. Five-year follow-up of a randomized multicenter trial of intensive early intervention vs standard treatment for patients with a first episode of psychotic illness: the OPUS trial. *Archives of general psychiatry*, 65, 762-771.
- BHOPAL, R. 2004. Glossary of terms relating to ethnicity and race: for reflection and debate. *Journal of epidemiology and community health*, 58, 441-445.

- BHUGRA, D., AYONRINDE, O., BUTLER, G., LEESE, M. & THORNICROFT, G. 2011a. A randomised controlled trial of assertive outreach vs. treatment as usual for black people with severe mental illness. *Epidemiol Psychiatr Sci*, 20, 83-9.
- BHUGRA, D. & BECKER, M. A. 2005. Migration, cultural bereavement and cultural identity. *World Psychiatry*, *4*, 18.
- BHUGRA, D., GUPTA, S., BHUI, K., CRAIG, T., DOGRA, N., INGLEBY, J. D., KIRKBRIDE, J., MOUSSAOUI,
 D., NAZROO, J. & QURESHI, A. 2011b. WPA guidance on mental health and mental health
 care in migrants. *World Psychiatry*, 10, 2-10.
- BHUGRA, D., HILWIG, M., HOSSEIN, B., MARCEAU, H., NEEHALL, J., LEFF, J., MALLETT, R. & DER, G. 1996. First-contact incidence rates of schizophrenia in Trinidad and one-year follow-up. Br J Psychiatry, 169, 587-92.
- BHUGRA, D., HILWIG, M., MALLETT, R., CORRIDAN, B., LEFF, J., NEEHALL, J. & RUDGE, S. 2000.
 Factors in the onset of schizophrenia: a comparison between London and Trinidad samples.
 Acta Psychiatr Scand, 101, 135-41.
- BHUGRA, D., LEFF, J., MALLETT, R., DER, G., CORRIDAN, B. & RUDGE, S. 1997. Incidence and outcome of schizophrenia in whites, African-Caribbeans and Asians in London. *Psychol Med*, 27, 791-8.
- BHUI, K. 2016. Brexit, social division and discrimination: impacts on mortality and mental illness? : RCP.
- BHUI, K. & BHUGRA, D. 2002. Mental illness in Black and Asian ethnic minorities: pathways to care and outcomes. *Advances in Psychiatric Treatment*, **8**, 26-33.
- BHUI, K., STANSFELD, S., HULL, S., PRIEBE, S., MOLE, F. & FEDER, G. 2003. Ethnic variations in pathways to and use of specialist mental health services in the UK Systematic review. *The British Journal of Psychiatry*, 182, 105-116.
- BHUI, K., ULLRICH, S. & COID, J. W. 2014. Which pathways to psychiatric care lead to earlier treatment and a shorter duration of first-episode psychosis? *BMC Psychiatry*, 14, 72.
- BIRCHWOOD, M. 1995. Early intervention in psychotic relapse: Cognitive approaches to detection and management. *Behaviour Change*.
- BIRCHWOOD, M., CONNOR, C., LESTER, H., PATTERSON, P., FREEMANTLE, N., MARSHALL, M., FOWLER, D., LEWIS, S., JONES, P. & AMOS, T. 2013. Reducing duration of untreated psychosis: care pathways to early intervention in psychosis services. *The British Journal of Psychiatry*, 203, 58-64.
- BIRCHWOOD, S. M. Z. I. M. 2008. Early interventions in psychosis. *Mental health: from policy to practice*, 137.
- BIRD, V., PREMKUMAR, P., KENDALL, T., WHITTINGTON, C., MITCHELL, J. & KUIPERS, E. 2010. Early intervention services, cognitive–behavioural therapy and family intervention in early psychosis: systematic review. *The British Journal of Psychiatry*, 197, 350-356.
- BLAKEMORE, K. & BONEHAM, M. 1994. *Age, Race and Ethnicity: A Comparative Approach. Rethinking Ageing Series*, ERIC.
- BOURQUE, F., VAN DER VEN, E. & MALLA, A. 2011. A meta-analysis of the risk for psychotic disorders among first- and second-generation immigrants. *Psychological Medicine*, 41, 897-910.
- BOYDELL, J., BEBBINGTON, P., BHAVSAR, V., KRAVARITI, E., VAN OS, J., MURRAY, R. M. & DUTTA, R. 2012. Unemployment, ethnicity and psychosis. *Acta Psychiatr Scand*, 23, 1600-0447.
- BOYDELL, J., OS, J. V., MCKENZIE, K., ALLARDYCE, J., GOEL, R., MCCREADIE, R. G. & MURRAY, R. M. 2001. Incidence of schizophrenia in ethnic minorities in London: ecological study into interactions with environment. *Bmj*, 323, 1336.
- BURNETT, R., MALLETT, R., BHUGRA, D., HUTCHINSON, G., DER, G. & LEFF, J. 1999. The first contact of patients with schizophrenia with psychiatric services: social factors and pathways to care in a multi-ethnic population. *Psychol Med*, 29, 475-83.
- CALLAN, A. F. 1996. Schizophrenia in Afro-Caribbean immigrants. *Journal of the Royal Society of Medicine*, 89, 253-256.

- CALLARD, F., BROADBENT, M., DENIS, M., HOTOPF, M., SONCUL, M., WYKES, T., LOVESTONE, S. & STEWART, R. 2014. Developing a new model for patient recruitment in mental health services: a cohort study using Electronic Health Records. *BMJ Open*, 4, e005654.
- CANTOR-GRAAE, E. & PEDERSEN, C. B. 2007. Risk of schizophrenia in second-generation immigrants: a Danish population-based cohort study. *Psychol Med*, 37, 485-94.
- CANTOR-GRAAE, E. & SELTEN, J. P. 2005. Schizophrenia and migration: a meta-analysis and review. *Am J Psychiatry*, 162, 12-24.
- CARDNO, A. G. & GOTTESMAN, I. I. 2001. Twin studies of schizophrenia: From bow-and-arrow concordances to Star Wars Mx and functional genomics. *American Journal of Medical Genetics*, 97, 12-17.
- CHAMBERLAIN, M. 2002. Caribbean migration: globalized identities, Routledge.
- CHENG, F., KIRKBRIDE, J., LENNOX, B., PEREZ, J., MASSON, K., LAWRENCE, K., HILL, K., FEELEY, L., PAINTER, M., MURRAY, G., GALLAGHER, O., BULLMORE, E. & JONES, P. 2011a. Administrative incidence of psychosis assessed in an early intervention service in England: First epidemiological evidence from a diverse, rural and urban setting. *Psychological Medicine: A Journal of Research in Psychiatry and the Allied Sciences*, 41, 949-958.
- CHENG, F., KIRKBRIDE, J. B., LENNOX, B. R., PEREZ, J., MASSON, K., LAWRENCE, K., HILL, K., FEELEY, L., PAINTER, M., MURRAY, G. K., GALLAGHER, O., BULLMORE, E. T. & JONES, P. B. 2011b. Administrative incidence of psychosis assessed in an early intervention service in England: first epidemiological evidence from a diverse, rural and urban setting. *Psychol Med*, 41, 949-58.
- CHEUNG, D., ROPER, L. & PURDON, S. E. 2014. Pathways to (specialized) care: patient costs and contacts en route to a first-episode psychosis clinic. *Early Interv Psychiatry*, *8*, 375-81.
- CHIEN, V. H. & COMPTON, M. T. 2008. The impact of mode of onset of psychosis on pathways to care in a hospitalized, predominantly African-American, first-episode sample. *Early Interv Psychiatry*, 2, 73-9.
- CHORLTON, E., MCKENZIE, K., MORGAN, C. & DOODY, G. 2012. Course and outcome of psychosis in black Caribbean populations and other ethnic groups living in the UK: a systematic review. *Int J Soc Psychiatry*, 58, 400-8.
- CLARKE, M. C., HARLEY, M. & CANNON, M. 2006. The role of obstetric events in schizophrenia. *Schizophrenia bulletin*, 32, 3-8.
- COCHRANE, R. & BAL, S. S. 1987. Migration and schizophrenia: an examination of five hypotheses. *Social Psychiatry*, 22, 181-191.
- CODD, E. 1998. A relational model of data for large shared data banks. 1970. *MD computing: computers in medical practice,* 15, 162.
- COID, J. W., KIRKBRIDE, J. B., BARKER, D., COWDEN, F., STAMPS, R., YANG, M. & JONES, P. B. 2008a. Raised incidence rates of all psychoses among migrant groups: Findings from the east london first episode psychosis study. *Archives of General Psychiatry*, 65, 1250-1258.
- COID, J. W., KIRKBRIDE, J. B., BARKER, D., COWDEN, F., STAMPS, R., YANG, M. & JONES, P. B. 2008b.
 Raised incidence rates of all psychoses among migrant groups: findings from the East
 London first episode psychosis study.[Erratum appears in Arch Gen Psychiatry. 2009
 Feb;66(2):161]. Archives of General Psychiatry, 65, 1250-8.
- COID, J. W., KIRKBRIDE, J. B., BARKER, D., COWDEN, F., STAMPS, R., YANG, M. & JONES, P. B. 2009. "Raised incidence rates of all psychoses among migrant groups: Findings from the East London first episode psychosis study": Erratum. *Archives of General Psychiatry*, 66, 161.
- COMMANDER, M. J., COCHRANE, R., SASHIDHARAN, S. P., AKILU, F. & WILDSMITH, E. 1999. Mental health care for Asian, black and white patients with non-affective psychoses: pathways to the psychiatric hospital, in-patient and after-care. *Soc Psychiatry Psychiatr Epidemiol*, 34, 484-91.

- COMPTON, M. T., ESTERBERG, M. L., DRUSS, B. G., WALKER, E. F. & KASLOW, N. J. 2006. A descriptive study of pathways to care among hospitalized urban African American first-episode schizophrenia-spectrum patients. *Soc Psychiatry Psychiatr Epidemiol*, 41, 566-73.
- COMPTON, M. T., GORDON, T. L., GOULDING, S. M., ESTERBERG, M. L., CARTER, T., LEINER, A. S., WEISS, P. S., DRUSS, B. G., WALKER, E. F. & KASLOW, N. J. 2011. Patient-level predictors and clinical correlates of duration of untreated psychosis among hospitalized first-episode patients. *J Clin Psychiatry*, 72, 225-32.
- COMPTON, M. T., KASLOW, N. J. & WALKER, E. F. 2004. Observations on parent/family factors that may influence the duration of untreated psychosis among African American first-episode schizophrenia-spectrum patients. *Schizophr Res*, 68, 373-85.
- COOPER, C., MORGAN, C., BYRNE, M., DAZZAN, P., MORGAN, K., HUTCHINSON, G., DOODY, G., HARRISON, G., LEFF, J., JONES, P., ISMAIL, K., MURRAY, R., BEBBINGTON, P. & FEARON, P.
 2008. Perceptions of disadvantage, ethnicity and psychosis. *British Journal of Psychiatry*, 192, 185-190.
- CORRIGALL, R. & BHUGRA, D. 2013. The role of ethnicity and diagnosis in rates of adolescent psychiatric admission and compulsory detention: a longitudinal case-note study. *J R Soc Med*, 106, 190-5.
- CRAIG, T. K., GARETY, P., POWER, P., RAHAMAN, N., COLBERT, S., FORNELLS-AMBROJO, M. & DUNN,G. 2004a. The Lambeth Early Onset (LEO) Team: randomised controlled trial of theeffectiveness of specialised care for early psychosis. *Bmj*, 329, 14.
- CRAIG, T. K., GARETY, P., POWER, P., RAHAMAN, N., COLBERT, S., FORNELLS-AMBROJO, M. & DUNN, G. 2004b. The Lambeth Early Onset (LEO) Team: randomised controlled trial of the effectiveness of specialised care for early psychosis. *Bmj*, 329, 1067.
- DAS-MUNSHI, J., ASHWORTH, M., GAUGHRAN, F., HULL, S., MORGAN, C., NAZROO, J., ROBERTS, A., ROSE, D., SCHOFIELD, P., STEWART, R., THORNICROFT, G. & PRINCE, M. J. 2016. Ethnicity and cardiovascular health inequalities in people with severe mental illnesses: protocol for the E-CHASM study. *Social Psychiatry and Psychiatric Epidemiology*, **51**, 627-638.
- DAS-MUNSHI, J., BECARES, L., BOYDELL, J. E., DEWEY, M. E., MORGAN, C., STANSFELD, S. A. & PRINCE, M. J. 2012. Ethnic density as a buffer for psychotic experiences: findings from a national survey (EMPIRIC). *Br J Psychiatry*, 201, 282-90.
- DE WIT, M. A., TUINEBREIJER, W. C., VAN BRUSSEL, G. H. & SELTEN, J. P. 2012. Ethnic differences in risk of acute compulsory admission in Amsterdam, 1996-2005. *Soc Psychiatry Psychiatr Epidemiol*, 47, 111-8.

DEPARTMENT OF EDUCATION. 2012. GCSE and Equivalent attainment by pupil characteristics in England [Online]. Available: https://www.gov.uk/government/statistics/gcse-andequivalent-attainment-by-pupil-characteristics-in-england-2010-to-2011 [Accessed].

- DEPARTMENT OF HEALTH 2002. Reform: the next 3 years' priorities and planning framework, 2003-2006. *In:* HEALTH, D. O. (ed.) *Appendix B. Department of Health.*
- DEPARTMENT OF HEALTH 2005. Delivering Race Equality in Mental Health Care: An Action Plan for Reform Inside and Outside Services and Government's Response to the Independent Inquiry into the Death of David Bennett.
- DOMINGUEZ, M. D., FISHER, H. L., MAJOR, B., CHISHOLM, B., RAHAMAN, N., JOYCE, J., WOOLLEY, J., LAWRENCE, J., HINTON, M., MARLOWE, K., AITCHISON, K., JOHNSON, S. & HODES, M. 2013. Duration of untreated psychosis in adolescents: ethnic differences and clinical profiles. *Schizophr Res*, 150, 526-32.
- DRAKE, R. J., ADDINGTON, J., VISWANATHAN, A. C., LEWIS, S. W., COTTER, J., YUNG, A. R. & ABEL, K. M. 2016. How age and gender predict illness course in a first-episode nonaffective psychosis cohort. J Clin Psychiatry, 77, e283-9.
- EHMANN, T. S., TEE, K. A., MACEWAN, G. W., DALZELL, K. L., HANSON, L. A., SMITH, G. N., KOPALA, L. C. & HONER, W. G. 2014. Treatment delay and pathways to care in early psychosis. *Early Interv Psychiatry*, 8, 240-6.

- EXCELLENCE, N. I. F. C. & HEALTH, N. C. C. F. M. 2002. Schizophrenia: core interventions in the treatment and management of schizophrenia in primary and secondary care. National Institute for Clinical Excellence.
- FARIS, H. D., H. 1939. *Mental Disorders in Urban Areas: an ecological study of schizophrenia and other psychoses,* Oxford, Nuniversity of Chicago Press.
- FEARON, P., KIRKBRIDE, J. B., MORGAN, C., DAZZAN, P., MORGAN, K., LLOYD, T., HUTCHINSON, G., TARRANT, J., FUNG, W. L., HOLLOWAY, J., MALLETT, R., HARRISON, G., LEFF, J., JONES, P. B. & MURRAY, R. M. 2006a. Incidence of schizophrenia and other psychoses in ethnic minority groups: results from the MRC AESOP Study. *Psychol Med*, 36, 1541-50.
- FEARON, P., KIRKBRIDE, J. B., MORGAN, C., DAZZAN, P., MORGAN, K., LLOYD, T., TARRANT, J., LUN ALAN FUNG, W., HOLLOWAY, J., MALLETT, R., HARRISON, G., LEFF, J., JONES, P. B., MURRAY, R. M., MUGA, F., MIETUNEN, J., ASHBY, M., HAYHURST, H., CRAIG, T., MCCABE, J., SAMELE, C., GWENZI, E., SHARPLEY, M., VEARNALS, S., HUTCHINSON, G., BURNETT, R., KELLY, J., ORR, K., SALVO, J., GREENWOOD, K., RAUNE, D., LAMBRI, M., AUER, S., ROHEBAK, P., MCINTOSH, L., DOODY, G., WINDOW, S., WILLIAMS, P., BAGALKOTE, H., DOW, B., BOOT, D., FARRANT, A., JONES, S., SIMPSON, J., MOANETTE, R., SIRIP SURANIM, P. Z., RUDDELL, M., BREWIN, J. & MEDLEY, I. 2006b. Incidence of schizophrenia and other psychoses in ethnic minority groups: Results from the MRC AESOP Study. *Psychological Medicine*, 36, 1541-1550.

FERNANDO, S. 2002. *Mental Health, Race and Culture,* China, Palgrave.

- FUNG, W. L. A., BHUGRA, D. & JONES, P. B. 2009. Ethnicity and mental health: the example of schizophrenia and related psychoses in migrant populations in the Western world. *Psychiatry*, 8, 335-341.
- FUSAR-POLI, P., MENEGHELLI, A., VALMAGGIA, L., ALLEN, P., GALVAN, F., MCGUIRE, P. & COCCHI, A.
 2009. Duration of untreated prodromal symptoms and 12-month functional outcome of individuals at risk of psychosis. *The British Journal of Psychiatry*, 194, 181-182.
- GABBIDON, J., FARRELLY, S., HATCH, S. L., HENDERSON, C., WILLIAMS, P., BHUGRA, D., DOCKERY, L., LASSMAN, F., THORNICROFT, G. & CLEMENT, S. 2014. Discrimination attributed to mental illness or race-ethnicity by users of community psychiatric services. *Psychiatric Services*.
- GARETY, P. A., CRAIG, T. K., DUNN, G., FORNELLS-AMBROJO, M., COLBERT, S., RAHAMAN, N., READ,
 J. & POWER, P. 2006. Specialised care for early psychosis: symptoms, social functioning and patient satisfaction: randomised controlled trial. *Br J Psychiatry*, 188, 37-45.
- GAYER-ANDERSON, C. 2013. *Childhood adversity, social support and psychosis.* Doctor of Philosophy, Kings College London.

GHALI, S., FISHER, H. L., JOYCE, J., MAJOR, B., HOBBS, L., SONI, S., CHISHOLM, B., RAHAMAN, N., PAPADA, P. & LAWRENCE, J. 2013. Ethnic variations in pathways into early intervention services for psychosis. *The British Journal of Psychiatry*, 202, 277-283.

GHALI, S., FISHER, H. L., JOYCE, J., MAJOR, B., HOBBS, L., SONI, S., CHISHOLM, B., RAHAMAN, N., PAPADA, P., LAWRENCE, J., BLOY, S., MARLOWE, K., AITCHISON, K. J., POWER, P. & JOHNSON, S. 2012. Ethnic variations in pathways into early intervention services for psychosis. *Br J Psychiatry*, 6, 6.

GOATER, N., KING, M., COLE, E., LEAVEY, G., JOHNSON-SABINE, E., BLIZARD, R. & HOAR, A. 1999. Ethnicity and outcome of psychosis. *The British Journal of Psychiatry*, 175, 34-42.

- GOLDBERG, D. P. & HUXLEY, P. 2003. *Mental illness in the community: the pathway to psychiatric care*, Psychology Press.
- GOULDING, S. M., LEINER, A. S., THOMPSON, N. J., WEISS, P. S., KASLOW, N. J. & COMPTON, M. T. 2008. Family strengths: a potential determinant of the duration of untreated psychosis among hospitalized African-American first-episode patients. *Early Interv Psychiatry*, 2, 162-8.
- GREATER LONDON AUTHORITY 2011. English Indecies of Deprivation: A London Perspective. HANOEMAN, M., SELTEN, J.-P. & KAHN, R. S. 2002. Incidence of schizophrenia in Surinam.

Schizophrenia Research, 54, 219-221.

- HARRISON, G., AMIN, S., SINGH, S. P., CROUDACE, T. & JONES, P. 1999. Outcome of psychosis in people of African-Caribbean family origin. Population-based first-episode study. *The British Journal of Psychiatry*, 175, 43-49.
- HARRISON, G., GLAZEBROOK, C., BREWIN, J., CANTWELL, R., DALKIN, T., FOX, R., JONES, P. & MEDLEY, I. 1997. Increased incidence of psychotic disorders in migrants from the Caribbean to the United Kingdom. *Psychological medicine*, 27, 799-806.
- HENDERSON, R., WILLIAMS, P., GABBIDON, J., FARRELLY, S., SCHAUMAN, O., HATCH, S., THORNICROFT, G., BHUGRA, D. & CLEMENT, S. 2015. Mistrust of mental health services: ethnicity, hospital admission and unfair treatment. *Epidemiology and psychiatric sciences*, 24, 258.
- HESLIN, M., LOMAS, B., LAPPIN, J. M., DONOGHUE, K., REININGHAUS, U., ONYEJIAKA, A., CROUDACE, T., JONES, P. B., MURRAY, R. M., FEARON, P., DAZZAN, P., MORGAN, C. & DOODY, G. A. 2015. Diagnostic change 10 years after a first episode of psychosis. *Psychol Med*, 45, 2757-69.
- HICKLING, F. W. & RODGERS-JOHNSON, P. 1995. The incidence of first contact schizophrenia in Jamaica. *The British Journal of Psychiatry*, 167, 193-196.
- HJERN, A., WICKS, S. & DALMAN, C. 2004. Social adversity contributes to high morbidity in psychoses in immigrants–a national cohort study in two generations of Swedish residents. *Psychological medicine*, 34, 1025-1033.
- HUTCHINSON, G., TAKEI, N., BHUGRA, D., FAHY, T., GILVARRY, C., MALLETT, R., MORAN, P., LEFF, J. & MURRAY, R. 1997. Increased rate of psychosis among African-Caribbeans in Britain is not due to an excess of pregnancy and birth complications. *The British Journal of Psychiatry*, 171, 145-147.
- JABLENSKY, A., SARTORIUS, N., ERNBERG, G., ANKER, M., KORTEN, A., COOPER, J., DAY, R. & BERTELSEN, A. 1992a. Schizophrenia: manifestations, incidence and course in different cultures A World Health Organization Ten-Country Study. *Psychological medicine*. *Monograph supplement*, 20, 1-97.
- JABLENSKY, A., SARTORIUS, N., ERNBERG, G., ANKER, M., KORTEN, A., COOPER, J. E., DAY, R. & BERTELSEN, A. 1992b. Schizophrenia: manifestations, incidence and course in different cultures. A World Health Organization ten-country study. *Psychol Med Monogr Suppl*, 20, 1-97.
- JARVIS, G. E., KIRMAYER, L. J., JARVIS, G. K. & WHITLEY, R. 2014. The role of Afro-Canadian status in police or ambulance referral to emergency psychiatric services. *Psychiatric Services*.
- JOHNS, L. C., CANNON, M., SINGLETON, N., MURRAY, R. M., FARRELL, M., BRUGHA, T., BEBBINGTON, P., JENKINS, R. & MELTZER, H. 2004. Prevalence and correlates of self-reported psychotic symptoms in the British population. *The British Journal of Psychiatry*, 185, 298-305.
- JOHNSON, S., ZINKLER, M. & PRIEBE, S. 2001. Mental health service provision in England. Acta Psychiatrica Scandinavica, 104, 47-55.
- JONES, P. 2011. Schizophrenia & Psychotic Disorders: Current epidemiological highlights. *Cutting Edge Psychiatry in Practice*.
- JORDAN, W. D. 1974. *The white man's burden: Historical origins of racism in the United States*, Oxford University Press, USA.
- JOSEPH, R. & BIRCHWOOD, M. 2005. The national policy reforms for mental health services and the story of early intervention services in the United Kingdom. *Journal of Psychiatry and Neuroscience*, 30, 362.
- KAFFASH, J. 2017. Average GP waiting times remain at two weeks despite rescue measures [Online]. Pulse. Available: http://www.pulsetoday.co.uk/your-practice/practicetopics/access/average-gp-waiting-times-remain-at-two-weeks-despite-rescuemeasures/20034534.article [Accessed 16 MArch 2018].
- KANESHIRO, B., GELING, O., GELLERT, K. & MILLAR, L. 2011. The challenges of collecting data on race and ethnicity in a diverse, multiethnic state. *Hawaii medical journal*, 70, 168.

KARLSEN, S., NAZROO, J. Y. & NAZROO, J. 2006. Defining and measuring ethnicity and 'race'. *Health* and social research in multiethnic societies, 20.

- KENDELL, R. E., MALCOLM, D. E. & ADAMS, W. 1993. The problem of detecting changes in the incidence of schizophrenia. *Br J Psychiatry*, 162, 212-8.
- KIRKBRIDE, J., BARKER, D., COWDEN, F., STAMPS, R., YANG, M., JONES, P. & COID, J. 2008. Psychoses, ethnicity and socio-economic status. *British Journal of Psychiatry*, 193, 18-24.
- KIRKBRIDE, J. B., CROUDACE, T., BREWIN, J., DONOGHUE, K., MASON, P., GLAZEBROOK, C., MEDLEY, I., HARRISON, G., COOPER, J. E., DOODY, G. A. & JONES, P. B. 2009. Is the incidence of psychotic disorder in decline? Epidemiological evidence from two decades of research.[Erratum appears in Int J Epidemiol. 2009 Dec;38(6):1753]. International Journal of Epidemiology, 38, 1255-64.
- KIRKBRIDE, J. B., ERRAZURIZ, A., CROUDACE, T. J., MORGAN, C., JACKSON, D., BOYDELL, J., MURRAY,
 R. M. & JONES, P. B. 2012a. Incidence of schizophrenia and other psychoses in England,
 1950-2009: a systematic review and meta-analyses. *PLoS One*, 7, 22.
- KIRKBRIDE, J. B., ERRAZURIZ, A., CROUDACE, T. J., MORGAN, C., JACKSON, D., BOYDELL, J., MURRAY,
 R. M. & JONES, P. B. 2012b. Incidence of schizophrenia and other psychoses in England,
 1950-2009: a systematic review and meta-analyses. *PLoS One*, 7, e31660.
- KIRKBRIDE, J. B., FEARON, P., MORGAN, C., DAZZAN, P., MORGAN, K., TARRANT, J., LLOYD, T., HOLLOWAY, J., HUTCHINSON, G., LEFF, J. P., MALLETT, R. M., HARRISON, G. L., MURRAY, R.
 M. & JONES, P. B. 2006. Heterogeneity in incidence rates of schizophrenia and other psychotic syndromes: Findings from the 3-center AESOP study. *Archives of General Psychiatry*, 63, 250-258.
- KIRKBRIDE, J. B., JONES, P. B., ULLRICH, S. & COID, J. W. 2014. Social deprivation, inequality, and the neighborhood-level incidence of psychotic syndromes in East London. *Schizophr Bull*, 40, 169-80.
- KIRKBRIDE, J. B., LUNN, D., MORGAN, C., LAPPIN, J. M., DAZZAN, P., MORGAN, K., FEARON, P., MURRAY, R. M. & JONES, P. B. 2010. Examining evidence for neighbourhood variation in the duration of untreated psychosis. *Health & place*, 16, 219-225.
- KIRKBRIDE, J. B., MORGAN, C., FEARON, P., DAZZAN, P., MURRAY, R. M. & JONES, P. B. 2007. Neighbourhood-level effects on psychoses: Re-examining the role of context. *Psychological Medicine: A Journal of Research in Psychiatry and the Allied Sciences*, 37, 1413-1425.
- KIRKWOOD, B. & STERNE, J. 2003a. Analysis of clustered data. *Essential Medical Statistics. Massachusetts: Blackwell Science Ltd*, 353-370.
- KIRKWOOD, B. R. & STERNE, J. A. 2003b. Medical statistics. *Medical statistics*.
- KIRMAYER, L. J., WEINFELD, M., BURGOS, G., DU FORT, G. G., LASRY, J. C. & YOUNG, A. 2007. Use of health care services for psychological distress by immigrants in an urban multicultural milieu. *Can J Psychiatry*, 52, 295-304.
- KOHLER, U. & KREUTER, F. 2005. *Data analysis using Stata*, Stata press.
- KUIPERS, E., HOLLOWAY, F., RABE-HESKETH, S. & TENNAKOON, L. 2004. An RCT of early intervention in psychosis: Croydon Outreach and Assertive Support Team (COAST). *Social psychiatry and psychiatric epidemiology*, 39, 358-363.
- LAGUERRE, M. 1987. Afro-Caribbean folk medicine, Bergin and Garvey Publishers, Inc.
- LAWLOR, C., JOHNSON, S., COLE, L. & HOWARD, L. M. 2012. Ethnic variations in pathways to acute care and compulsory detention for women experiencing a mental health crisis. *Int J Soc Psychiatry*, 58, 3-15.
- LAY, B., LAUBER, C. & ROSSLER, W. 2005. Are immigrants at a disadvantage in psychiatric in-patient care? *Acta Psychiatr Scand*, 111, 358-66.
- LEE, W., BINDMAN, J., FORD, T., GLOZIER, N., MORAN, P., STEWART, R. & HOTOPF, M. 2007. Bias in psychiatric case–control studies. *The British Journal of Psychiatry*, 190, 204-209.
- LEFF, J. 1988. *Psychiatry around the globe: A transcultural view*, Gaskell/Royal College of Psychiatrists.

- LEWIS, I. M. 1971. *Ecstatic religion: An anthropological study of spirit possession and shamanism*, Penguin (Non-Classics).
- LLOYD, T., KENNEDY, N., FEARON, P., KIRKBRIDE, J., MALLETT, R., LEFF, J., HOLLOWAY, J., HARRISON, G., DAZZAN, P., MORGAN, K., MURRAY, R. M. & JONES, P. B. 2005. Incidence of bipolar affective disorder in three UK cities: Results from the AESOP study. *British Journal of Psychiatry*, 186, 126-131.

LONDON BOROUGH OF LAMBETH 2012. State of the borough.

LONDON BOROUGH OF SOUTHWARK. 2011. *Census 2011 briefing* [Online]. Southwark Council. Available:

http://www.southwark.gov.uk/info/200223/census_2011/2723/census_2011_briefing [Accessed 2015].

- LUNDBERG, P., CANTOR-GRAAE, E., KAHIMA, M. & ÖSTERGREN, P.-O. 2007. Delusional ideation and manic symptoms in potential future emigrants in Uganda. *Psychological medicine*, 37, 505-512.
- MADEN, A., FRIENDSHIP, C., MCCLINTOCK, T. & RUTTER, S. 1999. Outcome of admission to a medium secure psychiatric unit. 2. Role of ethnic origin. *Br J Psychiatry*, 175, 317-21.
- MALLETT, R. 1997. MRC Sociodeomgraphic Schedule. Section of Social Psychiatry, IOP.
- MALLETT, R., LEFF, J., BHUGRA, D., PANG, D. & ZHAO, J. H. 2002. Social environment, ethnicity and schizophrenia. A case-control study. *Soc Psychiatry Psychiatr Epidemiol*, 37, 329-35.
- MANN, F., FISHER, H. L. & JOHNSON, S. 2014a. A systematic review of ethnic variations in hospital admission and compulsory detention in first-episode psychosis. *J Ment Health*, 23, 205-11.
- MANN, F., FISHER, H. L., MAJOR, B., LAWRENCE, J., TAPFUMANEYI, A., JOYCE, J., HINTON, M. F. & JOHNSON, S. 2014b. Ethnic variations in compulsory detention and hospital admission for psychosis across four UK Early Intervention Services. *BMC Psychiatry*, 14, 256.
- MARCH, D., HATCH, S. L., MORGAN, C., KIRKBRIDE, J. B., BRESNAHAN, M., FEARON, P. & SUSSER, E. 2008. Psychosis and Place. *Epidemiologic Reviews*, 30, 84-100.
- MARMOT, M. 2005. Social determinants of health inequalities. *The Lancet*, 365, 1099-1104.
- MARSHALL, M., LEWIS, S., LOCKWOOD, A., DRAKE, R., JONES, P. & CROUDACE, T. 2005. Association between duration of untreated psychosis and outcome in cohorts of first-episode patients: a systematic review. *Archives of General Psychiatry*, 62, 975.
- MARSHALL, M. & RATHBONE, J. 2011. Early Intervention for Psychosis. *Schizophrenia Bulletin*, 37, 1111-1114.
- MASON, D. 1995. Race and ethnicity in modern Britain, Oxford University Press.
- MATEOS, P., SINGLETON, A. & LONGLEY, P. 2009. Uncertainty in the analysis of ethnicity classifications: issues of extent and aggregation of ethnic groups. *Journal of Ethnic and Migration Studies*, 35, 1437-1460.
- MATHESON, S., SHEPHERD, A., PINCHBECK, R., LAURENS, K. & CARR, V. 2013. Childhood adversity in schizophrenia: a systematic meta-analysis. *Psychological medicine*, 43, 225-238.
- MCGORRY, P. D., KILLACKEY, E. & YUNG, A. 2008. Early intervention in psychosis: concepts, evidence and future directions. *World Psychiatry*, **7**, 148-156.
- MCGORRY, P. D., NELSON, B., AMMINGER, G. P., BECHDOLF, A., FRANCEY, S. M., BERGER, G., RIECHER-ROSSLER, A., KLOSTERKOTTER, J., RUHRMANN, S. & SCHULTZE-LUTTER, F. 2009. Intervention in individuals at ultra-high risk for psychosis: a review and future directions. *Journal of Clinical Psychiatry*, 70, 1206.
- MCGOVERN, D. & COPE, R. 1991. Second generation Afro-Caribbeans and young whites with a first admission diagnosis of schizophrenia. *Social psychiatry and psychiatric epidemiology*, 26, 95-99.
- MCGRATH, J. 1999. Hypothesis: is low prenatal vitamin D a risk-modifying factor for schizophrenia? *Schizophrenia research*, 40, 173-177.

- MCGRATH, J., SAHA, S., WELHAM, J., EL SAADI, O., MACCAULEY, C. & CHANT, D. 2004. A systematic review of the incidence of schizophrenia: the distribution of rates and the influence of sex, urbanicity, migrant status and methodology. *BMC Med*, *2*, 13.
- MCGRATH, J. J., SAHA, S., AL-HAMZAWI, A. O., ALONSO, J., ANDRADE, L., BORGES, G., BROMET, E. J., OAKLEY BROWNE, M., BRUFFAERTS, R., CALDAS DE ALMEIDA, J. M., FAYYAD, J., FLORESCU, S., DE GIROLAMO, G., GUREJE, O., HU, C., DE JONGE, P., KOVESS-MASFETY, V., LEPINE, J. P., LIM, C. C., NAVARRO-MATEU, F., PIAZZA, M., SAMPSON, N., POSADA-VILLA, J., KENDLER, K. S. & KESSLER, R. C. 2016. Age of Onset and Lifetime Projected Risk of Psychotic Experiences: Cross-National Data From the World Mental Health Survey. *Schizophr Bull*.
- MCGUFFIN, P. F., A. HARVEY, I. 1991. A polydiagnostic application of operational criteria in studies of psychotic illness: Development and reliability of the opcrit system. *Archives of General Psychiatry*, 48, 764-770.
- MCKENZIE, K. & ET AL. 2001. Comparison of the outcome and treatment of psychosis in people of Caribbean origin living in the UK and British Whites: report from the UK700 trial. *British Journal of Psychiatry. No.,* 178, 160-165.
- MCKENZIE, K., FAHY, T., JONES, P., HARVEY, I., TOONE, B. & MURRAY, R. 1995. Psychosis with good prognosis in Afro-Caribbean people now living in the United Kingdom. *Bmj*, 311, 1325-1327.
- MCKENZIE, K., FEARON, P. & HUTCHINSON, G. 2008. Migration, ethnicity and psychosis. *Society and Psychosis*, 143-160.
- MENTALHEALTHCARE. 2015. *Mental health law and compulsory treatment* [Online]. Available: http://www.mentalhealthcare.org.uk/mental_health_act#top [Accessed 18 Sept 2015 2015].
- MOHAN, R., MCCRONE, P., SZMUKLER, G., MICALI, N., AFUWAPE, S. & THORNICROFT, G. 2006. Ethnic differences in mental health service use among patients with psychotic disorders. *Social psychiatry and psychiatric epidemiology*, 41, 771-776.
- MORGAN, C., ABDUL-AL, R., LAPPIN, J. M., JONES, P., FEARON, P., LEESE, M., CROUDACE, T., MORGAN, K., DAZZAN, P., CRAIG, T., LEFF, J. & MURRAY, R. 2006a. Clinical and social determinants of duration of untreated psychosis in the AESOP first-episode psychosis study. *Br J Psychiatry*, 189, 446-52.
- MORGAN, C., CHARALAMBIDES, M., HUTCHINSON, G. & MURRAY, R. M. 2010. Migration, ethnicity, and psychosis: toward a sociodevelopmental model. *Schizophrenia Bulletin*, 36, 655-64.
- MORGAN, C. & FEARON, P. 2007. Social experience and psychosis. Insights from studies of migrant and ethnic minority groups. *Epidemiologia e Psichiatria Sociale*, 16, 118-123.
- MORGAN, C., FEARON, P., HUTCHINSON, G., MCKENZIE, K., LAPPIN, J. M., ABDUL-AL, R., MORGAN, K., DAZZAN, P., BOYDELL, J., HARRISON, G., CRAIG, T., LEFF, J., JONES, P. & MURRAY, R.
 2006b. Duration of untreated psychosis and ethnicity in the AESOP first-onset psychosis study. *Psychol Med*, 36, 239-47.
- MORGAN, C., FISHER, H., HUTCHINSON, G., KIRKBRIDE, J., CRAIG, T., MORGAN, K., DAZZAN, P., BOYDELL, J., DOODY, G., JONES, P., MURRAY, R., LEFF, J. & FEARON, P. 2009. Ethnicity, social disadvantage and psychotic-like experiences in a healthy population based sample. *Acta Psychiatrica Scandinavica*, 119, 226-235.
- MORGAN, C., KIRKBRIDE, J., HUTCHINSON, G., CRAIG, T., MORGAN, K., DAZZAN, P., BOYDELL, J., DOODY, G., JONES, P., MURRAY, R., LEFF, J. & FEARON, P. 2008a. Cumulative social disadvantage, ethnicity and first-episode psychosis: A case-control study. *Psychological Medicine: A Journal of Research in Psychiatry and the Allied Sciences*, 38, 1701-1715.
- MORGAN, C., KIRKBRIDE, J., LEFF, J., CRAIG, T., HUTCHINSON, G., MCKENZIE, K., MORGAN, K., DAZZAN, P., DOODY, G. A., JONES, P., MURRAY, R. & FEARON, P. 2007. Parental separation, loss and psychosis in different ethnic groups: a case-control study. *Psychol Med*, 37, 495-503.
- MORGAN, C., MALLETT, R., HUTCHINSON, G., BAGALKOTE, H., MORGAN, K., FEARON, P., DAZZAN, P., BOYDELL, J., MCKENZIE, K., HARRISON, G., MURRAY, R., JONES, P., CRAIG, T. & LEFF, J. 2005a.

Pathways to care and ethnicity. 1: Sample characteristics and compulsory admission. Report from the AESOP study. *Br J Psychiatry*, 186, 281-9.

- MORGAN, C., MALLETT, R., HUTCHINSON, G., BAGALKOTE, H., MORGAN, K., FEARON, P., DAZZAN, P., BOYDELL, J., MCKENZIE, K., HARRISON, G., MURRAY, R., JONES, P., CRAIG, T. & LEFF, J.
 2005b. Pathways to care and ethnicity. 2: Source of referral and help-seeking. *The British Journal of Psychiatry*, 186, 290-296.
- MORGAN, C., MALLETT, R., HUTCHINSON, G. & LEFF, J. 2004. Negative pathways to psychiatric care and ethnicity: the bridge between social science and psychiatry. *Soc Sci Med*, 58, 739-52.
- MORGAN, C., MCKENZIE, K. & FEARON, P. 2008b. Society and psychosis. *Society and psychosis*. New York, NY: Cambridge University Press; US.
- MORGAN, C., REININGHAUS, U., FEARON, P., HUTCHINSON, G., MORGAN, K., DAZZAN, P., BOYDELL, J., KIRKBRIDE, J. B., DOODY, G. A., JONES, P. B., MURRAY, R. M. & CRAIG, T. 2014. Modelling the interplay between childhood and adult adversity in pathways to psychosis: initial evidence from the AESOP study. *Psychol Med*, 44, 407-19.
- MORRISON, T. C., WAHLGREN, D. R., HOVELL, M. F., ZAKARIAN, J., BURKHAM-KREITNER, S., HOFSTETTER, C. R., SLYMEN, D. J., KEATING, K., RUSSOS, S. & JONES, J. A. 1997. Tracking and follow-up of 16,915 adolescents: Minimizing attrition bias. *Controlled Clinical Trials*, 18, 383-396.
- MORTENSEN, P. B. 1995. The untapped potential of case registers and record-linkage studies in psychiatric epidemiology. *Epidemiologic reviews*, **17**, 205-209.
- MULDER, C. L., KOOPMANS, G. T. & SELTEN, J. P. 2006. Emergency psychiatry, compulsory admissions and clinical presentation among immigrants to the Netherlands. *Br J Psychiatry*, 188, 386-91.
- MUNK-JØRGENSEN, P., KASTRUP, M. & MORTENSEN, P. B. 1993. The Danish psychiatric register as a tool in epidemiology. *Acta Psychiatrica Scandinavica*, 87, 27-32.
- NAZROO, J. Y. Ethnicity and mental health: findings from a national community survey. 1997. PSI. NEW POLICY INSTITUTE. 2014. *London's Poverty Profile* [Online]. Available:
 - http://www.londonspovertyprofile.org.uk/indicators/topics/housing-and-
 - homelessness/new-homeless-households-by-borough/ [Accessed].
- NICE. 2015. *Psychosis and schizophrenia in adults* [Online]. National Institute for Clinical Excellence. Available: https://www.nice.org.uk/guidance/qs80 [Accessed].
- ODEGAARD, O. 1932. Emigration and insanity. Acta Psychiatrica et Neurologica Scand, 4, 206.
- OLUWATAYO, O. & GATER, R. 2004. The role of engagement with services in compulsory admission of African/Caribbean patients. *Social psychiatry and psychiatric epidemiology*, 39, 739-743.
- ONS 2003. Ethnic group statistics: a guide for collection and classification of ethnicity data.
- ONS 2011a. 2011 Census Population and Household Estimates for England and Wales, March 2011, 2912. Office of National Statistics ONS.
- ONS. 2011b. 2011 Census: Key statistics fo England and Wales, Marach 2011 [Online]. Office of National Statistics. Available:

http://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populati onestimates/bulletins/2011censuskeystatisticsforenglandandwales/2012-12-11 [Accessed 03.08.2015 2015].

- ONS 2011c. Neighbourhood Statistics.
- PATEL, R., ODUOLA, S., CALLARD, F., WYKES, T., BROADBENT, M., STEWART, R., CRAIG, T. K. & MCGUIRE, P. 2017. What proportion of patients with psychosis is willing to take part in research? A mental health electronic case register analysis. *BMJ open*, **7**, e013113.
- PATEL, V. 1995. Explanatory models of mental illness in sub-Saharan Africa. *Social science & medicine*, 40, 1291-1298.
- PECK, E. & NORMAN, I. J. 1999. Working together in adult community mental health services: exploring inter-professional role relations. *Journal of Mental Health*, 8, 231.

- PERERA, G., BROADBENT, M., CALLARD, F., CHANG, C.-K., DOWNS, J., DUTTA, R., FERNANDES, A., HAYES, R. D., HENDERSON, M. & JACKSON, R. 2016. Cohort profile of the South London and Maudsley NHS Foundation Trust Biomedical Research Centre (SLaM BRC) Case Register: current status and recent enhancement of an Electronic Mental Health Record-derived data resource. *BMJ open*, 6, e008721.
- PERERA, G., SOREMEKUN, M., BREEN, G. & STEWART, R. 2009. The psychiatric case register: noble past, challenging present, but exciting future. *The British Journal of Psychiatry*, 195, 191-193.
- POWER, P., IACOPONI, E., REYNOLDS, N., FISHER, H., RUSSELL, M., GARETY, P., MCGUIRE, P. K. & CRAIG, T. 2007. The Lambeth Early Onset Crisis Assessment Team Study: general practitioner education and access to an early detection team in first-episode psychosis. *Br J Psychiatry Suppl*, 51, s133-9.
- PRIEBE, S., FROTTIER, P., GADDINI, A., KILIAN, R., LAUBER, C., MARTÍNEZ-LEAL, R., MUNK-JØRGENSEN, P., WALSH, D., WIERSMA, D. & WRIGHT, D. 2008. Mental health care institutions in nine European countries, 2002 to 2006. *Psychiatric Services*, 59, 570-573.
- PRINCE, M., STEWART, R., FORD, T. & HOTOPF, M. 2003. Practical psychiatric epidemiology. *New York: Oxford*.
- PUBLIC HEALTH ENGLAND. 2004. *Postcode Lookup Widget* [Online]. Available: https://www.ndtms.org.uk/emids/cgi-bin/ons_locale.cgi [Accessed].
- RAMSAY, C. E., STEWART, T. & COMPTON, M. T. 2012. Unemployment among patients with newly diagnosed first-episode psychosis: prevalence and clinical correlates in a US sample. *Social psychiatry and psychiatric epidemiology*, 47, 797-803.
- REININGHAUS, U., CRAIG, T. K., FISHER, H. L., HUTCHINSON, G., FEARON, P., MORGAN, K., DAZZAN, P., DOODY, G. A., JONES, P. B., MURRAY, R. M. & MORGAN, C. 2010a. Ethnic identity, perceptions of disadvantage, and psychosis: Findings from the AESOP study. *Schizophrenia Research*, 124, 43-48.
- REININGHAUS, U., CRAIG, T. K., FISHER, H. L., HUTCHINSON, G., FEARON, P., MORGAN, K., DAZZAN, P., DOODY, G. A., JONES, P. B., MURRAY, R. M. & MORGAN, C. 2010b. Ethnic identity, perceptions of disadvantage, and psychosis: findings from the AESOP study. *Schizophr Res*, 124, 43-8.
- REININGHAUS, U. A., MORGAN, C., SIMPSON, J., DAZZAN, P., MORGAN, K., DOODY, G. A., BHUGRA, D., LEFF, J., JONES, P., MURRAY, R., FEARON, P. & CRAIG, T. K. 2008. Unemployment, social isolation, achievement-expectation mismatch and psychosis: findings from the AESOP Study. *Soc Psychiatry Psychiatr Epidemiol*, 43, 743-51.
- RUMBAUT, R. G. & PORTES, A. 2001. *Ethnicities: Children of immigrants in America*, Univ of California Press.
- SASS, B., MOFFAT, J., BHUI, K. & MCKENZIE, K. 2009. Enhancing pathways to care for black and minority ethnic populations: a systematic review. *Int Rev Psychiatry*, 21, 430-8.
- SCHOFIELD, P., ASHWORTH, M. & JONES, R. 2011. Ethnic isolation and psychosis: Re-examining the ethnic density effect. *Psychological Medicine: A Journal of Research in Psychiatry and the Allied Sciences*, 41, 1263-1269.
- SELTEN, J.-P., CANTOR-GRAAE, E., SLAETS, J. & KAHN, R. S. 2002. Ødegaard's selection hypothesis revisited: schizophrenia in Surinamese immigrants to the Netherlands. *American Journal of Psychiatry*, 159, 669-671.
- SELTEN, J. P., VEEN, N., FELLER, W., BLOM, J. D., SCHOLS, D., CAMOENIE, W., OOLDERS, J., VAN DER VELDEN, M., HOEK, H. W., RIVERO, V. M., VAN DER GRAAF, Y. & KAHN, R. 2001. Incidence of psychotic disorders in immigrant groups to The Netherlands. *Br J Psychiatry*, 178, 367-72.
- SEWELL, H. 2009. *Working with Ethnicity, Race and Culture in Mental Health,* London, Jessica Kingsley Publishers.
- SHARPLEY, M., HUTCHINSON, G., MCKENZIE, K. & MURRAY, R. M. 2001. Understanding the excess of psychosis among the African-Caribbean population in England. Review of current hypotheses. *Br J Psychiatry Suppl*, 40, s60-8.
- SHAW, R. J., ATKIN, K., BECARES, L., ALBOR, C. B., STAFFORD, M., KIERNAN, K. E., NAZROO, J. Y.,
 WILKINSON, R. G. & PICKETT, K. E. 2012. Impact of ethnic density on adult mental disorders: narrative review. *Br J Psychiatry*, 201, 11-9.
- SINGH, S. P., BROWN, L., WINSPER, C., GAJWANI, R., ISLAM, Z., JASANI, R., PARSONS, H., RABBIE-KHAN, F. & BIRCHWOOD, M. 2015a. Ethnicity and pathways to care during first episode psychosis: the role of cultural illness attributions. *BMC psychiatry*, 15, 1.
- SINGH, S. P., BROWN, L., WINSPER, C., GAJWANI, R., ISLAM, Z., JASANI, R., PARSONS, H., RABBIE-KHAN, F. & BIRCHWOOD, M. 2015b. Ethnicity and pathways to care during first episode psychosis: the role of cultural illness attributions. *BMC Psychiatry*, **15**, 287.
- SINGH, S. P., BURNS, T., TYRER, P., ISLAM, Z., PARSONS, H. & CRAWFORD, M. 2014a. Ethnicity as a predictor of detention under the Mental Health Act. *Psychological medicine*, 44, 997-1004.
- SINGH, S. P., BURNS, T., TYRER, P., ISLAM, Z., PARSONS, H. & CRAWFORD, M. J. 2014b. Ethnicity as a predictor of detention under the Mental Health Act. *Psychological Medicine*, 44, 997-1004.
- SINGH, S. P. & GRANGE, T. 2006. Measuring pathways to care in first-episode psychosis: a systematic review. *Schizophrenia Research*, **81**, **75**-82.
- SINGH, S. P., GREENWOOD, N., WHITE, S. & CHURCHILL, R. 2007. Ethnicity and the Mental Health Act 1983. *Br J Psychiatry*, 191, 99-105.
- SLAM. 2014. *Research* [Online]. Available: http://www.slam.nhs.uk/research [Accessed 15 July 2014].
- SMITH, G., BOYDELL, J., MURRAY, R., FLYNN, S., MCKAY, K., SHERWOOD, M. & HONER, W. 2006. The incidence of schizophrenia in European immigrants to Canada. *Schizophrenia Research*, 87, 205-211.
- SNOWDEN, L. R. & YAMADA, A.-M. 2005. Cultural differences in access to care. *Annu. Rev. Clin. Psychol.*, **1**, 143-166.
- STATACORP, L. 2011. STATA-IC (Version 12) College Station. Texas: StatCorp LP.
- STERK, B., SLIEF, E. M., BLANKERS, M., LINSZEN, D. H. & DE HAAN, L. 2010. Duration of untreated psychosis and ethnicity. *Schizophrenia Research*, 124, 238-239.
- STEWART, R., SOREMEKUN, M., PERERA, G., BROADBENT, M., CALLARD, F., DENIS, M., HOTOPF, M., THORNICROFT, G. & LOVESTONE, S. 2009. The South London and Maudsley NHS Foundation Trust Biomedical Research Centre (SLAM BRC) case register: development and descriptive data. *BMC Psychiatry*, 9, 51.
- STILO, S. A., DI FORTI, M., MONDELLI, V., FALCONE, A. M., RUSSO, M., O'CONNOR, J., PALMER, E., PAPARELLI, A., KOLLIAKOU, A., SIRIANNI, M., TAYLOR, H., HANDLEY, R., DAZZAN, P., PARIANTE, C., MARQUES, T. R., ZOCCALI, R., DAVID, A., MURRAY, R. M. & MORGAN, C. 2013. Social disadvantage: cause or consequence of impending psychosis? *Schizophr Bull*, 39, 1288-95.
- SUSSER, E., FINNERTY, M., MOJTABAI, R., YALE, S., CONOVER, S., GOETZ, R. & AMADOR, X. 2000. Reliability of the life chart schedule for assessment of the long-term course of schizophrenia. *Schizophrenia Research*, 42, 67-77.
- SUSSER, E., SCHWARZ, S., MORABIA, A., BROMET, E. J., BEGG, M. D., GORMAN, J. M. & KING, M.-C. C. 2006. *Psychiatric epidemiology: Searching for the causes of mental disorders*, Oxford University Press.
- SUSSER, E., VELING, W., SELTEN, J. P. & HOEK, H. W. 2010. Age at migration and risk of psychoses in immigrant/ethnic minority groups. *Neuropsychopharmacology*, 35, S20.
- TAFARI, S., ABOUD, F. E. & LARSON, C. P. 1991. Determinants of mental illness in a rural Ethiopian adult population. *Social science & medicine*, 32, 197-201.
- TAKEI, N., SHAM, P., CALLAGHAN, E., GLOVER, G. & MURRAY, R. 1995. Early risk factors in schizophrenia: place and season of birth. *European Psychiatry*, 10, 165-170.
- TARRICONE, I., MIMMI, S., PAPARELLI, A., ROSSI, E., MORI, E., PANIGADA, S., CARCHIA, G., BANDIERI, V., MICHETTI, R. & MINENNA, G. 2012. First-episode psychosis at the West Bologna

Community Mental Health Centre: results of an 8-year prospective study. *Psychological medicine*, 42, 2255-2264.

- TARRICONE, I., TOSATO, S., CIANCONI, P., BRACA, M., FIORILLO, A., VALMAGGIA, L. & MORGAN, C.
 2015. Migration history, minorities status and risk of psychosis: an epidemiological explanation and a psychopathological insight. *Official Journal of the Italian Society of Psychopathology Organo Ufficiale della Società Italiana di Psicopatologia*, 46, 424-430.
- TEN HORN, G., GIEL, R., GULBINAT, W. H., HENDERSON, J. H. & ORGANIZATION, W. H. 1986. Psychiatric case registers in public health: a worldwide inventory 1960-1985/editors, GHMM ten Horn...[et al.].
- TERMORSHUIZEN, F., SMEETS, H. M., BRAAM, A. W. & VELING, W. 2014. Neighborhood ethnic density and psychotic disorders among ethnic minority groups in Utrecht City. *Soc Psychiatry Psychiatr Epidemiol*, 49, 1093-102.
- THOMAS, S. P. & NANDHRA, H. S. 2009. Early intervention in psychosis: a retrospective analysis of clinical and social factors influencing duration of untreated psychosis. *Prim Care Companion J Clin Psychiatry*, 11, 212-4.
- TORTELLI, A., MORGAN, C., SZOKE, A., NASCIMENTO, A., SKURNIK, N., DE CAUSSADE, E. M., FAIN-DONABEDIAN, E., FRIDJA, F., HENRY, M. & EZEMBE, F. 2014. Different rates of first admissions for psychosis in migrant groups in Paris. *Social psychiatry and psychiatric epidemiology*, 49, 1103-1109.
- TULLOCH, A. D. 2013. Using SQLCRIS. *NIHR Biomedical Research Centre, South London and Maudsley NHS Foundation Trust.* London: South London and Maudsley NHS Foundation Trust.
- TURNER, M., SMITH-HAMEL, C. & MULDER, R. 2006. Pathways to care in a New Zealand first-episode of psychosis cohort. *Australian and New Zealand Journal of Psychiatry*, 40, 421-428.
- TYRER, P. 2011. Has the closure of psychiatric beds gone too far? Yes. *BMJ: British Medical Journal,* 343.
- UCLA: STATISTICAL CONSULTING GROUP. 2015. *Survival Analysis with Stata* [Online]. Available: http://www.ats.ucla.edu/stat/stata/seminars/stata_survival/ [Accessed].
- UCLA: STATISTICAL CONSULTING GROUP. 2016. *Negative Binomial Regression* [Online]. Available: http://www.ats.ucla.edu/stat/stata/dae/nbreg.htm [Accessed 07.07.2016 2016].
- UK DATA SERVICE. 2017. UK Data Service [Online]. Available: https://www.ukdataservice.ac.uk/ [Accessed].
- UNITED NATIONS HIGH COMMISSIONER FOR REFUGEES (UNHCR). 2015. *Global appeal 2015: update* [Online]. Available: . http://www.unhcr.org/ga15/index.xml. [Accessed].
- VAN DER POST, L., VISCH, I., MULDER, C., SCHOEVERS, R., DEKKER, J. & BEEKMAN, A. 2012. Factors associated with higher risks of emergency compulsory admission for immigrants: a report from the ASAP study. *Int J Soc Psychiatry*, 58, 374-80.
- VAN OS, J. & KAPUR, S. 2009. Schizophrenia. Lancet, 374, 635-45.
- VAN OS, J., KENIS, G. & RUTTEN, B. P. 2010. The environment and schizophrenia. *Nature*, 468, 203-12.
- VAN OS, J., LINSCOTT, R. J., MYIN-GERMEYS, I., DELESPAUL, P. & KRABBENDAM, L. 2009. A systematic review and meta-analysis of the psychosis continuum: evidence for a psychosis proneness-persistence-impairment model of psychotic disorder. *Psychological Medicine*, 39, 179.
- VEEN, N., SELTEN, J. P., HOEK, H. W., FELLER, W., VAN DER GRAAF, Y. & KAHN, R. 2002. Use of illicit substances in a psychosis incidence cohort: A comparison among different ethnic groups in the Netherlands. *Acta Psychiatrica Scandinavica*, 105, 440-443.
- VELING, W., HOEK, H. W., SELTEN, J.-P. & SUSSER, E. 2011. Age at migration and future risk of psychotic disorders among immigrants in the Netherlands: a 7-year incidence study. *American Journal of Psychiatry*, 168, 1278-85.

- VELING, W., SELTEN, J.-P., VEEN, N., LAAN, W., BLOM, J. D. & HOEK, H. W. 2006. Incidence of schizophrenia among ethnic minorities in the Netherlands: a four-year first-contact study. *Schizophrenia research*, 86, 189-193.
- VELING, W., SUSSER, E., VAN OS, J., MACKENBACH, J., SELTEN, J.-P. & HOEK, H. 2008. Ethnic density of neighborhoods and incidence of psychotic disorders among immigrants. *American Journal of Psychiatry*, 165, 66-73.
- VELTHORST, E., NIEMAN, D., VELING, W., KLAASSEN, R., DRAGT, S., RIETDIJK, J., ISING, H.,
 WUNDERINK, L., LINSZEN, D., DE HAAN, L. & VAN DER GAAG, M. 2012. "Ethnicity and baseline symptomatology in patients with an At Risk Mental State for psychosis":
 Corrigendum. *Psychological Medicine: A Journal of Research in Psychiatry and the Allied Sciences*, 42, 257.
- WALSH, E., LEESE, M., TAYLOR, P., JOHNSTON, I., BURNS, T., CREED, F., HIGGIT, A. & MURRAY, R.
 2002. Psychosis in high-security and general psychiatric services: report from the UK700 and special hospitals' treatment resistant schizophrenia groups. *Br J Psychiatry*, 180, 351-7.
- WARD, M. M. 2005. Estimating rare disease prevalence from administrative hospitalization databases. *Epidemiology*, 16, 270-1.
- WHEELER, A., ROBINSON, E. & ROBINSON, G. 2005. Admissions to acute psychiatric inpatient services in Auckland, New Zealand: a demographic and diagnostic review. *N Z Med J*, 118, U1752.
- WHO 1996. Personal and Psychiatric History Schedule. Geneva: World Health Organisation.
- WHO 2005. Mental Health Atlas, Geneva, World Health Organisation.
- WORLD HEALTH ORGANIZATION 1993. The ICD-10 classification of mental and behavioural disorders: diagnostic criteria for research.
- ZANDI, T., HAVENAAR, J. M., SMITS, M., LIMBURG-OKKEN, A. G., VAN ES, H., CAHN, W., ALGRA, A., KAHN, R. S. & VAN DEN BRINK, W. 2010. First contact incidence of psychotic disorders among native Dutch and Moroccan immigrants in the Netherlands: Influence of diagnostic bias. *Schizophrenia Research*, 119, 27-33.

APPENDIX 1: FEP Screening Search Commands

select d.*,Gender_ID,ethnicitycleaned,Marital_Status_ID from

(

select brcid, start_date as date, comments as notes, Location_udf_6

from event

where contains (comments, "psychosis*" or "delusion" or "hallucinat*" or "voices" or "psychotic") and event_category='ward progress note'

union all

select brcid, start_date as date, comments as notes, Location_udf_6

from event

where contains (comments, "psychosis*" or "delusion" or "hallucinat*" or "voices" or "psychotic") and event_category='event'

) d

left join dbo.EPR_Form p on d.BrcId=p.BrcId

where

[date] between '24-Jul-2010' and '30-jul-2010'

and

location_udf_6 like 'lambeth'

and

d.Brcld not in

(Select Diagnosis.BrcId from Diagnosis where

(Primary_Diag like 'F2%'

or Primary_Diag like 'F30.2%'

or Primary_Diag like 'F31.2%'

or Primary_Diag like 'F31.5%'

or Primary_Diag like 'F32.3%'

or Primary_Diag like 'F33.3%')

and Diagnosis_Date < '01-May-2010')

and

d.brcid not in

(Select brcid from SQLCrisImport.dbo.FEP_Exclude_IDs\$)

APPENDIX 2: Psychosis Screening Schedule

If the answer to any of the following questions is 'yes' (*must be at least one yes on section A or two on section B*), the patient is considered to screen positive for psychotic symptoms and therefore, is potentially eligible for inclusion.

Section A

Has the patient ever presented any of the following (circle Y or N):

1)	Hallucinations or pseudo-hallucinations in any modality	Y	Ν
2)	Delusions	Y	Ν
3)	Marked thought and speech disorder (e.g. incoherence, irrelevance,		
	thought blocking, neologisms, incomprehensibility of speech) other		
	than simple retardation or acceleration	Y	Ν
4)	Marked psychomotor disorder (e.g. negativism, mutism or stupor,		
	catatonic excitement, constrained attitudes or unnatural posture maintained		
	for long periods) other than simple retardation or acceleration	Y	Ν
5)	Emergence of marked exacerbation of bizarre and grossly inappropriate		
	behaviour (e.g. talking or giggling to self, acts incomprehensible to others,		
	loss of social constraints etc.)	Y	Ν

Section B

Has the patient ever presented a definite <u>change</u> in personality and behaviour manifested in any of the following (circle Y or N):

1)	Marked reduction or loss of interests, initiative and drive, leading to		
	a serious deterioration of the performance of usual activities and tasks	Y	Ν
2)	Emergence or marked exacerbation of social withdrawal (active		
	avoidance of communication with others)	Y	Ν
3)	Severe excitement, purposeless destructiveness or aggression	Y	Ν
4)	Episodic or persistent states of overwhelming fear or severe anxiety	Y	Ν
5)	Gross and persistent self-neglect	Y	Ν

APPENDIX 3: Data Collection Booklet

Subject number: <u>1 9 </u> _	2EU02.	Date of Birth	- _ - -
<u>2 0 </u>		Date	- _ - -

Social (1) Sociodemographics (at first contact)

1.	Gender	[O -77 Not H	Recorded]	O1 Male	O ₂ Fem	nale
2.	Age	[O -77 Not H	Recorded]			
3.	Postcode	[O -77 Not H	Recorded]		-	
4.	Ethnicity	[O -77 Not H	Recorded]			
	O11White British	n	O12 White Irish		O13 White gypsy, tra	veller O14 Other White
	O15 Mixed (w, b	c)	O16 Mixed (w, ba)		O17 Mixed (w, as)	O18 Other Mixed
	O19 Indian		O20 Pakistani		O21 Bangladeshi	O22 Chinese
	O23 Other Asian		O24 Black African		O25 Black Caribbean	O26 Other Black
	O27 Arab		O28 Other, specify:			
5.	Place of Birth	[O -77 Not I	Recorded]			
	O1 Austria		O2 Belgium		O3 France	O4 Germany
	O5 Ireland		O6 Italy		O7 Spain	O8 Switzerland
	O9 The Netherlan	nds	O10 Turkey		O11 United Kingdom	O12 Brazil
	O13 Australia		O14 other, specify:			
6.	Age of migration ((if applicable) [O -77 Not Rec	orded]		
7.	Ever employed (pa O1 Yes	aid work)	[O -77 Not Rec	orded]		Oo No

8. Registered with a GP O1 Yes

9. Lives with

Alone	Alone, with children	Partner, Spouse	Partner, Spouse, with children	Parents	Other family	Friends	Other: specify (e.g. hostel, halls of residence)	Not Recorded
01	O2	O3	O4	O5	O6	07	O8	0 -77

10. Housing tenure

Privately owned (self)	Privately owned (family)	Rented (Private)	Rented (government)	Other, specify:	Not Recorded
01	02	O3	O4	05	0 -77

11. Ever had a long-term relationship (one year or more) [O -77 Not Recorded] Oo No O1 Yes

12. Number of children ...?

|__|_|

13. Relationship status ...?

Single	Married, living with someone	In a steady relationship	Divorced, separated	Widowed	Not Recorded
01	02	03	04	05	0 -77

[O -77 Not Recorded]

14. Highest level of education achieved ...? [O -77 Not Recorded]

01	School, no qualifications	(to end of compulsory education; passed no exams, tests, etc.)
02	School, with qualifications	(to end of compulsory education; passed one or more exams, tests, etc.)
03	Tertiary, Further	(first level of non-compulsory education; e.g. A-levels, Baccalaureate)
04	Vocational	(job related education, e.g. teacher training, plumber, electrician, etc.)
05	Higher (undergraduate)	(University; first degree)
06	Higher (postgraduate)	(University: higher than first degree level, e.g. Masters, PhD)

15. Employment status ...?

Unemployed	Economically inactive (i.e. house person, physical illness/disability, carer, retired)	Student	Part-time employee	Full-time employee	Self- employed	Not Recorded
01	02	03	04	05	06	0 -77

16. Weekly + O1 Ye	contact with family s	[O -77 Not Recorded]	O0 No
17. Weekly + O1 Ye	contact with friends	[O -77 Not Recorded]	Oo No
18. Any repo O1 Ye	rt of social isolation	[O -77 Not Recorded]	Oo No

Social (2) Childhood

1.	One or both of your biolo	gical parents died, befor	re age 17?			
	1.a. Mother died O1 Yes		[O -77 Not I	Recorded]	Oo No	
	1.b. Father died O1 Yes		[O -77 Not F	Recorded]	Oo No	
2. 17?	Separated from a biologic	al parent (longer than 6	months), befo	ore age		
	2.a. Separated from mother O1 Yes		[O -77 Not Recorded]		Oo No	
No	2.b. Separated from father O O1 Yes		[0	Oo		
	2.e. What was the main re	ason for the separation	? [O -77 Not I	Recorded]		
Illness	O1 Parental	O2 Divorce, Separation	O3 Work	O4 Never knew parent	O5 Own illness	
school	O6 Boarding	O7 Migration	O8 Other			
	2.g. Specify other:					
				Age		

Before the age of 17 years		0-11			12-16	
3. Ever change schools? (other than change to secondary)	O-77 NR	O0 No	O1 Yes	O-77 NR	O0 No	O1 Yes

Notes

Age

Befo	pre the age of 17 years		0-11			12-16	
4. caus	Ever hit/slapped on number of occasions, enough to se harm?	O-77 NR	O0 No	O1 Yes	O-77 NR	O0 No	O1 Yes
5.	Ever sexually abused?	O-77 NR	O0 No	O1 Yes	O-77 NR	O0 No	O1 Yes

If yes to Q7:

(1)	(2)	(3)	(4	1)	(5)	(6)	(8)
Perpetrat or	Age started	Age ended	Dura yr s	ntion mt hs	Frequenc y	Severity	Official contact

If yes to Q 8:

(1)	(2)	(3)	(4	1)	(5)	(6)	(8)
Perpetrat or	Age started	Age ended	Dura yr s	ntion mt hs	Frequenc y	Severity	Official contact

RATINGS						
Perpetrator						
1 Both parents; 2 household	Mother; 3 Father; 4 Sibling; 5 Other relative; 6 Family friend	d; 7 Other person in household; 8 Other person outside of				
Frequency						
0 Never; 1 Rarely (once/twice); 2 Occasionally (less than monthly); 3Frequently (monthly+); 4 Very frequently (weekly+)						
Severity	Physical	Sexual				
0 None	No abuse; Object used without possibility of causing	No abuse; Flashed by stranger; Willing sexual contact				

	injury; Pushed, grabbed or shoved	with someone the same age					
1 Some	Hit with open hand; Single incidents of being slapped around head or face; Hit hard or repeatedly around body, hard enough to cause injury	Single incidents of non-intercourse abuse; Forced to watch pornography/masturbation; Verbal solicitations by relative/authority figure					
2 Moderate	Punched, kicked, bitten, burnt; Implement or object used that could have caused injury	Repeated non-intercourse abuse (touching of own or others' private parts)					
3 Marked	Life threatened (e.g. gun or knife); Severe/multiple injuries likely; Abuse intense and frequent	Repeated sexual abuse that involved intercourse (vaginal or anal)					
Official contact 0 None; 1 Social s	Official contact 0 None; 1 Social services; 2 GP; 3 Police; 4 Other (specify)						

Notes

DUP

Please note the most accurate date!

• In case only information about the *year of onset* is available, please note the 1st of July of that year as date of onset

• In case only information about the *month of onset* is available, please note the 15th of that month as date of onset

Date of onset psychosis:

First day of onset of psychotic symptoms

Onset of psychotic symptoms is defined as: Clear evidence of delusions, hallucinations, first rank symptoms, catatonic symptoms (i.e. A score of 2 for a psychotic symptom in Part II of the SCAN OR a score \geq 4 on PANSS items P1, 'delusions', P3 'hallucinatory behaviour', P5 'Grandiosity', P6 'Suspiciousness' or A9 'Unusual thought content').

Date of contact with mental health services (for FEP)

(day/month/year):

[O -77 Not Recorded]



Date of onset psychosis:

[O -77 Not Recorded]



Mode of Onset

- O1 Abrupt onset definable to within hours or days
- O2 Acute onset definable to within 1 week
- O3 Moderately acute onset definable within 1 month
- O4 Gradual onset over period up to 6 months
- O5 Insidious onset over period greater than 6 months

Pathway to Care

1 Mode of Contact	0 = Community; 1 = Home treatment; 2 = In-patient (voluntary);				
(Secondary Mental Health Service)	3 = In-patient (co	mpulsory);	-77 = Not recorded		
2 MHA Section (if applicable)	1 = Section 2 recorded	6	= Section 37	-77 = Not	
	2 = Section 3	7	= Section 37/41		
	3 = Section 4	8	= Section 47		
	4 = Section 5(2)	9	= Section 48		
	5 = Section 5(4)				
2a. MHA Sec 136/135	0= No 1= Ye	es -77=	Not recorded		
3 Source of Referral	1 = General practitioner; 2 = Nurse, other health worker, or social worker;				
	3 = Accident and	Emergency	; 4 = Police; 5 = Courts;	6 = Prison;	
	7 = Other, specify	/; -77 = Not	recorded		
4 Contact out of hours	0 = No	1 = Yes	-77 = Not record	led	
5 Family Involvement	0 = No	1 = Yes	-77 = Not record	led	
6 Police or CJA Involvement	0 = No	1 = Yes	-77 = Not record	led	

Family History

[O -77 Not Recorded]

1. Evidence of history of mental illness in first degree relative? O0 No O1 Probable O2 Defin

If probable or definite, complete following for each affected relative:

No.	Relative (1 Father; 2 Mother; 3 Sibling; 4 Child)	Age	Treatment (1 GP; 2 Social worker; 3 Other)	Treatment setting (1 Inpatient; 2 Outpatient; 3 Medication only)	Reliability of information (1 Good; 2 Fair; 3 Poor)	Type of Disorder
1.						
2.						
3.						
4.						
5.						

Substance Use

1 Ever smoked/used cannabis?	[O -77 Not Recorded]	O1 Yes	O0 No

APPENDIX 4: Follow-up Data Extraction

Outcome data on hospital admission, service use, compulsory admission, length of stay and diagnosis.

SELECT [brcid]

,[gender]

,[postcode]

[ethnicity],

,[employment_status]

,[age]

,[higehesteducationlevel]

,[registeredwithgp]

,[hospitaladmissions]

,[Firstadmission]

,[noofadmissions]

,[Psychosocial_intervention]

,[Currentcontactwithpsychiatricservices]

,[CurrentTeamorward]

,[dischargedfromslamservices]

,[referraldischargedate]

,[referraldischargedestination]

,[CAT_Therapies]

,[CBT]

,[Counselling]

,[Eclectic_Approcahes]

,[Family_Therapies]

,[Group_therapy]

,[Mentalisation_therapy]

,[No_Therapy_Offered]

,[Other_therapy]

,[play_therapy]

,[Therapy_Offered]

FROM [SQLCrisImport].[dbo].[vw_sho_cohort_base]

GO

Appendix 4a, Data on hospital admission pattern

SELECT [BrcId]

,[dateofadmission]

,[dateofdischarge]

,[mhastatusonadmission]

,[wardtype]

,[admissionsource]

,[mhastatusduringadmission]

,[Police_or_CJA_involvement]

,[Adult_Acute_inpatient_wards]

,[Forensic_Acute_inpatient]

,[PICU136_Suite]

,[Rehab_wards]

,[Triage_Wards]

FROM [SQLCrisImport].[dbo].[vw_sho_cohort_admissions]

GO

Appendix 4b, Data on compulsory admission pattern

SELECT [BrcId]

,[dateofadmission]

,[dateofdischarge]

,[mha_section_start_date]

,[mha_section]

FROM [SQLCrisImport].[dbo].[vw_sho_cohort_mhasection]

GO

Appendix 4c, Data on community service use pattern

SELECT [BrcId]

,[episodestartdate]

,[episodeenddate]

,[locationname]

[group],

,[Referralsource]

,[Assessment_and_Treatment]

,[CMHTs],

,[Early_intervention]

,[Forensic_community_teams]

,[Home_Treatment_Crisis_Resolution]
,[Outpatients_Day_services]
,[Psych_Liaison_A_and_E]
,[Psychology_team]
,[Rehab]
,[Specialist_teams]
,[Supported_Housing]
,[Grp_136Suite]
FROM [SQLCrisImport].[dbo].[vw_sho_cohort_team_episode]
GO

Appendix 4d, Data on clinical course and diagnosis

SELECT [brcid]

[cn_doc_id],

,[source_table]

,[diagnosis_date]

,[primary_diagnosis]

,[Secondary_Diag_1]

,[Secondary_Diag_2]

,[Secondary_Diag_3]

,[Secondary_Diag_4]

,[Secondary_Diag_5]

,[Secondary_Diag_6]

FROM [SQLCrisImport].[dbo].[vw_sho_cohort_diagnosis_structured]

GO